Court File No.: CV-12-9545-00CL

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD. 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

### MOTION RECORD VOLUME II

DATE: September 14, 2018

FRED TAYAR & ASSOCIATES
Professional Corporation
Barristers and Solicitors
65 Queen Street West, Suite 1200
Toronto, Ontario M5H 2M5

Fred Tayar (23909N)
Tel.: (416) 363-1800
Fax: (416) 363-3356

Lawyers for Domfoam Inc.

TO: THE SERVICE LIST

### SERVICE LIST (as at November 23, 2017)

TO:	DELOITTE RESTRUCTURING INC.
10.	
	181 Bay Street, Suite 1400
	Toronto ON M5J 2V1
	Catherine Hristow
	416-601-6415
	christow@deloitte.ca
	Paul Casey
	paucasey@deloitte.ca
ANTO TIO	THE PARTY OF THE P
AND TO:	THORNTON GROUT FINNIGAN LLP
	TD West Tower, TD Centre
	100 Wellington Street West, Suite 3200
	Toronto ON M5K 1K7
	Grant B. Moffat
	416-304-0599
	416-304-1313 fax
	gmoffat@tgf.ca
	Townson Court Me to D I to D
	Lawyers for the Monitor, Deloitte Restructuring Inc.
AND TO:	BLANEY MCMURTRY LLP
	2 Queen Street East, Suite 1500
	Toronto, Ontario M5C 3G5
	Toronto, Ontario 14150 505
	David T. Ullmann
	416-596-4289
	416-594-2437 FAX
	dullmann@blaney.com
	Alexandra Teodorescu
	416-596-4279
	416-593-5437 FAX
	ateodorescu@blaney.com L
	Lisodoresealegoraney.com D
	Lawyers for the Applicants

AND TO:	TANNER & GUINEY
AID IO.	130 Adelaide Street West
	Suite 3425
**************************************	Toronto ON M5H 3P5
	Robert G. Tanner
	416-862-7745
	416-862-7874 fax
	rgtanner@tannerguiney.com La
	awyers for Tony Vallecoccia
AND TO:	ATTORNEY GENERAL OF CANADA
	c/o Department of Justice
	Ontario Regional Office
	The Exchange Tower, Box 36
	130 King Street West
	Suite 3400
	Toronto ON M5X 1K6
	TOTOTILO OIN WIJA TKO
	Jacqueline Dais-Visca
	416-952-6010
	416-973-0809 fax
	jacqueline.dais-visca@justice.gc.ca
	Jacqueime.uais-visca@justice.gc.ca
AND TO:	KRONIS, ROTSZTAIN, MARGLES, CAPPEL LLP
	25 Sheppard Ave. West
	Suite 1100
	Toronto, ON M2N 6S6
	Philip Cho
	416-218-5494
	416-225-6751 fax
	PCho@krmc-
	law.com Lawyers for B

AND TO:	BRANCH MACMASTER LLP
AND IO.	1410 –
	777 Hornby Street Vanco
	ouver BC V7G 3E2
	Luciana P. Brasil
	604-654-2999
	604-684-3429 fax
	lbrasil@branmac.com
	Lawyers for "Hi! Neighbor" Floor Covering Co. Limited, Majestic Mattress Mfg.
	Ltd., Trillium Project Management Ltd., Option Consummateurs and Karine Robi
	illard
AND TO:	MINISTRY OF FINANCE (ONTARIO)
	LEGAL SERVICES BRANCH
	33 King Street West 6 <sup>th</sup> floor
	Oshawa ON L1H 8H5
	Kevin J. O'Hara
	905-433-6934
	905-436-4510 fax
	kevin.ohara@fin.gov.on.ca
	<u>KCVIII.Olicia(WIII.gov.oli,ca</u>
AND TO:	KAMINSKY & COMPANY
	Associated Lawyers
	7525 King George Boulevard
	Suite 220
	Surrey BC V3W 5A8
	Hargo Mundi
	604-591-7877 ext. 114
	604-591-1978 fax
	hsm@kaminskyco.com Lawye
	0.0000016707.1
	ers for 0932916 BC Ltd.
AND TO:	DEVENUE OFFEE
AND IU:	REVENU QUEBEC 3800, rue de Marly
	Secteur 5-1-
	8 Quebec G1X 4A5
	5
	Jacques Duperron
	866-374-7286 fax
	Jacques.duperron@revenuquebec.ca
	Jacques.duperron@revenuquebec.ca

AND TO:	REVENU QUEBEC Complexe Desjardins Secteu
	ur D231L0, C. P.3000
	Montreal QC H5B 1A4
:	
	Yvon Julien
	yvon.julien@revenuquebec.ca
AND TO:	REVENU QUEBEC
	Complexe Desjardins, secteur D221LC
	C. P.5000, succursale Desjardins Mont
	treal QC H5B 1A7
	Normand Berube
	514-287-6161
	514-873-8992 fax
	Normand.Berube@revenuquebec.ca
AND TO:	SUTTS STROSBERG LLP
	600 Westcomt Place 215
	5 Goyeau Street Windso
	or ON N9A 6V4
	Harvey T. Strosberg
	Harvey@strosbergco.com
	Heather Rumble Peterson
	hpeterson@strosbergco.com
	519-561-6248
	519-561-6203 fax
	Lawyers for Ontario Class Plaintiffs

AND TO:	CAMP FIORANTE MATTHEWS MOGERMAN
	#400-
***************************************	856 Homer Street Vancouv ver BC V6B 2W5
	Ver be Vob 2 W 3
	J.J.Camp, Q.C.
	jjcamp@cfmlawyers.ca
	Reider Mogerman
	rmogerman@cfmlawyers.ca
	Sharon L. Wong
	swong@cfmlawyers.ca
	604-689-7555
	604-689-7554 fax
	Lawyers for B.C. Class Plaintiffs
AND TO:	BERKOW YOUD LEV-FARRELL DAS LLP
	141 Adelaide Street West
	Suite 400
	Toronto ON M5H 3L5
	Jack Berkow
	jberkow@byldlaw.com
	Scott Crocco
	scrocco@byldlaw.com
	416-364-4900
	416-364-3865 fax
	Lawyers for Remaining Individual Settling Parties
AND TO:	BLACK & ASSOCIATES
	2175 King Road   King City ON L7B 1G3
	Laith Hahn
	905-893-8050
	905-893-8025 fax
	lhahn@blackandassociates.ca Lawyer
	rs for Quality & Company Inc.

AND TO:	REVENU QUEBEC
AND IU:	,
	Complexe Desjardins, secteur D221LC
	C.P. 5000, succursale Place-
	Desjardins Montreal QC H5B 1A7
	Jean Duval
	514-287-8821
	514-285-5348 fax
	jean.duval@revenuquebec.ca
AND TO:	MINISTRY OF JUSTICE
	Legal Services Branch
	1675 Douglas Street S
	Suite 400
	Victoria BC V8W 9J7
	David J. Hatter
	250-387-1274
	250-387-0700 fax
	David.Hatter@gov.bc.ca
	Lawyers for BC Ministry of Finance
AND TO:	CHUBB INSURANCE COMPANY OF CANADA
	1250, boul. René-
	Lévesque Ouest, bureau 2700 Montréal QC H3B
	B 4W8
	Catherine Rayle-Doiron
	514-938-2300, 5325
	514-938-2288 fax
	crayledoiron@chubb.com
L	228/12/2010/04/05/04/05/04/05/04/05/05/05/05/05/05/05/05/05/05/05/05/05/

christow@deloitte.ca; paucasey@deloitte.ca gmoffat@tgf.ca; dullmann@blaney.com; ateodorescu@@blaney.com; rgtanner@tannerguiney.com; jacqueline.dais-visca@justice.gc.ca; PCho@krmc-law.com; lbrasil@branmac.com; kevin.ohara@fin.gov.on.ca; hsm@kaminskyco.com; Jacques.duper rron@revenuquebec.ca; yvon.julien@revenuquebec.ca; Normand.Berube@revenuquebec.ca; Harvey y@strosbergco.com; hpeterson@strosbergco.com; jjcamp@cfmlawyers.ca; rmogerman@cfmlawyer rs.ca; swong@cfmlawyers.ca; jberkow@byldlaw.com; scrocco@byldlaw.com; lhahn@blackandasso ociates.ca; jean.duval@revenuquebec.ca; David.Hatter@gov.bc.ca; crayledoiron@chubb.com;

# Index

Court File No.: CV-12-9545-00CL

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD. 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### **TABLE OF CONTENTS**

- 1. Notice of Motion dated September 14, 2018
- 2. Affidavit of Jacques Vincent sworn September 13, 2018

#### **EXHIBITS**

- A Asset Purchase Agreement #1
- B Asset Purchase Agreement #2
- C Final Asset Purchase Agreement dated March 8, 2012
- D Sale Approval and Vesting Order dated March 16, 2012
- E Letter from Jacques Vincent to David Ullman dated June 22, 2018
- F Notice of Assignment of Claims sent to Fine Kaplan and Black R.P.C. dated June 22, 2018
- G Copy of cheque dated March 21, 2018 for (US) \$3,741,639.62 payable to the Applicant [now 4362063 Canada Ltd.]

- H Motion Record of the Applicants which was returnable before Mr. Justice Wilton-Siegel on May 29, 2018 containing a draft Order
- 3. Order of the Honourable Justice Wilton-Siegel dated May 29, 2018
- 4. Seventh Report of the Monitor (without exhibits)

# Tab E

THIS IS EXHIBIT "E" TO THE AFFIDAVIT OF JACQUES VINCENT SWORN THIS 13<sup>th</sup> DAY OF

SEPTEMBER, 2018

Commissioner, etc.

Anny Grouix

### LOFROMBOISE UUTKIN.

Mtre. Jacques Vincent : (514) 938-1924 #4634

Email: j.vincent@laframboise-gutkin.com

BY EMAIL AND BY REGISTERED MAIL

WITHOUT PREJUDICE

June 22<sup>nd</sup>, 2018

Mr David T. Ullmann
BLANEY MCMURTRY LLP
2 Queen Street East, Suite 1500
Toronto, Ontario M5C 3G5

RE: 4362063 Canada Ltd. (f/k/a Domfoam International Inc.) and

Domfoam Inc. (f/k/a 8032858 Canada Inc.)

O/f: 23306-006

Sir:

As you may recall, the undersigned is acting as legal counsel for Domfoam Inc.

We have been informed by Mr. Don Perelman, from the firm Fine Kaplan & Black R.P.C. in Philadelphia, that a payment of approximately US \$3,000,000 has been sent to your attention in the name of Domfoam International Inc. as the first payment from the Dow Settlement - Urethane Antitrust Litigation Settlement.

As you are aware, effective as of the 26<sup>th</sup> day of March 2012, our client, Domfoam Inc., has purchased from your client 4362063 Canada Inc., all its assets, undertakings and properties of every nature and kind whatsoever (except for the Excluded Assets) (the "APA") pursuant to a Sale Approval and Vesting Order granted by Justice Brown of the Ontario Superior Court of Justice, Commercial list, in the Court file No. CV-12-9545-00CL, dated March 16<sup>th</sup>, 2012 and further to which an assignment agreement and a general conveyance and assumption of assumed liabilities have been executed on March 26<sup>th</sup>, 2012 between our respective clients, *inter alia*.

June 22<sup>nd</sup>, 2018 Our file: 23306-006

Please be advised that any such payment received by you from the Dow Settlement – Urethane Antitrust Litigation Settlement belongs to our client and any such payment shall be immediately forwarded to the undersigned in respect of the APA that intervened between our respective clients.

If any such payment has been partially or totally distributed, you are hereby required to inform the recipient that the funds distributed are the property of our client and they shall be immediately directed to return such funds to the undersigned, in accordance with the APA.

As per Section 7.4 of the APA, please be advised of the new coordinate of the undersigned as follow:

Mtre. Jacques Vincent Laframboise Gutkin, g.p. 2000 - 1 Westmount Square Montreal (Quebec) H3Z 2P9

Tel.:

+1 (514) 938-1924 ext. 4634

Fax:

+1 (514) 938-3676

Email:

i.vincent@laframboise-gutkin.com

Also, please be advised that we have notified Mr. Don Perelman and the Dow Settlement - Urethane Antitrust Litigation Settlement's administrator that our client shall be the beneficiary of any further payment to be made by such group. You will find enclosed (attached) copy of the notice sent to Mr. Perelman earlier today.

If you have any question, please do not hesitate to contact the undersigned.

We will expect to receive the payment mentioned above within the next five (5) business days.

In the meantime, please accept our best regards.

LAFRAMBOISE GUTKIN G.P.

by:

JACQUES VINCENT, lawyer

JV/pac Encl.

•		

De:

Jacques Vincent

Α:

dperelman@finekaplan.com

Cc:

Stephanie.Amin-Giwner@gardencitygroup.com; michelle.smith@chooseoca.com; nblakeslee@finekaplan.com;

Chapdelaine Camille (c.chapdelaine@laframboise-gutkin.com)

Cci:

Paula Andrea Castellino (LG)

Objet:

Domfoam International Inc. - The Urethane Antitrust Litigation Settlement (#23306-006)

Date:

22 juin 2018 16:08:00

Pièces jointes :

20180622 Notice to Urethane Antitrust Litigation Settlement (s).pdf

image001.jpg

Asset Purchase Agreement between 4037057 Domfoam March 8 2012 (with schedules).pdf

Approval and Vesting Order 20120316 (s).pdf

K Assignment of APA vExec (s).pdf General Conveyance 20120326 (s).pdf

Importance:

Haute

#### Sir:

As per your exchange of emails and telephone conversations with my colleague Camille Chapdelaine, please find attached the following documents:

- Notice Assignment of Claims Domfoam International Inc.
- Asset Purchase Agreement dated March 8, 2012
- Approval and vesting order dated March 16, 2012
- Assignment Agreement dated March 26, 2012
- General Conveyance dated March 26, 2012

If you have any question, please do not hesitate to contact the undersigned.

Please accept our best regards.

#### Me JACQUES VINCENT

Associé | Partner

#### Laframboise Gutkin Cabinet d'avocats



LAFRAMBOISE GUTKIN. CABINET D'AVOCATS - LAW FIRM /S.E.N.C.

1, WESTMOUNT SQUARE, 20e ÉTAGE

MONTRÉAL QC H3Z 2P9

COURRIEL:

i.vincent@laframboise-gutkin.com

TELEPHONE: (514) 938-1924 Ext. 4634

FAX:

(514) 938-3676

WWW.LAFRAMBOISE-GUTKIN.COM

Veuillez noter que, sauf avis contraire du correspondant ou circonstance particulière, le courriel est considéré par le cabinet comme un moyen adéquat de communication, au même titre que le courrier ordinaire. Le présent courriel est destiné uniquement au(x) destinataire(s) susmentionné(s). Son contenu est confidentiel et relève du secret professionnel de l'avocat. Si vous avez reçu cette communication par erreur, veuillez nous en aviser immédiatement et effacer l'original, sans en tirer de copie, en dévoiler le contenu ni prendre quelque mesure fondée sur celui-ci.

Please note that, save a notice to the contrary by the recipient or special circumstances, the use of e-mail is considered by the firm as an adequate means of communication, equivalent to regular mail. The present message is intended only for the named recipient(s) above. Its contents are confidential and protected by the solicitor-client privilege. If you have received this message in error, please notify us immediately and delete the original without making a copy, disclosing its contents or taking any action based thereon.

#### Notice

#### . Assignment of Claims

RE: Domfoam International Inc.

Montreal (Canada), June 22, 2018.

Mr. Don Perelman
FINE KAPLAN & BLACK R.P.C.

1 South Broad Street
23<sup>rd</sup> Floor
Philadelphia, PA 19107
USA

Sir:

We are acting as legal counsels for Domfoam Inc. (f/k/a 8032858 Canada Inc.). (the "Purchaser").

On March 8, 2012, an asset purchase agreement intervened between 4037057 Canada Inc. (the "Assignor") and 4362063 Canada Ltd. (f/k/a *Domfoam International Inc.*, hereinafter the "Vendor") (the "APA"), with respect to all assets, undertakings and properties of the Vendor, of every nature and kind whatsoever and wherever situated (the "Purchased Assets"), save and except the "Excluded Assets", as listed in Schedule 2.2. of the APA.

On March 16, 2012, the terms and provisions of the APA and the purchase and sale transaction provided for therein and all of the Vendor's right, title and interest in and to the Purchased Assets vested in the Assignor, have been approved and authorized by an Order of Justice Brown (the "Approval and Vesting Order"), from the Ontario Superior Court of Justice and the Purchased Assets were thereafter vested in the Assignor.

On March 26, 2012, the Purchaser, pursuant to an assignment agreement amongst the Assignor, the Vendor and the Purchaser (the "Assignment"), became the beneficiary of all interest of the Assignor under the APA, including all rights of action or other rights accruing to the Assignor, or which might after the Assignment takes effect or accrue to the Assignor under the APA.

Concurrently on March 26, 2012, a general conveyance and assumption of assumed liabilities was executed between the Vendor and the Purchaser (the "Conveyance") by which, *inter alia*, the Vendor conveyed, transferred and assigned to the Purchaser and the Purchaser did accept, all of the right, title and interest of the Vendor in and to the Purchased Assets.

For further reference, you will find attached copies of the following executed documents:

- 1. The APA;
- 2. The Approval and Vesting Order;
- 3. The Assignment; and
- 4. The Conveyance.

Therefore and in consideration of the APA, as approved by the Approval and Vesting Order and as assigned under the Assignment, be advised that any payment from the Urethane Antitrust Litigation Settlement that would have been payable to the Vendor shall be considered as part of the Purchased Assets and shall thereon be payable to the Purchaser and addressed to the undersigned as follow:

Cheque payable to:

Domfoam Inc.

Sent to:

a/s Mtre. Jacques Vincent, lawyer 2000 – 1 Westmount Square Montreal (Quebec) H3Z 2P9 Canada

PLEASE ACT AND GOVERN YOURSELF ACCORDINGLY.

LAFRAMBOISE GUTKIN G.P.

per:

JACQUES VINCENT, lawyer

Encl.

Cc Mrs Stéphanie Amine-Giwner (by email and Registered Mail)
Mrs Michele Smith (by email and Registered Mail)

	·	
•		

#### ASSET PURCHASE AGREEMENT

Made as of the 8th day of March, 2012

Between

4037057 CANADA INC.

- and -

DOMFOAM INTERNATIONAL INC.

#### ASSET PURCHASE AGREEMENT

This Agreement is made as of the 8th day of March, 2012,

#### BETWEEN:

4037057 CANADA INC., a corporation existing under the laws of Canada

(hereinafter referred to as the "Purchaser")

- and -

DOMFOAM INTERNATIONAL INC., a corporation existing under the laws of Canada

(hereinafter referred to as the "Vendor")

#### RECITALS

- A. The Vendor carries on the Business and is willing to sell the Purchased Assets to the Purchaser;
- B. The Vendor has commenced proceedings in the Ontario Superior Court of Justice of the judicial district of Toronto (the "CCAA Court") under the Companies' Creditors Arrangement Act (Canada) (the "CCAA") pursuant to which, among other things, Deloitte has been appointed as the monitor (the "Monitor"); and
- C. The Vendor has agreed to sell and transfer and assign to the Purchaser and, subject to the issuance of the Approval Vesting Order, the Purchaser has agreed to purchase from the Vendor, subject to the conditions contained herein, all of the Vendor's interest in and to the Purchased Assets;

NOW THEREFORE this Agreement witnesses that in consideration of the mutual covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged by each party to the other, the parties agree as follows:

#### Section 1 - INTERPRETATION

#### 1.1 Definitions

In this Agreement:

- (a) "Agreement" means this asset purchase agreement:
- (b) "Allocation of Purchase Price" has the meaning set out in Section 2.12;
- (c) "Approval and Vesting Order" means an Order of the CCAA Court, in form and content satisfactory to both the Purchaser and the Vendor, acting reasonably, which approves the

Vendor entering into this Agreement and the Transactions contemplated hereunder and and vesting the Purchased Assets in the Purchaser free and clear of all liens, charges, pledges, security interests and other encumbrances;

- (c.1) "Assigned Contracts" has the meaning set out in Section 2.3:
- (d) "Assumed Employees" has the meaning set out in Section 2.5;
- (e) "Assumed Obligations" has the meaning set out in Section 2.6;
- (e.1) "BASF Receivables" has the meaning set out in Section 2.9:
- "Books and Records" means all books and records and the data contained therein, whether in paper or electronic form, in the possession of the Vendor relating to the Purchased Assets or the Assumed Obligations (other than accounting records and books and records required by applicable law to be retained by the Vendor, copies of which shall be provided to the Purchaser at the Vendor's cost), including all policies, plans, audit reports, all personnel, employment, payroll and training records relating to the Assumed Employees, customer lists, Customer Contracts, sales records, price lists, and all other related files, records and information in the possession of the Vendor relating to the Purchased Assets or the Assumed Obligations, but, for certainty, not including (i) such books, records and data relating to any of the Excluded Assets or the Excluded Obligations or (ii) the corporate minute books of the Vendor or any of its Subsidiaries;
- (g) "Business" means collectively the business, affairs and operations of the Vendor, which for greater certainty, excludes the Subsidiaries businesses; the Vendor is in the business of manufacturing and selling polyurethane foam in Canada and, to a minimum extent, in the United States of America, for the bedding, carpet underlay, furniture and industrial industries;
- (h) "Business Day" means a day on which banks are open for business in the City of Toronto but does not include a Saturday, Sunday or statutory holiday in the Province of Ontario;
- (i) "CCAA" has the meaning set out in the recitals hereof;
- (j) "CCAA Court" has the meaning set out in the recitals hereof:
- (k) "Closing" means the successful completion of the Transaction;
- (l) "Closing Date" means March 26<sup>th</sup>, 2012 or such other date as the parties may agree to in writing;
- (m) "Contracts" means the agreements as described in Schedule 1.1(m), as the same may be modified by the Purchaser from time to time pursuant to Section 2.8;
- (n) "Customer Contracts" means the Vendor's contracts, agreements, instruments and other legally binding commitments or arrangements, written or oral, concerning the provision of services by the Vendor to its customers including but not limited to those contracts listed in Schedule 1.1(n) attached hereto;
- (o) "Deposit" has the meaning set out in Section 2.11(a);

- (p) "DIP Financing" has the meaning set out in Section 5.1(n);
- (q) "Email" has the meaning set out in Section 7.4;
- (r) "Equipment" means all of the Vendor's equipment (including the Vendor's computer and computer accessories), office equipment, furniture, fixtures, furnishings and supplies of all kinds in the possession or control of the Vendor on the Closing Date, whether owned by the Vendor or in which the Vendor has an interest pursuant to an equipment operating lease or a security agreement;
- (r.1) "Equipment Operating Leases" means all operating leases of equipment or other personal property used by the Vendor as a lessee or under which the Vendor has rights as lessee and listed on Schedule 1.1(r.1) attached hereto;
- (s) "ETA" means the Excise Tax Act (Canada);
- (t) "Excluded Assets" has the meaning set out in Section 2.2;
- (u) "Excluded Employees" has the meaning set out in Section 2.5;
- (v) "Excluded Obligations" has the meaning set out in Section 2.7;
- (w) "Initial Order" means the Order of the CCAA Court obtained on January 12, 2012 in the CCAA Court file no. CV-12-9545-00CL;
- (x) "Intellectual Property" means the following intellectual property:
  - (i) all designs, graphics, slogans and other commercial symbols and all registrations and applications therefor;
  - (ii) all patents, patent rights (including divisions, reissues, renewals, re-examinations, continuations, continuations in part and extensions) and all applications therefor;
  - (iii) all copyrights, writings and other copyrightable works of authorship, including computer programs, databases and documentation therefor, integrated circuit topographies, industrial designs and other industrial property rights and all applications and registrations therefor and all renewals or extensions of such applications and registrations;
  - (iv) all proprietary and non-public business information, including know-how, trade secrets, improvements, concepts, ideas, technical data, drawings, specifications therefor, business methodologies and processes, confidential information and any licensed property or technology; and
  - (v) all goodwill and trademarks in whatever format, including without limitation, registered and unregistered trademarks, trade names, brand names, service marks, logos, copyrights, certification marks, drawings, permits, internet and electronic email addresses, URLs, telephone, telex and facsimile numbers, content of websites and domain names related to or connected with the business carried on by the Vendor, all related software and electronic code to the extent assignable and other similar intellectual property or intangibles of the Vendor;

- (y) "Interim Period" means the period between the date of acceptance of this Agreement by the Vendor and the Closing Date;
- (z) "Leased Premises" means the premises leased and occupied by the Vendor at:
  - (i) 8785 Langelier Blvd., (Saint-Leonard) Montreal, Québec, H1P 2C9;
  - (ii) 5675 Des Grandes-Prairies Blvd., (Saint-Leonard) Montreal, Québec, H1R 1B3;
  - (iii) 7525 Henri Bourassa East, Montreal, Québec, H1E 1N9;
- (aa) deleted;
- (bb) "Monitor" has the meaning set out in the Recitals;
- (cc) "Objection Notice" has the meaning set out in Section 2.10;
- (dd) "Offer Date Inventory Value" has the meaning set out in Section 2.9;
- (ee) "Offer Date Receivable Value" has the meaning set out in Section 2.9;
- (ff) "Order" means any order, directive, judgment, decree, award or writ of any tribunal;
- (gg) "Purchase Price" has the meaning set out in Section 2.9:
- (hh) "Purchased Assets" means the right, title and interest of the Vendor in and to the assets described in Schedule 1.1(hh), provided that the Purchased Assets shall not include any Excluded Assets;
- (ii) "Purchased Inventories" has the meaning set out in Section 2.9;
- (jj) "Purchased Receivables" has the meaning set out in Section 2.9;
- (kk) "Purchased Working Capital" has the meaning set out in Section 2.10;
- (II) "QSTA" means an Act respecting the Quebec Sales Tax.
- (mm) Withdrawn;
- (nn) "Schedules" means any and all schedules referred to in this Agreement and attached thereto;
- (00) "Subsidiary(ies)" means each of (i) Valle Foam Industries (1995) Inc. having its registered office at 4 West Drive, Brampton (Ontario) L6T 2H7, and (ii) A-Z Sponge & Foam Products Ltd. having its registered office at 811 Cundy Ave., Annacis Island, Delta (British Columbia) V3M 5P6. Those corporations being referred to individually as a "Subsidiary" and collectively as "Subsidiaries";
- (pp) "Taxes" has the meaning set out in Section 2.13;
- (qq) "Tax Return" means a report, return or other information or form required to be supplied to a governmental entity with respect to any Taxes;

- (rr) "Time of Closing" means 2:00 p.m. (Toronto Time) on the Closing Date or such other time on the Closing Date as the parties may mutually agree;
- (ss) "Transaction" means the transaction of purchase, sale, assignment and assumption contemplated by this Agreement;
- (tt) "Unionized Employees" has the meaning set out in Section 2.5; and
- (uu) "Vendor's Intellectual Property" means all Intellectual Property used by the Vendor as at the Closing Date, including any Intellectual Property listed in Schedule 1.1(uu) attached hereto.

#### 1.2 Interpretation Not Affected by Headings, etc.

The division of this Agreement into sections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. The terms "this Agreement", "hereof", "herein" and "hereunder" and similar expressions refer to this Agreement and not to any particular section hereof.

#### 1.3 Extended Meanings

Words importing the singular include the plural and vice versa, words importing gender include all genders and words importing persons include individuals, partnerships, associations, trusts, unincorporated organizations, corporations and governmental authorities. The term "including" means "including, without limitation," and such terms as "includes" have similar meanings.

[Section 1.4 follows on next page]

#### 1.4 Schedules

The following schedules are incorporated in and form part of this Agreement:

Schedule 1.1(m)	Contracts
Schedule 1.1(o)	Customer Contracts
Schedule 1.1(r.1)	Equipment Operating Leases
Schedule 1.1(hh)	Purchased Assets
Schedule 1.1(uu)	Vendor's Intellectual Property
Schedule 2.2	Excluded Assets
Schedule 2.6	Assumed Obligations
Schedule 2.9(A)	Purchased Receivables
Schedule 2.9(B)	Purchased Inventories
Schedule 2.9(E)	Excess rebates to customers
Schedule 4.1(c)	Confidentiality Agreement
Schedule 5.1(o)	List of assets to be acquired from Valle Foam Industries (1995) Inc.

[Section 2 follows on next page]

#### 2.1 Sale and Purchase of Assets

Subject to the terms and conditions hereof, the Vendor shall sell to the Purchaser and the Purchaser shall purchase from the Vendor the Purchased Assets on the Closing Date. The Purchaser acknowledges that it is not purchasing any other property or assets of the Vendor other than the Purchased Assets. The Purchaser acknowledges and agrees that this Agreement contains a condition (not capable of waiver) that, prior to the Closing, the Vendor shall have obtained the Approval and Vesting Order, as provided in this Agreement. If the Vendor shall not have obtained the Approval and Vesting Order after making application for the same, this Agreement shall automatically terminate without any further action on the part of any of the Vendor or the Purchaser and neither party hereto shall be under any further obligation to the others to complete the Transactions and each party hereto shall be released immediately of all of its obligations hereunder, other than as set forth in Sections 6.6(b) (return of the Deposit), 4.1(c) (Confidentiality Agreement) and 7.3 (Cost and Expenses).

#### 2.2 Excluded Assets

The Purchased Assets shall not include (i) any assets of the Vendor other than the Purchased Assets or (ii) those assets of the Vendor that are listed or described in Schedule 2.2 on the date hereof and those assets of the Vendor which are added to such Schedule 2.2 by the Purchaser during the Interim Period (collectively, the "Excluded Assets"). If any of the Excluded Assets or any proceeds in respect thereof shall at any time come into the possession of or under the control of the Purchaser or any of its employees, officers or agents, such assets and proceeds shall be held by the Purchaser in trust for the benefit of the Vendor. Within fifteen (15) Business Days from the date on which the Purchaser, or any of its employees, officers or agents, comes into possession of or obtains control over any of such assets or proceeds, as applicable, the Purchaser shall (a) by notice in writing delivered to the Vendor and the Monitor in accordance with the provisions hereof, so advise the Vendor and the Monitor, and (b) forthwith account and deliver over to the Vendor or the Monitor any such assets or proceeds, less any reasonable costs incurred by the Purchaser in executing said obligations.

#### 2.3 Assignment and Assumption of Contracts

Subject to the conditions and terms hereof, the Vendor and the Purchaser agree to work cooperatively and to use their commercial best efforts to obtain the consents or approvals to the assignment of the Customer Contracts and the Contracts or any other contract that the Purchaser may require to be added to Schedule 1.1(m) during the Interim Period (collectively the "Assigned Contracts), as requested by the Purchaser acting reasonably, on terms and conditions satisfactory to the Purchaser. Except for any administration fee required to be paid to the person whose consent or approval has been requested in respect of any such Assigned Contract, the Vendor shall not be required to pay any amount or fee whatsoever in connection with the Vendor's covenant to attempt to obtain such consent or approval for such Assigned Contracts. The Purchaser agrees to supply all such information to the Vendor and the person whose consent or approval has been requested (including, without limitation, credit and financial information) as may be reasonably requested by the Vendor or the person whose consent or approval has been requested and to otherwise cooperate, acting reasonably, with the Vendor and the person whose consent or approval has been requested in connection with the foregoing. Notwithstanding any other provision of this Agreement, the failure of the Vendor or the Purchaser to have obtained any such consents or approvals requested by the Purchaser shall not entitle the Purchaser to terminate this Agreement or reduce the Purchase Price and shall not operate to release the Purchaser from any of its obligations hereunder.

To the extent the assignment of any of the Assigned Contracts assigned to the Purchaser pursuant to the provisions hereof shall require the consent or approval of any person and such consent or approval is not obtained at or prior to the Time of Closing, this Agreement shall not constitute a contract to assign such Assigned Contract if an attempted assignment would constitute a breach thereof. The Vendor shall co-operate with the Purchaser in any reasonable arrangement designed to provide the Purchaser with the benefit of such Assigned Contract including enforcement of any and all rights of the Vendor (if any) against the other party thereto arising out of any breach or cancellation thereof by such party or otherwise. Nothing contained herein shall be construed to negate or diminish, as between the Vendor and the Purchaser, the Vendor's covenants and obligations to transfer and deliver to the Purchaser the Purchased Assets as provided in this Agreement.

#### 2.4 "As is, Where is"

The Purchaser acknowledges that the Vendor is selling and the Purchaser is purchasing the Purchased Assets on an "as is, where is" basis as they shall exist on the Closing Date. The Purchaser further acknowledges that it has entered into this Agreement on the basis that the Vendor does not guarantee title to the Purchased Assets and that the Purchaser has conducted such inspections of the condition of and the title to the Purchased Assets as it has deemed appropriate and has satisfied itself with regard to these matters. Without limiting the generality of the foregoing, the Purchaser acknowledges that there are no representations, warranties, terms, conditions, undertakings or collateral agreements, express or implied, statutory or otherwise, with respect to the Purchased Assets or with respect to any other matter or thing whatsoever except as otherwise expressly stated herein, including as to existence, title, encumbrances, description, fitness for purpose and merchantability concerning the Purchased Assets. The description of the Purchased Assets contained in the Schedule 1.1(hh) is for the purpose of identification only. The Purchased Assets are being purchased as they exist at execution hereof and, except as provided in Section 2.10, no adjustment will be allowed to the Purchaser for any change in condition, value, or quality of the Purchased Assets.

#### 2.5 Employees

The Vendor shall provide to the Purchaser within five (5) Business Days of the acceptance of this Agreement a list of the names of the individuals of the Vendor that are full-time, part-time or casual employees or individuals engaged on contract to provide services to the Vendor together with all recruitment agencies' employees actually working for the Vendor. The Purchaser agrees that it shall offer employment to (i) all the unionized employees employed by the Vendor (the "Unionized Employees") and (ii) substantially all other employees of the Vendor other than those employees or individuals that the Purchaser has identified in writing to the Vendor ("the "Excluded Employees"), and in any event at least 60% of the non-unionized employees, five (5) Business Days prior to the Closing Date (the Unionized Employees and such other employee or individual who accepts the Purchaser's offer, the "Assumed Employees"), effective as at the Time of Closing, on terms and conditions of employment as may be agreed between the Purchaser and the Assumed Employees (except in the case of the Unionized Employees whose employment terms shall be governed by the applicable collective bargaining agreement). Until the Closing Date, the Vendor shall be responsible for all wages, outstanding claims for severance prior to the Closing Date, workers compensation contributions, any amount owed under the Pay Equity Act (R.S.Q., chapter E-12.001) and any regulation adopted under such Act, and other remuneration and benefits as may be payable, or related, to the employees of the Vendor. It is expressly understood that the Vendor shall not be responsible for any accrued vacaction which may be owing to the Assumed Employees at or before the Time of Closing. After the Closing Date, the Vendor shall remain responsible for all such amounts payable, or related to, the Excluded Employees.

#### 2.6 Assumed Obligations

In connection with the acquisition of the Purchased Assets, the following obligations and liabilities ("Assumed Obligations") shall be assumed by the Purchaser as of the Time of Closing:

- (a) all obligations related to the Assumed Employees for the period beginning immediately after the Time of Closing;
- (b) all obligations and liabilities of the Vendor as of the Time of Closing under the Customer Contracts and the Contracts:
- (c) all obligations and liabilities in connection with the Purchased Assets and related to the period after the Time of Closing;
- (d) such other liabilities as the Purchaser advises the Vendor prior to the Time of Closing that the Purchaser will assume;
- (e) those obligations listed on Schedule 2.6 hereto;
- (f) all obligation related to the Assumed Employees for any outstanding accrued vacation pay outstanding as at the Time of Closing.

The Purchaser shall indemnify and hold harmless the Vendor with respect to any claim which may, from time to time, be asserted against the Vendor relating to the Assumed Obligations after the Time of Closing and from any and all costs (including all reasonable legal costs on a solicitor and its own client basis), actions, losses, claims, damages and liabilities whatsoever which the Vendor may suffer or incur by virtue of the failure of the Purchaser to pay, perform and discharge the Assumed Obligations.

#### 2.7 Excluded Obligations

Other than the Assumed Obligations, the Purchaser shall not assume and shall not be liable for any other liabilities or obligations of the Vendor (the "Excluded Obligations").

#### 2.8 Purchaser's Right to Exclude

Notwithstanding anything to the contrary in this Agreement, the Purchaser may, at its option, exclude any of the Purchased Assets, other than any of the Equipment, from the Transaction prior to the Closing Date, whereupon such Purchased Assets shall be Excluded Assets, provided, however, that there shall be no corresponding reduction in the Purchase Price. For greater certainty, the Purchaser shall be entitled to exclude any of the Contracts described in Schedule 1.1(m) hereto (with the exception of the collective bargaining agreement relating to the Unionized Employees) and any of the Customer Contracts described in Schedule 1.1(n) hereto.

#### 2.9 Purchase Price

The Purchase Price for the Purchased Assets would be, calculated as described hereinafter, the aggregate amount of three million six hundred sixty two thousand nine hundred and seventy five dollars (\$3,662,975) plus the assumption of the Assumed Obligations, said Purchase Price to be established as of the Closing Date (the "Purchase Price").

The Purchase Price is divided as follow:

(A)	Purchased Receivables:	\$2,450,976
(B)	Purchased Inventories:	\$946,586
(C)	Withdrawn	
(D)	All other Purchased Assets:	\$300,000
(E)	Excess rebates to customers	(\$34,587)

For more clarity, the Purchase Price has been established on the following method of calculation as for the Purchased Receivables and the Purchased Inventories, and same method of calculation shall be used to establish the final amount of the Purchase Price as of the Closing Date:

#### (A) Purchased Receivables

As of January 31, 2012, the total amount of all the Vendor's accounts receivable (the "Purchased Receivables") is five million nine hundred ninety six thousand six hundred and ninety two dollars (\$ 5,996,692) (the "Offer Date Receivable Value"), and is detailed in Schedule 2.9(A) – Section 1.

The portion of the Purchase Price attributed to the Purchased Receivables is two million four hundred fifty thousand nine hundred and seventy six dollars (\$2,450,976) and is calculated as described in Schedule 2.9(A) – Section 2.

#### (B) Purchased Inventories

As of January 11, 2012 (and for Chemicals (majors) and Small Chemicals, as of January 31, 2012), the total amount of all the Vendor's inventory (raw material, work in progress and finished goods inventory) (the "Purchased Inventories") is three million fifty six thousand and eleven dollars (\$3,056,011) (the "Offer Date Inventory Value") in Schedule 2.9(B) – Section 1.

The portion of the Purchase Price attributed to the Purchased Inventories is nine hundred forty six thousand five hundred and eighty six dollars (\$946,586) and is calculated as described in Schedule 2.9(B) – Section 2.

#### (C) BASF Receivables

Withdrawn

#### (D) Other Purchased Assets

The Purchaser is ready to pay the Vendor the amount of three hundred thousand dollars (\$300,000) for all the other Purchased Assets.

#### (E) Excess rebates to customers

The Purchase Price shall be reduced by the total amount of rebates earned by the Vendor's customers for the calendar year 2011 and not already applied to the outstanding invoices for each said customer at the Closing Time. As of January 27, 2012, the excess rebates to customers total thirty four

#### 2.10 Settlement of the Purchase Price

The Vendor and the Purchaser agree that the Purchase Price shall be adjusted as for the Purchased Receivables and the Purchased Inventories (collectively, the "Purchased Working Capital") as follows:

#### (A) Re: Accounts receivable:

Subject to the limit established hereinafter in this paragraph, the Purchase Price payable in relation with the Purchased Receivables shall be adjusted downward or upward, as the case may be, as per the formula established in Schedule 2.9(A) — Section 2, in the event and to the extent that, on the Closing Date, the aggregate value of the Purchased Receivables shall be lower or higher, as the case may be, than the Offer Date Receivable Value. Nonetheless, if the total amount of the Purchased Receivables, as the total appears in the first column on page 4 of Schedule 2.9(A) - Section 1, is, as of the Closing Date, over five million nine hundred thousand dollars (\$5,900,000), then the difference between the total amount of the Purchased Receivables as of the Closing Date and the limit herein established shall be deducted from the calculation of the current portion of the Purchased Receivables for the Vendor as of the Closing Date.

#### (B) Re: Inventory:

The Purchase Price payable in relation with the Purchased Inventories shall be adjusted downward or upward, as the case may be, as per the formula established in Schedule 2.9(B) – Section 2, in the event and to the extent that, on the Closing Date, the aggregate value of the Purchased Inventories shall be lower or higher, as the case may be, than the Offer Date Inventory Value.

On or before the fifth (5) Business Day after the Closing Date, the Vendor shall deliver to the Purchaser a statement setting forth a detailed calculation of the Purchased Working Capital as of the Closing Date, with all the relevant supporting evidence, and prepared by the Vendor with the same parameters as used for the evaluation of the Purchase Price as for the Purchased Receivables and the Purchased Inventories as detailed in Schedules 2.9(A) and 2.9(B).

The value of the Purchased Inventories shall be based on a physical inventory count and valued on a basis consistent with the Vendor's costing methodologies used for the establishment of the Offer Date Inventory Value. The physical inventory count will be jointly supervised by the Vendor, the Purchaser and the Monitor.

The Purchaser may dispute the value of the Purchased Working Capital as presented by the Vendor by notice in writing (the "Objection Notice") given to the Vendor within ten (10) Business Days after the delivery of the Vendor's statement as to the amount of the Purchased Working Capital. Such Objection Notice shall specify, in detail, the basis for the dispute and the Vendor's adjustments to the value of the Purchased Working Capital shown by the Vendor in its statement.

If the Purchaser and the Vendor cannot agree on the value of the Purchased Working Capital within five (5) Business Days after the delivery of the Objection Notice, then either the Purchaser or the Vendor may make a motion to the CCAA Court to resolve this dispute.

Forthwith after agreement as to the amount of the Purchased Working Capital or forthwith upon the decision of the CCAA Court as to the amount of the Purchased Working Capital, then, as the case may be, the Vendor shall refund to the Purchaser such portion of the Purchase Price which is equal to the amount by which the respective Purchased Working Capital values are less than the respective Offer Date Receivable Value and Offer Date Inventory Value, or alternatively, the Purchaser shall pay to the Vendor, by way of certified cheque or wire transfer, such additional amount which is equal to the amount by which the respective Purchased Working Capital values are higher than the respective Offer Date Receivable Value and Offer Date Inventory Value.

The Monitor and the Vendor agree that they shall not make a motion to the CCAA Court for distribution of any funds received by either of them under this Agreement, or otherwise distribute any such funds, unless the Purchase Price adjustments contained in this Section have been completed and the covenant contained in Section 4.2(b) has been fulfilled. However, in the event that the Purchase Price adjustments have not been settled by the date that is ninety (90) days following the Closing Date, the Monitor shall be entitled to bring a distribution motion on appropriate notice to the Purchaser and the Vendor and each of the Purchaser, the Vendor and the Monitor shall be entitled to make submissions to the CCAA Court in this regard.

#### 2.11 Method of Payment

The Purchase Price shall be satisfied as follows:

- (a) a deposit in the amount of three hundred and sixty thousand dollars (\$360,000) (the "Deposit") by dank draft or certified cheque payable to the Monitor, in trust, to be held in trust by the Monitor until the Time of Closing and then credited towards the Purchase Price, or in accordance with any other provision herein;
- (b) the amount of three million one hundred and two thousand nine hundred and seventy five dollars (\$3,102,975) by certified cheque or wire transfer payable to the Monitor at the Time of Closing;
- (c) the amount of two hundred thousand dollars (\$200,000) (the "Holdback") to be paid no later than a sixty (60) days following the Closing Date; and
- (d) by the assumption of the Assumed Obligations.

The Purchaser shall have the ability to offset any portion of the Holdback against any downward adjustment of the Purchase Price per Section 2.10 herein. The Purchaser shall also have the ability to offset any portion of the Holdback against any claim made against the Purchaser from any of the Vendor's employees for any period up to the Closing Time. In such case, the Purchaser shall have the right to retain the Holdback as long as final judgments have not been rendered vis à vis such claim.

#### 2.12 Allocation of Purchase Price

The Purchaser and the Vendor shall, acting reasonably, agree on the allocation of Purchase Price (other than for the Purchased Receivables and the Purchased Inventories) on or before the Closing Date, and they shall execute and file any Tax Returns required on the basis of such allocation.

#### 2.13 Taxes

The Purchaser shall be liable for and shall pay any and all federal, provincial and other sales, goods and services, value added, and other transfer taxes which are properly payable by the Purchaser in connection with the transfer of the Purchased Assets by the Vendor to the Purchaser, together with all duties, registration fees or other charges properly payable by the Purchaser upon or in connection with the conveyance or transfer of the Purchased Assets (collectively, the "Taxes").

#### 2.14 Value Added Tax

- (a) The Purchaser shall be liable for and shall pay any Taxes which may be exigible under Part IX of the ETA;
- (b) The Vendor hereby declares that it is duly registered under Subdivision (d) of Division V of Part IX of the ETA;
- (c) The Vendor hereby declares that it is duly registered under the QSTA;
- (d) Without limiting the Purchaser's indemnity obligations under Section 2.13, the Vendor and the Purchaser will on the Closing Date jointly execute and deliver to each other an election, in the prescribed form and containing the prescribed information, pursuant to subsection 167(1) of the ETA and the corresponding provision of the QSTA so that no tax is payable under any such laws in respect of the sale and purchase of the Purchased Assets hereunder, and the Purchaser shall file any joint election in accordance with the ETA and QSTA.

#### 2.15 Section 22 Election — Accounts Receivable

The Purchaser and the Vendor agree to elect jointly in the prescribed form under Section 22 of the *Income Tax Act* (Canada) as to the sale of the accounts receivable and other applicable Purchased Assets described in Section 22 of the *Income Tax Act* (Canada) and the corresponding provision of the statute that is the provincial equivalent thereof and to designate in such election an amount equal to the portion of the Purchase Price allocated to such assets pursuant to Section 2.12 as the consideration paid by the Purchaser therefor.

#### 2.16 Risk of Loss

During the Interim Period, up to the Time of Closing, the Purchased Assets shall be and remain at the risk of the Vendor. If, prior to the Closing Date, all or substantially all of the Purchased Assets or any plant operated by the Vendor is destroyed or damaged by fire or any other casualty or shall be appropriated, expropriated or seized by any governmental entity or other lawful authority, the Purchaser shall have the option, at its absolute discretion, exercisable by notice in writing given within fifteen (15) Business Days after the Purchaser receives notice in writing from the Vendor of such destruction, damage, appropriation, expropriation or seizure:

(a) In the event of damage or destruction, to complete the Transaction subject to a reduction of the Purchase Price to be mutually agreed to by the parties;

- (b) In the event of appropriation or expropriation, to complete the Transaction without reduction of the Purchase Price, in which event all proceeds of insurance or compensation for appropriation, expropriation or seizure shall be payable to the Purchaser and all right and claim of the Vendor to any such amounts not paid on the Closing Date shall be assigned to the Purchaser; or
- (c) In either event, to terminate this Agreement, receive a refund of the Deposit and not complete the Transaction, in which case all obligations of the Purchaser shall terminate forthwith upon the Purchaser giving notice as required herein.

[Section 3 follows on next page]

#### 3.1 Purchaser's Representations

The Purchaser represents and warrants to the Vendor that:

- (a) the Purchaser is a corporation duly incorporated, organized and subsisting under the laws of Canada;
- (b) the Purchaser has all necessary corporate power, authority and capacity to enter into this Agreement and to perform its obligations and the execution and delivery of this Agreement and the consummation of the transactions contemplated herein and have been duly authorized by all necessary corporate action on the part of the Purchaser;
- (c) the Purchaser is not a party to, bound or affected by or subject to any indenture, agreement, instrument, charter or by-law provision, order, judgment or decree which would be violated, contravened or breached by the execution and delivery by it of this Agreement or the performance by it of any of the terms contained herein;
- (d) this Agreement and all other documents contemplated hereunder to which the Purchaser is or will be a party have been or will be, as at the Time of Closing, duly and validly executed and delivered by the Purchaser and constitutes or will, as at the Time of Closing, constitute legal, valid and binding obligations of the Purchaser, as the case may be, enforceable in accordance with the terms hereof or thereof;
- (e) the Purchaser is not a non-Canadian person as defined in the Investment Canada Act; and
- (f) the Purchaser is or will be prior to Closing Date registered under Part IX of the ETA and under the QSTA.

#### 3.2 Representations and Warranties of the Vendor

The Vendor represents and warrants to the Purchaser as follows and acknowledges that the Purchaser is relying thereon in completing this Agreement:

- (a) the Vendor is a corporation duly incorporated, organized and subsisting under the laws of Canada;
- (b) the Vendor has all necessary corporate power, authority and capacity to enter into this Agreement and to perform its obligations and the execution and delivery of this Agreement and the consummation of the transactions contemplated herein and have been duly authorized by all necessary corporate action on the part of the Vendor;
- (c) the Vendor is not a party to, bound or affected by or subject to any indenture, agreement, instrument, charter, by-law provision, order, judgment or decree which would be violated, contravened or breached by the execution and delivery by it of this Agreement or the performance by it of any of the terms contained herein;

- (d) this Agreement and all other documents contemplated hereunder to which the Vendor is or will be a party have been or will be, as at the Time of Closing, duly and validly executed and delivered by the Vendor and constitutes or will, as at the Time of Closing, constitute legal, valid and binding obligations of the Vendor, as the case may be, enforceable in accordance with the terms hereof or thereof; and
- (e) the Vendor is a registrant for the purposes of Part IX of the ETA.

### 3.3 Survival of Representations and Warranties

The representations and warranties of the Vendor shall survive the completion of the Transaction and shall continue in full force and effect for the benefit of the Purchaser (which term for the purposes of this Section 3.3 shall include the Purchaser's directors and officers) for a period of nine (9) months from the Closing Date.

The representations and warranties of the Purchaser shall survive the completion of the Transaction and shall continue in full force and effect for the benefit of the Vendor (which term for the purposes of this Section 3.3 shall include the Vendor's respective directors and officers) for a period of nine (9) months from the Closing Date.

[Section 4 follows on next page]

### 4.1 Covenants of the Purchaser

- (a) The Purchaser agrees that from and after the Time of Closing, it shall preserve the Books and Records actually received from the Vendor at the Time of Closing, and at the reasonable request of the Vendor, the Monitor or another party acting in a court-appointed capacity (such as a receiver, interim receiver, receiver and manager or bankruptcy trustee, or any of their respective authorized representative) (a "Requestor") from time to time, the Purchaser shall co-operate fully with the Requestor and make available to the Requestor in a reasonable timely fashion during normal business hours (i) such Books and Records, (ii) the premises of the Purchaser for the Requestor's review of the Books and Records, and (iii) at the Requestor's expense, the personnel of the Purchaser, in each case to the extent the Requestor reasonably consider necessary (A) to permit the Requestor to prepare and file necessary Tax Returns and any other statutory documents relating to the operation by the Vendor of its business up to the Time of Closing, (B) to properly fulfil the Vendor's obligations in bankruptcy, as the case may be, (C) for the preparation by any Requestor of any Tax Returns or in connection with any assessment, reassessment or other form of document assessing liability of the Vendor for taxes under applicable tax legislation in respect of any taxation year, (D) in connection with the restructuring proceedings, or (E) to ensure compliance by the Vendor with applicable law;
- (b) The Purchaser agrees that from and after the Time of Closing, it shall permit the Requestor to make copies, at the Requestor's expense, of the Books and Records actually received by the Purchaser from the Vendor at the Time of Closing and shall provide reasonable assistance to the Requestor, at the Requestor's expense, to prepare T4 forms and records of employment for the employees of the Vendor who accept offers of employment from the Purchaser. In the case of the Books and Records actually received from the Vendor at the Time of Closing, the Purchaser agrees to preserve same for a minimum of 6 years from the Closing Date;
- (c) Until the Time of Closing, and in the event of the termination of this Agreement without consummation of the Transaction, the Purchaser shall keep confidential any information obtained from the Vendor in accordance with the confidentiality agreement executed between S.P. Holdings or Nominee and the Vendor on November 25, 2011 (the "Confidentiality Agreement") and attached hereto as Schedule 4.1(c) as if the Purchaser was a party thereto, and the Purchaser hereby agrees in favour of the Vendor to be bound by the terms of such confidentiality agreement as if it were an original signatory thereto. If this Agreement is terminated without consummation of the Transaction contemplated, promptly after termination, all documents, work papers and other written material obtained in connection with this Agreement and not made public (including all copies), shall be returned to the Vendor.

### 4.2 Covenants of the Vendor

(a) The Vendor shall allow the Purchaser to have reasonable access to the Vendor's management team, on a daily basis, and the Vendor's management team shall answer Purchaser's questions about the operation of the Business, the management of the Business and the decisions to be taken by the Vendor's management team during the Interim Period;

- (c) The Vendor shall use its best efforts to obtain such releases as may be required by the Purchaser, acting reasonably, regarding the termination of management employees, a list of which shall be provided by the Purchaser no later than ten (10) Business Days prior to the Closing Date; and
- (d) The Vendor agrees that during the Interim Period, it shall continue to conduct its Business in the ordinary course and, without limiting the generality of the foregoing, continue to serve its customers on a daily and normal basis without interruption, purchase inventory to satisfy the needs of the Business and deliver the finished goods on the timely manner as done in the past year.

[Section 5 follows on next page]

### Section 5 – CONDITIONS

### 5.1 Conditions - Purchaser

The obligation of the Purchaser to complete the Transaction is subject to the following conditions being fulfilled or performed by the Vendor or prior to the Time of Closing:

- (a) all representations and warranties of the Vendor contained in this Agreement shall be true as of the Closing Date with the same effect as though made on and as of that date;
- (b) deleted;
- (c) Withdrawn;
- (d) the Vendor shall not be bankrupt as at the Time of Closing;
- (e) the obtaining of the Approval and Vesting Order to the entire satisfaction of the Purchaser;
- (f) the Vendor shall have performed its obligations under this Agreement to the extent required to be performed on or before the Closing Date;
- (g) no action or proceedings shall be pending or threatened to restrain or prohibit the completion of the Transaction;
- (h) Withdrawn;
- (i) the Vendor shall not have transferred any inventory or equipment to any of its Subsidiaries, shall not have assigned any of its Contracts to its Subsidiaries or have sold to its Subsidiaries any of the Purchased Assets without having obtained from the Purchaser a prior written approval of said transaction which the Purchaser may refuse at its complete discretion;
- (j) Withdrawn;
- (k) the Vendor shall have paid all rent due under the Lease for the period from January 12, 2012 up to and throughout the Interim Period;
- (1) the Vendor shall have filed, before the Closing Date, articles of amendment to change its corporate name "Domfoam International Inc." to a numbered corporation, and shall have sought an order of the Court authorizing the change of name required herein if such Order is determined to be necessary by the Purchaser, and shall provide to the Purchaser, at the Time of Closing, (a) with a copy of the certificate of amendment evidencing said change and (b) a written authorization for the Purchaser to use said corporate name and any derivative of it;
- (m) deleted;
- (n) Withdrawn;
- (o) In the event that any or all of the assets of Valle Foam Industries (1995) Inc., listed in Schedule 5.1(o) are not, as at the Closing Date, subject to an agreement of purchase and sale and otherwise remain the assets of Valle Foam Industries (1995) Inc., the Vendor

(p) the obtaining on or before the date which is fifteen (15) Business Days from acceptance of this Agreement by the Vendor of a commitment letter from the actual general manager of the Vendor (John C. Howard) on the terms and conditions of a three (3) year employment agreement with the Purchaser.

The foregoing conditions are for the exclusive benefit of the Purchaser. Any such condition (except 5.1(e)) may be waived by the Purchaser in whole or in part. Any such waiver shall be binding on the Purchaser only if made in writing.

### 5.2 Conditions - Vendor

The obligation of the Vendor to complete the Transaction is subject to the following conditions being fulfilled or performed at or prior to the Time of Closing:

- (a) all representations and warranties of the Purchaser contained in this Agreement shall be true as of the Closing Date with the same effect as though made on and as of that date;
- (b) the Purchaser shall have performed each of its obligations under this Agreement to the extent required to be performed on or before the Closing Date; and
- (c) no action or proceedings shall be pending or threatened to restrain or prohibit the completion of the Transaction contemplated by this Agreement.

The foregoing conditions are for the exclusive benefit of the Vendor. Any such condition may be waived by the Vendor in whole or in part. Any such waiver shall be binding on the Vendor only if made in writing.

### 5.3 Mutual Conditions

The obligations of the Vendor on the one hand, and the Purchaser on the other hand are subject to the conditions that:

- the Approval and Vesting Order shall have been made by the CCAA Court on or before the 28<sup>th</sup> day of March, 2012 (or such later date as the parties may agree to, acting reasonably); and
- (b) the Approval and Vesting Order shall not have been stayed, varied or vacated, and no order shall have been issued which restrains or prohibits the completion of the Transaction.

The parties hereto acknowledge that the foregoing conditions are for the mutual benefit of the Vendor and the Purchaser.

### 5.4 Actions to Satisfy Closing conditions

Each of the parties agrees to diligently take all actions as are within its power to control, and to use commercially reasonable efforts to cause other actions to be taken which are not within its power to control, so as to ensure compliance with the conditions set forth in sections 5.1, 5.2 and 5.3 and, generally, the conditions set forth in this Agreement.

### 5.5 Non-Satisfaction of Conditions

If any condition set out in this Section 5 is not satisfied or performed on or prior the Closing Date, or at any date provided for herein, then the Vendor or the Purchaser, depending whose benefit the condition is inserted, may:

- (a) waive compliance with the condition in whole or in part in its sole discretion by written notice to the other party and without prejudice to any of its rights of termination in the event of non-fulfilment of any other condition in whole or in part; or
- (b) choose to give the other party an additional delay for said condition to be satisfied or performed; or
- (c) elect on written notice to the other party to terminate this Agreement before Closing.

[Section 6 follows on next page]

### 6.1 Closing

The completion of the Transaction shall take place on the Closing Date at the Time of Closing at a location to be determined by the mutual agreement of the parties in writing or at such other time as otherwise determined by mutual agreement of the parties in writing.

### 6.2 Purchaser's Deliveries on Closing

At or before the Time of Closing, the Purchaser shall execute and deliver to the Vendor the following, each of which shall be in form and substance satisfactory to the Vendor, acting reasonably:

- (a) payment of the portion of the Purchase Price payable at the Time of Closing;
- (b) payment or evidence of payment of applicable federal and provincial Taxes or alternatively, appropriate exemption certificates, as required by this Agreement;
- (c) general conveyance and assumption of liabilities agreement pursuant to Subsection 2.6 duly executed by the Purchaser, in form and substance mutually satisfactory to the Vendor and the Purchaser;
- (d) the Allocation of Purchase Price pursuant to Section 2.12:
- (e) an assignment or assignments of Intellectual Property agreements, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Purchaser;
- (f) an assignment or assignments of Customer Contracts and Contracts, in form and substance mutually satisfactory to the Purchaser and the Vendor, duly executed by the Purchaser;
- (g) an assignment of leases for the Leased Premises, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Purchaser;
- (h) a certificate, dated the Closing Date, confirming that all of the representations and warranties of the Purchaser contained in this Agreement are true as of the Closing Date, with the same effect as though made on and as of the Closing Date;
- (i) an acknowledgement dated the Closing Date, that each of the conditions precedent in Section 5.1 of this Agreement have been fulfilled, performed or waived as of the Closing Date; and
- (j) such further and other documentation as is referred in this Agreement or as the Vendor may reasonably require to give effect to this Agreement including pursuant to Sections 2.2, 2.6 and 2.8.

## 6.3 Vendor's Deliveries on Closing

At or before the Time of Closing, the Vendor shall execute and deliver to the Purchaser the following, each of which shall be in form and substance satisfactory to the Purchaser, acting reasonably:

- (a) a bill of sale pursuant to which the Vendor shall assign, sell, transfer and convey all of its right, title and interest in and to its Purchased Assets, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- (b) an assignment or assignments of Intellectual Property agreements, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- an assignment of leases for the Leased Premises, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- (d) an assignment or assignments of Customer Contracts and Contracts, in form and substance mutually satisfactory to the Purchaser and the Vendor, duly executed by the Vendor;
- (e) the Approval and Vesting Order in form and substance satisfactory to the Purchaser, acting reasonably;
- (f) a certificate, dated the Closing Date, confirming that all of the representations and warranties of the Vendor contained in this agreement are true as of the Closing Date, with the same effect as though made on and as of the Closing Date;
- (g) an acknowledgement dated the Closing Date, that each of the conditions precedent in Section 5.2 of this Agreement have been fulfilled, performed or waived as of the Closing Date; and
- (h) such further and other documentation as is referred to in this Agreement or as the Purchaser may reasonably require to give effect to this Agreement which shall be prepared by the Purchaser at its sole expense and in form and substance satisfactory to the Vendor, acting reasonably.

### 6.4 Possession of Assets

The Vendor shall remain in possession of the Purchased Assets until the Time of Closing. On Closing, the Purchaser shall take possession of the Purchased Assets which shall be located in the Leased Premises. In no event shall the Purchased Assets be sold, assigned, transferred or set over to the Purchaser until the conditions set out in the Approval and Vesting Order have been satisfied and the Purchaser has satisfied all delivery requirements outlined in Section 6.2.

### 6.5 Access to Assets

deleted

### 6.6 Termination

If either the Vendor on the one hand, or the Purchaser on the other hand, validly terminates this Agreement pursuant to the provisions of Sections 5.5, or 2.16,

- (a) all the obligations of the Vendor and Purchaser pursuant to this Agreement shall terminate;
- (b) the Purchaser shall be entitled to have the Deposit and all the monies paid hereunder returned immediately with any interest earned thereon, as the case may be, but without deduction provided; and
- (c) neither party shall have any right to specific performance or other remedy against, or any right to recover damages or expenses from, the other.

6.7	[Break Fe	e] With	drawn
-----	-----------	---------	-------

[Section 7 follows on next page]

### Section 7 – GENERAL

### 7.1 Further Assurances

Each of the parties shall, from time to time after the Closing Date, at the request and expense of the other, promptly take or cause to be taken such action and execute and deliver or cause to be executed and delivered to the other such documents and further assurances as may be reasonably necessary to give effect to this Agreement.

### 7.2 Obligations to Survive

Notwithstanding the completion of the transactions contemplated hereunder or the delivery of documents pursuant to this Agreement, the obligations, covenants, representations and warranties of the parties hereto shall survive such completion and shall remain in full force and effect and shall not merge as a result thereof.

### 7.3 Costs and Expenses

Each party hereto shall be responsible for all costs and expenses (including the fees and disbursements of legal counsel, bankers, investment bankers, accountants and other advisors) incurred by it in connection with this Agreement and the transactions contemplated herein.

### 7.4 Notice

Any notice or other communication under this Agreement shall be in writing and may be delivered personally or transmitted by fax or electronic mail ("Email"), addressed to:

in the case of the Purchaser, as follows:

4037057 CANADA INC. 8600 Decarie Blvd. Suite 200 Mont-Royal, QC H4P 2N2

Attention: Terry Pomerantz Telephone No.: (514) 341-8600

Fax No.: (514) 340-1389

Email: tpomerantz@tramsmgmt.com

With a copy to:

Lamarre Perron Lambert Vincent g.p. 480 Saint-Laurent Blvd. Suite 200 Montreal, QC H2Y 3Y7

Attention: Jacques Vincent
Telephone No.: (514) 798-0990
Fax No.: (514) 798-5599
Email: j.vincent@lplv.com

and in the case of the Vendor, as follows:

### DOMFOAM INTERNATIONAL INC.

8785, boul. Langelier Montréal, QC H1P 2C9

Attention:

Tony Vallecoccia

Telephone No.:

(514) 325-8120

Fax No.:

(514) 325-6477

Email:

tvallecoccia@vallefoam.com

with a copy to:

Minden Gross LLP 145 King Street West Suite 2100 Toronto, ON M5H 4G2

Attention:

Raymond M. Slattery

Telephone No.:

(416) 369-4149

Fax No.:

(416) 864-9223

Email:

rslattery@mindengross.com

and to:

the Monitor

### DELOITTE & TOUCHE INC.

181 Bay Street Suite 1400 Toronto, ON M5J 2V1

Attention: Telephone No.: Robert J. Bougie (416) 601-4509

Fax No.:

(416) 601-6690

Email:

rbougie@deloitte.ca

with a copy to:

Thornton Grout Finnigan LLP 100 Wellington Street West Suite 3200

Toronto, ON, M5K 1K7

Attention: Telephone No.: Grant B. Moffat (416) 304-0599

Fax No.:

(416) 304-1313

Email:

gmoffat@tgf.ca

Any such notice or other communication, if given by personal delivery, will be deemed to have been given on the day of actual delivery thereof and, if transmitted by fax or Email before 5:00 p.m. (Toronto time) on a Business Day, will be deemed to have been given on the Business Day, and if transmitted by fax or Email after 5:00 p.m. (Toronto time) on a Business Day, will be deemed to have been given on the Business Day after the date of the transmission.

### 7.5 Time

Time shall, in all respects, be of the essence hereof, provided that the time for doing or completing any matter provided for herein may be extended or abridged by an agreement in writing signed by the Vendor and the Purchaser or by their respective solicitors.

### 7.6 Currency

Except where otherwise indicated, all references herein to money amounts are in Canadian currency.

### 7.7 Benefit of Agreement

This Agreement shall enure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns.

### 7.8 Third Party Beneficiaries

Each party hereto intends that this Agreement shall not benefit or create any right or cause of action in or on behalf of any person other than the parties hereto and their successors and permitted assigns, and no person, other than the parties hereto and their successors and their permitted assigns shall be entitled to rely on the provisions hereof in any action, suit, proceeding, hearing or other forum.

### 7.9 Entire Agreement

This Agreement and the attached Schedules constitute the entire agreement between the parties with respect to the subject matter and supersede all prior negotiations and understandings. This Agreement may not be amended or modified in any respect except by written instrument executed by the parties.

### 7.10 Paramountey

In the event of any conflict or inconsistency between the provisions of this Agreement, and any other agreement, document or instrument executed or delivered in connection with this Transaction or this Agreement, the provisions of this Agreement shall prevail to the extent of such conflict or inconsistency.

### 7.11 Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the Province of Quebec and the laws of Canada applicable therein and each of the parties irrevocably attorns to the non-exclusive jurisdiction of the courts of the Province of Quebec. Notwithstanding the foregoing, the parties acknowledge that the Approval and Vesting Order will be sought before the Ontario Superior Court in Toronto and that any disputes related thereto as it relates to court approval or the current Companies' Creditors Arrangement Act proceedings of the Vendor shall be determined by the said Ontario Court. The parties acknowledge that any disputes related to the interpretation of this Agreement (including but not limited to its interpretation or its execution) shall be sought before the Quebec Superior Court in Montreal.

### 7.12 Severability

If any provision of this Agreement or any document delivered in connection with this Agreement is partially or completely invalid or unenforceable, the invalidity or unenforceability of that provision shall not affect the validity or enforceability of any other provision of this Agreement, all of which shall be construed and enforced as if that invalid or unenforceable provision were omitted. The invalidity or unenforceability of any provision in one jurisdiction shall not affect such provisions validity or enforceability in any other jurisdiction.

### 7.13 Commission - No Broker

The Purchaser agrees to indemnify the Vendor against any claim for compensation or commission by any third party or agent retained by the Purchaser in connection with, or in contemplation of, the Transaction. It is understood and agreed that no broker, agent or other intermediary has acted for the Vendor in connection with the sale of the Purchased Assets and the Purchaser shall not be liable for any commission or other remuneration payable or alleged to be payable to any broker, agent or other intermediary who purports to act or have acted for the Vendor.

### 7.14 Assignment

Except as hereinafter provided, no party may assign this Agreement without the prior written consent of the other parties, which consent may not be unreasonably withheld or delayed. This Agreement enures to the benefit of and binds the parties and their respective heirs, executors, administrators, personal and legal representatives, successors and permitted assigns. Notwithstanding the foregoing, this Agreement shall be assignable by the Purchaser if the assignee is a "related entity" to the Purchaser within the meaning of that expression under the *Income Tax Act* (Canada). In the event of such assignment, the Purchaser shall remain liable to the Vendor in respect of its obligations hereunder.

### 7.15 Counterparts

This Agreement and any amendment, supplement, restatement or termination of any provision of this Agreement may be executed and delivered in any number of counterparts, each of which when executed and delivered is an original but all of which taken together constitute one and the same instrument. A party's transmission by facsimile or by Email of a copy of this Agreement duly executed by that party shall constitute effective delivery by that party of an executed copy of this Agreement to the party receiving the transmission. A party that has delivered this Agreement by facsimile or by Email shall forthwith deliver an originally executed copy to the other party or parties.

### 7.16 Language

The parties acknowledge that they have requested and are satisfied that the foregoing be drawn up in English. Les parties reconnaissent qu'elles ont exigé que ce qui précède soit redigé en anglais et s'en déclarent satisfaites.

### [THE REMAINDER OF THIS PAGE HAS BEEN LEFT INTENTIONALLY BLANK]

The Vendor shall indicate its acknowledgement of and its agreement with the foregoing by signing both copies of this Agreement in the space provided below and returning one fully executed copy to the Purchaser (which return may be made by fax or Email) to:

LPLV

c/o:

Mtre. Jacques Vincent

Fax:

(514) 798-5599

Email:

j.vincent@lplv.com

by no later than 3:00 pm, on March 9, 2012, after which time this Agreement shall be considered null and void as to the Purchaser.

IN WITNESS WHEREOF the Purchaser has duly executed this Agreement on March 8, 2012 at 5:30 p.m.

4037057 CANADA INC.

By:

Name: Terry Pomerantz

Title: President

I have authority to bind the Purchaser.

IN WITNESS WHEREOF the Vendor has duly executed this Agreement on March 7, 2012 at 13/41/2010

DOMFOAM INTERNATIONAL INC.

By:

Name: ANTONIO /A-LECS=C/A

Title: CFO

I have authority to bind the Vendor.

The Vendor shall indicate its acknowledgement of and its agreement with the foregoing by signing both copies of this Agreement in the space provided below and returning one fully executed copy to the Purchaser (which return may be made by fax or Email) to:

LPLV

c/o:

Mtre. Jacques Vincent

Fax:

(514) 798-5599

Email:

j.vincent@lplv.com

by no later than 3:00 pm, on March 9, 2012, after which time this Agreement shall be considered null and void as to the Purchaser.

By:

IN WITNESS WHEREOF the Purchaser has duly executed this Agreement on March 8, 2012 at 5:30 p.m.

4037057 CANADA INC.

Title: President

I have authority to bind the Purchaser.

Name: Terry Pomerantz

IN WITNESS WHEREOF the Vendor has duly executed this Agreement on March \_\_\_\_\_, 2012 at \_\_\_\_\_.

DOMFOAM INTERNATIONAL INC.

Ву:

Name:

Title:

I have authority to bind the Vendor.

## Schedule I.1(in)

### Contracts.

- 1. Subject to Section 2.3, the following Contracts shall form part of the Purchased Assets:
  - 1.1. All lease agreements in relation with the Leased Premises;
  - 1.2. Collective bargaining agreement with the Unionized Employees for the period of 2009-2014 (Teamsters Québec, Local 1999);
  - 1.3. All Equipment Operating Leases, to the extent that the Purchaser wishes to avail itself thereof and are not part of the Excluded Assets;
  - 1.4. Exclusive agreement for the Canadian use of specific chemical and formula for Gel Visco;
  - 1.5. All contractual arrangements with the suppliers to the extent that the Purchaser wishes to avail itself thereof;
  - 1.6. The annual Microsoft maintenance contract with PYA Solutions Inc.;
  - 1.7. Microsoft (Navision Licence agreement);
  - 1.8. Insurance policy re: Property and Liability;
  - 1.9. Insurance policy re: 2002 Ford;
  - 1.10. Coface (receivables);
  - 1.11. Insurance policy re: Directors and Officers liability;
  - 1.12. CN non freight leases for tracks; and
  - 1.13. Canadiens hockey tickets...

## Schedule 1.1(n).

### Customer Contracts

- 1. Subject to Section 2.3, the following Customer Contracts shall form part of the Purchased Assets:
  - 1.1. Customer bearing number #U11771 in the Books and Records;
  - 1.2. Customer beating number #11682 in the Books and Records:
  - 1.3. Customer bearing number #10005 in the Books and Records;
  - 1.4. Customer bearing number #10072 in the Books and Records;
  - 1.5. Customer bearing number #U12129 in the Books and Records;
  - 1.6: Customer bearing number #10020 in the Books and Records;
  - 1.7. Customer bearing number #10244 in the Books and Records!
  - 1.8. Customer bearing number #11665 in the Books and Records;
  - 1.9. Customer bearing number #11924 in the Books and Records;
  - 1.10. Customer bearing number #11164 in the Books and Records;
  - 1.11. Customer bearing number #10413 in the Books and Records;
  - 1.12. Customer bearing number #10585 in the Books and Records;
  - 1.13. Customer bearing number #U12164 in the Books and Records;
  - 1.14. Customer bearing number #11209 in the Books and Records;
  - 1.15. Customer bearing number #11982 in the Books and Records:
  - 1.16. Customer bearing number #UI1429 in the Books and Records;
  - 1.17. Customer bearing number #11177 in the Books and Records;
  - 1.18. Customer bearing number #11268 in the Books and Records;
  - 1.19. Customer bearing number #11848 in the Books and Records;
  - 1.20. Customer bearing number #00202 in the Books and Records;

DÓMFOAM INTERNATIONAL INC. LEASÉ COMMITMENTS WORKSHEET Dec.15.2011 (forfiscal year ended Apc20th)

•					Minimu	m Future Leas	Minimum Future Lease Commitments	st	•
		Lease	2011	2012	2013	2014	2015	₽. *	2017
	<u>-</u>	Expiry	Current Yr,	į	Zild	3rd	4	त्त	,
Lease No.	Description of Property	MM/YYYY	Expenses	'Year	Year	Year	Year	Year.	Balance
	332467 1FUJA6CK97LY49529	05/2013	21.204	21:204	21 204	1.767	1	ı	,
	332468 1FUJA6CK57LY49530	05/2013	21 204	21 204	21 204	1.22	ſ		
	32469 1FUJA6CK77LY49531	05/2013	21.204	21 204	21 204	1767	71		
	29316 1FUJA6CK37LY49526	11/2013	20:964	20 964	20.964	12 229	1	1	
	29317 1FUJA6CK57LY49527	11/2013	20:148	20 148	20 148	11 753	1		
		11/2013	20 148	20 148	201748	11 753.	1	i	
	36256 1HTSCAAMXXH227588	11/2013	20 148	20 148	20:148	11 753	1	÷2	
Rvider . 3.	36257 1FUJA6CK97LY61325	11/2013	20-148	20 148	20.148	11 753	1	· Y	-
	336256 1FUJA6CK77/LY61324	11/2013	20-148	20:148	20-148	(1753.	1	Î	
	25584 1FUJA6CK87LY49523	05/2012	26.922	28 952	1.224E	1	-	ì	
	325582 1FUJA6CKX7LY49524	05/2012	26 952	26 952	2.248:	Î	,		
	324854 1FUJA6CK47LY49521	05/2013	24 780	24.780	24.780	2 065	1	ï	
	324855 1FUJA6CK67LY49522	05/2013	24 780	24 780	24.780	2 065		, -	
	13423 Rental # A4EC380520	09/2014	14 220	14 220	14,220.	14 220	5,925		
	13424 Rental # A3EC380443.	09/2014	14.220	:14:220	14 220	14.220	5.925		
	13422 Rental # A3EC380225	09/2014	14:220	14 220	14 220	14 220	.6.925		•
			12.900				. ,		
Tru Source 344371-212772	Ţ,	09/2016	1	1 176	1.176	1176	1:178	1,176	490
123	4	08/2016	ſ	1 5(2	1512	1 512	1.512	1.512	504
Fosfiba 344371-205533		111/2016		.9 086	9 912	.8 812	9.912	.2.8312	5 782
Toshibe 344371-205455	455 Plant & Sales Photocopier	11/1/2018-		.8 382	. 9 144	. 6 144	U 144.	1.44	5,334
			-			-		4	-
	TOTALS		344340	351.598	303.772	144 829·	39 519	21.744	12 1.1ò
PREPARED BY:	S				TANTE	TATE: 2011-07-11	•		

REVIEWED BY:

### Schedule 1.1(hh)

### Purchased Assets

- 1. All assets, undertakings and properties of the Vendor of every nature and kind whatsoever, and wherever situated; including without limitation the following:
  - 1.1. The Purchased Receivables:
  - 1.2. The Purchased Inventories;
  - 1.3. All goods other than goods listed in section 1.2 hereinabove that are held for sale, lease or rental (whether such goods are saleable, obsolete or damaged) in relation to the Business carried on by the Vendor including, materials and supplies:
  - 1.4. The Contracts:
  - 1.5. The Customer Contracts;
  - 1.6. The Equipment:
  - 1.7. The Vendor's Intellectual Property;
  - 1.8. All customer lists and lists of prospective customers and marketing and sales materials and databases not already included in the Vendor's Intellectual Property;
  - 1.9: All corporate names used by the Vendor;
  - 1.10. All rights, title and interest in orders on hand and other contractual arrangements with the customers to the extent that they can be assigned and that the Purchaser wishes to avail itself thereof;
  - 1.11. All the benefit of all unfulfilled orders received by and in favour of the Vendox:
  - 1.12. All other tangible and intangible assets and property used in connection with the Business;
  - 1.13. The exclusive right of the Purchaser to represent itself as carrying on such business in continuation of and in succession to the Vendor;
  - 1.14. All Books and Records:
  - 1.15. All plans, sketches and other documents related to (i) the operations of the Business and (ii) the customers of the Vendor;
  - 1.16. all other property, assets and rights, real or personal, tangible or intangible, owned by the Vendor or to which they are entitled to but excluding the Excluded Assets;

- 1.17. All R & D receivables, income tax receivables, tax refunds, deposit and deposit letters;
- 1.18. All the prepaid amounts related to:
  - 1.18.1. Insurance Property and Liability
  - 1.18:2. Insurance 2002 Ford
  - 1.18.3, Coface (receivables):
  - 1.18.4. Directors and Officers liability
- 1.18.5. All the taxes (municipal, business and school) for all the Leased Premises.
- 1.18.6. CN non freight tracts
  - 1.18.7. Canadiens hockey tickets.
  - 1:18.8. Microsoft (Navision Licence)
- 1.19. The season tickets from Club de Hockey Canadien for the seats 104-1, #13 and #14 together with all the rights attached to said season tickets and any renewal thereof;
- 1.20. All the Equipment Operating Leases that the Purchaser wishes to avail itself thereof and not specifically an Excluded Assets as per Schedule 2.2 attached to this Agreement; and
- 1.21. The assets described in Schedule 5.1(o) of this Agreement.

# Schedule 1.1(uu)

# Vendor's Intellectual Property

# Canadian Trademarks.

See document attached Total of three (3) pages

# **US** Trademarks

See document attached Total of one (1) page

CANADIAN REGISTERED TRADEMARKS: DOMFDAM:

C Millian Cartina	grapplication number	date of Name of applicant registration ["Ri"]	Name of the current ownersor applicant	date	Name of the agent	isnieus
SPRINGFLEX FOAM	1531270	A: 2011-06-10	Domípam International Inc.		N/A	formalized on 2014-05-15
CONTOUR-MATIC	1MC541558	A; 2005-06-07		2020-06-07	Ilm Sproule, A.Z.Sponge.	irgistereth
	LINC203073	R: 1974-11-08	Damfoam International Inc.	2019-11-08	Stikeman Ellioft (1.P	renewed on 2004-11-08.
· cụppiệ Foam·	9960ZSDWIT	R: 2002-11:-20.	Damfoam International Inc	2017-11-20	Jim Sproufe, A-Z Sponger & Foam Products Ltd.	registered
VELVAFOAM	LMC272231.	R-1982-08-27	Domfoam International Inc.	2012:08-27	Stikeman Ellott I.P.	renewed on 3997-08-27-
	-LMC623917	R: 2004-10-28	Domfoam International Inc.	2019-10-28	N/A	· registered
	LMC288288	8: 1584-02-24	Dominam International Inc.	2014-02-24	Stikeman Ellinte (LP ·	renewed on 1949-02-24.
	LMC271131	R: 1982-07-16	Pomfoum International Inc.	2012-07-16	Stikeman Fillott LLP	Tenewed on 1997-07-15
	1.MC292308	R: 1984-06-22	Domioam International Inc.	2014-06-22	Stikeman Elliott Li.R	. renewed on 1999-06-22.
	.LMC205857	R: 1975-03-14	Domfoam International Inc.	2020-03-14	Stikeman Ellott LLP	renewed on 2005-03-14-
	LMC145959.	R: 1966-06-30.	Domfoam International Inc.	2025-05-30	Stikeman Elliott Lip	nenewed on 2011-05-30.
,	1MC205858	R: 1975:03-1£	Domfoam International Inc.	2020-03-14	Stikeman Ellott (LP.	nenewed on 2005-03-34
	LMC202908	R: 1974-11-01	Domform International Inc.	2019-11-01	Stikeman Elliott LLP	renewed on 2004-11-01.
	LMC624903	R: 2004-11-08	Domigam International Inc.	2019-11-08	Jim Spraule: A.Z.Sponge.	jegjstered
	EM©89023	R: 1984-03-16	Domfoam International Inc.	2014-03-16	Stikeman Elliatt LLo	rehewed.on 1999-03-16
7	LMC202309-	R-1974-11-01	Domitoam International Inc.	2019-11-01	Stikeman Elliott (L.P.	· renewed on 2004-11:01
PERMA-BOND	LMC180311.	R: 1971-12:24	Domfoam-International Inc.	2016-12-24	Stikeman Elliott ILP	renewed on 2001-12-24:
Snuggle-Pedic	LMC649665	61-70-2002-8	Domfoaញ (កេខ្មែរភាងបែ្មាន! ព្រះ:	3020-07-19	Jim sprayle, N-Z Spanse. Z Form Products trd.	iegisterad
	LMC275254	1	Domitoam International Inc.	2012-12-31	Stikeman Ellort LP	ranewed-pn/1997-12-31
FREEDOM FOAM	, LMC729834.	R: 2008-12-01.	Domfoam International Inc.	2023-22,01.	Jim Sproule, A-Z Sponse, & Foam Products Ltd.	) registered
	SER6ZZJWT.	10-21-8002:4	Damioam International Inc.	2023-12-01:	Jim Sproulé, 4-2 Sponge & Form Products trd:	registared
•	LNICGÓS748	N: 2006-06-06	້ ວ່າກຸ່ງເຂນາງທີ່ຄວາມງານເພາະອຽນທົ່ວ	. 90:90-tžáž	Ifm Sproule, A-Z Sponge. & Foam Products (td.	relistered
NORTHERN TREASURE	IMCG38221	R: 2005-04-25	Domfoam International Inc.	2020:04-25	Silkeman Ellott LLP	registered
LA COLLECTION ZODIAC	LMC294245	R1 1984-08-17	Domicam international inc.	7014-08-17	Stikeman Elilot Lip	renaWed on 1999-08-17
WUNDER-BOND	LMC180310	R: 1973-12-24	Dominam International Inc.	2016-12-24	Stikeman Elliott LLP	renewed on 2001-12-24
-	LMC298957	R: 1985-01-11	Domiosm International Inc.	2015-01-11	Stikeman Elliott LP	renewed on 2000-01-11
	LMC292311.	R: 1984-06-72	Domioam International Inc.	2014-06-22	Stikeman Ellibit ELP	renewed:on 1999-06-22
-	.LMC265867	R-1582-02-35	Domiosm international inc.	2027-02-26	Stikeman Elliort LLP.	renewed on 2012-02-26
V <u>isco</u> nkadic Memory Fo	LMC688655	8-2007-05-31	Dönyfoam International Inc.	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	Jim Sproule, A-Z Sponge. & Foam Products Ltd.	registered
Çuddle-Pediç	LNICEd4418	R: 2005-07-14	Domfoam, international inc.	#t-£0-020Z	Ilm Sproule, A-Z Sponge & Foalm Products Itd.	registered.
VODIAC COLLECTION	JAA COURTE	R: 1984-05-11	Domfoam International Inc.	2034-05-37	Selveman Elliott 11.8	Tenemed on 1999,05-71

choococlet. Marst-3200/3187(5187403)Trademarks sale Canada

THE TO THE PER MEN THE PER ALL WITH

DOINTONAN MAPLE LEAP & DESIGNAL MICE 2006-077-09   Dointonan International Inc.   2006-077-07-37	ē	Trademack	Registration or application number	Registration or date of A or Realization	Name of the ឈុំកុស៊ី owner or epplicant	date of cancellation or abandonment	Name of the agent.
SECURIFOAM'R DESIGN   1960194   N. 1989-11-19   Domform International Inc.   2010-01-29	÷		LMC375564	R: 1590-17-09	Domfoam International Inc.	.50/20-5002	Stikėman Elliotė (LP
SECURIFOAM'& DESIGN         832300         A: 1998-12-20         Domfoam International Inc.         2001-01-10           PRÓTECT-A-RUG.         IMCIDATOUT         R: 1932-11-26         Domfoam Inc.         1983-01-28           PRÓTECT-A-RUG.         IMCIDATOUT         R: 1932-11-26         Domfoam Inc.         1983-01-28           FIZAREX         IMCAGISSZ         R: 1932-11-26         Domfoam Inc.         1986-09-20           FEAREX         IMCAGISSZ         R: 1932-11-26         Domfoam Inc.         1986-09-21           INCOGNA         R: 1932-11-26         Domfoam International Inc.         1986-09-21           INCOGNA         R: 1932-11-26         Domfoam International Inc.         1986-09-22           INCOGNA         R: 1932-11-26         Domfoam International Inc.         1986-09-22           INCOGNA         R: 1932-11-26         Domfoam International Inc.         1995-07-08           SUNINSE         A: 1982-11-26         Domfoam Inc.         1995-07-08           ALTO-CUMULUS         A: 1982-11-26         Domfoam Inc.         1995-07-08           ASTOCAINE         IMCISSON         R: 1982-11-26         Domfoam Inc.         1995-07-08           FEACEI         IMCISSON         R: 1982-11-26         Domfoam Inc.         1995-07-08           FEACEI<	7		1,460194	A: 2608-11-19	Domicam International Inc.	2010:12-17	Ilm Sprouse, 8-Z Sponge
NIMBUS   1983-012-28   1983-01-28   1983-0	in.	SIGN.	832300	A: 1996-12-20	Domfoam International Inc.	2001-01-10	Stikeman Ellotett Pr
PRÓTECT-A-RUG.         LIMC177017         R.1971-07-23         Domfoam inc.         1986-09-10           STIATUS.         496-288         A.1987-07-23         Domfoam inc.         1986-09-10           FINATUS.         496-288         A.1987-03-07         Domfoam inc.         1986-09-10           FARBES         LANCIOSER         R.1976-03-07         Domfoam international inc.         1989-04-27           FABRIAT         SG012.1         A.1983-03-11         Comfoam international inc.         2005-07-21           Domfoam & design         LMC203.76         R.1974-12-06         Domfoam international inc.         2005-07-21           Domfoam & design         LMC203.77         R.1957-11-05         Domfoam international inc.         2005-07-21           SUNISE         LMC203.76         R.1982-11-05         Domfoam international inc.         2005-07-21           A LOCCLING         LMC203.04         R.1982-11-05         Domfoam inc.         1997-07-08           SURINISE         LMC203.04         R.1982-11-05         Domfoam inc.         1987-07-08           VISCOAIRE         1204-09-22         Domfoam inc.         1987-07-08           VISCOAIRE         1388-33         R.1982-11-26         Domfoam inc.         1988-07-07-08           R         FINKPANTHER         <	. <b>'4</b>		495427		Սգարժաղ Լոգ.	1983-01-28	Seal, Shapdanick
STIATUS   495428   A. 1982-13-36   Domfoam Inc.   1983-42-19	หา	PROTECT-A-RUG.	LMC177017		Domfoam.int:		Seal, Shaposnick:
FLAREX         FLAREX         R: 1973-03-07         Domfoam International Inc.         1590-04-27           CFC& DESIGN         LMCZ03869         R: 1974-12-06         Domfoam International Inc.         1500-04-22           HABILIA         LMCZ0376         R: 1974-12-06         Domfoam International Inc.         2005-07-21           UNCCELL IR         LMC20375         R: 1982-11-06         Domfoam International Inc.         2005-07-22           ODMSoam & design         LMC20404         R: 1982-11-06         Domfoam International Inc.         2005-07-07           SUNRISE         ALTO-COMULUS         R: 1982-11-06         Domfoam International Inc.         1583-12-12           COLINGLUS         A: 1982-11-06         Domfoam International Inc.         1583-12-12         Domfoam Inc.         1583-12-12           COLINGLUS         A: 1982-11-16         Domfoam Inc.         1583-12-12         Domfoam Inc.         1583-12-12           VISCOAIRE         ILACA369         A: 2004-02-12         Domfoam Inc.         2005-02-14         DOS-00-14           PINK PANKTHER         LMC236736         R: 1975-10-19         Domfoam Inc.         1984-01-06         PROFOARD           GIRRUS         A: 1982-11-26         Domfoam Inc.         1984-01-06         PROFOARD	LQ.	o,	495428		Domitorm inc.		Scal, Shaposnick
CFC & DESIGN         LMC203689         R.13974-12-06         Domfoain International Inc.         2605-07-21           HABITAT         50012.1         Ar 1383-03-11         Qenticam Inc.         2005-07-21           UNOCELL HR         LMC203756         R. 1974-12-06         Domfoam International Inc.         2005-07-05           Dunnfoam & design*         LMC375563         R. 1982-11-06         Domfoam International Inc.         2005-07-05           SUNRISE         LMC375563         R. 1982-11-06         Domfoam International Inc.         1983-12-10           ALTO-COMMULUS         Ar 1982-11-06         Domfoam Inc.         1983-12-10           ALTO-COMMULUS         Ar 1982-11-06         Domfoam Inc.         1983-12-10           ALTO-COMMULUS         Ar 1982-11-06         Domfoam Inc.         1983-12-10           VISCOAIRE         Ar 1982-11-06         Domfoam Inc.         1983-12-10           VISCOAIRE         Ar 1985-12-0         Domfoam Inc.         2005-02-14           FINKYPANTHER         LMC33659         Ar 2403-12-0         Domfoam Inc.         1990-03-0           GIRRUS         Ar 1987-11-26         Domfoam Inc.         1984-01-06         1984-01-06	ж.		LMC205782	R: 1975-03-07			Seal, Shaposnick
HABITAT   S001221	. ∞		LMCZ03684	R:1974-12-05			Stikeman Elliott LI P.
UROCELL HR         LMC03776         R: 1974-12-06         Domfoam International Inc.         2005-07-21           Domfoam & design:         LMC137553         R: 1934-12-05         Domfoam International Inc.         2005-07-21           SUNRISE         (MC264104         R: 1852-11-05         Domfoam International Inc.         1993-07-08           ASTO-COMULUS         495-425         A: 1982-11-26         Domfoam Inc.         1983-12-19           CUMULUS         495-425         A: 1982-11-26         Domfoam Inc.         1983-10-02           EVER/ROND         LMC184974         R: 1972-06-18         Domfoam Inc.         1983-10-02           VISCOAIRE         13204369         A: 2004-01-22         Domfoam Inc.         2005-02-14           PINK PANTHER         LAG03-12-02         Domfoam Inc.         1989-03-46           GIRRUS         A: 1982-11-26         Domfoam Inc.         1984-02-06	۵Ł		500121	A: 1983-03-11			Seal, Shaposhick
Dom/oam & design         EMCS/35563         R: 1980-11-05         Dom/oam International Inc.         2005-07-05           SUNINSE         LMC264104         R: 1981-11-05         Dom/oam International Inc.         1997-07-08           ALTO-CEMULUS         495475         R: 1982-11-05         Dom/oam International Inc.         1983-12-19           CLIMULUS         495475         R: 1972-12-15         Dom/oam Inc.         1983-12-19           EVERIBOND         LMC18497A, R: 1972-02-18         Dom/oam Inc.         1988-12-19           FlorkCell         Ties/Cell         Dom/oam Inc.         2005-02-14           FlorkPARTHER         LMC13679-12-12         Dom/oam Inc.         1989-03-16           FINKPARTHER         LMC236726_A         R: 1972-11-26         Dom/oam Inc.         1989-03-16           GIRRUS         A: 1987-11-26         Dom/oam Inc.         1989-03-16         1989-03-16	10		LMC203776	R: 1974-12-06			Stikeman Elfort LLP
SUNUISE         Lyd.Z64104         R: 1981-11-06         Domfoam International Inc.         1997-07-08           ALTO-CHMULUS         485426         A: 1982-11-26         Domfoam Inc.         1383-12-39           CLIMULUS         485426         A: 1982-11-26         Domfoam Inc.         1383-12-39           EVER-RODAIRE         1304369         A: 2004-03-2         Domfoam Inc.         1398-12-36           VISCOAIRE         1204369         A: 2004-03-2         Domfoam Inc.         2005-02-14           FINK PANTHER         1.MC236726         R: 1279-12-19         Domfoam Inc.         1390-03-16           GIRRUS         495425         A: 1382-11-26         Domfoam Inc.         1384-01-06	ਜ਼		LMC375563	R-1950-11-09;			Stikeman Elllott LLP.
ALTO-CUMULUS         485,426         At 1982-31-26         Domfoam Inc.         1483-32-39           CUMULUS         485,425         At 1982-31-26         Domfoam Inc.         1583-42-19           CUMULUS         147,148,1426         At 1982-31-26         Domfoam Inc.         1583-42-19           VEVERALIS         150,040,01-22         Domfoam Inc.         2005-02-40         Inc.           VEX.CO. IT IN TABLES         At 2003-02-22         Domfoam Inc.         2005-02-40         Inc.           FINK PANTHER         1 MC236776;         R: 1978-11-26;         Domfoam Inc.         1980-03-46         Inc.           CIRRUS         495428         A: 1987-11-26;         Domfoam Inc.         1980-03-16         Inc.	12	,	UMCZ64104	R:1981-11-06			Stikeman Elllöst LLP.
CUMULUS         495425.         At 1982-11-26         Dómfoain finc.         15983-12-19           EVER-BOND         LMC18497A.         R. 1972-08-18         Dómfoain linc.         1987-40-42           VISCOAIRE         1204369         At 2004-02-2.         Dómfoain linc.         2005-02-44           FENCEL         TASSS94         At 2003-12-02         Dómfoain linc.         2005-02-48           FINKPANTHER         1 MC236776.         B: 1978-10-12         Dómfoain linc.         1984-01-06           GIRRUS         495429.         At 1982-11-26         Dómfoain linc.         1984-01-06	13	24 24 4	495425	A: 1982-11-26	:		Seel, Shaposnick
EVER.BOND         LMC184974.         R. 1972.08-18         Domfoam Inc.         1306-32.           VISCOAIRE         1304-369         A. 2004-02-22         Domfoam Inc.         2005-02-14           *Textoal         1398-37-10-12         Domfoam Inc.         2005-02-14           *FINK*PANTHER         14 MC236776:         R: 1379-10-15         Domfoam Inc.         1580-03-16           *GIRRUS         435429         A: 1987-11-26         Domfoam Inc.         1984-01-16	14.		495425	A: 1982-11-25			Seal Shabosnick
VISCOAIRE         1204369         At 2004-01-22         Domfoam Insernational Inc.         2005-02-14           *Flexcel**         T1598594         At 2004-01-22         Domfoam Inc.         2005-03-08         At 1905-03-16           *FINK*PANTHER         *LIAC236736;**         R: 1979-10-19         Domfoam Inc.         1380-03-16         At 1984-01-06           *GIRRUS         *At 1982-11-26;**         Domfoam Inc.         1384-01-06         At 1984-01-06	(2)	EVERIBOND	LMC184974.	R1972:08-18	Domfoam Inc.		Seal, Shaposnick
** Tex,Cel***     1139594     A. 2003-12-02     Unifoam Inc.     2005-03-08       FINK PANTHER     1.MC236726;     R: 1979-10-19     Domfoam Inc.     1390-03-16       GIRRUS     495429     R: 1387-11-26;     Domfoam Inc.     1384-01-06	101		1204369	A 2004-01-22	Daminam international Inc.		N/A.
FINK PANTHER 1.MC23677.6; R; 1575-10-15 Domform Inc. 1990-03-16. GIRRUS A; 1987-11-26; Domform Inc. 1984-01-06	<b>1</b> 7	•	1198594	V: 2003-12-02 ·	ပြစ်ကစ်ချွဲကျ International Inc.	\$ <del>0</del> -5002	Mrt Sproule, A-Z Springe
FINK PANTHER 1482-57.26; R: 1282-11-26; Domfoan Inc. 1394-01-05			1				S room reporter for
GIRRUS 485428 A. 1982-11-26: Domfosn tnis. 1984-01-06	ė	NTHER	11MC236776.	8: 1979-10-19	Domform inc.		Seal: Shaposnick
	ğ		495429.	A: 1982-11-26:	Domfoam this:		Seal, Shaposnick
					,		
				<del></del>	٠		

11/00000 LP.LV3152-3200/3137\3187-003\Trademarks\_table\_tanada

CANADIAN SEARCH OF TRADEMARKS: DOMFDAM

	ľ	
śnieś	Notice of approbation sent on 2011-10-31.	,
Name of the applicant,	Domfqam International Inc	
date	2011-03-07	
Application pumber	1518007	.~,
Trademark	GELFLEX	

0-LP.LM3151-3700/31#7/31#7-003\Tradem2rks\_lable\_C2/2da\_.

# US REGISTERED TRADEMARK: DOMFOAM

TO CO CO CO

agent Status	:		Meier registered
Name of the ag			Lawrence H. 1
ier.gr Expiry	date	,	2015-12-18: Lawrence H. Meier regist
Registration date Name of the current owner of	applicant		Domfoam liternational line.
Registration date			2005-12-13
Serial.number			78508999.
Trademark			1 NORTHERN TREASURE
			څانې

# US CANCELLED TRADEMARK: DOMFOAM

 Trademark	Serial number date of applicat	date of application	Name of the current owner or date of cancellation or Name of the agent applicant	date of cancellation or abandonment	Name of the agent
 CURDLE FOAM	78219889	2003-02-27	Domfoam International Ing.	2004-02-11	N/A
SECURIFOAM & DESIGN.	SIGN 75311611	1997-06-19.	Domfoam International Inc. 2000-10-12	2000-10-12	Jess M Collen

### Schedule 2.2.

### Excluded Assets

- The Purchaser acknowledges and agrees that the following assets shall be considered excluded of the Transaction:
  - 1.1. The accounts payable, except as otherwise provided in this Agreement;
  - 1.2. The tax losses, except as otherwise provided in this Agreement;
  - 1.3. The cash on hand or on deposit with banks or other depositaries;
  - 1.4. The debts due to the Vendor from any shareholder, director, officer or employee of the Vendor or any other person who does not deal at arm's length with the Vendor.
  - 1.5. The following Equipment Operating Leases:

The second secon

- 1.5.1. Ryder Lease No. 332467 1FUJA6CK97LY49529
- 1,5.2. Ryder-Lease No. 332468 1FUJA6CK57LY49530
- 1.5.3. Ryder Lease No. 332469 1FUJA6CK77LY49531
- 1.5.4. Ryder Lease No. 329316 1FUJA6CK37LY49526
- 1.5.5. Ryder Lease No. 329317 1FUIA6CK57LY49527
- 1.5.6. Ryder Lease No. 329318 1FUJA6CK77LY49528
- 1.5.7. Ryder-Lease No. 336256 1HTSCAAMXXH227668
- 1,5.8. Ryder Lease No. 336257 IFUJA6CK97LY61325
- 1.5.9. Ryder-Lease No. 336256 1FUIA6CK77LY61324
- 1.5.10: Ryder-Lease No. 325581 1FUIA6CK87LY49523
- 1.5:11. Ryder-Lease No. 325582 1FUJA6CKX7LY49524
- 1.5.12. Ryder Lease No. 324854 1FUTA6CK47LY49521
- 1.5.13. Ryder Lease No. 324855 1FUJA6CK67LY49522
- 1.5.14. Toshiba Lease No. 344371-205533 Main photocopicur
- 1.5.15. Toshiba Lease No. 344371-205455 Plant & Sales Photocopier
- 1.6. The shares issued and outstanding of the capital of "YALLE FOAM INDUSTRIES (1995) INC."
  registered in the name of the Purchaser; and
- 1.7. The shares issued and outstanding of the capital of "A-Z Sponge & Foam PRODUCTS LTD." registered in the name of the Purchaser.

# Schedule 2.6

# Assumed Obligations

- 1. The obligation of the Vendor under the Contracts as of the Time of Closing, and
- 2. The obligation of the Vendor under the Customer Contracts as of the Time of Closing;

# Schedule 2.9(A).

# Purchased Receivables

### Section 1

# List of Purchased Receivables as of January 31, 2012.

See document attached Total of four (4) pages

# Section 2

Calculation of the Purchase Price attributed to the Purchased Receivables.

See document attached Total of fourteen (14) pages DOMFOAM INTERNATIONAL LIMITED Aged Accounts Receivable as of 01-31-2012

Customer No.	Balance Due	Current	31-60 Days:	61-90 Days	Över 90 Days
00157	704,58.	704.58,	-	<del>-</del> -	-
00202	25,631.17	25,431,85	199.32	-	<b></b> ·
00291	5,362,49	1,032,48	3,320.06	1,009,95	-
00308	4,316.16	4,316.16	-	<b>*</b>	-
00316	1,952,01	1,736.65	215.36	بذ	-
10005	511,571.37	303,779,87	180,349.98	.27,441.52	<del>.</del>
10020	128,444.89	128,444,89	,÷	-	-
10024	2,028.58	, ,	2,028.58	_	t.
10052	7,668:75	6,949.67	719,08.	₹	-
10072	528,925,52	258,533.73	155,454,72	114,937,07	س
10080	41,326.70	41,326.70	-	•	
10120	1,158.95	1,158.95	_	-	wh.
10136	17,679.48	17,679,48		-	-
10145	5,746.68	3,072.41	2,674.27	-	<u></u>
10222	1,004.61	1,004.61	,-		<u> </u>
10244	101,343.87	101,343.87	<b>-</b> ,	_	-
10273	15,318.85	13,356.08	2,962.78	_	<b></b>
10305	6,438.19	3,402.81	2,141.95	893,43	_
10331;	2,254.11	2,254.11	٠.,,	<b>2.</b> ·	- ·
10351	3,754:38	٠.	3,754.38	مث	_
10358	970.58ر29	11,583.52	10,024.90.	8,412.16	¥,
10371	32,216.28	19,133.32	1,056.44	11,342.99	683,53
10378	7,105.73	1,598.57	1,859.97	_	3,647,19
10379	3,680.61	3,680,61	•	<u></u>	
10384	45,107.85	27,771.01	16,306:37	1,030.47	_
10385	10,927,29.	10,927.29	-	¥	*
10387	7,615.54		1,173,56	410:29	6,031.69
10413	153,635.36	42,935,08	19,370.39	55,168.67	36,161.22
10426	3,529,98	3,529.98	~	=	ىد
10465	833.89	833,89	,uic	*	
10494	32,749,54	\$2,749. <b>5</b> 4	-	_	_
10502	11,893,25	3,266.84	1,087.18	3,549.67	3,989.56
10509	1,729.11	1,729,11	, ~	-	-
10510	18,715,15	18,715,15	<del>-</del>	<u>.</u>	-
10521	2,826.31	2,826,31		<del>.</del>	<del>-</del>
10545	28,034.38	28,034,38	<del></del>	~-	
10547	15,568.17	5,498.67	10,069.50	-	<del>-</del> ,
10551	74,572.24	74,572.24			; <del>-</del>
10570	6,278.42	6,278.42	<del></del>	-	-
10585	•	55,233,52	32,888,35	65,818.84	67,225.29
	XXT,100,00	J-7/2J-1,J-2		020020107	
10595	221,166.00 344.24	344,24	- Captacian	52/040104	معامل المالي المالية

Customer No.	Balance Due	Current	31-60 Days	.61-90 Days	Over 90 Days
10785	6,075.15	6,075.15		•	· +
10848:	3,937,95	. 7	1,580.27	2,357.68	₹
10864	284.40	.284.40		_	-
11049	9,484.48	6,739,35	2,745,13		-
.11150	3,850.31	· <del>,</del>	3,850.31	•	<b>5</b> -
11155	2,615,92.	2,615.92	-	<del>~</del> .	°~
,11164	48,779.05	51,488.31	(2,709.26)		÷
11177	89,437.38	30,746,66	25,400.93	24,863,69	8,426,10
11197	626.81	626,81	ä	<del>7</del> -	, , , , , , , , , , , , , , , , , , ,
11209	29,097,79	29,097;79	. 2	•	N. Nest
11210	1,778.75	1,115.72	663.03	i-	. 15:
11268	`36,681,28	36,681.28	-	è	<b>र</b>
11272	672.18	1,034.89	(362,71)	-	غ
11276	15,524.84	÷	15,524.84	<b></b>	₹
11289	16,671.06	5,017,77	5,249.62	4,461.72	1,941.95
11399	80,315.23	6,706.11	9,359.14	7,755.01	56,494,97
11415	11,174,51	9,312,23	1,862.38	¥,	₩.
11421	15,293,31	905,43	8,702,70	5,685,18	
11457	26,525.81	25,525.81	÷ .	Ī	-
11558	3,645.51	3,121.23	.525 <b>.</b> 28	÷	-
11577	5,027.53	4,979.82	· ·	·=	47 <u>.71</u>
11580	442.19	442.19	<u>ئ</u>	=	~
11627	1.80	1.80	ے		**
11665	387,973,57	138,756.63	112,323:23	117,018.61	19,875:10
11676	42,927.94	19,314.91	17,404.97	6,208.06	بشد.
11682	254,229.37	253,895,10	· ·	334.27	· -
11684	5,358,46	5,758.76	(400.30)	5	ے
11714	8,989,08	8,715.63	353.56	₹*	(81:11)
.11760	885,20	÷	885,20	d ÷	- · · · · · · · · · · · · · · · · · · ·
1177 <u>9</u>	10,301.45	10,301:45	~ .	•	÷
11811	18,946 <i>.7</i> 1	18,946.71	<b>'</b> a.	•	-
11828	4,191,55	4,191.55	-		ž
11833 <sup>-</sup>	1,682.72	1,098.94	• -	583.78	<b>-</b>
11844	5,903.05	5,903.05		-	جب
11848	93,242.34	24,928:53	30,248.93	22,139,94	15,924,94
11859	6,594.80	. 6,694.80		,	· -
11871	1;171.60	1,171.60	<b>F</b> 1	· · ·	.=
11914	.36,917,36	23,905.21	9,420.51	3,591.64	
11,924	231,421,59	30,900,74	31,178.08	73,739,94	95,602:83
11927	42,533,30	25,496,59	17,036.71	₹.	*
11948	3,133.52	3,133.52		7.	
11974	(11.50)	(11.50)	-	<b></b>	
11976	23,398.27		11,839.97		- 2:
11979	29,254,14	27,397,56	1,856.58		-
11982	42,209,54		*	(290.90)	· · · · · · · · · · · · · · · · · · ·

	Customer No.	Balance Due	Current	31-60 Days	61-90 Days	Qver 90 Days
	12021	6,359,25	5,618.26	740.99	ŧ <del>-</del> -	_
	12101	965.55	~	965.55	<u>.</u>	<b>-</b>
	12102	71,175,61	27,028.49	13,870,53	30,276.59	<del>,</del>
	12104	40,670,61	20,573,97	20,181.29	(84.65):	
	12122	(32,80)		,		•
	12140	17,913.60	17,913.60	<del>-</del>	_	_
	12146	20,962,95	4,908.38	5,781.24	.5,890,81	4,382.52
•	12158.	17,823.38	17,823,38	- <b>,</b>	÷	• • • • • • • • • • • • • • • • • • •
	12230:	6,506,76	4,144.61	2,362,15	÷	<del>.</del>
	12242	9,119.76	8,420.15	699.61	<b>-</b>	• -
	12253	5,080,77	1,217.35	3,863,42	<b></b>	-
	12256	17,887,78	17,887,78	- ·	<u></u>	#
	12270	4,819.79	=	4,819.79	-	<del>.</del>
	12277	2,346.65	1,225,63	747.35	373,67	<del></del>
	12282	5,362,13	5,338,88	1	23:25	<u>.</u>
	12290	2,002,76	2,002.76	_		-
	12291	596.03	596.03	_	~	-
	12296	29,232.85	- 384.08	- 15,929.32	10,493,91	2,425,54
	12298	59,795.77	25,619.87	14,419.36		2,297.82
	12301	13,668.05	9;923,90	3,744.15	, (=,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	्रस् <sub>र</sub> स्वयं राज्यन
	12303	1,184,25	1,184.25	-,, ,	· =	<u>-</u>
	12308	12,344,90	12,344.90	-	2	_
	12314.	2,330.86	1,275.30	1,055.56	=	<del>"</del>
	12326	916.76	524.35	392.41	u¢	2
	12330.	4,543.77		~	1,005.11	3,538,66
	12331	23,556.06	7,529.09	6,507.75	8,366.47	1,152,75
	12334	3,620.86	3,620.86		ar a	₹8100 <b>₹8</b> 104 <b>-</b> 0
	12341	1,179,80	1,179,80	<b>-</b> .	-	<u></u> •
	12343	28,957.45	19,659.45	9,079.05	209.90	9,05
	.12353	1,131,12	1,131,12	•	<u>+</u>	<u>-</u>
	12355	4,681,06	4,681.06	<del>.</del>		-
	12357	11,881.35	1,573.48	<b>~</b>	2,016.81	8,291.06
	12358	1,294.39	1,294,39	·		
	12367	9,722.17	9,722.17	,	<b>-</b>	-
	12373	5.00	5.00	÷	_	. :
	1,2380	1,454.04	1,454,04		<u>.</u> .~	_
	12390	3,304.57	3,304.57	4.	_	_
	U11429	15,044,90	15,044,90	÷	4	_
	U11771	1,079,755.37.	556,732,35	358,231.67	164,791.35	-
	U12129	544,154.40	168,733.64	160,191.76	178,981,57	36,247.43
	U12164	75,413.54	20,979.01	53,559.58	874.95	- 200 cm - mg
	U12171	8,393.63	7,170.44	1,223,19	-	
	U12180	3,884,22	190.10	3,694.12	ם	<b></b>
	U12183	,236.00	*		236,00	-
	U12184	12,578,70	12,578.70	<del>-</del>		_
			1.75 *** *** *** ***			

Customer No.	Balance Due	Current	31-60 Days	61-90 Days O	ver 90 Days
U12187	12,428,10	12,428,10	=	~	<b></b>
U12188.	17,757.90			<u>.</u>	
Total	5,996,691.62	3,208,741.60	1,434,256,08	979,378.14	374,315.80

0

# AGED ACCOUNTS RECEIVABLE DOMEDAM INTERNATIONALING

-

จอบอากรับ! :		
TOTAL	### ### ### ### ### ### ### ### ### ##	38 68128. 672.18 15.224.84 16.671.06 11.774.81 25.555.81 35.555.81 35.555.81 35.627.53 442,19.
	(c) (a) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	o en in en in en en en en en en
OVER 90 DAYS		C 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
s,	A A A A A A A A A A A A A A A A A A A	0)
61-90 DAYS	1000,47 11030,47 1030,47 1030,47 1030,47 1030,47	4461,77
s S		. E. F. 10 10 10 10 10 10 10 10 10 10 10 10 10
31-60 DAYS	199,38 9,220,066 2,028,38 15,141,39 15,141,39 16,143,38 10,027,48 1,1863,97 1,18	. (36277) 1 (46277) 1 (46277) 1 (46277) 1 (46277) 1 (46277) 1 (46277) 1 (46277) 1 (46277) 1 (46277)
ţ	See See Fee Fee Fee Fee Fee Fee Fee Fee	
CURRENT	28.401,48 1.092,48 1.092,48 1.092,48 1.092,49 1.092,70 1.092	35 661 28 1034,84 401,77 9 37,523 26 525,81 3 121,83 4 97,83 6 48,19
· <b>:</b>		
сверіт ціміт	24 000,000 15 000,000	85 000,000 25 000,000 30,000,00 30,000,00 35,000,00 10,000,00 25,000,00 25,000,00
F.	සුමු සු	25 25 25 25 25 25 25 25 25 25 25 25 25 2
2		
	457 VEN; REZPRADDUTS. 297 DINE RRTING. 299 DINE RRTING. 299 DINE RRTING. 299 DINE RRTING. 398 ROSITECH INNOVATION ING. 316 LIND FURNITURE (CANADALING). 10022 ALPHANICING. 10022 ALPHANICING. 10022 ALPHANICING. 10030 GALLITY RIGS. OF CANADA LI 10030 GALLITY RIGG. 10030 GALLITY RIGG. 10030 GALLITY RIGG. 10030 GALLITY RIGG. 10030 GALLITY SPORTS ING. 10030 MATELAS GAULHAN ING. 11030 MATELAS THEN ING. 11030 MATELAS GAULHAN ING. 11030 MATELAS GAULHAN ING. 11130 MATELAS GAULHAN ING. 11131 MATELAS GAULHAN ING. 11131 MATELAS GAULHAN ING.	itzer roull Land inc. Tyrz inter roungent 1927 inter connection 1928 Matelas Group Limited 1928 Matelas Sissta 1927 Matelas Pecifice 1929 Matoric Fechiologies inc. 1957 Matoric Pechiologies inc.
NAME	"45" VEN, JEEZ PRODUITS. "802 BUGATTI DESIGN ING" "899 DINE ÄRTING. "399 EQSITECHTING AND ADDOLES. "390 EQSITECHTING ADDOLES. "390 EQSITECHTING ADDOLES. "390 EQSITECHTING ADDOLES. "390 EQSITECHTING FOR ADDOLES. "390 EQSITECHTING FOR ADDOLES. "390 EQSITECHTING FOR ADDOLES. "390 EQSITECHTING FOR ADDOLES. "390 EQSITECHTING. "390 EQSITECH	11268 ROUIL LARD INC. 11272 INTER CONDONT. 11276-STEERS GROUP LINITED. 11278 MATELAS SIESTA. 11447 MATELAS PESTIDE. 11458 MATELAS PESTIDE. 11558 MATELAS PESTIDE. 11558 MATELAS PESTIDE. 11558 MATELAS PESTIDE. 11558 MATELAS PESTIDE. 11569 PETITON DESIGN.

뜿

# AGED ACCOUNTS RECEIVABLE DOMFORM WITERNATIONAL INC.

-

-							
11627 PRO TAC INDUSTRIES INC.	S 09			. <b>9</b> .			, ,
11676 LITERIE DORMATEX INC.	909		16 215	17.504.97	. K. 204' DR: 4	<b>9-</b> ¥	100°C
11684 FOUISTAR CANADAING	, 15 15 15 15		758.76	(400,30) S	,		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
11714 TRICA:	9.09		L	. 353,56	. 1	~=	AC 080 St
11750 VEHANDA JARDIN INC	\$ 09					•	00,480
11779 MANUFACTURE DE MATELAS	69		30 301.45 3				TO OF
17811 BROLAIN DISTRIBUTORS LTD.	. 60 . 60	125 dog,oo \$	3 12.948.	1	. 44	Y-44	18.848.71.
11828 DOMFORT FOAM & FIBRE	64		¥ 181,55 S				4191.55
11839 MATELASS SELECT	G.		1.098.94	ys ·	583.78 \$		1 882,72
11844 PEL INTERNATIONAL LTEE	30.08		5 903,05	•			5 303,05
11848 INDUSPACING	70.00		24.928,63	30 248,99 35	22 139,94 \$	4	.53 242,34
T1859 WESTBORG FLOORING INC	S.		\$ 08'4'89 g	•	ı	17	6 694,80
1 1871 MILLENAIRE WILLIAM INC.	30.00		1 177,60 \$	1			171.60
11914 EXCEL RSVC MQUSSE FOAM!	90 S		23,905,21, \$	420.51	3.591.64	40	36.917.35
11927 MEUBLES BELISLE INC.	s.09		· 25 496,69 · S	17 036,71 \$	, 1		42 533.30
11948 MEUBLES TOMALING.	8		3.133,62 \$	,			48,680 pt
11974 STYLTEC INC.	8		6 (11,50)	.*	: <b>(/)</b>		(0,11)
11975 MATELAS ORTHOPEDIQUE	in OB		11 558,30 .5	.\$ 28.668.1L			23.398.27
11979 BECPRO:MEDICAL INC	300		397.56	855 58	4 69 14		20.05
11982 DIAMOND DIVERSIFIED IND. L	30.08		500.44		ő		42-209-54
12021 SEARS CANADA INC.	600		618.28		-		P35925
12101 CREO PACK INTERNATIONAL	9	(5,000,00		965,55	•	· iō	965.55
AD MATS CORPORATION	25.57		4		(84.85). \$		40.670.61
12122 ATLANTIC FABRICS LIMITED	7.5		(12 80)				(A) (E)
12140 DOWNEAST PLASTICS LTD.	\$. 0g		•	· •			17.019.50
32146 MATELAS BEAU REVEING	S CB	20 000.00	2008 GB GB GB	5.781.24	-	ģ	20.000.00
		50,000,00			1 11444		3
TOTAL BEERN STORE THE EAGERS	, 4 , 4	175 000.00			-		47,000,00
-	7 5	ON COLUMN	A THE VETT	24 CAC		9- <u>0</u>	1 044,00.
SLAND CONTRACT STATES OF SACRE	9. J	יי מניסטטער		7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7			a took to
	# 4 2 2 2 3 4	o od od akti					C 1000 L
A SOLE THE SANDVATIONS	9 6	4. 00 000 pc	D 20114	Ar coo			2 1,000 E
19970 COUNTIBE CARRY ET ING CW	Poet dated CHI A		2 2 3 4	270			A.810.70
POWER SPECIAL PROPERTY OF THE	T.		ď		ķ		Take Fr
Y2282 VICTINE 2008 INC.	i un	50 000.00	338.88	<u>}</u>	2.5	)	0.000 m
12290 LUGIFLEX-MOBIDE BUREAU	S. OF		ģ	ŝ			. 2 002.7E
12291 PAKTEKING,	S. Co		586,03	Ó	;	,	596,03
12296 PAVARIFURNITURE IND.	0 <u>6</u>				10493,91 \$	2.425,54	,29,232,85
12298 G. HOMANO INC.	\$	00,000			458,72	7,87	22'98'69
12308 WES INDUSTRIES INC.	नि	00,000			-tys		1.184,25
12308 PYRA MEDIC INC. (W)		000,000			.igs.	iji	12 344,90
12314 GROUPE SM PELLETIER	5 69		1275,30 \$.		(I)	i)	2,330,45
둫				392,41 5	Ą	Ç	916,76
12230 GREATECH INC. 90690744 QUE	\$ 09		į	r	.005,11	3.53B,66° \$	4.548,77
12331 ORTHEX CANADA ENR W			7 529,09. \$	6 507,75 S	8.355,47 S.		23,555,06
12384 SHOLA INC.	8	15.000,00	620,86		<b>63</b> .		3.620,86
72341 PEVOLUTION SANTE 9196445	.000		179,80	, 1	٠ ,	is a	1 179,80
12343_JAYMAR brix	છ. જ			'n.	ė		28 957,45
TZ363 EQUIPMENT SANITAIRE CHER		00'000	131,12	<i>of .</i>			1.131.12
18355 ADJ ARTDESIGN INTERNATIO		00000	891.08		1	.,	4.581,06
TRUE VALUE INC. W	, .	So notono oz	•		S		11 881
TAGOR TINGS FAIN FAIN ALIGNAL INC.			244				T. KKA, OS.
12367 CONFORT EXCELLEX INC			9.722,17				9.722,17
J2373 PPD FOAM WELD TON INC.	49, C	00'000'01	S DOS	<i>(</i> 2.	i,	₹i	Öö'n
1958年では、1月18日 11日本・公司に対し、公子の元本からなっていまして、日本の一日の一日の一日の一日の一日の一日の一日の日本の一日の日本の一日の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本	•			<i>y_</i> 1			1 404 CR
יעניים בייניים ביינים ביינים ביינים ביינים ביינים ביינים ביינים בייניים ביינים ביינים ביינים ביינים ביינים ביי		rt.	A CONTRACT				Je'bing to

(7).

siparaty con

₹

# AGED ACCOUNTS RECEIVABLE DOMFOAMINTERATIONAL INC.

•							
TOODS ELRAN	er (		-	\$ 27.441,52			00'000 050
1957 STECO INC. 19387 LTERIE DRUMMOND LTEF 19883 AFROND TO BARTHERSHIP	5 6 6 9 8 8	80 000,000 80 000,000 80 000,000	10.100.01 10.000.01	4 173,56 S 410,29 W	7 000 PEO 0	7 (S15,54 W	14,000,00
ATTRA DAR FOAM	64 65 65			r 1.		48 779,05	75.000,00
11389 SIEGES DUCHARME INTERNA	25. 52. 54. 55.	350 000,00	6.706i11 's	9.359.14 8 7755.01 8	56 494,97 \$	40 315,20	75 000,00
よしなし される ほほびひきにゅう グンドランドロ 1000	9 V			S 78789 B4			200,000,000
Tego, L. DAVIS TEXTILES (1991) INC	19		\$ 923,80	, ,	2.03	13 668,05	00,000,38
SUBITOTAL CDN lastred	8	1773 000,000 5	: 497 409,92 \$	232 854,79 \$ 126-374,93 \$	158 815,02 · S	1.015,452,65 \$	844,000,00
10413 MATELAS MARTIN	80	375 000,000 \$		78,891 25, \$		159.635,38	80,000,00
JOSES SUBAR	.s. 06	325 000,000 \$		\$ 65 818,84		227 166,00 . \$	75 000,00
11177 TISSUS MORICO INC.	8	150 000,00	30746,85 \$	25 400,93 \$ 24,863,69 \$	8 426,10 . 5	.89 437,38 ·S	00,000,001
Jiess Bozanjojno.	6. 6.	\$ 00,000,000	138-756,63	5.117.018.63		382.973.57	25 000,00
SUBTOTAL DUNINGUIEG Net 90	ķ	1.250 000,00 \$	267 671 69: 'S.	189 982 90 \$ 262,869,81 \$	131 687,71 \$.	. '852-212,31- \$	280 000,00
The second secon		,	,		•		
SUB TOTAL GANNA GOUNT					Š.		
SECOND SECOND		ַנֵּע	2.416.202.87 ·S	843 485 23 .5.503(883),41, .5	338 088.37 S	3 901-639-88	
7. kg		].					
TOTALOFFER			-				
The state of the s			80%	90.09	20%		
SUB TOTAL ODN		50	810 672,64" \$:	252/388/62 \$128 783,20 \$	9513,53 \$	1 201 357,89	
			RAMA	BO%.	ROK		
SUB TOTAL CDN Instred:		is	298.44	S 7882	95.287,81. \$	. 609.271,60	
			809 809 809 809 809 809 809 809 809 809	50% 505 50 50 50 50 50 50 50 50 50 50 50 50	50%	F14"1957-98	
יחם זה וער כחש שצחנים שבניזה		8		מומו ומו מ		2010011	
		استنت				30.8	
SUB TOTAL GANNA COUNT					3		
TOTAL OFFER		[0]	1,269,721,72 : 5'	506 091,14 \$362 330,05 8	183 813,97. \$:	.2.321:956,87	
Hydroy Vat Int EO's M familianting 1800		1.750 nón 00 . \$.	253.8957.0°.5	\$ 834.27 \$		.254.229.37	-
12(02.4.2)PONGER FORM	. 8. . 9.	125.000.00	27 028 49.	3 30 276,59	- ja	71175,61	-

(M)

YABLE,	ģ
8	7
竝	중
DUNTS RECEIV	F
'n	ž
Š	뿣
	-
₹	Ö.
AGED	×
ď	ă.

Comment William Comments of the Comments of th

NAME	NET	CREDITLIMIT	CURRENT	\$1-60.DAYS. 61:90.DAYS	ÖVER BU DAYS	TOTAL
ULI 1771 LEFECO FIBRES'INC. UT216 CITS BED MANUFACTURING C UT2171 FRANKLIN AUGUST TRADING UT21818 FOANT FACTORY INC. UT21818 GREAT NORTHERN ASSOCIAT UT21817 CUINN DISTRIBUTORS UT21818 FRAUS, USA-INC.	\$ 2 8 8 8 8 8 8 8 8 9 8 8 8 9 8 9 8 8 9 8 9 8 9	2,030,325,00 % \$038,00 % \$1,035,00 % \$1,035,00 % \$1,73,00 % \$1,000,00 % \$1,000,00 % \$1,000,00 % \$1,000,00 %	\$56782,35. \$20.975,01. \$7170,44. \$12.576,70. \$1777,50. \$1777,50. \$1777,50. \$1777,50. \$1707,50. \$	358 231,677 \$164,791,355 \$5 53 559,68 \$7 \$125,495 \$6 \$74,985 \$6 \$75,985 \$6 \$75,985 \$6 \$75,985 \$6 \$75,985 \$6 \$75,985 \$6 \$75,985	An un top un un ib, un ib.	1,079,745,37 75 413,64 8 43,64 8 43,63 8 73,60 12 42,61,10 17,757,90 18 884,62
sus Total usp	[ <del>63</del> ]	2 645 787,00 \$	627.886,60 \$	416 708,56 S165 902,30 S	s ,	1.210 447.46
U14429. NRF-DISTRIBUTORS INC.	.s. Di	255 410,00 \$	* 043,90 %	. i . i . i . i . i . i . i . i . i . i	<b>9</b> 0	15 044.90 St. 35 100.000
SUB TOTAL USD Insured	<b>[4</b> ]	256 410,00 .\$	15.044,90 - \$	8 . 8	8	15 044.90 S40.000,00
Uleise wow inc.	\$ 06	1,380,671,00° \$	188 735,84. \$	3, 72,198,871,2, 37,191,037	36 247.43 \$	544,154,40 \$. 150,000,00
Sub Total USD Insured Net 90	\$	\$ 1:380-671,00 \$	168 703,64 .\$	160 191,76 · S 178 881,57 \$	36 247,43 'S	544.154,40 S. 150,000,00
SUB TOTAL GANNA COUNT			4.	ير و و و و و و و و و و و و و و و و و و و	American de la Lega de La Carte de La Cart	
the state of the second of the		ř				

	.423.656,61		5 265.72	11	190.454,04	80	,	76,376,919
10%		35%	'S	3638	12 686,60 \$		S	12 685,60 \$
.35%	145 848,00; \$ 58 065,81: :\$	35%	5	.38%	56 067,12 \$ .62-643,55 \$ .	- Landanian Company		120709,35 \$.
35%	145 848,00: \$	35%	S	36%	58 067,12 \$			201-915,11 \$ 120-709,35
35%	219.742,81 'S	3.5%	5 265,72 'S	35%	-69.066,777.18			1284.055,30 .\$
	so:		S		is			iń

576 900,32 \$344 883,87 \$

811 615,14 \$

971.975 \$ 576.900,32 \$344.883,87 \$ 36.247,52 \$ 9. 17.0

úì.

USD CONVERTED TO CON.
EXCHANGE BATE Jan 34, 2012
TOTAL CON GROSS
TOTAL CON GREEF.

SUR TOTAL GANNA COUNT

TOTALOFFER

SUB TOTAL USD Insured

SUB TOTAL USD Insured

TOTAL OFFER.

TOTALGROSS

**(E**)

Æ

Insurance											
TOTAL	(14-şûB,00).	(24,369,08),	(127,739,99) (1,158,95)	162,69	(1.454,44)	(1-403,14) (7'139,28)	F E 31 71	(4.801,00) (6.153347)	(\$14(\$01)).	(2 615,92) (16.269,76) (5.189,92)	(15-984,94) (6-523,36), (73,380,06).
DVER 90.DAXS		ώ .ω σ. <del>(</del> ά	rui or igrio is i	ક લક્ષ્યું કહે. જ	ny sip th	G to th	in iq in iq i	ક્ષું છે <b>છ</b> ેલ છે.	4 30 xx is ri-th rit	ત્યુ માં. મર્ટ મર્ટે. મર્ટ નેટ	44.05) 44.05) 44.05.00.00.00.00
61-90 DAYS			Ú14.937,971	162,69	(1.554,44)	(1 030,47)			.(614;01).		. (4461,72). S
31-50 DAYS	(189,52)		(12 802,92)  ફ. (14,937,97)	· 0j·	ωi·	(6-108,81) \$	•		b		(115,524,84) \$ (115,49)
CURBENT	\$ (89'808.41)	(24.868.08)	(56'85) ().		,	(1'403,14)	·	(4 eth. co) (5.193,47)		(2.615.92) (16.269,76) (5.194,22)	(900ge st)
		ų,	is is			ú) ·		49 (J):		<i>ம்</i> . மு. மு	la.
ÇREDIT'LIMIT'	25.000;00 25.000;00 15.000,00	00,000 \$ 00,000 350 00,000 \$	30,000,00 976,000,00 725,000,00 50,000,00 50,000,00	175 000,00 175 000,00 40,000,00	5.000,00 30.000,00 10.000,00	50-000,00 75:500,00 75:000,00	90,000,000 20,000,000 20,000,000 20,000,00	15,030,00 50,000,00 16,000,00 60,000,00 33,000,00	2,000-02 8,000-02 90,000-03 8,000-00 9,000-00 9,000-00 9,000-00	15 000,00 27'000,00 27'000,00 100,000,00 25 000,00	25 600,00 26 600,00 26 600,00 27 600,00 27 600,00 28 600,00 28 600,00
NET			<u> </u>		00 00 00 00 00 00 00 00 00 00 00 00 00	9 4 5 8 4 8 8 4 8	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	a or		<b>ଜ୍ଞାନ୍ତ ଓ ପ୍ରତ୍ର</b> ପ୍ରତିଷ୍ଠ କୃତ୍ରି ପ୍ର	384848499 3848489
NAME	GDN 157 VENREZ PRODUITS 202 BUGATTI DESIGN INC 281 DINE ART INC	SUB FLUSHIELE, TINNOVET IDVINOV. SUB LIND FURNITURE (CANADA)LT 10220 LBCCANADA ING 10024 ARWAY SUBGICAL APPLICAN.	10052 ALPHAVIC INC. 10072 KRAUS CANADA INC.(HO) 10080 CUALITY PLOSS OF CANADA U. 10120.DRAVERS CARPET L.TEE. 10138. BOBLEN INC.	10222, MATELAS CONDOR ENRG. 10244 MATELAS DAUPHININO. 10273. DOUBLETEX INO. 10273. GOSO BERNINETT FLEET (QUEREC) IN	TOSST MACHENIA FOR TOSST	70379 MATELAS LANOLETTE:NG. 10304 TEXTILES LETOURNEAU 10385 MATELAS LION DX3R INC.	10426: CANADEL. 10465: COUPONS PATENT IND. 10494: BENBOUPIRAGE A.N.P. 10502: LITERIE PROVINCIALE, L'IEE.	1(150) AMEUBLINGURI, FENAISSANG 1(150) MATELAS RENE ING. 1(152) REMBLICHIS ROUSSEAUING. 1(1545: SEAWAY RIASTICS UTER 1(1547: SERV, CORIR. DU CANIDA.	105/0 SPC941 MASKA NAU: 10596 THANISOMADA INDUSTRIES 10737 FUTON TTH HENVENUTOPIA ( 10745. GROUPE GIBO:INC. 10745. PRINCESSE 10646. PTEXTILES ENRG. 11044.LITERIE:SEDUCTION ING.	11 150. JABOIN DE VILÇE (NG. 11162. JABOIN DE VILÇE (NG. 11197 MSGAN ING. 11210; ARROQUATINO. 11210; PROQUATINO. 11289. ROQUILLAPO ING.	11272 INTER CONFORT 11226, STEERS, SRODP LIMITED 11229, MATELAS SIESTA, 11415 HOKKLAND TEXTILES INC. 11457: MATELAS PRESTIGE 11558: QUATRO, TECHNOLOGIES INC. 11577: MATOPES PLUS.INC, 11690 FUTON DESIGN.

(m)

:Rebales

(3)

(50 230,99) \$ (122,358,27)

α. ω.	<b>∽</b>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6. 1 1 1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	. எப்படிரும் இது இதிசு ஒரு ஒது இது இது இது இது இது இது இது இது இது இ
			(83,85)	
	(14 128,04)	(40,99)	uş.	,
(ප භාගීම).		\$ (92,181,28)	.; .; .; .; .; .; .; .; .; .; .; .; .;	
2 000,00 2 000,00 3 000,00 3 000,00 12 000,00 50 00,00 50 00,00 50 00,00 50 00,00 50 00,00 50 00,00 50 00,00 50 00,00 50 00,00	116 000 000 000 000 000 000 000 000 000	145,000,100 & 14	15 (00) 40 15 (00) 40 20 (20) 40 20 (20) 40 20 (20) 40 20 (20) 40 20 (20) 50 20 (20) 50 20 (20) 50 21 (20	1 \$ 000,000 75 000,000 20 000,000 20 000,000 10 000,000 10 000,000 15 000,000
· 罗岛北宫盟坚坚备穹垒; 吨分分外外和加州地	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	Post 25 6 8 8 8 8 8 8 8 9 9 9 9 9 9 9 9 9 9 9 9	00 00 8.4888878 8.8888 0000000000000000000000000000000	88888 844 AAAAAAAAAAAAAAAAAAAAAAAAAAAAAA
11627 PRO TACINDUSTRIES'ING. 11676 ILISTIE DORMATEX ING. 11676 ILISTAR, CANADA ING. 1174 TRICA. 11760 VERANDA: UNC. 11776 WANUFACTURE DE MATELAS. 11817 BROLAIN DISTRIBUTORS, LED. 11828: COMPCORT, FOAM, & FIBRE. 11838, MATELAS SELECT. 11848, PRELINTERED. 11846, PRELINTERED.	11999 NUCKERAKINO, 11999 WESTBORD FLOORING INC. 11914 EXCEL RSVQ MOUSSE FOAM I 11994 WESTBORD FLOORING INC. 11992 MEUBLES POMALI INC. 11997 STRITEC INC. 11997 STRITEC INC. 11997 MATELAS ORTHOPEDIQUE 11998 PELPRO MEDICAL INC. 11998 MATELAS ORTHOPEDIQUE 11999 MATELAS ORTHOPEDIQUE 11998 MATELAS ORTHOPEDIQUE	12021 SEARS JANKUA ING. 1201) GRE O PAGKINTERNATIONAL, 12104 AO MATS CORPORATIONAL, 12104 AO MATS CORPORATIONAL, 12104 DOWNLAST. PLASTICS LIMITED 12104 DOWNLAST. PLASTICS LIMITED 12104 DOWNLAST. PLASTICS LIMITED 12104 DOWNLAST. PLASTICS LIMITED 12106 MATELAS BEAU REVEING 12106 MATELAS BEAU REVEING 12206 WESCO, NA, ING. 12206 SPORT SYSTEMS CANADA. IN 12213 GRADIEU CANADA. 12215 GEOLIUMOVATIONS 12210 COULTURE GAPILYLE ING. GW	1227 DOMINION OUTTRIG.SERVICE 12282 VIÇLINE 2006 INC. 12290 LOGIFLEK MOB DE BUREAU 12291 PAKTEK INC. 12292 BAKTEK INC. 12203 WES.INDUSTRIES INC. 12203 GREATECH INC. 42205 WARTHER. EMBRAL AGE WINL 12230 GREATECH INC. 9090744 CUE 12230 GREATECH INC. 9080744 CUE 12230 GREATECH INC. 9080744 CUE	1234: REVOLUTON SANTE 9198445 12343: EQUIPMENT SANTAJRE-CHER 1235: ADJ, ART DESIGN INTERNATIO. 1235: VALENTINO ÍNO. 1235: UNGRA P. INTERNATIOAL INC. 1236: UNGRE P. INTERNATIOAL INC. 12373: PPD FGAM, SOLUTION INC. 1236: ONFORT EXCELELEX INC. 1236: ONFORT EXCELELEX INC. 1236: ONFORT EXCELELEX INC. 1239: QONOUEST CARRET CORR.

SUB TOTAL CON-

IED ACCOUNTS RECEIVABLE	DOMFOAM INTERNATIONAL INC.
Ž.	Š.
9	6

Rébaigs

	1771 143/58) \$ 350 000,000 \$ 50 000,00 14 000,00 75	(256.570,57) \$ 844,000,00	00,000 257 00,000 201 00,000 201	(10,577,75) \$. 280,000,00		(556.989,05)	(178118,68)	(153:942,34)	(6°346,65)		(333 404,65)	i r
	14 10 04 05 10 10 05 16 10. (GE 692 ±1).	5) .\$ (05,097,71).	# 80 (BL/V/\$00)	(10 577,75) .5	\$	(30.280,00) s. (6	ż	7,06: (10.656.18)	50% (6.348,65) \$	99	(17 e91,22) S (55	ig ig.
*	(143.702.06); \$ (27.44°1,52);	(143 702,05) S. (27 441,52) \$	ų.	φ		(193 933,05) 'S (149 799,78) -\$	s (7341	60% 60% 60% 60% (16.464,91) 5:	\$ \$ \$ \$ \$		(118 359,89) \$ (89 879,87)	
;	(18 833,96) (48 832,73):	· (67-885,69) \$		4		(182 956,21)	60% (69.173.71) \$	(40.600,01)	50%		\$ (67,677,601)	
	15. 5. 45.000,00 30. 45.000,00 30. 45.000,00 30. 45.000,00 30. 45. 45.000,00 45. 45. 45. 450.00,00 45. 45. 45. 450.00,00 45. 45. 45. 45. 450.00,00 45. 45. 45. 45. 45. 450.00,00 45. 45. 45. 45. 45. 45. 45. 45. 45. 45.	\$ 00,000,000 \$	30 \$ 375,000,00 90 \$ 425,000,00 90 \$ 450,000,00 30 \$ 400,000,00	. s 1.250.000,000 S.		on	[9]	9			n	45 .\$. 1.750.000,00 60. \$ 125.000,00
•	10006 JELRAN 10371 SYLCOJING. 10387 LITERIE DRUMMOND LTEE. 10551 GESCO LIDPARTNERISHIR. 11364 O. & R. POAM 11399-SIBGES DUCHARME INTERNA 11421 LESMEUBLES VANTYGE 1000 11824 MATELAS D'OR 9083-9834 QO 11324 LLOAVIS, TEXTILES (1991) INC	SUB TOTAL CON Insured	10413 MAJELAS MAFTIN 10885-SUBAR- 10127-TISSUS-MORICO INC. 11685-BOZANTO INC.	SUBTOTAL CON Insured Net 90	SUBTOTAL GAINA COUNT	TOTAL GROSS.	TOTAL OFFER. SUB-TOTAL CDN:	SUBTOTAL CONTRUCED	SUB TOTÁL RON insured Met 90.	SUB TOTAL GANNA GOUNT	TOTAL OFFER	11882 value fóam indústries unc 12102 a 2 spongéa. Foam

4

Rebates

(b)

Consplidated

(g)

Cansolidated

22 22 22 23 24 24 24 25 25 25 25 25 25 25 25 25 25 25 25 25	1.139, 129 4.681, 10. 11.681, 23. 17.22, 17 14.50, 04 14.50, 04
Company of the compan	H. C. L.
22 22 22 23 24 24 24 25 25 25 25 25 25 25 25 25 25 25 25 25	A CONTRACTOR
7.4.04.7.7.4.04.0.0.0.0.0.0.0.0.0.0.0.0.	
を	• • • • • • • • • • • • • • • • • • • •
4	000000 0000000 00000000000000000000000
යි. යි. යි. සිම්බන්ගේ සිම්බන්ගේ සිම	පිලිල <u>පුර්ණී</u> අදෑ කිස්ව
11027 PRO-TAO INDUSTRIES INC; 1104-TRICA 1750 WERAKISANADAINO; 1104-TRICA 1750 WERAKISA LARDIN INC 11011 GRCLAIN DISTRIBUTORS.LTD. 11022 COMFORT FOAM & FIBRE 11032 COMFORT FOAM & FIBRE 11034 PEL. INTERNATIONAL LIFE 11046 INDUSTACE INC; 11057 WILLES YOUR OF LOOPING INC; 11057 WILLES YOUR OF LOOPING INC; 11057 WILLES SELECT TOWAL INC; 11076 MATELAS CELECT TOWAL INC; 11076 MATELAS CHAPLED IND. L. 12017 CAR DELES BELISLE INC; 11076 DE PACKNED OF LOOPING, 11076 SERBY CANADAINO; 12101 CAR DE PACKNED OF TRICE, 12104 AD MATELAS DE PASTOCE LTD. 12104 AD MATELAS DE PASTOCE LTD. 12105 MATELAS DE PASTOCE LTD. 12105 MATELAS DE PASTOCE LTD. 12205 WESTOC NA INC; 12205 BEANDOMRAS PLASTOCE LTD. 12205 WESTOC NA INC; 12205 BEANDOMRAS PLASTOCE LTD. 12205 WESTOC NA INC; 12205 GARRING CARLYLE INC; 12205 COUTURE CARLYLE INC; 12206 CARLES ON INC. 12206 CARLES ON INC. 12206 CARLES ON INC; 12206 CARLES ON INC. 12206 FANAR PUBLING INC; 12206 FANAR PUBLING INC; 12206 FANAR PUBLING INC; 12206 FANAR ECH INC SUBSOTA OUE 12207 GRIPHES AN RELETINE INC; 12206 CARLES CANADA ENR W 12207 GRIPHES AN RELETINE INC; 12208 GRIPHES AN RELETINE INC; 12209 CARLES CANADA ENR W 12209 LARDING INC. 12201 GRIPHES AN RELETINE INC; 12201 GRI	12868 FCU IPMENT SANTARE CHER. 12857 MALENTINO INCW. 12858 LINEAR: INTERNATIONAL INC. 12857 PEP FORM SOLUTION INC. 12897 PEP FORM SOLUTION INC. 12899 CONFULION SECURIA SEGRETICA SEGRETICA SECURIA SEGRETICA SECURIA SEGRETICA SE

**©** 

SUB TOTAL CON

min had

# AGED ACCOUNTS RECEIVABLE. DOMFOAM INTERNATIONAL INC.

Consolidated

And the second s

	00 00 00 00 00 00 00 00 00 00 00 00 00	0000	000,000						
	350.000,00 14,000,00 14,000,00 75,000,00 75,000,00 25,000,00 35,000,00 35,000,00	80.000.00 90.000.00 75.000.00 100.000 25.000.00	00 082.						•
<b>*</b>	344,427,79 5 22.246,79 7 815,54 6 55 738,28 5 160,681 7 15,283,31 6 2,5881,29 7 15,888,05 8 15,888,05 8	158 882,09 °S 166,00 °S 221 166,00 °S 89 457,38 °S 377 095,82 °S 377 095,82 °S	841,634,56 'S'	344 670,83	028 242.23	455'329,25	504.980,74	1.988.552,22	254,529,37
	683,683 693,683 693,683 684,893 684,803 684,803 684,803 684,803 684,803 684,803 684,803 684,803 684,803 684,803 684,803 684,80	36 (51,22 S. 67,225,28 S. 87,225,28 S. 87,225,39 S. 87,395 S. 87,3	121/109,96 .\$	307.788.37 S 3	20% 8 125,14 S	631.63 631.63	72.665,98. S.	166 422,75 \$	ii iir
•	1056,44 \$ 11342,99 \$ 1178,56 \$ 410,29 \$ (2.709,29) \$ 7.755,01 \$ 8.702,70 \$ 5.805,18 \$ 31.178,08 \$ 73.739,94 \$ 5.3744,15 \$	99 152,73 \$ 98 83 841 \$ 19 370,39 \$ 15 166 £7 \$ 32 883,37 \$ 65 818 64 \$ 25 400,88 \$ 24 863,69 \$ 112,323,38 \$ 117,018,51 \$	189 982,90 \$ 262 869,81 \$	649.552,7B .\$454.083,62 .\$	\$ 55.06	60% 691,64 \$ 59 60%	146.889,74 \$ 157.721,89 . \$	389731,31* \$272,450,17, \$	3 870,53 - \$ 30.276,59 \$
	2057729.E8 19 1593.29 2657.29 2657.29 2657.29 20 20 20 20 20 20 20 20 20 20 20 20 20 2	42.935,08 \$ 55,236,52 \$ 90,746,68 \$ 138,756,19 \$ 138,756,19 \$ 138,756,19 \$ 138,756,19 \$ 1	267 671,89 18.	71.933 248,66 S:	w	is,	160 603.13° S	7 159 948,00 .\$	263,895,10 \$ -27,028,48: \$
	500,000,00 % 30,00	375-000.00 \$ 375-000.00 \$ 575-000.00 \$ 400.000.00 \$	* 1'250'000,00 \$		(9)	5	8	us .	\$ 1750.050,00 \$ \$ 125.000,00 \$
	在民宫留宫花园花花	9 8 8 8 9	.c.					. i	8. 60 3. 60
	10005 ELPAN. 10371 SYLOO INO. 10371 SYLOO INO. 10387 LITERIE DRUMMOND LTEE 10551 GESCOLTD PARTINERSHIR 11164 DE R FOAM 11398 SIEGES DUCKARME INTERNA- 11424 LES MEDIBLESVANTAGE 1000 11924 MATELAS D'OR 8053-9834:QC -72301 L. DAVIS TEXTILES (1997) INC-	SUBTOTAL CONTREPTED 1055 SUBAR 1055 SUBAR 11177-115SUS MORIOGING. 11655 BOZANTO, INC.	SUB TOTAL CONTINED Net 30.	SÚB TOTAL GANNA COUNT TÓTAL GROSS.	TOTAL OPFER	SUBTOTAL CDM Insured	SUBTOTAL CDM,Insured Net 89 SUB TOTAL GANNA COUNT	ТОТАЦ ОРРЕВ	T1682 VALLE FOAM INDUSTRIES ING 12102. A Z SPONGE RFOAM

(=)

H	ن
景	垩
	Ž
品品	ē,
	₹
SE	딾
S	峑
ၓၟ	歪
ã	D.
병	8
٠.	73

Consolidated

Dated dan 91,2012

	ואַנוסו.	20.55 86.50 10.00	1.11. 1,86 S. AP.000109.	,86 . \$ .40.000:00	140 8: 150 000,00	.40 S 150 000,00	·	37.		09	. 40		86	in the second se
	<u>[G</u> ]	934.846,02 75.413,44 8:393,69 235,04 12.428,10 12.428,10 17.757,90 3:484,52	765 638,10	11 415,86	544 154:40	544,154,40		1,321-211,37	267.970.34	3 958,60	190 454,04		462 423,98	1 321.211,37 467.403.98
	OVER 90.DAYS	糖锅切碗的锅箱 1.1. 年7. 平年1. 年	ii ii	S	36 247,43 5	36.247.43 \$	•	36.247,43 \$	.40%	2	35% 12.686.60 S	တ	12,686,60 \$	36-247,43 - \$
<del>-</del>	61-90 DAYS	792 88 89 70 88 74 89 70 88 70	156 · S · 1110,95 S:	(A)	\$ 72,189.37,18. 37,181.57	160 191,76 \$ 178 981,57 \$		295 882,32 .\$180 092,52 .\$	- is	3576 3576	35%; 35% 7,12 '\$ 62 643,55 \$		181 5, 63 032,38 · 5	\$ ද්ර <sup>2</sup> ර්ජ රජ් <sup>2</sup> ද ද 180 රජ් <sup>2</sup> ද ද
	IT 31-60 DAYS	00000000000000000000000000000000000000	\$ 126 590	J6 &	æ- ₩	Ş.		c <sub>2</sub>	48.74 &	6	90 98 .S.	,	103 608;81	
	CURRENT	568,732,35 20,979,01 7170,44 5 12,579,01 5 12,429,10 130,10	\$ 627.B36.60	. s 11418.86	49)EEL 1891.64	. \$ .168753,64	nt and significant and signifi	5, 807 989,10	1 3	3 396.60	35) 5. 59 055,77		\$ 282.795,19	. 8
	स्तः टब्र्ह्वार LIMIT	45 S. 2 000 325,00 60 S. 406 055,00 50 S. 7 083,00 50 S. 15 172,00 60 S. 46 110,00 45 S. 73 886,00 50 S. 73 886,00	S 2.645767.00	\$ 256 410,00	.90, \$' 1.380,671,00	\$ 1.380'671,00.				•	-			F3
	NAME.	USD USTATA, JEFECO FIBRESING USTATA FRANKUN AUGUST TRADING UZTATA FRANKUN AUGUST TRADING UZTAS FOAM FACTORY INC. UTZIRG GREAT NORTHERN ASSOCIAT UTZIRG GUINN DISTREMIN ASSOCIAT UTZIRG KRAUS USA INC.	SUB-TOTAL USD U11429. NRF DISTRIBUTORS INC.	SUBLIOIAL USD INSUIRE	Uzita Wawing	SUB TOTAL USD Insured Net 90	SUB TOTAL GANNA GOUNT	TOTAL GROSS.	notal opeer sub total, usd	SUBTOTAL USPInsured	SUB TOTAL USD lasured	SUB TOTAL GANNA COUNT	TOTALOFFER	USD CONVERTED TO CDN. EXCHANGE RATE (4n, 3), 2012 TOTAL CON GROSS TOTAL CON OFFICE

# AGED ACCOUNTS RECEIVABLE DOMFOAM INTERNATIONAL INC.

GROSS	-			
Amount Before Rebates (AR)	Ì		. !	; ;
SUBTOTAL	<del>69.</del>		USD to CDN	Total: 3.244.422,37
SUB TOTAL losured	69 <del>(</del>	1.015.452,66	<del>(1)</del>	1.030.497,56
SOB LOTAL Insured Net 90	<del>//</del> :		1	1.396.366,71
SUB TOTAL GANNA COUNT	: <del>/)</del> -	<del>i</del>	<b>!</b>	ì
TOTAL GROSS	<del>63</del>	3 901 639,88 \$	1 769 646,76 \$ . \$	5 671 286,64
WALLE FOAM INDUSTRIES INC.			₩	254.229.37
AZ SPONGE & FOAM			· <del>69.</del>	71 175,64
MISC. EXCHANGE RATE			<del>63</del> .	I.
TOTAL GROSS			<del>ф</del>	5 996 691,62
Rebate Amount (Rebates)		į	1	
			USD to CDN	Total
SUBITOTAL SUBITOTAL Insured	स् <del>ड स्डे</del> सं	(289 820,73) - \$ (256 570,57) - \$ (10 577 75) - \$	(444 809,35) \$ \$	(734-630,08) (260-196,61) (10-527-75)
SUB TOTAL GANNA COUNT	<del>, e9.</del>	Part I	ri R <del>igi</del> t	**************************************
TOTAL GROSS	67	(556 969,05)	(448 435,39) \$ \$	(1 005 404,44)
			·	
Combined Amount (Combined)	*	CDN	1 USD USD to CDN	Total
SUB TOTAL	. <del>6'9.</del>	•	į.	2,509.792,29
SUB TOTAL Insured	€42		i i	770 300,95
SUB FOTAL Insured Net 90.	. <del>69</del> .	84,1 634,56 \$	544 154,40 \$	1.385,788,96
SUB TOTAL GANNA COUNT	<del>(4)</del>		ï.	
FOTAL GROSS	s)	3.344.670,83 \$	1321211,37 \$ - \$	4.665 882,20
			•	<b>.</b>

2.450.976,20

462 423,98

1 988 552,22

671

TOTAL OFFER

# AGED ACCOUNTS RECEIVABLE DOMEGAM INTERNATIONAL INC.

II.	į	Y U
L	ī	ľ,
	Ĭ	L

Amount Before Rebates (AR)	***************************************	CDN	dsn	USD to CDN	Tofal
SUBTOTAL	éў				1 625 014,50
SUB TOTAL insured	ĠΒε	271,60		ı	614 537,31
SUB TOTAL Insured Net 90	<del>69</del>	511 327,39 \$		· <del>69</del>	701 781.43
SUB TOTAL GANNA COUNT	69-	. t	· <del>(A)</del>	te de la composition della com	- 4·
TOTAL OFFER	65	2,321,956,87	619 376,37 \$	Ġ	2:941:333,24

					v
Rebate Amount (Rebates)				<del>**</del>	•
	1	CON	asn	USD to CDN.	Total
SUBITOTAL	<del>6</del> 9	(173 115,66) \$	·(155.683,27) \$		(328 798,93)
SUB TOTAL Insured	ij.	(153'942,34)' &		· <del>é?</del>	(1,55.211,46)
SUB TOTAL Insured Net 90	69	(6.346,65): \$		1	(6.346,65)
SUB'TOTAL GANNA COUNT	<del>69</del> -	<del>(9</del> )	<del>ta</del>	i ( <del>C)</del>	, k
тото (меня	4	(333,404,65) \$	(156 952 39)	£ .	/4490 357 04)
		.1			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Combined Amount (Combined)		-		<del>-</del>	
		CDN	ūsn	USD to CDN	Total
SUBITOTAL	69	1:028:242;23\$	267 973,34 \$		1296 215,57
SUB TOTAL Insured	.es	455 329,25. \$	\$ 09,966 E	6 <del>9</del>	.459.325,86
SUB TOTAL Insured Net 90	. <del>64</del>	504-980,74	190 454,04 \$	49	695 434,78
SUB TOTAL GANNA COUNT	₩	1 <del>69</del> .	· i	<del>⊀€9.</del> I	i i

## Schedule 2.9(B)

# Purchased Inventories:

#### Section 1

# List of Purchased Inventories as of January 11, 2012

(and for some items as of January 31, 2012)

See document attached Total of one (1) page

#### Section 2

Calculation of the Purchase Price attributed to the Purchased Inventories.

See document attached Total of two (2) page

	ĭ	- Juneau	e 2:9(B) - Sec	HUIFT	
	<u>.</u>	. :		<del></del>	
	-	w9.6			_ _
	· XIII jā				
	1	le estronte les			
_:					
			SEC.	January 11, 2012	
	W MATER	RIAL			T
SĆF	RAP		1	155,026	
IL,N	<b>f</b>		2	52 547	T
JNE	ERLAY TUE	SING.	3	36 551	$\top$
(Ó)	I ČHEMIÇAL	ŝ	.5.	26 656	十
301	DED CHEM	IICAL	6.	7354	_
	MICALS (M/		7	810,266	(1
	IERS	1	·8 .	86 756	
_	LLCHEMIC	A1 S:	9:	505 644	I A
	YESTER FIL	<del></del>	. 40.	0 505 644	(1
		1	sub total:		
		-	ann rotat.	1 680 800 1 680 800	-
	RK IN PR	) OCESS	<del> </del>	1 Ιούο όσος	+
·					+
	IDED BLOC		11	10372	+
	DED BUNS		12	0	<u> </u>
		K/ETHER BLOCKS:	13	1 010 428	
	ND ESTER		14	25′374	+
-		UNDERLAY BUNS	15	0	1
OE	Ý ETHER BI	UNS'	16	143 382	1
<u></u> ,			sub total:	1 189 556	
	<u> </u>			1 189 556	_
0	TAL R.M 8	k W. IN P.	total:	2 870 356	
	<u> </u>			2 870 356	
NI)	ISHED GO	DODS: ,			
ÒŅ	DED UNDE	RLAY	17	0	
HIF	PING INVE	NTORY (GOODS)	18,23	113 574	(2
_			sub total.	113 574	
				113.574	1
<b>'</b> O'	TAL INVE	NTORY	TOTAL:	2 983 930	
	, , , , , , , , , , , , , , , , , , , ,			2 983.930	
					-
EC	TION 1	SUPPLIES	. 4	33.407	1
	TION 2	FIBER	2	3 763	$\top$
	TION:3:	BLOCS	3:	13:286	_
_	TION 4	BUN ENDS	4	1 827	+-
	TION 5	COMMANDES/stoc		19 798	
- Y		1-11/10/2	<del>-</del>  -		- -
'n	ral inve	NTORY	TOTAL:	72 081	+-
٠,٠	THE ILVE		· I.m. range	, 17 AOI	+
	<u>-</u>	·			- -
<u> </u>	ERALL	VALUE:	<u> </u>	3 056 011	+
<u>۷</u>	LIXALL	Awring		4.030.011	+
<del></del> -,	·				+
43.	القدائد سيثما	A Hagunanak a Hallah	in the Deep	nons on Ealistic 2012	_ _
T).				com on Feb. 14, 2012.	- -
~ ·					F-

	1	•	figer Purchase (SE)	
			Price 2	
	ŠEÇ.	January 11, 2012	offered:	
RAW MATERIAL		+		
SCRAP	.15	155 026	727,930 lbs @: 15	/lb.
FILM	Ź	.52.547		•
UNDERLAYTUBING	3.	36,551		
NON CHEMICALS	5	26.656	10% 2 666 See note 2 below	,
HONDED CHEMICAL -	. <u>a</u>	7 354		
CHEMICALS (MAJORS)	7 *	810 266	30% 204 168 See notes 1 & 2 b	เครื่อรับ
OTHERS	8	86 756	0%	C.013
SMALL CHEMICALS	· <b>9</b> •	505.644	The state of the s	
POLYESTER FIBRE	<b>3</b> :	•	50% 252 822 See note 2 below	
POLICOIENTENE	مساحدة عاديد	0.	- Care (1986)	
	śub lotal;	1 680 800 1 680 800		
wóbych spożeka.		1 090 000-		
WORK IN PROCESS				
BONDED BLOCKS equate:	11	10372		
BONDED BUNS round	12	Ď.		
SOUNTE BLOCK / ETHER BLOCKS		1010 428	30% 7 7 7 303 128	
ROUND ESTER BUNS	14	25 974	30% 7612	
ROUND ETHER UNDERLAY BUNS		0		
POLY ETHER BUNS	18	143 382	30% 7 3 78 43 015	
-	sup total:	1,189 556		
	_	1 189 556	<b>元</b> 學學學	
TOTAL R.M & W. IN P	lòlal:	2 870 356		
		2 870 356		
FINISHED GOODS	•		25 S S S S S S S S S S S S S S S S S S S	
BONDEĎ UNDERŁAÝ	17	O.	30%	
SHIPPING INVENTORY (GOODS)	18,23	113:574	See note 3 below	,
	sub total:	113 574		
	. ,	113,574		
TOTAL INVENTORY	TOTAL:	2,983,930		
•	, , , , , , , , , , , , , , , , , , , ,	2 983 930		
•				
SECTION 1 SUPPLIES	1	33 407		
SECTION 2 FIBER 1	2	3 763		
SECTION 3' BLOCS'	3	13.286	30%	
SECTION 4' BUN ENDS	4.	1 827		
SECTIONS COMMANDESA	· · · · · · · · · · · · · · · · · · ·	19798	305448	
TOTAL INVENTORY	TOTAL:	72.081		
10//4/11/0//10()(	io ium	1244)		•
Add purchase of uncounted II	tems with na son	cilicorice above.	20 000	
1 4 b bentage of gitablities in		a		
OVERALL VALUE		3 056 011	946 586	
The same section of Control of the same of		2 200 4.1	1 2500 1 24 260 1 .	

#### \*\* Designates counted inventory

Note 1. Total Chemical in inventory-		810 266
less TDI in inv (value \$0.00)	93 313 lbs @ 1.39/lb	129 705
Remaining inventory		680 561
•	Value at 30%	204 168

Note 2

The inventory regarding those items have been evaluated as of January 31, 2012.

Note:3

The Purchaser shall pay 100% of the latex based on the cost presented on page 107 of the document "Equipment and Inventory" posted on February 14, 2012 in the Data Room (see attachment) and the final count of Latex as of the Closing Date.



Sauf sur indication contraire, tout le matériel compté est bon et disponible immédiatement.:

21 (LAST SHEET)

# LATEX INVENTORY

TYPE	QUANTITY	(	COST	TOTAL	
Single Soft (TXL 65)		\$.	115:30	\$	٠.
Single Medium (TXL 76)		\$:	135,30	\$.	
Single Firm (TXL 85)		\$	204,95	\$	4
Single XTRA FIRM (TXL 95	)	\$:	185:00	\$	<del>-</del>
Twin Soft (T 65)	·	\$.	115.30	\$	
Twin Medium (T 75)	1	\$:	135,30	<b>*</b>	بن
Twin Firm (T 85)		\$	204:95	\$	
Twin XTRA FIRM (T 95)		\$	185.00	\$	~
Double Soft (F 65)		\$.	216,21	\$	_
Double Medium (F 75)		\$	230.40	\$	-
Double Firm (F85)		\$'	236.40	\$:	,,
Double XTRA FIRM (F 95)		\$	256.00 <sup>-</sup>	\$	
Queen Soft (Q 65)		\$	185.85	\$	<u>^</u>
Queen Medium (Q-75)		\$.	198.45	\$	
Queen Firm (Q 85)		\$.	315.31	\$	_
Queen XTRA FIRM (Q95)		\$	352,40	\$	
King Soft (K 65)		\$.	235.20	\$	-
King Medium (K75)		\$	250.95	\$	
King Firm (K 85)		\$	284.55	\$	÷
King XTRA Firm (K95)	J		330.75	\$	
CK SOFT (CK 65)		\$	325.00	\$	
CK MED (CK 75)		\$	325,00	<b>.\$</b> .	<u> </u>
CK FIRM (CK 85)		\$	360.00	\$	•
CK XTRA FIRM (CK 95)		\$	390,00	\$	-4.
5 ZONE		\$:	390.00	\$	-
7 ZONE -		\$	400.00·	\$.	,
PILLOW-		\$	20.00	\$	1

# Schedule 2.9(E)

# Excess rebates to customers

See document attached Total of two (2) pages

EXCESS UNAPPLIED REBATES

34 587,40

Control of the second of the s

Domfoam International Rebates As of Jan 27 2012

	REBATE AMOUNT
	OUTSTANDING OFFSET
OHETA	AGAINST ACCOUNTS
CUST # 10072	RECEIVABLE
10120	\$127,739.99
	\$1,769.76
10192	\$591.27
	\$22,390,75
10551	\$18,833.96
11207	\$2,249.62
U11429	\$3,626.04
12000	\$7;267.85
12253	\$3,606.38
00202	\$14,808,00
10005	\$171,143.58
1020	\$24,868.08
10305	-\$1,62,69
10358	\$1,554.44
10379	\$4,939,56
10384	\$7,139.28
10510	\$4,601.00
10545	\$6,193.47
10848	\$514,01
11155	\$4 <sub>1</sub> 186.45
11164	\$66,036.80
11209	\$16,269.76
11268	\$5,184.22
11289	\$6,523.36
11457	\$13,380,06
11666	\$10,577.75
11771	\$444,809,35
11844	\$5,922,65
11924	\$17,760,30
11927	\$11,128.04
12158	\$8,822.25
12282	\$5,716.50
	\$1,039,991.84

## Schedule 4.1(c)

# Confidentiality Agreement

See document attached nine (9) pages

#### CONFIDENTIAL

Donifoam International Inc. 8785 Langeller Blvd St Leonard; QC H1P 2C9

#### Attention John: Howard

Rer

Service and the first transplate and the service and

Domfoam International Inc., Valle Foam Industrice (1995) Inc. and A-Z Sponge & Foam Products Ltd. (collectively, the "Corporation") — Proposed Sale of Assets ("Proposed Transactions")

We have requested that the Corporation provide us (the "Interested Party") with the Confidential Information (defined below) to assist in determining for ourselves the desirability and feasibility of proceeding with the Proposed Transactions pursuant and subject to the terms and conditions set by the Corporation (the "Permitted Purpose") and contained having.

It is acknowledged that but for delivery of this letter the Corporation would not provide the interested Party such Confidential Information.

Upon acceptance of this letter by the Corporation and subject to the provisions thereof, the Corporation shall provide or cause to be provided to the Interested Party such Confidential Information as the Corporation in the exercise of its unfattered discretion may determine.

"Confidential Information" means all financial statements and information and all data, compilations, analysis, documentation, and other information and every part thereof, whether in electronic form or otherwise, which now or hereafter comes into the possession or under the control of the Information, and other information, and including, without limitation, that relating to the Corporation's business organization, business operations and systems, Know-how, business plans, Research Information, Customer Information, suppliers, financial structure, Financial Information, Supplied Information, Marketing Information, assets, employees, employees relationships, Software and other information of any kind relating to any of the foregoing and includes all data, compilations, analysis, and other documents and information prepared by or on behalf of the Interested Party or by any of the Representatives based, in whole of in part, on any of the foregoing, but expressly excluding any reports or analysis of the Corporation conducted by any professionals related by or with respect to the Corporation, including, without limitation, Deloitte LLP, in counsesion with the current financial state of the Corporation, any restructuring scenarios available to the Corporation, and any analysis of the value of the Corporation or its assets.

For the purposes of this letter:

(a) "Corporation" shall include and extend to each subsidiary of the Corporation, each partnership controlled by the Corporation or their

respective subsidiaries, each of whom shall be entilled to directly enforce the provisions of this letter against the interested Party not with standing that this letter is not directly addressed to or required to be accepted by any one or more of such subsidiaries and partnerships:

- (b) "Customer Information" includes any information pertaining to the Corporation's customers, prospective customers, pustomer base and markets, prospective customer mames and addresses and the names of cimployees of customers or prospective customers with whom the Corporation is in contact in its business, or proposes to contact, customer or prospective customer requirements and the Corporation's contracts with its customer requirements and the Corporation's contracts with its customers or prospective customers, including details as to pricing, supply of services; and royally and/or commission rates;
- (c) "Figure is Information" includes any and all information pertaining to the Corporation's actual or anticipated sales, income, projections, profit, profitability, pricing, salaries and wages;
- (d) "Know-how" includes the accumulated skills, experience, knowledge, patents, intellectual property, information, data, patterns, designs, engineering specifications, schematics and other information with respect to the design; production and provision of the products and/or services offered by the Corporation's business or proposed business and of the Software related thereto;
- (é) "Marketing Information" includes information including but not limited to the Corporation's marketing programs, plans, strategies and proposed future: products, services, adventising and promotions.
- (f) "person" shall be broadly interpreted to include, without limitation, any individual, corporation, limited liability corporation, company, group, partnership, limited liability partnership or other entity.
- (g) "Representatives" means, collectively, each of the directors, officers, employees, agents and other representatives of the Interested Party or the Corporation tas the case may be including, without limitation, their respective lawyers, accountants, consultants and financial advisors;

Suppfimet and

- (h) "Research Information" includes information pertaining to any research, development, investigation, study analysis, experiment or lest carried on or proposed to be carried on by the Corporation;
- (i) "Software" includes all methods, programs, techniques and other information; and materials relating to the handling or treatment of data by computers and all other necessary information utilized in order to provide the services offered by the Corporation's business or proposed business and all documentation thereto, and includes all computer programs, including

HTML, TAVA, UNIX, LINUX, GNU, G++, and other intercet oriented, computer programs, systems software and application programs, and all related manuals, documentation and materials relating to the systems software and application programs;

(j) "Supplied Information" includes all business information, computer software and technology which is proprletary to any other person doing business with the Corporation and which is made available to the Corporation under conditions of confidentiality.

At the request of the Corporation and for other good and valuable consideration (the receipt and sufficiency of which is hereby acknowledged) the Interested Party coverants and agrees with the Corporation as follows:

- 1. Deemed Confidential Property: As between the Corporation and the Interested Party, the Confidential Information; and every part thereof, is and shall at all times be deemed to be the confidential property and constitute valuable trade secrets of the Corporation, whether or not, but for this paragraph, the same would otherwise be considered confidential and/or trade secrets and whether or not it is marked or otherwise expressed on its face to be confidential.
- 2. Ownershipt The Confidential Information is and shall be deemed to be owned solely by the Corporation and the right to maintain confidential the Confidential information constitutes an exclusive proprietary right of the Corporation which it is entitled to project. The Interested Party does not and shall be deemed not to have any right to or proprietary interest in the Confidential Information. All of the Confidential Information which is disclosed to or otherwise comes into the possession or under the control of the Interested Party shall; subject to paragraph 5 below, he received and held by the Interested Party in trust safely for the Corporation notwithstanding the Interested Party's right to use it for the Permitted Purpose.
- 3. Permitted Use: The Confidential Information shall only be used directly by the Interested Party and its Representatives for the Permitted Purpose. Except as otherwise provided for herein, the Confidential Information will be kept strictly confidential by the Interested Party, whether or not such Confidential Information is marked or otherwise expressed on its lace to be confidential. Other than as permitted herein, the Interested Party will not, directly or indirectly, without the prior written consent of the Corporation (which consent may be arbitrarily withheld) disclose, publish, reproduce or otherwise disseminate or communicale the Confidential Information, nor will it permit, cause or acquiesce in such disclosure or other dissemination or communication of the Confidential Information by the Representatives of the Interested Party or by any other person, all in any manner whatsoever, in whole or in part.

The Interested Party agrees to restrict the disclosure, dissemination or other communication of the Confidential Information solely to such of the Interested Party's Representatives who need to know the Confidential Information for the Permitted Purpose and the Interested Porty shall inform its Representatives of the ownership and strict confidential nature of the Confidential Information.

Supprimer of

Supprime: who sie

Supprimés éd

Supprimes in witing by the Interested.

Supprinté; and who agree in writing with the impiested Barly to be bound by the terras of this letter

- Disclosure List & Responsibility, for Representatives: The Interested Party shall maintain a current listing of each of its Representatives to whom any of the Confidential Information has been disclosed or otherwise disseminated or communicated and on each and every request of the Corporation; the Interested Party shall without delay deliver to the Corporation a complete and accurate copy of such list. The Interested Party shall similarly deliver on request, the following information:
  - (2) name of the any law limits, accounting firms and consulting firms (the "Firms" Vinvatural with the Interested Party as its Representatives;
  - (b) riame of the person in charge for each Pirm:
  - (c) business address and telephone number to reach the person in charge for each

Supprimerof any or all present set out on the read list.

The Interested Party shall be responsible and liable to the Corporation for any and all acts or omissions of each of its Representatives which if done or omitted directly by the Interested Party would be a breach of the provisions of this lotter.

The Interested Party shall safeguard the Confidential Information from disclosure or other dissemination or communication to any person other than as expressly permitted herein.

- Exceptions to Prohibited User. The obligations and limitations imposed on the Interested Party and its Representatives hereunder shall not apply to any Confidential Information:
  - (a) which is or becomes generally available to the public other than as a result of its disclosure or other dissentination or communication by the interested Rarty or its Representatives; or
  - (b) which becomes available to the interested Party or its Representatives on a non-confidential basis from a source other than the Corporation or its Representatives, provided that such source is not then bound by a confidentiality agreement with the Corporation or otherwise prohibited from transmitting the Confidential Information to the Interested Party or its Representatives by contractual, legal or fiduciary obligation or

át such

Supprimetr

(9) which is already in possession of the interested Party: provided that such Confidential Information is not subject to another confidentiality agreement of non-disclosure agreement with the Corporation.

In addition, the obligations and restrictions imposed on the interested Party and its Representatives hereunder shall not extend to prohibit the interested Party from discussing the Proposed Transactions with and disclosing any of the Confidential Information to:

(i) any other person who has signed a confidentiality agreement with the Corporation in substantially life form of the agreement constituted by this letter regarding any of the Proposed Transactions; and

(ii) any governmental or regulatory body in Canada having jurisdiction over the operations of the Corporation or any part thereof or over the Interested Party, in connection with its participation in any of the Proposed Transactions, provided that the disclosure, dissemination or other communication of any of the Confidential Information to such governmental or regulatory body is made on a confidential basis and such confidential basis is either provided for by applicable law or recognized in writing by the relevant body:

Notwillstanding the foregoing provisions of this paragraph 5, in the event that the Interested Party or its Representatives are required by law or governmental action to disclose any of the Confidential Information, the Interested Party and/or its Representatives will on a hest effort law is deliver to the Corporation prompt prior notification of such requirement(s) so that the Corporation may seek an appropriate protective order and/or waive compliance by the Interested Party or its Representatives with the provisions of this letter. If, failing the obtaining of a protective order or the delivery of such a waiver, the Interested Party is, in the opinion of its legal counsel, compelled to disclose, disseminate, or otherwise communicate the Confidential Information, the Interested Party may disclose, disseminate, or otherwise communicate the Confidential Information, but only to the extent so compelled, without liability hereunder.

6. Return of Informations. The Interested Party shall, on receipt of the written request of the Corporation, promptly deliver to the Corporation all of the Confidential Information which is in recorded form (whether such form he in writing or in electronically retrievable form or electronically stored data or otherwise), and each of the Interested Party and its Representatives shall not retain any copies thereof.

7. No Liability for Information: At the time(s) any of the Confidential Information is delivered to the Interested Party, the Corporation will attempt to include in such Confidential Information such materials which the Corporation considers to be reliable and relevant for the Permitted Purpose; however, the Corporation shall not have or incur any, and the Interested Party hereby waives any right to claim any, liability to the Interested Party or its Representatives in connection with or arising from the Confidential Information and the use thereof by the Interested Party or its Representatives.

The Interested Party acknowledges and agrees for itself and its Representatives that neither the Corporation nor any of its Representatives in providing the Confidential Information has or will be taken to have made either expressly or impliedly any representations or warranties as to the accuracy or completeness of the Confidential Information and the interested Party confirms that it is not relying on any representations of warranties of the Corporation of its Representatives in making use of the Confidential Information.

8. Indemnity: The interested Party hereby unconditionally indemnities and shall forever save harmless the Corporation from and against any and all lusses, damages, liabilities (whether actual, contingent of otherwise), proceedings, costs and expenses including, Without limitation, legal fees (on a scale as between a solicitor and his own client), which the Corporation or its Representatives may suffer, incur or sustain in

Supprime; written

connection will or as a result of a breach by the Interested Party or any of its Representatives of any of the provisions contained in this letter:

All of the provisions contained herein are reasonable in the circumstances and valid and the interested Party hereby walves all defences, equities and rights to and against the strict enforcement thereof by the Corporation and its Representatives.

9. Continuity of Agreement: The prohibitions, limitations and obligations of the Interested Party herein contained shall continue in full force and effect notwithstanding the Interested Party ceasing to proceed with any of the Proposed Fransactions and will survive any termination or cancellation of its involvement, if any, in the Proposed Transactions.

10. Use of Information Post Involvement: The Interested Parly shall not, at any time after it ceases to proceed with or be involved in the Proposed Transactions and except as otherwise expressly provided for herein; (i) use any of the Confidential Information in furtherance of the business of the Interested Parly or the business of any other person; (ii) interfere in any way with any confractual or other business relationship of the Corporation; (iii) disclose; disseminate or otherwise communicate the Confidential Information or any part thereof to any person or utilize the Confidential Information in any way which is adverse to the Corporation; or (iv) directly or indirectly solicit for employment any person who is now employed by the Corporation, unless the Corporation otherwise agrees in writing prior to such solicitation for employment.

11. Right to Enjoin: It is acknowledged that the provisions of this letter are essential for the protection of the Corporation and, in the event of the non-performance of any provisions hereof in strict accordance with their specific terms or any breach of any provisions hereof in strict accordance with their specific terms or any breach of any provisions of this letter, the same would enuse immediate and irreparable harm and damage to the Corporation for which monetary relief would be inadequate or impossible to ascertain. The Interested Party agrees that upon any breach or threatened breach of the provisions hereof, the Corporation shall be entitled to obtain from any court of competent jurisdiction, interim and permanent injunctive relief or other appropriate form of equitable relief to effectively enforce the provisions hereof, and an accounting of all profits and benefits which may have been or which may be derived both directly or indirectly, as a direct or indirect result of such breach by the Interested Party or any of its Representatives who have committed or who have threatened to commit such breach, which rights and remedies shall be cumulative and in addition to any other rights of remedies to which the Corporation may be entitled.

No failure or delay by the Corporation in exercising or taking steps to enforce any of its rights or entitlements under the provisions of this letter shall operate as a woiver thereof, unless such waiver is in writing and signed by the Corporation, nor shall any single of partial exercise thereof proclude any other or future exercise of any other right or entitlement of the Corporation hereunder.

12. Notices: Any notice desired; contemplated or required by this letter to be given shall be delivered by hand or facsimile to:

Supprimet rights of set off,

(a):	the Corporation at:	•	
	8785 Langelier Blvd.		- (Suppriméi
	St Leonard; QC HIP	2C9	- Supprimés .
	Attentions	The state of the s	Supprimer y
•	Telephone No.	John Howard 514-325-8120	
	Facsimile No.	514-325-6477	Supprimő:
	r destriction	7 (4-255-044 K	Supprimé:
	• :	•	Supprimé:
· (Þ)	the Interested Party at	:	Supprinté:
	,8600 Decarle Blvd		
	Sinto 300	سمحيدوستين ولانتان والمحاورة	- Supprimé:
	Town of Mount Royal	LING LLID OND	(ma-1-4-
	trofin of historic reals	FXX 1341. XIX	Supprimés
	Attention:	Frank I. Gattinger	Supprimé: 9
	Telephone No.	314-341-8600	Supprimé: Quelisc
	Facsimilé No.:	\$14-340-1389	Supprime:
	•	A STATE OF THE STA	Sùpprimés
	With copy to:	,	
	480 Saint-Laur Suite 200 Montreal, OC Allention: Tulephone No. Facsimile No.	112Y 3Y7	
enforc	le Law: The provisions sed exclusively in accord in applicable therein:	s of this letter shall be governed by and construed and dance, with the laws of <u>Quebec</u> and the federal laws of	- Supprimé: Catallo
ne letter by s nterested Par	signing and returning the rty, whereupon this let	ence its agreement and acceptance of the provisions of eenclosed copy thereof which has been signed by the er will constitute the binding agreement between the with respect to the subject matter thereof.	
	•	Yours very truly,	•
• •		S.P. Holdings Canada Inc. or Nominee	
÷	7	Per TI PENERINATE	
			Supprimé: Ter:

TOE

S.P. Holdings Canada Inc. or Nomince

Accepted and agreed to with effect as of the date set out above.

DOMFOAM INTERNATIONAL INC.

Per:

Per: TOHN C. HOWARD

| 18DV-730

ge 7 : [1] Supprimé			2011-11-25 08:42;
			·
,	Pēr:		
•		· · · · · · · · · · · · · · · · · · ·	
•			•
			•
• :			
•			
			•
•			
• •			
	•		
•			
		•	
•	•		
			~ -
	•		
,		-	
4			
-			,
•			
•			,

# Schedule 5.1(o)

List of assets to be acquired by the Vendor

From Valle Form Industries (1995) Inc.

Prior to the Closing Date

See document attached one (I) page

### Equipment to be Acquired From Valle Foam

Bonded foam peeler and poring machine
Foam stredding line c/w blower
2 shredded foam storage bins c/w flutter arms, gearboxes and drives (2 units @ \$5,000 each)
Baumer contour saw c/w infeed and outleed conveyors.
Wynter contour saw c/w infeed and outleed conveyors
Convoluter - Edge Saw 88" capacity
Carrousel (of Chinese manufacture)
Automatic wrapping machine (built at Domfoam in 1996)
Automatic vertical saw
Zwick-foam tester

Dismanile, transport and reassemble at Domfoam.

Additional Items.
Foam shredding line ofw blower floor scale
Laminator unwind set-up
Laminator heating system.
Underlay sealer
Web align unit
Air compressorH.P. boiler 50 HP

Dismantle, transport and reassemble at Domfoam

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE MR.	)	FRIDAY, THE 16th DAY
	)	
JUSTICE BROWN	)	OF MARCH, 2012

ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

ARRANGEMENT OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES (1995) INC., DOMFORM INTERNATIONAL INC., and A-Z SPONGE & FOAM ARRONDETS LTD.

(the "Applicants")

# SALE APPROVAL AND VESTING ORDER (re Domfoam International Inc.)

THIS MOTION, made by the Applicants for an Order approving the sale transaction (the "Transaction") contemplated by an asset purchase agreement (the "Sale Agreement") between Domfoam International Inc. (the "Debtor") and 4037057 Canada Inc. (the "Purchaser") dated March 8, 2012 and Confidential Supplement to the Third Report of Deloitte & Touche Inc., in its capacity as Courtappointed monitor (the "Monitor"), dated March 13, 2012 (the "Report"), and vesting in the Purchaser the Debtor's right, title and interest in and to the assets described in the Sale Agreement (the "Purchased Assets"), was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the affidavit of Tony Vallecoccia sworn March 13, 2012, and the exhibits thereto, the Report dated March 13, 2012, and the Confidential Supplement attached thereto, and on hearing the submissions of counsel for the Applicants, counsel for the Monitor, and no one appearing for any other person on the service list, although properly served as appears from the affidavit of Victoria Stewart sworn March 13, 2012, filed,

- 1. THIS COURT ORDERS AND DECLARES that the Transaction is hereby approved, and the execution of the Sale Agreement by the Debtor is hereby authorized and approved, with such minor amendments as the Debtor may deem necessary. The Debtor and the Monitor are hereby authorized and directed to take such additional steps and execute such additional documents as may be necessary or desirable for the completion of the Transaction and for the conveyance of the Purchased Assets to the Purchaser.
- 2. THIS COURT ORDERS AND DECLARES that upon the delivery of a Monitor's certificate to the Purchaser substantially in the form attached as Schedule A hereto (the "Monitor's Certificate"), all of the Debtor's right, title and interest in and to the Purchased Assets described in the Sale Agreement shall vest absolutely in the Purchaser, or such related entity as it may direct, free and clear of and from any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, trusts or deemed trusts (whether contractual, statutory, or otherwise), liens, executions, levies, charges, or other financial or monetary claims (direct, indirect, absolute or contingent), assignments, judgments, executions, writs of seizure and sale, options, adverse claims, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured or otherwise (collectively, the "Claims") including, without limiting the generality of the foregoing: (i) any encumbrances or charges created by the Order of the

Honourable Justice Newbould dated January 12, 2012; (ii) all charges, security interests or claims evidenced by registrations pursuant to the *Personal Property Security Act* (Ontario) or any other personal property registry system; and (iii) the Claims (all of which are collectively referred to as the "Encumbrances", which term shall not include the permitted encumbrances, easements and restrictive covenants listed in the Sale Agreement) and, for greater certainty, this Court orders that all of the Encumbrances affecting or relating to the Purchased Assets are hereby expunged and discharged as against the Purchased Assets.

- 3. THIS COURT ORDERS that for the purposes of determining the nature and priority of Claims, the net proceeds from the sale of the Purchased Assets shall stand in the place and stead of the Purchased Assets, and that from and after the delivery of the Monitor's Certificate all Encumbrances shall attach to the net proceeds from the sale of the Purchased Assets with the same priority as they had with respect to the Purchased Assets immediately prior to the sale, as if the Purchased Assets had not been sold and remained in the possession or control of the person having that possession or control immediately prior to the sale.
- 4. **THIS COURT ORDERS AND DIRECTS** the Monitor to file with the Court a copy of the Monitor's Certificate, forthwith after delivery thereof.
- 5. THIS COURT ORDERS that, pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act, the Debtor and the Monitor are authorized and permitted to disclose and transfer to the Purchaser all human resources and payroll information in the Company's records pertaining to the Debtor's past and current employees, including personal information of those employees listed in the Sale Agreement. The Purchaser shall maintain and protect the privacy of such information and shall be entitled to use the personal

information provided to it in a manner which is in all material respects identical to the prior use of such information by the Debtor.

### 6. THIS COURT ORDERS that, notwithstanding:

- (a) the pendency of these proceedings;
- (b) any applications for a bankruptcy order now or hereafter issued pursuant to the *Bankruptcy and Insolvency Act* (Canada) in respect of the Debtor and any bankruptcy order issued pursuant to any such applications; and
- (c) any assignment in bankruptcy made in respect of the Debtor;

the vesting of the Purchased Assets in the Purchaser, as well as the execution of all agreements pursuant to this Order shall be binding on any trustee in bankruptcy that may be appointed in respect of the Debtor and shall not be void or voidable by creditors of the Debtor, nor shall it constitute nor be deemed to be a settlement, fraudulent preference, assignment, fraudulent conveyance, transfer at undervalue, or other reviewable transaction under the *Bankruptcy and Insolvency Act* (Canada) or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

7. **THIS COURT ORDERS AND DECLARES** that the Transaction is exempt from the application of the *Bulk Sales Act* (Ontario).

- 8. **THIS COURT ORDERS** that the Debtor is authorized and directed to change its corporation name from Domfoam International Inc. in order to complete the Transaction, if necessary.
- 9. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States to give effect to this Order and to assist the Debtor and the Monitor and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Debtor and the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order or to assist the Debtor and the Monitor and its agents in carrying out the terms of this Order.

#1863901 | 4079509

ENTERED AT / INSCRIT À TORONTO ON / BOOK NO: LE / DANS LE REGISTRE NO.:

MAR 1 6 2012

#### Schedule A - Form of Monitor's Certificate

Court File No. CV-12-9545-00CL

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES (1995) INC., DOMFOAM INTERNATIONAL INC., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### MONITOR'S CERTIFICATE

#### RECITALS

- A. Pursuant to an Order of the Honourable Justice Newbould of the Ontario Superior Court of Justice (the "Court") dated January 12, 2012, Deloitte & Touche Inc. (the "Monitor") was appointed to monitor the business and financial affairs of Domfoam International Inc. (the "Debtor").
- B. Pursuant to an Order of the Court dated March 16, 2012, the Court approved the asset purchase agreement made as of March 8, 2012 (the "Sale Agreement") between the Debtor and 4037057 Canada Inc. (the "Purchaser") and provided for the vesting in the Purchaser, or such related entity as it may direct, of the Debtor's right, title and interest in and to the Purchased Assets, which vesting is to be effective with respect to the Purchased Assets upon the delivery by the Debtor or Monitor to the Purchaser of a certificate confirming (i) the payment by the

Purchaser of the Purchase Price for the Purchased Assets, save and except for the Holdback; (ii) that the conditions to Closing as set out in the Sale Agreement have been satisfied or waived by the Debtor and the Purchaser; and (iii) save and except for the Holdback and the settlement of the Purchase Price as per Section 2.10 of the Sale Agreement, the Transaction has been completed to the satisfaction of the Debtor and the Monitor.

C. Unless otherwise indicated herein, terms with initial capitals have the meanings set out in the Sale Agreement.

### THE MONITOR CERTIFIES the following:

- 1. The Purchaser has paid and the Monitor has received the Purchase Price for the Purchased Assets payable on the Closing Date pursuant to the Sale Agreement, save and except for the Holdback;
- 2. The conditions to Closing of the Sale Agreement have been satisfied or waived by the Debtor and the Purchaser; and
- 3. Save and except for the Holdback and the settlement of the Purchase Price as per Section 2.10 of the Sale Agreement, the Transaction has been completed to the satisfaction of the Monitor.
- 4. This Certificate was delivered by the Monitor at [TIME] on \_\_\_\_\_ 2012.

Deloitte & Touche Inc., in its capacity as Court-appointed monitor of the Applicants, and not in its personal capacity

Per:			
	Name:	 	
	Title:		

#1863901 | 4079509

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES IN THE MAT'TER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED (1995) INC., DOMFOAM INTERNATIONAL INC., and A-Z SPONGE & FOAM PRODUCTS LTD.

Court File No. CV-12-9545-00CL

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

Proceeding commenced at TORONTO

# APPROVAL & VESTING ORDER (re Domfoam International Inc.)

### MINDEN GROSS LLP

145 King Street West, Suite 2200 Toronto ON M5H 4G2

# Raymond M. Slattery (LSUC #20479L)

416-369-4149

rslattery@mindengross.com

## David T. Ullmann (LSUC#423571)

416-369-4148

dullmann@mindengross.com

### Sepideh Nassabi (LSUC#60139B) 416-369-4323 snassabi@mindengross.com

Lawyers for the Applicants

416-864-9223 fax

### ASSIGNMENT AGREEMENT effective as of the Closing Date.

AMONG:

4037057 CANADA INC., corporation duly incorporated, having its head office at 8600 Decarie Blvd., Suite 200, Town of Mont-Royal, in the Province of Québec, H4P 2N2, herein represented by Terry Pomerantz, its President, duly authorized has he so declares;

Hereinafter called (the "Assignor")

AND:

8032858 CANADA INC., corporation duly incorporated, having its head office at 8600 Decarie Blvd., Suite 200, Town of Mont-Royal, in the Province of Québec, H4P 2N2, herein represented by Terry Pomerantz, its President, duly authorized has he so declares;

Hereinafter called (the "Assignee")

AND:

DOMFOAM INTERNATIONAL INC., a corporation duly incorporated, having its head office at 8785 Langelier Blvd. (Saint-Leonard), City of Montreal, herein represented by Tony Vallecoccia, its duly authorised representative;

Hereinafter called (the "Vendor")

### PREAMBLE:

### WHEREAS:

- the Assignor entered into an asset purchase agreement as of March 8, 2012 (the "APA") with the Vendor by which the Assignor purchased from the Vendor the Purchased Assets (as described in the APA);
- by Order of Justice Brown dated March 16, 2012 (the "Approval and Vesting Order"), the Ontario Superior
  Court of Justice (the "Court") approved the terms and provisions of the APA and the purchase and sale
  transaction provided for therein and all of the Vendor's right, title and interest, if any, in and to the Purchased
  Assets were vested in the Purchaser, conditional upon the filing by the Monitor of a certificate with the Court;
- the Assignor has the ability to assign the APA (as per Section 7.14 of the APA) if the assignee is a "related entity" to the Assignor within the meaning of that expression under the *Income Tax Act* (Canada). In the event of such assignment, the Assignor shall remain liable to the Vendor in respect of its obligations hereunder;
- the Assignee has agreed to take over all the obligations of the Assignor pursuant to the APA and the Assignee
  declares that it is a "related entity" to the Assignor within the meaning of that expression under the *Income Tax*Act (Canada);
- it is the intention of the parties that all the rights and liabilities as per the APA be assigned from the Assignor to the Assignee;

the Vendor covenants and agrees to this Assignment Agreement;

NOW THEREFORE in consideration of the preamble and the mutual agreements and covenants herein contained, the parties hereto hereby covenant and agree as follows:

### 1 INTERPRETATION - DEFINED TERMS

- 1.1 Preamble The preamble forms an integral part of the present Agreement.
- 1.2 Defined terms Other than as defined herein, all capitalized terms shall have the meanings ascribed to them in the APA.

### 2 ASSIGNMENT AND DIRECTION

In consideration of the sum of TWO DOLLARS (\$2.00) and other valuable consideration, the receipt of which is acknowledged, the Assignor assigns and directs to the Assignee all its interest in and to the APA, including all rights of action or other rights accruing to the Assignor, or which might after this assignment takes effect or accrue to the Assignor under the APA. Notwithstanding the assignment of the APA to the Assignee, the Assignor shall remain liable to the Vendor in respect of its obligations as per the APA.

### 3 INTERVENTION

The Vendor intervenes to this Assignment Agreement to acknowledge and consent to the assignment.

### 4 GENERAL PROVISIONS

- 4.1 Further Assurances Each of the parties hereby covenants and agrees that at any time and from time to time after the Closing Date it will, upon the request of the others, do, execute, acknowledge and deliver or cause to be done, executed, acknowledged and delivered all such further acts, deeds, assignments, transfers, conveyances and assurances as may be required for the better carrying out and performance of all the terms of this Assignment Agreement.
- 4.2 Counterparts This Assignment Agreement may be executed in several counterparts, each of which so executed shall be deemed to be an original, and such counterparts together shall constitute but one and the same instrument.
- 4.3 Successors and Assigns This Assignment Agreement shall be binding upon and enure to the benefit of the parties hereto and their respective successors and permitted assigns. Nothing herein, express or implied, is intended to confer upon any Person, other than the parties hereto and their respective successors and assigns, any rights, remedies, obligations or liabilities under or by reason of this Assignment Agreement.
- 4.4 Amendments No modification or amendment to this Assignment Agreement may be made unless agreed to by the parties hereto in writing.

4.5 Governing Law - This Assignment Agreement shall be governed by and construed in accordance with the laws of the Province of Quebec and the laws of Canada applicable therein and each of the parties irrevocably attorns to the non-exclusive jurisdiction of the courts of the Province of Quebec. The parties acknowledge that any disputes related to the interpretation of this Assignment Agreement (including but not limited to its interpretation or its execution) shall be sought before the Quebec Superior Court in Montreal.

### 5 CONFIRMATION OF THE APA

Except as otherwise provided in this Assignment Agreement, the parties confirm and reiterate all the terms and conditions contained in the APA and the parties acknowledge being bound by said terms and conditions.

### 6 LANGUAGE

The parties acknowledge that they have required that this Assignment Agreement and all related documents be prepared in English. / Les parties reconnaissent avoir exigé que la présente convention et tous les documents connexes soient rédigés en anglais.

(Signature page follows.)

### IN WITNESS WHEREOF, the parties hereto have executed the present Assignment Agreement:

SIGNED BY THE ASSIGNOR, in Town of Mont-Royal, this  $26^{th}$  day of March 2012.

4037057 Canada Inc.

Per:

Terry Pomerantz, President
I have authority to bind the corporation

SIGNED BY THE ASSIGNEE, in Town of Mont-Royal, this 26th day of March 2012.

8032858 Canada Inc.

Terry Pomerantz, President

I have authority to bind the corporation

SIGNED BY THE VENDOR, in Toronto, this 26th day of March 2012.

Domfoam International Inc.

Per:

Tony Vallecoccia, President
I have authority to bind the corporation

K\_Assignment\_of\_APA\_vExec

IN WIT	INESS WHEREOF, the parties hereto have executed the present Assignment Agreement:
	D BY THE ASSIGNOR, in Town of Mont-Royal, this 26 <sup>th</sup> day of March 2012.  7 Canada Inc.
	Terry Pomerantz, President have authority to bind the corporation
	D BY THE ASSIGNEE, in Town of Mont-Royal, this 26th day of March 2012.  B Canada Inc.
	Cerry Pomerantz, President have authority to bind the corporation

SIGNED BY THE VENDOR, in Toronto, this 26th day of March 2012.

Domfoam International Inc.

Der.

Tony Vallecoccia, President I have authority to bind the corporation

K\_Assignment\_of\_APA\_vExec

### Tab F

THIS IS EXHIBIT "F" TO THE AFFIDAVIT OF JACQUES VINCENT

SWORN THIS 13<sup>th</sup> DAY OF SEPTEMBER, 2018

Commissioner, etc.

Anny Groulx

**Notice** 

389

### **Assignment of Claims**

RE: Domfoam International Inc.

Montreal (Canada), June 22, 2018.

Mr. Don Perelman
FINE KAPLAN & BLACK R.P.C.
1 South Broad Street
23<sup>rd</sup> Floor
Philadelphia, PA 19107
USA

Sir:

We are acting as legal counsels for Domfoam Inc. (f/k/a 8032858 Canada Inc.). (the "Purchaser").

On March 8, 2012, an asset purchase agreement intervened between 4037057 Canada Inc. (the "Assignor") and 4362063 Canada Ltd. (f/k/a *Domfoam International Inc.*, hereinafter the "Vendor") (the "APA"), with respect to all assets, undertakings and properties of the Vendor, of every nature and kind whatsoever and wherever situated (the "Purchased Assets"), save and except the "Excluded Assets", as listed in Schedule 2.2. of the APA.

On March 16, 2012, the terms and provisions of the APA and the purchase and sale transaction provided for therein and all of the Vendor's right, title and interest in and to the Purchased Assets vested in the Assignor, have been approved and authorized by an Order of Justice Brown (the "Approval and Vesting Order"), from the Ontario Superior Court of Justice and the Purchased Assets were thereafter vested in the Assignor.

On March 26, 2012, the Purchaser, pursuant to an assignment agreement amongst the Assignor, the Vendor and the Purchaser (the "Assignment"), became the beneficiary of all interest of the Assignor under the APA, including all rights of action or other rights accruing to the Assignor, or which might after the Assignment takes effect or accrue to the Assignor under the APA.

Concurrently on March 26, 2012, a general conveyance and assumption of assumed liabilities was 390 executed between the Vendor and the Purchaser (the "Conveyance") by which, *inter alia*, the Vendor conveyed, transferred and assigned to the Purchaser and the Purchaser did accept, all of the right, title and interest of the Vendor in and to the Purchased Assets.

For further reference, you will find attached copies of the following executed documents:

- 1. The APA;
- 2. The Approval and Vesting Order;
- 3. The Assignment; and
- 4. The Conveyance.

Therefore and in consideration of the APA, as approved by the Approval and Vesting Order and as assigned under the Assignment, be advised that any payment from the Urethane Antitrust Litigation Settlement that would have been payable to the Vendor shall be considered as part of the Purchased Assets and shall thereon be payable to the Purchaser and addressed to the undersigned as follow:

Cheque payable to:

Domfoam Inc.

Sent to:

a/s Mtre. Jacques Vincent, lawyer 2000 – 1 Westmount Square Montreal (Quebec) H3Z 2P9 Canada

PLEASE ACT AND GOVERN YOURSELF ACCORDINGLY.

LAFRAMBOISE GUTKIN G.P.

per:

JACQUES VINCENT, lawyer

Encl.

Cc Mrs Stéphanie Amine-Giwner (by email and Registered Mail)
Mrs Michele Smith (by email and Registered Mail)

### ASSET PURCHASE AGREEMENT

Made as of the 8th day of March, 2012

Between

4037057 CANADA INC.

- and -

DOMFOAM INTERNATIONAL INC.

This Agreement is made as of the 8th day of March, 2012,

### BETWEEN:

4037057 CANADA INC., a corporation existing under the laws of Canada

(hereinafter referred to as the "Purchaser")

- and -

DOMFOAM INTERNATIONAL INC., a corporation existing under the laws of Canada

(hereinafter referred to as the "Vendor")

### RECITALS

- A. The Vendor carries on the Business and is willing to sell the Purchased Assets to the Purchaser;
- B. The Vendor has commenced proceedings in the Ontario Superior Court of Justice of the judicial district of Toronto (the "CCAA Court") under the Companies' Creditors Arrangement Act (Canada) (the "CCAA") pursuant to which, among other things, Deloitte has been appointed as the monitor (the "Monitor"); and
- C. The Vendor has agreed to sell and transfer and assign to the Purchaser and, subject to the issuance of the Approval Vesting Order, the Purchaser has agreed to purchase from the Vendor, subject to the conditions contained herein, all of the Vendor's interest in and to the Purchased Assets;

NOW THEREFORE this Agreement witnesses that in consideration of the mutual covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged by each party to the other, the parties agree as follows:

### Section 1 - INTERPRETATION

### 1.1 Definitions

In this Agreement:

- (a) "Agreement" means this asset purchase agreement;
- (b) "Allocation of Purchase Price" has the meaning set out in Section 2.12;
- (c) "Approval and Vesting Order" means an Order of the CCAA Court, in form and content satisfactory to both the Purchaser and the Vendor, acting reasonably, which approves the

Vendor entering into this Agreement and the Transactions contemplated hereunder and and vesting the Purchased Assets in the Purchaser free and clear of all liens, charges, pledges, security interests and other encumbrances;

- (c.1) "Assigned Contracts" has the meaning set out in Section 2.3;
- (d) "Assumed Employees" has the meaning set out in Section 2.5;
- (e) "Assumed Obligations" has the meaning set out in Section 2.6;
- (e.1) "BASF Receivables" has the meaning set out in Section 2.9;
- "Books and Records" means all books and records and the data contained therein, whether in paper or electronic form, in the possession of the Vendor relating to the Purchased Assets or the Assumed Obligations (other than accounting records and books and records required by applicable law to be retained by the Vendor, copies of which shall be provided to the Purchaser at the Vendor's cost), including all policies, plans, audit reports, all personnel, employment, payroll and training records relating to the Assumed Employees, customer lists, Customer Contracts, sales records, price lists, and all other related files, records and information in the possession of the Vendor relating to the Purchased Assets or the Assumed Obligations, but, for certainty, not including (i) such books, records and data relating to any of the Excluded Assets or the Excluded Obligations or (ii) the corporate minute books of the Vendor or any of its Subsidiaries:
- (g) "Business" means collectively the business, affairs and operations of the Vendor, which for greater certainty, excludes the Subsidiaries businesses; the Vendor is in the business of manufacturing and selling polyurethane foam in Canada and, to a minimum extent, in the United States of America, for the bedding, carpet underlay, furniture and industrial industries;
- (h) "Business Day" means a day on which banks are open for business in the City of Toronto but does not include a Saturday, Sunday or statutory holiday in the Province of Ontario;
- (i) "CCAA" has the meaning set out in the recitals hereof;
- (j) "CCAA Court" has the meaning set out in the recitals hereof;
- (k) "Closing" means the successful completion of the Transaction:
- (I) "Closing Date" means March 26<sup>th</sup>, 2012 or such other date as the parties may agree to in writing;
- (m) "Contracts" means the agreements as described in Schedule 1.1(m), as the same may be modified by the Purchaser from time to time pursuant to Section 2.8;
- (n) "Customer Contracts" means the Vendor's contracts, agreements, instruments and other legally binding commitments or arrangements, written or oral, concerning the provision of services by the Vendor to its customers including but not limited to those contracts listed in Schedule 1.1(n) attached hereto;
- (o) "Deposit" has the meaning set out in Section 2.11(a);

- (p) "DIP Financing" has the meaning set out in Section 5.1(n);
- (q) "Email" has the meaning set out in Section 7.4;
- (r) "Equipment" means all of the Vendor's equipment (including the Vendor's computer and computer accessories), office equipment, furniture, fixtures, furnishings and supplies of all kinds in the possession or control of the Vendor on the Closing Date, whether owned by the Vendor or in which the Vendor has an interest pursuant to an equipment operating lease or a security agreement;
- (r.1) "Equipment Operating Leases" means all operating leases of equipment or other personal property used by the Vendor as a lessee or under which the Vendor has rights as lessee and listed on Schedule 1.1(r.1) attached hereto;
- (s) "ETA" means the Excise Tax Act (Canada);
- (t) "Excluded Assets" has the meaning set out in Section 2.2;
- (u) "Excluded Employees" has the meaning set out in Section 2.5;
- (v) "Excluded Obligations" has the meaning set out in Section 2.7;
- (w) "Initial Order" means the Order of the CCAA Court obtained on January 12, 2012 in the CCAA Court file no. CV-12-9545-00CL;
- (x) "Intellectual Property" means the following intellectual property:
  - (i) all designs, graphics, slogans and other commercial symbols and all registrations and applications therefor;
  - (ii) all patents, patent rights (including divisions, reissues, renewals, re-examinations, continuations, continuations in part and extensions) and all applications therefor;
  - (iii) all copyrights, writings and other copyrightable works of authorship, including computer programs, databases and documentation therefor, integrated circuit topographies, industrial designs and other industrial property rights and all applications and registrations therefor and all renewals or extensions of such applications and registrations;
  - (iv) all proprietary and non-public business information, including know-how, trade secrets, improvements, concepts, ideas, technical data, drawings, specifications therefor, business methodologies and processes, confidential information and any licensed property or technology; and
  - (v) all goodwill and trademarks in whatever format, including without limitation, registered and unregistered trademarks, trade names, brand names, service marks, logos, copyrights, certification marks, drawings, permits, internet and electronic email addresses, URLs, telephone, telex and facsimile numbers, content of websites and domain names related to or connected with the business carried on by the Vendor, all related software and electronic code to the extent assignable and other similar intellectual property or intangibles of the Vendor;

- (y) "Interim Period" means the period between the date of acceptance of this Agreement by the Vendor and the Closing Date;
- (z) "Leased Premises" means the premises leased and occupied by the Vendor at:
  - (i) 8785 Langelier Blvd., (Saint-Leonard) Montreal, Québec, H1P 2C9;
  - (ii) 5675 Des Grandes-Prairies Blvd., (Saint-Leonard) Montreal, Québec, H1R 1B3;
  - (iii) 7525 Henri Bourassa East, Montreal, Québec, H1E 1N9;
- (aa) deleted;
- (bb) "Monitor" has the meaning set out in the Recitals;
- (cc) "Objection Notice" has the meaning set out in Section 2.10;
- (dd) "Offer Date Inventory Value" has the meaning set out in Section 2.9;
- (ee) "Offer Date Receivable Value" has the meaning set out in Section 2.9;
- (ff) "Order" means any order, directive, judgment, decree, award or writ of any tribunal;
- (gg) "Purchase Price" has the meaning set out in Section 2.9;
- (hh) "Purchased Assets" means the right, title and interest of the Vendor in and to the assets described in Schedule 1.1(hh), provided that the Purchased Assets shall not include any Excluded Assets;
- (ii) "Purchased Inventories" has the meaning set out in Section 2.9:
- (jj) "Purchased Receivables" has the meaning set out in Section 2.9;
- (kk) "Purchased Working Capital" has the meaning set out in Section 2.10;
- (ll) "QSTA" means an Act respecting the Quebec Sales Tax.
- (mm) Withdrawn;
- (nn) "Schedules" means any and all schedules referred to in this Agreement and attached thereto;
- (00) "Subsidiary(ies)" means each of (i) Valle Foam Industries (1995) Inc. having its registered office at 4 West Drive, Brampton (Ontario) L6T 2H7, and (ii) A-Z Sponge & Foam Products Ltd. having its registered office at 811 Cundy Ave., Annacis Island, Delta (British Columbia) V3M 5P6. Those corporations being referred to individually as a "Subsidiary" and collectively as "Subsidiaries";
- (pp) "Taxes" has the meaning set out in Section 2.13;
- (qq) "Tax Return" means a report, return or other information or form required to be supplied to a governmental entity with respect to any Taxes;

- (rr) "Time of Closing" means 2:00 p.m. (Toronto Time) on the Closing Date or such other time on the Closing Date as the parties may mutually agree;
- (ss) "Transaction" means the transaction of purchase, sale, assignment and assumption contemplated by this Agreement;
- (tt) "Unionized Employees" has the meaning set out in Section 2.5; and
- (uu) "Vendor's Intellectual Property" means all Intellectual Property used by the Vendor as at the Closing Date, including any Intellectual Property listed in Schedule 1.1(uu) attached hereto.

### 1.2 Interpretation Not Affected by Headings, etc.

The division of this Agreement into sections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. The terms "this Agreement", "hereof", "herein" and "hereunder" and similar expressions refer to this Agreement and not to any particular section hereof.

### 1.3 Extended Meanings

Words importing the singular include the plural and vice versa, words importing gender include all genders and words importing persons include individuals, partnerships, associations, trusts, unincorporated organizations, corporations and governmental authorities. The term "including" means "including, without limitation," and such terms as "includes" have similar meanings.

[Section 1.4 follows on next page]

397

### 1.4 Schedules

The following schedules are incorporated in and form part of this Agreement:

Schedule 1.1(m)	Contracts
Schedule 1.1(o)	Customer Contracts
Schedule 1.1(r.1)	Equipment Operating Leases
Schedule 1.1(hh)	Purchased Assets
Schedule 1.1(uu)	Vendor's Intellectual Property
Schedule 2.2	Excluded Assets
Schedule 2.6	Assumed Obligations
Schedule 2.9(A)	Purchased Receivables
Schedule 2.9(B)	Purchased Inventories
Schedule 2.9(E)	Excess rebates to customers
Schedule 4.1(c)	Confidentiality Agreement
Schedule 5.1(o)	List of assets to be acquired from Valle Foam Industries (1995) Inc.

[Section 2 follows on next page]

### 2.1 Sale and Purchase of Assets

Subject to the terms and conditions hereof, the Vendor shall sell to the Purchaser and the Purchaser shall purchase from the Vendor the Purchased Assets on the Closing Date. The Purchaser acknowledges that it is not purchasing any other property or assets of the Vendor other than the Purchased Assets. The Purchaser acknowledges and agrees that this Agreement contains a condition (not capable of waiver) that, prior to the Closing, the Vendor shall have obtained the Approval and Vesting Order, as provided in this Agreement. If the Vendor shall not have obtained the Approval and Vesting Order after making application for the same, this Agreement shall automatically terminate without any further action on the part of any of the Vendor or the Purchaser and neither party hereto shall be under any further obligation to the others to complete the Transactions and each party hereto shall be released immediately of all of its obligations hereunder, other than as set forth in Sections 6.6(b) (return of the Deposit), 4.1(c) (Confidentiality Agreement) and 7.3 (Cost and Expenses).

### 2.2 Excluded Assets

The Purchased Assets shall not include (i) any assets of the Vendor other than the Purchased Assets or (ii) those assets of the Vendor that are listed or described in Schedule 2.2 on the date hereof and those assets of the Vendor which are added to such Schedule 2.2 by the Purchaser during the Interim Period (collectively, the "Excluded Assets"). If any of the Excluded Assets or any proceeds in respect thereof shall at any time come into the possession of or under the control of the Purchaser or any of its employees, officers or agents, such assets and proceeds shall be held by the Purchaser in trust for the benefit of the Vendor. Within fifteen (15) Business Days from the date on which the Purchaser, or any of its employees, officers or agents, comes into possession of or obtains control over any of such assets or proceeds, as applicable, the Purchaser shall (a) by notice in writing delivered to the Vendor and the Monitor in accordance with the provisions hereof, so advise the Vendor and the Monitor, and (b) forthwith account and deliver over to the Vendor or the Monitor any such assets or proceeds, less any reasonable costs incurred by the Purchaser in executing said obligations.

### 2.3 Assignment and Assumption of Contracts

Subject to the conditions and terms hereof, the Vendor and the Purchaser agree to work cooperatively and to use their commercial best efforts to obtain the consents or approvals to the assignment of the Customer Contracts and the Contracts or any other contract that the Purchaser may require to be added to Schedule 1.1(m) during the Interim Period (collectively the "Assigned Contracts), as requested by the Purchaser acting reasonably, on terms and conditions satisfactory to the Purchaser. Except for any administration fee required to be paid to the person whose consent or approval has been requested in respect of any such Assigned Contract, the Vendor shall not be required to pay any amount or fee whatsoever in connection with the Vendor's covenant to attempt to obtain such consent or approval for such Assigned Contracts. The Purchaser agrees to supply all such information to the Vendor and the person whose consent or approval has been requested (including, without limitation, credit and financial information) as may be reasonably requested by the Vendor or the person whose consent or approval has been requested and to otherwise cooperate, acting reasonably, with the Vendor and the person whose consent or approval has been requested in connection with the foregoing. Notwithstanding any other provision of this Agreement, the failure of the Vendor or the Purchaser to have obtained any such consents or approvals requested by the Purchaser shall not entitle the Purchaser to terminate this Agreement or reduce the Purchase Price and shall not operate to release the Purchaser from any of its obligations hereunder.

To the extent the assignment of any of the Assigned Contracts assigned to the Purchaser pursuant to the provisions hereof shall require the consent or approval of any person and such consent or approval is not obtained at or prior to the Time of Closing, this Agreement shall not constitute a contract to assign such Assigned Contract if an attempted assignment would constitute a breach thereof. The Vendor shall co-operate with the Purchaser in any reasonable arrangement designed to provide the Purchaser with the benefit of such Assigned Contract including enforcement of any and all rights of the Vendor (if any) against the other party thereto arising out of any breach or cancellation thereof by such party or otherwise. Nothing contained herein shall be construed to negate or diminish, as between the Vendor and the Purchaser, the Vendor's covenants and obligations to transfer and deliver to the Purchaser the Purchased Assets as provided in this Agreement.

### 2.4 "As is, Where is"

The Purchaser acknowledges that the Vendor is selling and the Purchaser is purchasing the Purchased Assets on an "as is, where is" basis as they shall exist on the Closing Date. The Purchaser further acknowledges that it has entered into this Agreement on the basis that the Vendor does not guarantee title to the Purchased Assets and that the Purchaser has conducted such inspections of the condition of and the title to the Purchased Assets as it has deemed appropriate and has satisfied itself with regard to these matters. Without limiting the generality of the foregoing, the Purchaser acknowledges that there are no representations, warranties, terms, conditions, undertakings or collateral agreements, express or implied, statutory or otherwise, with respect to the Purchased Assets or with respect to any other matter or thing whatsoever except as otherwise expressly stated herein, including as to existence, title, encumbrances, description, fitness for purpose and merchantability concerning the Purchased Assets. The description of the Purchased Assets contained in the Schedule 1.1(hh) is for the purpose of identification only. The Purchased Assets are being purchased as they exist at execution hereof and, except as provided in Section 2.10, no adjustment will be allowed to the Purchaser for any change in condition, value, or quality of the Purchased Assets.

### 2.5 Employees

The Vendor shall provide to the Purchaser within five (5) Business Days of the acceptance of this Agreement a list of the names of the individuals of the Vendor that are full-time, part-time or casual employees or individuals engaged on contract to provide services to the Vendor together with all recruitment agencies' employees actually working for the Vendor. The Purchaser agrees that it shall offer employment to (i) all the unionized employees employed by the Vendor (the "Unionized Employees") and (ii) substantially all other employees of the Vendor other than those employees or individuals that the Purchaser has identified in writing to the Vendor ("the "Excluded Employees"), and in any event at least 60% of the non-unionized employees, five (5) Business Days prior to the Closing Date (the Unionized Employees and such other employee or individual who accepts the Purchaser's offer, the "Assumed Employees"), effective as at the Time of Closing, on terms and conditions of employment as may be agreed between the Purchaser and the Assumed Employees (except in the case of the Unionized Employees whose employment terms shall be governed by the applicable collective bargaining agreement). Until the Closing Date, the Vendor shall be responsible for all wages, outstanding claims for severance prior to the Closing Date, workers compensation contributions, any amount owed under the Pay Equity Act (R.S.Q., chapter E-12.001) and any regulation adopted under such Act, and other remuneration and benefits as may be payable, or related, to the employees of the Vendor. It is expressly understood that the Vendor shall not be responsible for any accrued vacaction which may be owing to the Assumed Employees at or before the Time of Closing. After the Closing Date, the Vendor shall remain responsible for all such amounts payable, or related to, the Excluded Employees.

In connection with the acquisition of the Purchased Assets, the following obligations and liabilities ("Assumed Obligations") shall be assumed by the Purchaser as of the Time of Closing:

- (a) all obligations related to the Assumed Employees for the period beginning immediately after the Time of Closing;
- (b) all obligations and liabilities of the Vendor as of the Time of Closing under the Customer Contracts and the Contracts;
- (c) all obligations and liabilities in connection with the Purchased Assets and related to the period after the Time of Closing;
- (d) such other liabilities as the Purchaser advises the Vendor prior to the Time of Closing that the Purchaser will assume;
- (e) those obligations listed on Schedule 2.6 hereto;
- (f) all obligation related to the Assumed Employees for any outstanding accrued vacation pay outstanding as at the Time of Closing.

The Purchaser shall indemnify and hold harmless the Vendor with respect to any claim which may, from time to time, be asserted against the Vendor relating to the Assumed Obligations after the Time of Closing and from any and all costs (including all reasonable legal costs on a solicitor and its own client basis), actions, losses, claims, damages and liabilities whatsoever which the Vendor may suffer or incur by virtue of the failure of the Purchaser to pay, perform and discharge the Assumed Obligations.

### 2.7 Excluded Obligations

Other than the Assumed Obligations, the Purchaser shall not assume and shall not be liable for any other liabilities or obligations of the Vendor (the "Excluded Obligations").

### 2.8 Purchaser's Right to Exclude

Notwithstanding anything to the contrary in this Agreement, the Purchaser may, at its option, exclude any of the Purchased Assets, other than any of the Equipment, from the Transaction prior to the Closing Date, whereupon such Purchased Assets shall be Excluded Assets, provided, however, that there shall be no corresponding reduction in the Purchase Price. For greater certainty, the Purchaser shall be entitled to exclude any of the Contracts described in Schedule 1.1(m) hereto (with the exception of the collective bargaining agreement relating to the Unionized Employees) and any of the Customer Contracts described in Schedule 1.1(n) hereto.

### 2.9 Purchase Price

The Purchase Price for the Purchased Assets would be, calculated as described hereinafter, the aggregate amount of three million six hundred sixty two thousand nine hundred and seventy five dollars (\$3,662,975) plus the assumption of the Assumed Obligations, said Purchase Price to be established as of the Closing Date (the "Purchase Price").

Purchased Receivables.

/ A \

(14)	i dichased receivables.	\$2,430,976
(B)	Purchased Inventories:	\$946,586
(C)	Withdrawn	
(D)	All other Purchased Assets:	\$300,000
(E)	Excess rebates to customers	(\$34,587)

For more clarity, the Purchase Price has been established on the following method of calculation as for the Purchased Receivables and the Purchased Inventories, and same method of calculation shall be used to establish the final amount of the Purchase Price as of the Closing Date:

**\$2.450.076** 

### (A) Purchased Receivables

As of January 31, 2012, the total amount of all the Vendor's accounts receivable (the "Purchased Receivables") is five million nine hundred ninety six thousand six hundred and ninety two dollars (\$5,996,692) (the "Offer Date Receivable Value"), and is detailed in Schedule 2.9(A) – Section 1.

The portion of the Purchase Price attributed to the Purchased Receivables is two million four hundred fifty thousand nine hundred and seventy six dollars (\$2,450,976) and is calculated as described in Schedule 2.9(A) – Section 2.

### (B) Purchased Inventories

As of January 11, 2012 (and for Chemicals (majors) and Small Chemicals, as of January 31, 2012), the total amount of all the Vendor's inventory (raw material, work in progress and finished goods inventory) (the "Purchased Inventories") is three million fifty six thousand and eleven dollars (\$3,056,011) (the "Offer Date Inventory Value") in Schedule 2.9(B) – Section 1.

The portion of the Purchase Price attributed to the Purchased Inventories is nine hundred forty six thousand five hundred and eighty six dollars (\$946,586) and is calculated as described in Schedule 2.9(B) – Section 2.

### (C) BASF Receivables

Withdrawn

### (D) Other Purchased Assets

The Purchaser is ready to pay the Vendor the amount of three hundred thousand dollars (\$300,000) for all the other Purchased Assets.

### (E) Excess rebates to customers

The Purchase Price shall be reduced by the total amount of rebates earned by the Vendor's customers for the calendar year 2011 and not already applied to the outstanding invoices for each said customer at the Closing Time. As of January 27, 2012, the excess rebates to customers total thirty four

thousand five hundred and eighty seven dollars (\$34,587) as calculated in Schedule 2.9(E) and shall be adjusted as at the Closing Date using the same parameters.

### 2.10 Settlement of the Purchase Price

The Vendor and the Purchaser agree that the Purchase Price shall be adjusted as for the Purchased Receivables and the Purchased Inventories (collectively, the "Purchased Working Capital") as follows:

### (A) Re: Accounts receivable:

Subject to the limit established hereinafter in this paragraph, the Purchase Price payable in relation with the Purchased Receivables shall be adjusted downward or upward, as the case may be, as per the formula established in Schedule 2.9(A) – Section 2, in the event and to the extent that, on the Closing Date, the aggregate value of the Purchased Receivables shall be lower or higher, as the case may be, than the Offer Date Receivable Value. Nonetheless, if the total amount of the Purchased Receivables, as the total appears in the first column on page 4 of Schedule 2.9(A) - Section 1, is, as of the Closing Date, over five million nine hundred thousand dollars (\$5,900,000), then the difference between the total amount of the Purchased Receivables as of the Closing Date and the limit herein established shall be deducted from the calculation of the current portion of the Purchased Receivables for the Vendor as of the Closing Date.

### (B) Re: Inventory:

The Purchase Price payable in relation with the Purchased Inventories shall be adjusted downward or upward, as the case may be, as per the formula established in Schedule 2.9(B) – Section 2, in the event and to the extent that, on the Closing Date, the aggregate value of the Purchased Inventories shall be lower or higher, as the case may be, than the Offer Date Inventory Value.

On or before the fifth (5) Business Day after the Closing Date, the Vendor shall deliver to the Purchaser a statement setting forth a detailed calculation of the Purchased Working Capital as of the Closing Date, with all the relevant supporting evidence, and prepared by the Vendor with the same parameters as used for the evaluation of the Purchase Price as for the Purchased Receivables and the Purchased Inventories as detailed in Schedules 2.9(A) and 2.9(B).

The value of the Purchased Inventories shall be based on a physical inventory count and valued on a basis consistent with the Vendor's costing methodologies used for the establishment of the Offer Date Inventory Value. The physical inventory count will be jointly supervised by the Vendor, the Purchaser and the Monitor.

The Purchaser may dispute the value of the Purchased Working Capital as presented by the Vendor by notice in writing (the "Objection Notice") given to the Vendor within ten (10) Business Days after the delivery of the Vendor's statement as to the amount of the Purchased Working Capital. Such Objection Notice shall specify, in detail, the basis for the dispute and the Vendor's adjustments to the value of the Purchased Working Capital shown by the Vendor in its statement.

If the Purchaser and the Vendor cannot agree on the value of the Purchased Working Capital within five (5) Business Days after the delivery of the Objection Notice, then either the Purchaser or the Vendor may make a motion to the CCAA Court to resolve this dispute.

Forthwith after agreement as to the amount of the Purchased Working Capital or forthwith upon the decision of the CCAA Court as to the amount of the Purchased Working Capital, then, as the case may be, the Vendor shall refund to the Purchaser such portion of the Purchase Price which is equal to the amount by which the respective Purchased Working Capital values are less than the respective Offer Date Receivable Value and Offer Date Inventory Value, or alternatively, the Purchaser shall pay to the Vendor, by way of certified cheque or wire transfer, such additional amount which is equal to the amount by which the respective Purchased Working Capital values are higher than the respective Offer Date Receivable Value and Offer Date Inventory Value.

The Monitor and the Vendor agree that they shall not make a motion to the CCAA Court for distribution of any funds received by either of them under this Agreement, or otherwise distribute any such funds, unless the Purchase Price adjustments contained in this Section have been completed and the covenant contained in Section 4.2(b) has been fulfilled. However, in the event that the Purchase Price adjustments have not been settled by the date that is ninety (90) days following the Closing Date, the Monitor shall be entitled to bring a distribution motion on appropriate notice to the Purchaser and the Vendor and each of the Purchaser, the Vendor and the Monitor shall be entitled to make submissions to the CCAA Court in this regard.

### 2.11 Method of Payment

The Purchase Price shall be satisfied as follows:

- (a) a deposit in the amount of three hundred and sixty thousand dollars (\$360,000) (the "Deposit") by dank draft or certified cheque payable to the Monitor, in trust, to be held in trust by the Monitor until the Time of Closing and then credited towards the Purchase Price, or in accordance with any other provision herein;
- (b) the amount of three million one hundred and two thousand nine hundred and seventy five dollars (\$3,102,975) by certified cheque or wire transfer payable to the Monitor at the Time of Closing;
- (c) the amount of two hundred thousand dollars (\$200,000) (the "Holdback") to be paid no later than a sixty (60) days following the Closing Date; and
- (d) by the assumption of the Assumed Obligations.

The Purchaser shall have the ability to offset any portion of the Holdback against any downward adjustment of the Purchase Price per Section 2.10 herein. The Purchaser shall also have the ability to offset any portion of the Holdback against any claim made against the Purchaser from any of the Vendor's employees for any period up to the Closing Time. In such case, the Purchaser shall have the right to retain the Holdback as long as final judgments have not been rendered vis à vis such claim.

### 2.12 Allocation of Purchase Price

The Purchaser and the Vendor shall, acting reasonably, agree on the allocation of Purchase Price (other than for the Purchased Receivables and the Purchased Inventories) on or before the Closing Date, and they shall execute and file any Tax Returns required on the basis of such allocation.

### 2.13 Taxes

The Purchaser shall be liable for and shall pay any and all federal, provincial and other sales, goods and services, value added, and other transfer taxes which are properly payable by the Purchaser in connection with the transfer of the Purchased Assets by the Vendor to the Purchaser, together with all duties, registration fees or other charges properly payable by the Purchaser upon or in connection with the conveyance or transfer of the Purchased Assets (collectively, the "Taxes").

### 2.14 Value Added Tax

- (a) The Purchaser shall be liable for and shall pay any Taxes which may be exigible under Part IX of the ETA;
- (b) The Vendor hereby declares that it is duly registered under Subdivision (d) of Division V of Part IX of the ETA;
- (c) The Vendor hereby declares that it is duly registered under the OSTA;
- (d) Without limiting the Purchaser's indemnity obligations under Section 2.13, the Vendor and the Purchaser will on the Closing Date jointly execute and deliver to each other an election, in the prescribed form and containing the prescribed information, pursuant to subsection 167(1) of the ETA and the corresponding provision of the QSTA so that no tax is payable under any such laws in respect of the sale and purchase of the Purchased Assets hereunder, and the Purchaser shall file any joint election in accordance with the ETA and QSTA.

### 2.15 Section 22 Election — Accounts Receivable

The Purchaser and the Vendor agree to elect jointly in the prescribed form under Section 22 of the *Income Tax Act* (Canada) as to the sale of the accounts receivable and other applicable Purchased Assets described in Section 22 of the *Income Tax Act* (Canada) and the corresponding provision of the statute that is the provincial equivalent thereof and to designate in such election an amount equal to the portion of the Purchase Price allocated to such assets pursuant to Section 2.12 as the consideration paid by the Purchaser therefor.

### 2.16 Risk of Loss

During the Interim Period, up to the Time of Closing, the Purchased Assets shall be and remain at the risk of the Vendor. If, prior to the Closing Date, all or substantially all of the Purchased Assets or any plant operated by the Vendor is destroyed or damaged by fire or any other casualty or shall be appropriated, expropriated or seized by any governmental entity or other lawful authority, the Purchaser shall have the option, at its absolute discretion, exercisable by notice in writing given within fifteen (15) Business Days after the Purchaser receives notice in writing from the Vendor of such destruction, damage, appropriation, expropriation or seizure:

(a) In the event of damage or destruction, to complete the Transaction subject to a reduction of the Purchase Price to be mutually agreed to by the parties;

- (b) In the event of appropriation or expropriation, to complete the Transaction without reduction of the Purchase Price, in which event all proceeds of insurance or compensation for appropriation, expropriation or seizure shall be payable to the Purchaser and all right and claim of the Vendor to any such amounts not paid on the Closing Date shall be assigned to the Purchaser; or
- (c) In either event, to terminate this Agreement, receive a refund of the Deposit and not complete the Transaction, in which case all obligations of the Purchaser shall terminate forthwith upon the Purchaser giving notice as required herein.

[Section 3 follows on next page]

### 3.1 Purchaser's Representations

The Purchaser represents and warrants to the Vendor that:

- (a) the Purchaser is a corporation duly incorporated, organized and subsisting under the laws of Canada;
- (b) the Purchaser has all necessary corporate power, authority and capacity to enter into this Agreement and to perform its obligations and the execution and delivery of this Agreement and the consummation of the transactions contemplated herein and have been duly authorized by all necessary corporate action on the part of the Purchaser;
- (c) the Purchaser is not a party to, bound or affected by or subject to any indenture, agreement, instrument, charter or by-law provision, order, judgment or decree which would be violated, contravened or breached by the execution and delivery by it of this Agreement or the performance by it of any of the terms contained herein;
- (d) this Agreement and all other documents contemplated hereunder to which the Purchaser is or will be a party have been or will be, as at the Time of Closing, duly and validly executed and delivered by the Purchaser and constitutes or will, as at the Time of Closing, constitute legal, valid and binding obligations of the Purchaser, as the case may be, enforceable in accordance with the terms hereof or thereof:
- (e) the Purchaser is not a non-Canadian person as defined in the Investment Canada Act; and
- (f) the Purchaser is or will be prior to Closing Date registered under Part IX of the ETA and under the QSTA.

### 3.2 Representations and Warranties of the Vendor

The Vendor represents and warrants to the Purchaser as follows and acknowledges that the Purchaser is relying thereon in completing this Agreement:

- (a) the Vendor is a corporation duly incorporated, organized and subsisting under the laws of Canada;
- (b) the Vendor has all necessary corporate power, authority and capacity to enter into this Agreement and to perform its obligations and the execution and delivery of this Agreement and the consummation of the transactions contemplated herein and have been duly authorized by all necessary corporate action on the part of the Vendor;
- (c) the Vendor is not a party to, bound or affected by or subject to any indenture, agreement, instrument, charter, by-law provision, order, judgment or decree which would be violated, contravened or breached by the execution and delivery by it of this Agreement or the performance by it of any of the terms contained herein;

- (d) this Agreement and all other documents contemplated hereunder to which the Vendor is or will be a party have been or will be, as at the Time of Closing, duly and validly executed and delivered by the Vendor and constitutes or will, as at the Time of Closing, constitute legal, valid and binding obligations of the Vendor, as the case may be, enforceable in accordance with the terms hereof or thereof; and
- (e) the Vendor is a registrant for the purposes of Part IX of the ETA.

### 3.3 Survival of Representations and Warranties

The representations and warranties of the Vendor shall survive the completion of the Transaction and shall continue in full force and effect for the benefit of the Purchaser (which term for the purposes of this Section 3.3 shall include the Purchaser's directors and officers) for a period of nine (9) months from the Closing Date.

The representations and warranties of the Purchaser shall survive the completion of the Transaction and shall continue in full force and effect for the benefit of the Vendor (which term for the purposes of this Section 3.3 shall include the Vendor's respective directors and officers) for a period of nine (9) months from the Closing Date.

[Section 4 follows on next page]

### 4.1 Covenants of the Purchaser

- (a) The Purchaser agrees that from and after the Time of Closing, it shall preserve the Books and Records actually received from the Vendor at the Time of Closing, and at the reasonable request of the Vendor, the Monitor or another party acting in a court-appointed capacity (such as a receiver, interim receiver, receiver and manager or bankruptcy trustee. or any of their respective authorized representative) (a "Requestor") from time to time, the Purchaser shall co-operate fully with the Requestor and make available to the Requestor in a reasonable timely fashion during normal business hours (i) such Books and Records, (ii) the premises of the Purchaser for the Requestor's review of the Books and Records, and (iii) at the Requestor's expense, the personnel of the Purchaser, in each case to the extent the Requestor reasonably consider necessary (A) to permit the Requestor to prepare and file necessary Tax Returns and any other statutory documents relating to the operation by the Vendor of its business up to the Time of Closing, (B) to properly fulfil the Vendor's obligations in bankruptcy, as the case may be, (C) for the preparation by any Requestor of any Tax Returns or in connection with any assessment, reassessment or other form of document assessing liability of the Vendor for taxes under applicable tax legislation in respect of any taxation year, (D) in connection with the restructuring proceedings, or (E) to ensure compliance by the Vendor with applicable law;
- (b) The Purchaser agrees that from and after the Time of Closing, it shall permit the Requestor to make copies, at the Requestor's expense, of the Books and Records actually received by the Purchaser from the Vendor at the Time of Closing and shall provide reasonable assistance to the Requestor, at the Requestor's expense, to prepare T4 forms and records of employment for the employees of the Vendor who accept offers of employment from the Purchaser. In the case of the Books and Records actually received from the Vendor at the Time of Closing, the Purchaser agrees to preserve same for a minimum of 6 years from the Closing Date;
- (c) Until the Time of Closing, and in the event of the termination of this Agreement without consummation of the Transaction, the Purchaser shall keep confidential any information obtained from the Vendor in accordance with the confidentiality agreement executed between S.P. Holdings or Nominee and the Vendor on November 25, 2011 (the "Confidentiality Agreement") and attached hereto as Schedule 4.1(c) as if the Purchaser was a party thereto, and the Purchaser hereby agrees in favour of the Vendor to be bound by the terms of such confidentiality agreement as if it were an original signatory thereto. If this Agreement is terminated without consummation of the Transaction contemplated, promptly after termination, all documents, work papers and other written material obtained in connection with this Agreement and not made public (including all copies), shall be returned to the Vendor.

### 4.2 Covenants of the Vendor

(a) The Vendor shall allow the Purchaser to have reasonable access to the Vendor's management team, on a daily basis, and the Vendor's management team shall answer Purchaser's questions about the operation of the Business, the management of the Business and the decisions to be taken by the Vendor's management team during the Interim Period;

- (b) The Vendor shall provide the Purchaser, within fifteen (15) Business Days following the Closing Date, confirmation that all wages, outstanding claims for severance prior to the Closing Date, workers compensation contribution, any amount owed under the Pay Equity Act (R.S.Q., chapter E-12.001) and any regulation adopted under such Act, and other remuneration and benefits as may be payable, or related, to the employees of the Vendor, immediately before the Closing Date, have been paid up to and including the Closing Date. The Vendor shall not terminate any employees during the Interim Period without cause. The Vendor shall give 48 hours notice to the Purchaser of any termination with cause of an employee during the Interim Period.
- (c) The Vendor shall use its best efforts to obtain such releases as may be required by the Purchaser, acting reasonably, regarding the termination of management employees, a list of which shall be provided by the Purchaser no later than ten (10) Business Days prior to the Closing Date; and
- (d) The Vendor agrees that during the Interim Period, it shall continue to conduct its Business in the ordinary course and, without limiting the generality of the foregoing, continue to serve its customers on a daily and normal basis without interruption, purchase inventory to satisfy the needs of the Business and deliver the finished goods on the timely manner as done in the past year.

[Section 5 follows on next page]

### Section 5 – CONDITIONS

### 5.1 Conditions - Purchaser

The obligation of the Purchaser to complete the Transaction is subject to the following conditions being fulfilled or performed by the Vendor or prior to the Time of Closing:

- (a) all representations and warranties of the Vendor contained in this Agreement shall be true as of the Closing Date with the same effect as though made on and as of that date;
- (b) deleted;
- (c) Withdrawn;
- (d) the Vendor shall not be bankrupt as at the Time of Closing;
- (e) the obtaining of the Approval and Vesting Order to the entire satisfaction of the Purchaser;
- (f) the Vendor shall have performed its obligations under this Agreement to the extent required to be performed on or before the Closing Date;
- (g) no action or proceedings shall be pending or threatened to restrain or prohibit the completion of the Transaction;
- (h) Withdrawn;
- (i) the Vendor shall not have transferred any inventory or equipment to any of its Subsidiaries, shall not have assigned any of its Contracts to its Subsidiaries or have sold to its Subsidiaries any of the Purchased Assets without having obtained from the Purchaser a prior written approval of said transaction which the Purchaser may refuse at its complete discretion;
- (j) Withdrawn;
- (k) the Vendor shall have paid all rent due under the Lease for the period from January 12, 2012 up to and throughout the Interim Period;
- (l) the Vendor shall have filed, before the Closing Date, articles of amendment to change its corporate name "Domfoam International Inc." to a numbered corporation, and shall have sought an order of the Court authorizing the change of name required herein if such Order is determined to be necessary by the Purchaser, and shall provide to the Purchaser, at the Time of Closing, (a) with a copy of the certificate of amendment evidencing said change and (b) a written authorization for the Purchaser to use said corporate name and any derivative of it;
- (m) deleted;
- (n) Withdrawn;
- (o) In the event that any or all of the assets of Valle Foam Industries (1995) Inc., listed in Schedule 5.1(o) are not, as at the Closing Date, subject to an agreement of purchase and sale and otherwise remain the assets of Valle Foam Industries (1995) Inc., the Vendor

shall have acquired said assets from Valle Foam Industries (1995) Inc., prior to the Closing Date, which assets will therefore form part of the Purchased Assets. The inclusion or exclusion of the assets on Schedule 5.1(o) will not adjust the Purchase Price payable hereunder by the Purchaser; and

(p) the obtaining on or before the date which is fifteen (15) Business Days from acceptance of this Agreement by the Vendor of a commitment letter from the actual general manager of the Vendor (John C. Howard) on the terms and conditions of a three (3) year employment agreement with the Purchaser.

The foregoing conditions are for the exclusive benefit of the Purchaser. Any such condition (except 5.1(e)) may be waived by the Purchaser in whole or in part. Any such waiver shall be binding on the Purchaser only if made in writing.

### 5.2 Conditions - Vendor

The obligation of the Vendor to complete the Transaction is subject to the following conditions being fulfilled or performed at or prior to the Time of Closing:

- (a) all representations and warranties of the Purchaser contained in this Agreement shall be true as of the Closing Date with the same effect as though made on and as of that date;
- (b) the Purchaser shall have performed each of its obligations under this Agreement to the extent required to be performed on or before the Closing Date; and
- (c) no action or proceedings shall be pending or threatened to restrain or prohibit the completion of the Transaction contemplated by this Agreement.

The foregoing conditions are for the exclusive benefit of the Vendor. Any such condition may be waived by the Vendor in whole or in part. Any such waiver shall be binding on the Vendor only if made in writing.

### 5.3 Mutual Conditions

The obligations of the Vendor on the one hand, and the Purchaser on the other hand are subject to the conditions that:

- the Approval and Vesting Order shall have been made by the CCAA Court on or before the 28th day of March, 2012 (or such later date as the parties may agree to, acting reasonably); and
- (b) the Approval and Vesting Order shall not have been stayed, varied or vacated, and no order shall have been issued which restrains or prohibits the completion of the Transaction.

The parties hereto acknowledge that the foregoing conditions are for the mutual benefit of the Vendor and the Purchaser.

### 5.4 Actions to Satisfy Closing conditions

Each of the parties agrees to diligently take all actions as are within its power to control, and to use commercially reasonable efforts to cause other actions to be taken which are not within its power to control, so as to ensure compliance with the conditions set forth in sections 5.1, 5.2 and 5.3 and, generally, the conditions set forth in this Agreement.

### 5.5 Non-Satisfaction of Conditions

If any condition set out in this Section 5 is not satisfied or performed on or prior the Closing Date, or at any date provided for herein, then the Vendor or the Purchaser, depending whose benefit the condition is inserted, may:

- (a) waive compliance with the condition in whole or in part in its sole discretion by written notice to the other party and without prejudice to any of its rights of termination in the event of non-fulfilment of any other condition in whole or in part; or
- (b) choose to give the other party an additional delay for said condition to be satisfied or performed; or
- (c) elect on written notice to the other party to terminate this Agreement before Closing.

[Section 6 follows on next page]

#### 6.1 Closing

The completion of the Transaction shall take place on the Closing Date at the Time of Closing at a location to be determined by the mutual agreement of the parties in writing or at such other time as otherwise determined by mutual agreement of the parties in writing.

### 6.2 Purchaser's Deliveries on Closing

At or before the Time of Closing, the Purchaser shall execute and deliver to the Vendor the following, each of which shall be in form and substance satisfactory to the Vendor, acting reasonably:

- (a) payment of the portion of the Purchase Price payable at the Time of Closing;
- (b) payment or evidence of payment of applicable federal and provincial Taxes or alternatively, appropriate exemption certificates, as required by this Agreement;
- (c) general conveyance and assumption of liabilities agreement pursuant to Subsection 2.6 duly executed by the Purchaser, in form and substance mutually satisfactory to the Vendor and the Purchaser;
- (d) the Allocation of Purchase Price pursuant to Section 2.12;
- (e) an assignment or assignments of Intellectual Property agreements, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Purchaser;
- (f) an assignment or assignments of Customer Contracts and Contracts, in form and substance mutually satisfactory to the Purchaser and the Vendor, duly executed by the Purchaser;
- (g) an assignment of leases for the Leased Premises, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Purchaser;
- (h) a certificate, dated the Closing Date, confirming that all of the representations and warranties of the Purchaser contained in this Agreement are true as of the Closing Date, with the same effect as though made on and as of the Closing Date;
- (i) an acknowledgement dated the Closing Date, that each of the conditions precedent in Section 5.1 of this Agreement have been fulfilled, performed or waived as of the Closing Date; and
- (j) such further and other documentation as is referred in this Agreement or as the Vendor may reasonably require to give effect to this Agreement including pursuant to Sections 2.2, 2.6 and 2.8.

# 6.3 Vendor's Deliveries on Closing

At or before the Time of Closing, the Vendor shall execute and deliver to the Purchaser the following, each of which shall be in form and substance satisfactory to the Purchaser, acting reasonably:

- (a) a bill of sale pursuant to which the Vendor shall assign, sell, transfer and convey all of its right, title and interest in and to its Purchased Assets, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- (b) an assignment or assignments of Intellectual Property agreements, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- an assignment of leases for the Leased Premises, in form and substance mutually satisfactory to the Vendor and the Purchaser, duly executed by the Vendor;
- (d) an assignment or assignments of Customer Contracts and Contracts, in form and substance mutually satisfactory to the Purchaser and the Vendor, duly executed by the Vendor;
- (e) the Approval and Vesting Order in form and substance satisfactory to the Purchaser, acting reasonably;
- (f) a certificate, dated the Closing Date, confirming that all of the representations and warranties of the Vendor contained in this agreement are true as of the Closing Date, with the same effect as though made on and as of the Closing Date;
- (g) an acknowledgement dated the Closing Date, that each of the conditions precedent in Section 5.2 of this Agreement have been fulfilled, performed or waived as of the Closing Date; and
- (h) such further and other documentation as is referred to in this Agreement or as the Purchaser may reasonably require to give effect to this Agreement which shall be prepared by the Purchaser at its sole expense and in form and substance satisfactory to the Vendor, acting reasonably.

#### 6.4 Possession of Assets

The Vendor shall remain in possession of the Purchased Assets until the Time of Closing. On Closing, the Purchaser shall take possession of the Purchased Assets which shall be located in the Leased Premises. In no event shall the Purchased Assets be sold, assigned, transferred or set over to the Purchaser until the conditions set out in the Approval and Vesting Order have been satisfied and the Purchaser has satisfied all delivery requirements outlined in Section 6.2.

#### 6.5 Access to Assets

deleted

#### 6.6 Termination

If either the Vendor on the one hand, or the Purchaser on the other hand, validly terminates this Agreement pursuant to the provisions of Sections 5.5, or 2.16,

- (a) all the obligations of the Vendor and Purchaser pursuant to this Agreement shall terminate;
- (b) the Purchaser shall be entitled to have the Deposit and all the monies paid hereunder returned immediately with any interest earned thereon, as the case may be, but without deduction provided; and
- (c) neither party shall have any right to specific performance or other remedy against, or any right to recover damages or expenses from, the other.

6.7	[Break Fee]	Withdrawn

[Section 7 follows on next page]

#### 7.1 Further Assurances

Each of the parties shall, from time to time after the Closing Date, at the request and expense of the other, promptly take or cause to be taken such action and execute and deliver or cause to be executed and delivered to the other such documents and further assurances as may be reasonably necessary to give effect to this Agreement.

#### 7.2 Obligations to Survive

Notwithstanding the completion of the transactions contemplated hereunder or the delivery of documents pursuant to this Agreement, the obligations, covenants, representations and warranties of the parties hereto shall survive such completion and shall remain in full force and effect and shall not merge as a result thereof.

### 7.3 Costs and Expenses

Each party hereto shall be responsible for all costs and expenses (including the fees and disbursements of legal counsel, bankers, investment bankers, accountants and other advisors) incurred by it in connection with this Agreement and the transactions contemplated herein.

#### 7.4 Notice

Any notice or other communication under this Agreement shall be in writing and may be delivered personally or transmitted by fax or electronic mail ("Email"), addressed to:

in the case of the Purchaser, as follows:

4037057 CANADA INC. 8600 Decarie Blvd. Suite 200 Mont-Royal, QC H4P 2N2

Attention: Terry Pomerantz Telephone No.: (514) 341-8600 Fax No.: (514) 340-1389

Email: tpomerantz@tramsmgmt.com

With a copy to:

Lamarre Perron Lambert Vincent g.p. 480 Saint-Laurent Blvd. Suite 200 Montreal, QC H2Y 3Y7

Attention: Jacques Vincent
Telephone No.: (514) 798-0990
Fax No.: (514) 798-5599
Email: j.vincent@lplv.com

and in the case of the Vendor, as follows:

#### DOMFOAM INTERNATIONAL INC.

8785, boul. Langelier Montréal, QC H1P 2C9

Attention:

Tony Vallecoccia

Telephone No.:

(514) 325-8120

Fax No.:

(514) 325-6477

Email:

tvallecoccia@vallefoam.com

with a copy to:

Minden Gross LLP 145 King Street West Suite 2100 Toronto, ON M5H 4G2

Attention:

Raymond M. Slattery

Telephone No.:

(416) 369-4149

Fax No.:

(416) 864-9223

Email:

rslattery@mindengross.com

and to:

the Monitor

#### DELOITTE & TOUCHE INC.

181 Bay Street Suite 1400 Toronto, ON M5J 2V1

Attention: Telephone No.:

Robert J. Bougie (416) 601-4509

Fax No.:

(416) 601-6690

Email:

rbougie@deloitte.ca

with a copy to:

Thornton Grout Finnigan LLP 100 Wellington Street West Suite 3200

Toronto, ON, M5K 1K7

Attention: Telephone No.:

Grant B. Moffat (416) 304-0599

Fax No.:

(416) 304-1313

Email:

gmoffat@tgf.ca

Any such notice or other communication, if given by personal delivery, will be deemed to have been given on the day of actual delivery thereof and, if transmitted by fax or Email before 5:00 p.m. (Toronto time) on a Business Day, will be deemed to have been given on the Business Day, and if transmitted by fax or Email after 5:00 p.m. (Toronto time) on a Business Day, will be deemed to have been given on the Business Day after the date of the transmission.

#### 7.5 Time

Time shall, in all respects, be of the essence hereof, provided that the time for doing or completing any matter provided for herein may be extended or abridged by an agreement in writing signed by the Vendor and the Purchaser or by their respective solicitors.

### 7.6 Currency

Except where otherwise indicated, all references herein to money amounts are in Canadian currency.

## 7.7 Benefit of Agreement

This Agreement shall enure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns.

### 7.8 Third Party Beneficiaries

Each party hereto intends that this Agreement shall not benefit or create any right or cause of action in or on behalf of any person other than the parties hereto and their successors and permitted assigns, and no person, other than the parties hereto and their successors and their permitted assigns shall be entitled to rely on the provisions hereof in any action, suit, proceeding, hearing or other forum.

#### 7.9 Entire Agreement

This Agreement and the attached Schedules constitute the entire agreement between the parties with respect to the subject matter and supersede all prior negotiations and understandings. This Agreement may not be amended or modified in any respect except by written instrument executed by the parties.

#### 7.10 Paramountey

In the event of any conflict or inconsistency between the provisions of this Agreement, and any other agreement, document or instrument executed or delivered in connection with this Transaction or this Agreement, the provisions of this Agreement shall prevail to the extent of such conflict or inconsistency.

#### 7.11 Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the Province of Quebec and the laws of Canada applicable therein and each of the parties irrevocably attorns to the non-exclusive jurisdiction of the courts of the Province of Quebec. Notwithstanding the foregoing, the parties acknowledge that the Approval and Vesting Order will be sought before the Ontario Superior Court in Toronto and that any disputes related thereto as it relates to court approval or the current Companies' Creditors Arrangement Act proceedings of the Vendor shall be determined by the said Ontario Court. The parties acknowledge that any disputes related to the interpretation of this Agreement (including but not limited to its interpretation or its execution) shall be sought before the Quebec Superior Court in Montreal.

## 7.12 Severability

If any provision of this Agreement or any document delivered in connection with this Agreement is partially or completely invalid or unenforceable, the invalidity or unenforceability of that provision shall not affect the validity or enforceability of any other provision of this Agreement, all of which shall be construed and enforced as if that invalid or unenforceable provision were omitted. The invalidity or unenforceability of any provision in one jurisdiction shall not affect such provisions validity or enforceability in any other jurisdiction.

#### 7.13 Commission - No Broker

The Purchaser agrees to indemnify the Vendor against any claim for compensation or commission by any third party or agent retained by the Purchaser in connection with, or in contemplation of, the Transaction. It is understood and agreed that no broker, agent or other intermediary has acted for the Vendor in connection with the sale of the Purchased Assets and the Purchaser shall not be liable for any commission or other remuneration payable or alleged to be payable to any broker, agent or other intermediary who purports to act or have acted for the Vendor.

#### 7.14 Assignment

Except as hereinafter provided, no party may assign this Agreement without the prior written consent of the other parties, which consent may not be unreasonably withheld or delayed. This Agreement enures to the benefit of and binds the parties and their respective heirs, executors, administrators, personal and legal representatives, successors and permitted assigns. Notwithstanding the foregoing, this Agreement shall be assignable by the Purchaser if the assignee is a "related entity" to the Purchaser within the meaning of that expression under the *Income Tax Act* (Canada). In the event of such assignment, the Purchaser shall remain liable to the Vendor in respect of its obligations hereunder.

### 7.15 Counterparts

This Agreement and any amendment, supplement, restatement or termination of any provision of this Agreement may be executed and delivered in any number of counterparts, each of which when executed and delivered is an original but all of which taken together constitute one and the same instrument. A party's transmission by facsimile or by Email of a copy of this Agreement duly executed by that party shall constitute effective delivery by that party of an executed copy of this Agreement to the party receiving the transmission. A party that has delivered this Agreement by facsimile or by Email shall forthwith deliver an originally executed copy to the other party or parties.

#### 7.16 Language

The parties acknowledge that they have requested and are satisfied that the foregoing be drawn up in English. Les parties reconnaissent qu'elles ont exigé que ce qui précède soit redigé en anglais et s'en déclarent satisfaites.

#### [THE REMAINDER OF THIS PAGE HAS BEEN LEFT INTENTIONALLY BLANK]

The Vendor shall indicate its acknowledgement of and its agreement with the foregoing by signing both copies of this Agreement in the space provided below and returning one fully executed copy to the Purchaser (which return may be made by fax or Email) to:

LPLV

c/o:

Mtre. Jacques Vincent

Fax:

(514) 798-5599

Email:

j.vincent@lplv.com

by no later than 3:00 pm, on March 9, 2012, after which time this Agreement shall be considered null and void as to the Purchaser.

IN WITNESS WHEREOF the Purchaser has duly executed this Agreement on March 8, 2012 at 5:30 p.m.

4037057 CANADA INC.

By:

Name: Terry Pomerantz

Title:

President

I have authority to bind the Purchaser.

IN WITNESS WHEREOF the Vendor has duly executed this Agreement on March 7, 2012 at 1914 1919

DOMFOAM INTERNATIONAL INC.

By:

Name: ANTONIO VALLEC

Title: CFO

I have authority to bind the Vendor.

The Vendor shall indicate its acknowledgement of and its agreement with the foregoing by signing both copies of this Agreement in the space provided below and returning one fully executed copy to the Purchaser (which return may be made by fax or Email) to:

LPLV

c/o:

Mtre. Jacques Vincent

Fax:

(514) 798-5599

Email:

j.vincent@lplv.com

by no later than 3:00 pm, on March 9, 2012, after which time this Agreement shall be considered null and void as to the Purchaser.

IN WITNESS WHEREOF the Purchaser has duly executed this Agreement on March 8, 2012 at 5:30 p.m.

4037057 CANADA INC.

By:

Name: Terry Pomerantz

Title:

President

I have authority to bind the Purchaser.

IN WITNESS WHEREOF the Vendor has duly executed this Agreement on March \_\_\_\_\_, 2012 at \_\_\_\_\_.

DOMFOAM INTERNATIONAL INC.

Ву:

Name:

Title:

I have authority to bind the Vendor.

## Schedule 1:1(m)

#### Contracts.

- 1. Subject to Section 2.3, the following Contracts shall form part of the Purchased Assets:
  - 1.1. All lease agreements in relation with the Leased Premises;
  - 1.2. Collective bargaining agreement with the Unionized Employees for the period of 2009-2014 (Teamsters Québec, Local 1999);
  - 1.3. All Equipment Operating Leases, to the extent that the Purchaser wishes to avail itself thereof and are not part of the Excluded Assets;
  - 1.4. Exclusive agreement for the Canadian use of specific chemical and formula for Gel Visco;
  - 1.5. All contractual arrangements with the suppliers to the extent that the Purchaser wishes to avail itself thereof;
  - 1.6. The annual Microsoft maintenance contract with PYA Solutions Inc.;
  - 1.7. Microsoft (Navision Licence agreement);
  - 1.8. Insurance policy re: Property and Liability;
  - 1.9. Însurance policy re: 2002 Ford;
  - 1.10. Coface (receivables);
  - 1.11. Insurance policy re: Directors and Officers liability;
  - 1.12. CN non freight leases for tracks; and
  - 1.13. Canadiens hockey tickets...

# Schedule 1.1(n).

### Customer Contracts

- 1. Subject to Section 2.3, the following Customer Contracts shall form part of the Purchased Assets:
  - 1.1. Customer bearing number #U11771 in the Books and Records:
  - 1.2. Customer bearing number #11682 in the Books and Records:
  - 1.S. Customer bearing number #10005 in the Books and Records;
  - 1.4. Customer bearing number #10072 in the Books and Records.
  - 1.5. Customer bearing number #U12129 in the Books and Records;
  - 1.6: Customer bearing number #10020 in the Books and Records;
  - 1.7. Customer bearing number #10244 in the Books and Records;
  - 1.8. Customer bearing number #11665 in the Books and Records;
  - 1.9. Customer bearing number #11924 in the Books and Records;
  - 1.10. Customer bearing number #11164 in the Books and Records;
  - I.11. Customer bearing number #10413 in the Books and Records;
  - 1.12. Customer bearing number #10585 in the Books and Records:
  - 1.13. Customer bearing number #U12164 in the Books and Records;
  - 1.14. Customer bearing number #11209 in the Books and Records;
  - 1.15. Customer hearing number #11982 in the Books and Records;
  - 1.16. Customer bearing number #U11429 in the Books and Records;
  - 1.17. Customer bearing number #11177 in the Books and Records;
  - 1.18. Customer bearing number #11268 in the Books and Records;
  - 1.19. Customer bearing number #11848 in the Books and Records;
  - 1.20. Customer bearing number #00202 in the Books and Records;

The state of the state of the second state of

Domfoam international inc. Lease commitments worksheet Dec.15.2011 (for fiscal year ended Aprison)

( North Company of the Company of th

And the second s

[_] \_m					Minimu	m Future Leas	Minimum Future Lease Commitments	Slo	
332467 1 332468 1 332469 1 329315		"Fase	2011	2012	2013	2014	2015		2017
332468 1 332468 1 329316 1	·	Expiry	Current Yr.	is:	274	3rd	득	in.	
		MMPYYYY	Expenses	'Year	Year	Year.	Year	Year	Balance
		05/2013	21.204	21.204	21 204	1.767	1		
		05/2013	21 204	21 204	21 204	1 767	•	\$	
		05/2013	21.204	21 204	21 204	7.97	1		
6		11/2013	20:964	20 964	20.964	12 229	-		
↽		11/2013	20:148	.20 148	20 148	11,753		i	
*	ì	11/2018	20 148	20 148	20'148	11 753.	1	i.	
336256 1HTSCAAMXXH227668		11/2013	20 148	20 148	20:148	11 753	j	· ,	
		11/2013"	20-148	20 148	20 148	11.753	1	·ď.	
		11/2013	20-148	20 148	20 148	14 753,	1	· )	
		05/2012	26.922	~ 28 952	. 2.246	1	1	j	
325582 1FUJA6CKX7LY49524	٠	05/2012	26 952	26 952	2.248:	ī	j		
324854 1FUJA60K47LY49521		05/2013	24 780	24.780	24.780	2 065	ł	ï	
324856[1FUJA6CK67LY49522		05/2013	24 780	24 780	24.780	2 065	ţ		
13423 Rental # A4EC380620		09/2014	14 220	14 220	14.220.	14 220	5.925		
13424 Rental # A3EC380443		09/2014	14220	14:220	14 220	14:220	5.825		
. 13422 Rental # A3EC38		09/2014	14.220	14 220	14 220	14 220	5.925		
Rental # A4EC380738			12,900 (.			,			
Tru Sours 344371-212772 Water dispenser Plant	,	91,02/60		¥ 176	1.476	1.176	1:178	1,176	490
14		08/2016	_	-1 512	1512	1 512	1.512	1.512	504
344371-205533: Main Photocopier		11/2016	1	.9 086	9 912	9912	9.912	. 9.912	5 782
344377-205455 Plant & Sales Photocopier	Ĩ.	14/2018-	-	.8 382	8144	. 6 144	g 144	9144	5334
						_		4	
TOTALS		 -	344340	351.596	303.772	144 829	39 519	21.744	12 110
PRÉPARED BY: LS.					DATE	2011-07-11		. :	

## Purchased Assets

- 1. All assets, undertakings and properties of the Vendor of every nature and kind whatsoever, and wherever situated including without limitation the following:
  - 1.1, The Purchased Receivables;
  - 1.2. The Purchased Inventories;
  - All goods other than goods listed in section 1.2 hereinabove that are field for sale, lease or rental (whether such goods are saleable, obsolete or damaged) in relation to the Business carried on by the Vendor including, materials and supplies:
  - 1.4. The Contracts:
  - 1.5. The Customer Contracts;
  - 1.6. The Equipment:
  - 1.7. The Vendor's Intellectual Property;
  - 1.8. All customer lists and lists of prospective customers and marketing and sales materials and databases not already included in the Vendor's Intellectual Property;
  - 1,9. All corporate names used by the Vendor;
  - 1.10. All rights, title and interest in orders on hand and other contractual arrangements with the customers to the extent that they can be assigned and that the Purchaser wishes to avail itself thereof;
  - 1.11. All the benefit of all unfulfilled orders received by and in favour of the Vendon;
  - 1.12. All other tangible and intangible assets and property used in connection with the Business;
  - 1.13. The exclusive right of the Purchaser to represent itself as carrying on such business in continuation of and in succession to the Vendor;
  - 1.14. All Books and Records;
  - 1.15. All plans, sketches and other documents related to (i) the operations of the Business and (ii) the gustomers of the Vendor;
  - 1.16. all other property, assets and rights, real or personal, tangible or intangible, owned by the Vendor or to which they are entitled to but excluding the Excluded Assets;

- 1.17. All R & D receivables, income tax receivables, tax refunds, deposit and deposit letters;
- 1.18. All the prepaid amounts related to:
  - 1.18.1. Insurance Property and Liability
  - 1.18.2. Insurance 2002 Ford
  - 1.18.3. Coface (receivables):
  - 1.18.4. Directors and Officers liability
- 1.18.5. All the taxes (municipal, business and school) for all the Leased Premises.
- 1.18.6.- CN non freight tracts
  - 1.18.7. Canadiens hockey tickets.
  - 1:18.8. Microsoft (Navision Licence)
- 1.19. The season tickets from Club de Hockey Canadien for the seats 104-I, #13 and #14 together with all the rights attached to said season tickets and any renewal thereof.
- 1.20. All the Equipment Operating Leases that the Purchaser wishes to avail itself thereof and not specifically an Excluded Assets as per Schedule 2.2 attached to this Agreement; and
- 1.21. The assets described in Schedule 5.1(o) of this Agreement.

# Schedule 1.1(uu)

# Vendor's Intellectual Property

# Canadian Trademarks.

See document attached Total of three (3) pages

# **US** Trademarks

See document attached Total of one (1) page Constitution of the state of th

Spitus	tormalized on 2014-06-15	मुंबाहाई।हेंच	renewed on 2004-11-08:	resista	renewed on 1997-08-27	registered	renewed on 1959-02-24.	renewed.on.1997-07-16	henewed on 1999-06-22.	repeived on 2005-03-14.	renewed on 2011-05-30.	renewed on 2005-03-14	renewed of 2004-11-01-	paratsigai	rehewed on 1999-03-16	renewed on 2004-11:01	renewed on 2001-12-24	iegistered.	Janewed-an 1997-12-31	relistered	registered	्रिक्टीश्रम्बन्धेत् स	registreed.	renewed on 1999-08-17	renewed:00,2001-12,24	renewed on 2000-01-11	renewed on 1999-06-22	renewed by 2012-02-26	registered	registered	renewed on 1999-05-11
Name of the agent	N/A	Jim Sproule, A.Z.Sponge.	Stikeman Elliott 11.P.	Jim Sproule, A-2 Sponger R. Foam Products Ltd.	Stikeman Elliott LLP.	N/A	Stikeman Elliott L.P ·	Stikeman Elliott LLP	Stikeman Elliöte LLP.	Stikeman Ellott LLP	Stikeman Elliott LLF	Stikeman Elliott [LP.	Stikeman Elliott LLP	Jim Spraule, A-Z Sponge & Foam Products trd:	Stikeman Elifatt LLo	Stikeman Elliott LLR	Stikeman Elliott ILP	Jim Spraule, A-Z Sponge. & Foam Products Etd.	Stikeman Elliptt LIP	Jim Sproule, A.Z.Sponge. & Foam Products Ltd.	Jim Sproule, 4-Z Sponge. & Form Products End.	Jim Spiolule, A.Z.Sponge & Foarm Products Ltd.	Stikeman Elliott L.P. "	Stikeman Elliot; LLP	Stikeman Elliott LLP	Stikeman, Elliott LLP	Stikeman Ellibit IIP	Stikemen Elitort LIP.	Jim Spraule, A-Z Sponge & Foam Products Itd.	Jim Sproule, A-Z Sponge & Foam Products Ltd.	Sukeman Elliott ELP
Exploy. Hate	í	2020-05-07	2019-11-08	02-11-202	2012:08-27	2019-10-28	2014-02-24	2012-07-15	2014-05:22	2020-03-14	2026-06-30:	2020-03-14	2019-11-01	2019-41-08	2014-03-16	2019-11-01	2016-12-24	2020-07-19	12-21-2102	2025-12,01,	10.21-2202	. 2051-05:05	2020-04-25	2014-08-17	2016-12-24	2015-01-11	2014-06-22	2027-02-26	, KE-50-250Z	.2020-02-14	2034-05-11
Name of the current owner.or applicant	Domínam int≛rnational inc.		Damfoam International Inc.	Oppiram International Inc.	Domform International Inc.	Domicam International Inc.	Domícam International Inc.	Domfoaminternational inc.	Domfózm International Inc.	Domfoam international inc.	Domfoam interpational inc.	Domition International Inc.	Domfoem international inc.	Donifqam intemational រណៈ	Domfoam International Inc.	Domfoam International Inc.	Domfoam-International Inc.	บัดการ์อสุก ให้ดูการน้อกระไทย:	Domitoam Internațional Inc.	Domfoam International Inc.	Damfoam International Inc.	Domicam International Inc.	Domitoam International Inc.	Domidari International Inc.	Dominate International Inc.	Comform International Inc.	Dominarh International Inc.	Domiosm International Inc.	Dörnfoam international inc.	Domíoam International Inc.	Romfoam International Inc.
date of application ("A") or registration ("R")	.A: 2011-06-10	8; 2005:05-02	R: 1974:11-08	R: 2002-11-20.	R-1982-08-27	R: 2004-10-28	8: 1584-02-24.	R: 1982-07-16	R: 1984-06-22	R: 1975-03-14	·R:1366-06-30.	.Rt 1975-03-14	R: 1974-11-01	R: 2004-11-08	R: 1984-03-16	R-1974-11-01	8:1971-12-24	8: 200S-07-19	R: 1982-12-31	R: 2008-12-01.	10-źr-800ż:18	Pc 2005-06-06	R: 2005-04-25	Rt 1984-05-17	R: 2973-12-24	R: 1985-01-11	R; 1984-06-72	R-1982-02-26	R:2007-05-31	R: 2005-07-14	R: 1984-05-11
Registration : or application number	1531270	LMC641558	LN4C203073	LMC570966	LMC272231(	LMC623917	LMC288288	LMC271131	LMC292308	1MC205857	LMC145959;	LMC205BSB	LMC2029D8	LMC624903	: LMC289023	LMC202909-	LMC180311.	LMC644665	LMC275264:	.LMC729834.	TMC729835	LMC665748	LMC638221	LMC294245.	LMC180310	LMC258957	LMC292311.	LMCZ65867	. Szággadni	-LNC694418	EMC290646
Tragemark	SPRINGFLEX FOAM	CONTOLIR-MATIC	VILLA	· chabit Foam·	VELVAFOAM	ComfyCel	SPECTRUM	· INSULGARD	CELCOR	PRO-BOND	DOMFORM	Copee	CHĀTEAU	RoyalCel	EXCALIBUR	BIVIERA	PERMA-BOND	Spag-alganus.	VELTEX	FREEDOM FORM	нізотадія	Hug.A.P.galo	NORTHERNTREASURE	LA COLLECTION ZODIAC	W'UNDER-BOND	BONDEK	QUANTUM:	MAGNUM	VISCOMAGIC Memory Fo	Cuddle-Pedic	ZODIAC COLLECTION
	rt	ŭ١٠	'n	<b>A.</b>	L/J	iè	<b>,</b>	ф¢	ρĥ	ģ	Д	77	13,	ች	'n	19	Ä	EĞ H	15	8	17.	Ħ	žį.	75	\$2	. 55	127	.23	 	S.	표:

Lipopolice. Walsi-azografanyiyaenogitademanka table Canada

CANADIAN CANCELLED OR ABANDONED TRADEMARKS: DGMFGAM

i acgressing.	Registration or date of A. or R. application	date gf A or R	Name of the current owner or applicant	date of cancellation or abandonment	Name of the agent
DOMFORM MAPLE LEAF & DESIGN MC375564	41MC375564	R: 1590-11-09	Domfount International Inc.	2006-07-06	Stikoman Elliotit Lip
VISGOPTUȘH	1960194	A: 2009-111-19	Domfoam Interpational Inc.	2010-12-27	Jim Sproule, 8-Z Sponge
SECURIFOAM'S DESIGN	832300	A: 1996-12-20	Domfoam International Inc.	2001-01-10	Stikeman Ellfotet I Pr
NIMBUS	495427		Domfoam Inc.	1983-01-28	Seal, Shaposnick
PRÓTECT-A-RUG.	LMC177017		Domfoam,Int.	1986-09-10	Seal, Shaposolck:
STRATUS	495428		Domfoam Inc.	1983-12-19	Seal, Shaposnick
FLAREX	LMC205782		Domfeam Inc.	1990-04-27	Seal, Shaposnick
CFC& DESIGN	LMC203689	R:1974-12-05	Domfoam International Inc.	2005-07-21	Stikeman Elifott L'I.R.
HABITAT	500121	A: 1983-03-11	Comform Inc.	1986-08-22	Seal, Shaposhick
UROCELL HR	LMC203776	R: 1974-12-06	Dominam International Inc.	2005-07-21	Stikeman Elfort ILP
Domfoam & design	LMC375563	R-1930-11-09	Domfoam International Inc.	2005-07-05	Stikeman Elliott LEP.
SUNRISE	LMC264304	R: 1981-11-05	Domicam International Inc.	1997-07-08	Stikeman Ellibit LLP
ALTO-CUMULUS	495426	A: 1982-11-26	Domfasim line.	1983-12-79	Seel, Shaposnick
CUMULUS	495425		Dómfaam Inc.		Seal Shaposnick
EVER-BOND	LMC184974.	t.	Domfoam Inc.		Seal, Shaposnick
VISCOAIRE.	1204369	A: 2004-01-22	Dominam International Inc.	2005-02-14	N/A.
FlexÇelî **	1198594	י אי אַנסיבי־די־סיב	են ունջոլդ Լուեւրունոց Լուբ		Jim Sproule, A-2 Springe
	•	•	-		S road Republics (to
PINKPANTHER	LMC236776	8:1979-10-19	Domfoam inc.	3990-03-16	Seat: Shabosnick
CIRRUS	495429	A: 1982-11-26:	Domfoam Intc.	1984-01-06	Scal, Shaposnick
		•	1		
			-		

1440000.1.P.1.473152.52001318775187-603177-demarks\_table\_Gunga

CANADIAN SEARCH OF TRADEMARKS DOMFDAM

Trademark	Applicationinumber	date	Name of the applicant,	śnież
GELFLEX	1518007	. 10-50-LIOZ	2011-03-07 Dominam International Inc.	Notice of approbation sent on 2011-10-31.
	•			

.tq00004.R.LV\31513200\3187\3187-003\Trademarks\_lable\_Larada.

Serial number Re TREASURE 78508999. Ze		te Name of the current owner of	nilicalita (aste		Domtoam, International Inc.   2015-12-18   Lawrence H. Meier   registered
Serial, number TREASURE (78508999.		kegistration date		1 4	2005-12-13
TREASI		Serial, number		: : : :	/x50x999.
Tradema Tradema	•	Trademark			NORTHERN TREAST

US CANCELLED TRADEMARK: DOMFOAM

Trademark	Serial number date of	date of	Name of the current conner of date of cancellation or Name of the agent	date of cancellation or	Name of the agent
		application .	applicant	abandonment	
•		•	-		
CUDDLE FOAM	78219889	2003-02-27	Domfoam International Inc.	2004-02-11	N/A
SECURIFOAM & DESIGN.	51GN 75311611	61-90-7661	Domfoam International Inc.	2000-10-12	Jess M Collen

# Excluded Assets

- The Purchaser acknowledges and agrees that the following assets shall be considered excluded of the Transaction:
  - 1.1. The accounts payable, except as otherwise provided in this Agreement;
  - 1.2. The fax losses, except as otherwise provided in this Agreement;
  - 1.3. The cash on hand or on deposit with banks or other depositaries;
  - 1.4. The debts due to the Vendor from any shareholder, director, officer or employee of the Vendor or any other person who does not deal at arm's length with the Vendor.
  - 1.5. The following Equipment Operating Leases:

The second section of the second seco

- 1.5.1. Ryder Lease No. 332467 1FUJA6CK97LY49529
- 1,5.2. Ryder-Lease No. 332468 1FUJA6CK57LY49530
- 1.5.3. Ryder-Lease No. 332469 1FUJA6CK77LY49531
- 1.5.4. Ryder Lease No. 329316 IFUJA6CK37LY49526
- 1.5.5. Ryder Lease No. 329317 1FUJA6CK57LY49527
- 1.5.6. Ryder Lease No. 329318 1FUJA6CK77LY49528
- 1.5.7. Ryder Lease No. 336256 1HTSCAAMXXH227668
- 1.5.8. Ryder Lease No. 336257 IFUJA6CK97LY61325
- 1.5.9. Ryder-Lease No. 336256 1FUIA6CK77LY61324
- 1.5.10: Ryder Lease No. 325581 1FUIA6CK87LY49523
- 1.5:11. Ryder-Lease No. 325582 1FUJA6CKX7LY49524
- 1.5.12. Ryder Ledse No. 324854 1FUTA6CK47LY49521
- 1.5.13. Ryder Lease No. 324855 1FUJA6CK67LY49522
- 1.5.14. Toshiba Lease No. 344371-205533 Main photocopieur
- 1.5,15. Toshiba Lease No. 344371-205455 Plant & Sales Photocopier
- 1.6. The shares issued and outstanding of the capital of "VALLE FOAM INDUSTRIES (1995) INC." registered in the name of the Purchaser; and
- 1.7. The shares issued and outstanding of the capital of "A-Z Sponce & FOAM PRODUCTS LTD." registered in the name of the Purchaser.

# Schedule 2.6

# Assumed Obligations

- 1. The obligation of the Vendor under the Contracts as of the Time of Closing, and
- 2. The obligation of the Vendor under the Customer Contracts as of the Time of Closing;

# Purchased Receivables

# Section 1

List of Purchased Receivables as of January 31, 2012.

See document attached Total of four (4) pages

## Section 2

Calculation of the Purchase Price attributed to the Purchased Receivables.

See document attached Total of fourteen (14) pages. DOMFOAM INTERNATIONAL LIMITED Aged Accounts Receivable as of 01-31-2012

00157       704;58       704;58       -         00202       25,631,17       25,431,85       199,32         00291       5,362,49       1,032,48       3,320,06       1,009,95         00308       4,316,16       4,316,16       -       -         00316       1,952,01       1,736,65       215,36       -         10005       511,571,37       303,779,87       180,349,98       27,441,52         10020       128,444,89       128,444,89       -       -         10024       2,028,58       -       2,028,58       -         10052       7,668,75       6,949,67       719,08       -	
00291:       5;362,49       1;032,48       3,320,06       1,009,95         00308       4,316,16       -       -         00316       1,952,01       1,736,65       215,36       -         10005       511,571,37       303,779,87       180,349,98       27,441,52         10020       128,444,89       128,444,89       -       2,028,58       -	1 1 1 1
00308       4,316.16       4,316.16         00316       1,952.01       1,736.65       215,36         10005       511,571.37       303,779.87       180,349.98       27,441.52         10020       128,444.89       128,444.89       -       2,028.58       -         10024       2,028.58       2,028.58       -       -	7
00316     1,952,01     1,736,65     215,36       10005     511,571,37     308,779,87     180,349,98     27,441,52       10020     128,444,89     128,444,89     -       10024     2,028,58     -     2,028,58	1. 1. 1.
10005     511,571.37     303,779.87     180,349.98     27,441.52       10020     128,444.89     128,444.89     2,028.58     2,028.58	
10020 128,444.89 128,444.89 10024 2,028:58 - 2,028:58 -	: :- :-
10020 128,444.89 128,444.89 2,028.58 - 2,028.58 -	۳ ت
10024 2,028:58 - 2,028:58 -	<b>-</b>
10052 7 66979 6 976 67 976 97	
within therein afficial trainer is	_
10072 528,925,52 258,533.73 155,454,72 114,937,07	4
- 10080 41,326.70 41,326.70 -	
10120 1,158.95 1,158.95	***
10136 17,679.48 17,679.48	-
10145 5,746.68 3,072.41 2,674.27	<u> -</u>
10222 1,004,61 1,004.61 -	
10244 101,343,87 101,343,87	_
10273 15,318.85 13,356.08 2,952.78 -	
10305 6,438.19 3,402.81 2,141.95 893.43	_
10331 2,254.11 2,254.11	<u>.</u> ·
10351 3,754,38 - 3,754,38 -	_
10358 29,970.58 11,583.52 10,024.90: 8,412.16	÷,
10371 82,216.28 19,133.32 1,056.44 11,342.99	683.53
10378 7,105.73 1,598.57 1,859.97 -	3,647.19
10379 3,680.61 3,680.61	ugar.
10384 45,107.85 27,771.01 16,306.37 1,030.47	_
10385 10,927,29 10,927.29 - +	
10387 7,615,54 - 1,173,56 410.29	6,031.69
10413 153,635.36 42,935.08 19,370.39 55,168.67	36,161.22
10426 3,529.98 3,529.98 -	ىك
10465 833.89 833.89	-
10494 32,749,54 32,749,54	_
10502 11,893.25 3,266.84 1,087.18 3,549.67	3,989,56
10509 1,729,11	·
10510 18,715,15 18,715,15 -	-
10521 2,826,31 2,826,31	÷
10545 28,034.38 28,034.38 + ~	_
10547 15,568:17 5,498.67 10,069.50 -	,
10551 74,572.24 74,572.24	5 <b></b> ·
10570 6,278.42 6,278.42 -	-
	67,225.29
10595 344.24 344,24 -	#
10737 1,249.53 1,249.53	

Customer No.	Balance Due	Current	31-60 Days	.61-90 Days	Over 90 Days
10785	6,075.15	6,075.15	÷	-	· ·
10848:	3,937,95	<del>-</del>	1,580.27	2,357.68	<del>*</del>
10864	284.40	.284.40	4 🚑	_	-
11049	9,484.48	6,739,35	2,745,13		~
11150	3,850.31		3,850.31	~	<b>:</b>
11155	2,615,92	2,615.92	-	۲٠	· <del>-</del>
11164	48,779.05	51,488,31	(2,709.26)		÷
11177	89,437.38	30,746:66	25,400.93	24 <sub>2</sub> 863,69	8,426,10
11197	626.81	626,81	4	<del></del>	· .
11209	29,097,79	29,097,79		-	2
11210	1,778.75	1,115.72	663.03	<i>!-</i>	. 121
11268	36,681,28	36,681,28	,	÷	<del>κ</del>
11272	672.18	1,034.89	(362.71)	-	4
11276	15,524.84	÷	15,524.84	<del>7</del>	7
11289	16,671.06	5,017,77	5,249.62	4,461.72	1,941.95
11399	80,315.23	6,706.11	9,359.14	7,755.01	56,494,97
11415	11,174.51	9,312,23	1,862.38	4.	are.
11421	15,293,31	905.43	8,702,70	5,685.18	·
11457	26,525.81	26,525.81	÷	7	<b>-</b> -,
11558	3,646.51	3,121.23	.525.28	<b>*</b> ***	-
11577	5,027.53	4,979.82	·		47,71
11580	442.19	442.19	<b></b>	<b>±</b> ,	<del>-</del>
11627	1.80	1.80′	۵.		# .
11665	387,973.57	138,756.63	112,323.23	117,018.61	19,875.10.
11676	42,927.94	19,314.91	17,404.97	6,208.06	,:-
11682	254,229,37	253,895,10	<b>4</b> 4	334.27	
11684	5,358:46.	5,758.76	(400.30)		ے
11714	8,989,08	8,716.63	353.56		(81:11)
.11760.	.885,20	÷	885,20	.∳	;
11779	10,301.45	10,301:45	. ~	~	<b>~</b>
11811	18,946.71	18,946,71	<b>5</b> 4.	<del>-</del>	-
11828	4,191,55	4,191.55	<del></del>		å
11833	1,682.72	1,098.94	· <del>-</del>	583.78	4
11844	5,903.05	5,903.05	. • -	-	خ
11848	93,242.34	24,928:53	30,248.93	22,139,94	15,924,94
11859	6,694.80	6,694.80		. ~~	.=
11871	1,171.60	1,171.60	#·_	·÷•	.=
11914	.36,917,36	23,905,21	9,420.51	3,591.64	4
11,924	231,421.59	30,900.74	31,178.08		95;602;83
11927	42,533,30	25,496.59	17,036.71	÷.	
11948	3,133.52	3,133.52		₽.	L
11974	(11.50)				rup.
11976	.23,398.27	11,558.30	11,839,97		* <b>3</b>
1:1979	29,254,14	27,397,56	1,856.58	, <del>-</del> -	· —
11982	42,209,54	42,500,44	•	(290.90)	· -

	Customer No.	Balance Due	Current	31-60 Dáys	61-90 Days	Qver 90 Days
	12021	6,359,25	5,618:26	740.99	£,	
	12101	965.55	<b>-</b> i	'965.55	÷	÷
	12102	71,175,61	27,028.49	13,870,53	30,276.59	<b>F</b>
	12104	40,670.61	20,573,97	20,181.29	(84,65).	. <del></del>
	12122	(32,80)		7	Agent 13 mg/ga	-
	12140	17,913.60	17,913,60	<del>-</del>		_
	12146	20,962,95	4,908.38	5,781,24	5,890,81	4,382.52
•	12158.	17,823,38	17,823;38	÷	, - <b>,</b>	7 tan arawa —
	12230:	6,506,76	4,144,61	2,362.15	<del></del>	. =
	12242	9,119,76	8,420.15	699.61	_	•
	12253	5,080.77	1,217.35	3,863,42	<b>-</b>	_
	12256	17,887,78	17,887.78		L	<b>#</b>
	12270	4,819.79	(3) ( = 1) (3)	4,819.79	-	-
	12277	2,346.65	1,225,63	747.35	373,67	
	12282	5,362,13	5,338,88	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	23:25	<u>.</u>
	12290	2,002,76	2,002.76	_		_
	12291	596.03	596.03	~	-	_
	12296	29,232.85	- 384.08	- 15,929,32	10,493.91	2,425.54
	12298	59,795.77	25,619.87	14,419.36	17,458.72	2,297.82
	12301	13,668.05	9;923,90	3,744.15	—- <b>7 (5</b> 7,50 mg	-स्युक्ताराक्त
	12303	1,184,25	1,184.25	-,, ,	· <del>-</del>	_
	12308	12,344,90	12,344.90	<b>→</b>	a	
	12314	2,330.86	1,275.30	1,055,56	-	<b>~</b>
	12326	916.76	·524.3 <del>5</del>	392.41	·	-
	12330.	4,543.77		~	1,005.11	3,538,66
	12331.	23,556.06	7,529.09	6,507.75	8,366.47	1,152,75
	12334	3,620.86	3,620.86	÷,	**	
	12341	1,179.80	1,179.80	<b>-</b> .	-	-
	12343	28,957.45	19,659.45	9,079.05	209.90	9,05
	12353	1,131,12	1,131,12		e e	
	12355	4,681,06	4,681.06	₩	_	•
	12357	11,881.35	1,573.48	ani:	2,016.81	8,291.06
	12358-	1,294,39	1,294,39	· =>	7 7	
	12367	9,722.17	9,722.17	, + m	-	_
	12373	5:00	5.00	-	<del>-</del>	
	12380	1,454.04	1,454.04		2,-	_
	12390	3,304.57	3,304.57	· .	<u> </u>	_
	U11429	15,044,90	15,044.90	÷	÷	_
	U11771	1,079,755.37	556,732,35	358,231.67	164,791,35	_
	U12129	544,154.40	168,733.64	160,191.76	178,981,57	36,247.43
	U12164	75,413.54	20,979.01	53,559.58	874.95	ar and the state of the
	U12171	8,393.63	7,170.44	1,223,19		: : 4
	U12180	3,884.22	190.10	3,694.12	2	
	U12183	236.00	ুল কলাই ক	-x -m' 4	236.00	-
	U12184	12,578,70	12,578.70	<b></b>	74 T-10%00.	-
	·:		ा <b>रस्य र</b> ाष्ट्री			

Customer No.	Balance Due	Current	31-60 Days	61-90 Days Ov	er 90 Days
U12187	12,428,10	12,428.10	<del></del>	~	-
U12188.	17,757.90	17,757.90		<u> </u>	
Total	5,996,691.62	3,208,741.60	1,434,256.08	979,378,14	374,315.80

AGED ACCOUNTS RECEIVABLE	DOMFOAM INTERNATIONAL INC.
SED ACCOUNTS	DAM INTERNA

เลยเกรเกรา	
TOTAL	25 625 1. 1 1. 125.20 1. 125.20 1. 12
OVER 90 DAYS	
61-90 DAYS	
31-60 DAYS 61-90 DAYS	26.661 27.662
CURRENT	25 400 40 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1
. <b>T</b>	**************************************
СЯЕОП СІМП	25 000,000  15 600,000  16 600,000  17 5 000,000  17 5 000,000  17 5 000,000  17 5 000,000  17 5 000,000  18 5 000
NET.	ឧទ្ទុខឧទ្ទុធមន្ត្រីកាន្ទនាង ខ្លួន ដូច្នាន់ក្នុងខ្លួនគ្នាធ្ងន់ក្នុងខាងខ្លួនគ្នាន់ក្នុងខ្លួនទំនួនទំនួងក្នុងទំនួ ឯកស្រុកការសំសុស្ត្រស្ត្រស្ត្រ ស្ត្រីស ឯសភាពសុក្សសុកការប្រសុសសកល្ខេសសក្សសកមកប្រភពសុភាគមិត្ត
	• :
NAME.	1457 VENÍ: REZ. PRODUITS. 297 DINE ÁRTÍNG. 297 DINE ÁRTÍNG. 398 POSITECKTINNOVATION ING. 398 POSITECKTINNOVATION ING. 398 POSITECKTINNOVATION ING. 10052 ACHAVIO ING. 10053 ACHAVIO ING. 10053 ACHAVIO ING. 10054 ACHAVIO ING. 10054 ACHAVIO ING. 10054 ACHAVIO ING. 10055 ACHAVIO ING. 10055 ACHAVIO ING. 10055 ACHAVIO ING. 10055 ACHAVIO ING. 10056 ACHAVIO ING. 11057 ACHOVIO ING. 11057 ACHOVIO ING. 11057 ACHOVIO ING. 11057 ACHAVIO ING.

420 647,54 ·· S 214 638,67 \$-

SUB TOTAL CON

1,80, 42 927,94	5.35g/46.	OSIGNATION OF	18.946.71	4-191,55	6 903,05	46,442.68	6.64.85	36.017.10.00.	42 563.30	. 133 gr	(41,50)	23.398527	42.209.54	035925	965,55	40.670,61	(32,80)	100 CAO CC	0.000	823,3	8.508,78	19767.5	2 USU, 77	481979	2.348,53	5 362,19	606.03	29,292,85	22,285,22	10 34 5 CH	2330,86	919,78	4.040177	20 000 00 00 00 00 00 00 00 00 00 00 00	179,80	28.557,45	1131	4.581,06	1, 201,645	9.722,47	ÒO Y	1.454,04	3:304,57
देश-स्थाप ४१ ५			(-148) (-1			15.924,94		n e			<i>V</i>					υ <u>σ</u> :		A 1080 TV	301301		AM (S				i)		à ti	40	297,82	4 <b>u</b> ;		ť	8.538,66	اعراب ا	* i.a	9,05				· (a			
17.404.97 \$ .9 208.06 \$	)- j.	اء د رويون م	k	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	) )	3 \$ 22.139.94	in i	3 3391.64	036,71 55	r m	es à	1000 to 1000 t	(280.80)	9,		(84,85)		5.800 Rt	Tolland St. D. Latina	ja	.1 •	5 19 19 07 07 07 0	, i	יי. קיי	\$. 679,67	23,25	• ;	\$ 10493,91	36 .5. 17/458,22	ę a	Š	392,41 5	1.005,11	49.66	. 10	209,90	r of .	19. C.	in the second		1.	ì us_	r 
19314.91	1 10 pt		18.946.71		٠ _	5.	5 684 80 · 5		25 496,59		(11,50) 0		\$00 4#	618/2B.	e)i		7. (32,80)	80 800		323,38		20.15	287.78	); ) )	225,63		585,03	384,08	619,87	22 344 97 S	275,30	524.35 5			9 08 67 L.1		131,12		2. 95 Apr.				
8 000000 c		30,000,00	125 600,00	25-000:00	45 000,00	100:000:00	30,000,00	60,000,00	95 000,00	10.000,01	7.000.00 20.000.00	00,000,86	75 000,00	4\$ 000,00	15,000,00	65,000,00	20 000 00.	20.000.02	50 000,00	175 000,00	40 000,00 %	מס מסט של זיי	30,000,000		15 000,00	59 000,00	5 300,00	25.000,00	00,000,00	5 000 00	20 000,00	5,	00'00\$ £	15 000 pu	00,000 &	75 000,00	10,000,00	30,000,00	Adding or Au		00,000	15 000,00.	
9.00 K	8 2			, G	96	75					G 1					2	6 6	3 19				n ç		Post dated			200	.08		QDD.	i		•	2 5	000	<i>(</i> ,				30	Đ,	Ą.	
11627 PRO TACINDUSTRIES INC. 11675 LICERIE DORMATEXING. 11674 FOLIETAR CANADA INC.	14714 TRICK	11779 WANUFACTURE DE MATELAS	14811 BROLAIN DISTRIBUTORS (TD.	19838 WATELAS SELECT	11844 PEL INTERNATIONAL LIZE	1784B INDUSPACING	11859 WILL BOARD AND LINE	11914 EXOEL RSVC MOUSSE FOAM	14927 MEUBLES BELISLEING.	1184B.MEUBLES TOMALING.	41974 STYLTEC INC.	11979-BELLPHONEDICAL INC.	11982 DIAMOND DIVERSIFIED IND. L.	12021 SEARS CANADA INC.	12101 CREO PACKINTERNATIONAL	2104 AD MATS CORPORATION	12122 GILAN (G FABRICS LIM LED	22146 MATELAS BEAU BEVEING	12/50 REMBOURRAGE JA MARTINE	12158 BARRY SUPPLIES & FABRIC!	12230 WESOO NA INC.	172252 OF OH! OTO I EMPLOADA IN	ACCUMULATIONS OF THE THE TANAL OF THE TANAL	12270 COUTURE CARLYCEING, CW	12277 DOMINION CUTTING SERVICE	12282 VICENE 2006 INC.	14450, FASTER INC.	12286 PAVARIFURNITUBEIND.	12299 G. HOMANO INC.	12500 WHO INDUO ( TIED WO.	12314 GROUPE'SM PELLETIER	12326 HABWELL EMBALLAGEW NI	12330 GREATECH INC, 90690744 QUE	Add	12341 REVOLUTION SANTE 919645	Teads JAYMAR Brix	12363 EQUIPMENT SANITAJAE CHER	TREES ADI AR CODSIGN INTERNATION TROPES VALENTINO INC. NO.	12358 LINEA PUNTERNATIONAL INC.	12367, CONFORT EXCELFLEX INC	12373 PPD FOAM SOUDTION INC.	12380 @STRIBUTION SECURI - SPOR	12899 GONGUEST CABRET CORP.

25 900,00 30 900,00 30 900,00 31 900,00 38 900,00 38 900,00 37 600,00 37 600,00 37 900,00 37 900,00 37 900,00	19371 SYLLOO INC. 19387 LITERIE DRIUMMOND LTEE 1938 SE 50 000.00 ST 172 000.00 ST 172 000.00 ST 172 000.00 ST 173	303 779 87 \$ 160349.36 \$ 27.441.52 \$ 689.59 \$ .92 15, 27 \$ .950 000, 00 10 173, 65 \$ 410, 24 \$ .95 15, 24 \$ .92 15, 24 \$ .173, 65 \$ 410, 24 \$ .93 15, 24 \$ .775, 124 \$ .775, 124 \$ .775, 124 \$ .775, 124 \$ .775, 125 \$ .77	429506 \$ 222 854,79 \$ 126374.93 \$ 158 813.02 \$ 11015.452.55 \$ 42935.08 \$ 19.370,34 \$ 52.966,57 \$ 39 161,22 \$ 19.370,34 \$ 22.016,00 \$ 30.746,88 \$ 22.0166,00 \$ 30.746,88 \$ 22.0166,00 \$ 30.746,88 \$ 22.0166,00 \$ 30.746,88 \$ 112.923,23 \$ 112.016,61 \$ 19.825,10 \$ 30.747,38 \$ 126.767,89 \$ 112.923,23 \$ 112.923,81 \$ 112.923,81 \$ 112.923,81 \$ 112.923,81 \$ 12.923	60% 60% 60% 60% 60% 736.0563.41 \$ 390.639.88 60% 800.639.88 60% 60% 60% 60% 60% 60% 60% 60% 60% 60%	253.895.10 % 34.77 % 334.77 \$ 5.254.28 % % % % % % % % % % % % % % % % % % %
		50 000,00 30 000,00 40 000,00 175 000,00 185 000,00 38,000,00 200,000,00 25,000,00	378 (00),00 \$ 228 (00),00 \$ 150 (00),00 \$ 400 (00),00 \$ 1250 000,00 \$	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1,750,000,00

(M)

뜣

ECEIVABLE	TIONAL INC.
AGED ACCOUNTS RECEIVABLE.	DOMFORM INTERNAT

NAME	RET	CREDITLIMIT	CURRENT	SI-60 DAYS. 61:90 DAYS	OVER DO DAYS	TOTAL
USD JTTM LEHECO FIRRES'ING.	ं छो (तुः पुर	2.030.325,00 \$.	\$56,38,35	358 231,67 \$ 164,791,35 . \$	e, 2 Age 2 Jo	1,079,755,37
Utzieł otis bed manufacturing c utzit franklin august trading	6 6 6 6	\$ 00,580 %	717045	1 223.19. \$ 874.95 .\$	9 V.	75 413,54
ZTBB FOAM FACTORY INC.	. 09 8	\$ 60,598 6			rus t	236,00
2184 GREAT NORTHERN ASSOCIAT	8 6 N3	59 172,00 %	12.578,70 \$	10 to	φ.† • t	12.578.70
STAR KRAUS USAINO.	, <del>, ,</del>		17 767 90 'S	#: 69 1 - 3 79 - 60 1 - 1	h; 44	12,428,10
ZIBOIL ATEX INTERNATIONAL	90	s. 00,000 pt:	190,10 . \$		Ç <b>şi</b> r L	-3 884,22
SUB TOTAL USD	[rj]	2 645:787,00 \$	627.896;50 \$	416 708,56 · \$165 902,30 \$	· S	T.210 447,46
11429. NRFDISTRIBUTORS JNG:	\$ 00	255 410,00 .5	15.043,80	9) 1) 2)	<b>69</b>	15 044.90 S 40,000,00
SUB TOTAL USD Insured	W	255410,00 .5	15.044.90 \$	9 . 9	V	15.044.90 \$ . 40.000,00
UIZIZO WOW INC.	\$ :06	1,380,671,001,5	188 735 84 S	. \$. 75,189,871.2. 87,191.081	36247.43 \$	564.189,40 .\$. 150,000,50
Sup Total USD Inspred Net 90	S	'S. ' 1:380.671,00 \$	168 733,64 · .\$	160 191,76 \$178 981,57 \$	.36.247,43 'S.	544.154,40 \$. 150,000,00

219.742/81 'S 145.948,00' S 58.065,81 :S 423.656,61,  5.265,72 S 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5.	32%	35%	.35%.	10%	
35% 35% 528 S 5 28 36% 36 067,72 5 .62 643,55 5, 12 886,60 5 190.45 \$ 201.915,11 \$ 120.709,35 \$ 7.2 886,60 \$ 419.371	219,742,81	145 848,00; \$	58 065,81: :S	ía ·	423 656,61,
\$5.85. 35% 35% 35% 55.85% 55.85% 35% 190.45% 55.85%	35%	332%	35%.	32%	
36% 36% 36% 36% 36% 36% 36% 190.4% 36% 190.4% 36% 36% 36% 36% 36% 36% 36% 36% 36% 36	5 265,72 \$	3.	65	S	\$ 265.72
56 067,12 \$ .62 643,55 \$12 686,60 \$ 190 A8. \$ 201 915,11 \$ 120 709,35 \$12 686,60 \$ 619,371	35%	38%	38%	35%	*****
\$ 20f 916,11 \$ 120709,35 \$,	59.056,77.13	56 067,12 5	62-643,55 .\$	12,686,60 \$	190.454,04
20f 915,11 \$ 120709,35 \$,	*			***************************************	
201 815,11 \$120709,35 \$				,	%0
201 915,11 \$ 120709,35 \$ . 12 666,60 \$	,	,		w	
201 915,11 \$120,709,35 \$ 12,686,60 \$		:			
	1284.055,30 .\$	201 915,11 \$1	20-709.35 .\$	12 686,60 \$	619.376.37

35247,43 \$ 1769,646,76

576 900,32 \$344 883,87 \$

811 615,14 \$

USD CONVERTED TO CON.
EXCHANGE RATE JODGA, 2012
TOTAL CON GROSS
TOTAL CON OFFER.

SUBTOTAL GANNA COUNT

TOTALOFFER

SUB TOTAL USD Insured

SUB TOTAL USD Insured

TOTAL OFFER. SUBTOTALUSD

TOTAL GROSS

1,769:646,76, 619:378,37 36247.43. S.

576 900,32 - (\$344 983,87 \$

811.615,14 \$

ŵ.

Ŧ

냌.	Ď.	
3	AL.	
1100	ă	
7	RNA	
Š	본	
} }	₹.	
į	FF	
É	ğ	

Repaies

fnsúrance									
Total	(14-\$08,00);	(24.868,08).	(127,739) (1,158,95)	162,69	(1.454,44). (1.403.14). (7.139.28).	iges is to se	(4.801,00). (6.198/47).	(\$'615,92); (\$'626,76); (16269,76)	(15.524,84) (6.523,39) (13.380,09)
DVER 40 DAYS	ઉંકપો ઇક્સો •	us do iti (	કું જે <del>હે</del> લ્લું જે લે	ના કે કુક કુક જો કુક કુક કુક કુક કુક કુક કુક કુક કુક કુ	<i>છે. છે. છે. છે</i>	ળ જેલ્લું છે છે	ાં. ભ નાંતુલ <b>લ</b> ે. લે જે જે જે	કું સ્વનાય કર્યા કર્ય -	(2017-0-4) (2017-0-4)
61-90 DAYS			ý 14 saz, ez	162,69	(1-554,44).		•	,(ġ٩4;oṭ),	. (4 <del>4 6</del> 1,7 2) . \$
31-50 DAYS	(25,861)		(12 802,927) și (14,937,97)	· vý	\$ \$ (6:108,81)			<b>.</b>	(15 52,64) \$ (26,64)
CURRENT	(14:608,68) S	(24.868,08)	(36'851 1).		(1'405;14)		(4 au.,00) (6 195,47)	(2 <u>6</u> 16,92) (16,269,76)	(18 380 08)
19	% <u>8</u> 888	8.8.8.8 4	න ද දෙස් පිළිබු	0000	0,000,00	S S S S S	# # 888888888	, , , , , , , , , , , , , , , , , , ,	* * * * * * * * * * * * * * * * * * *
OREDIT LIMIT	25.000,00 25.000,00 15.000,00 7.000,00	350 000,000 350 000,000 350 000,000	25.000,000 25.000,000 25.000,000 25.000,000 25.000,000	175 000,00 .175,000,00 40,000,00	30.000 30.000 30.000 30.000 37.000 37.000	25.000 35.000 35.000 30.000 30.000	15,000,00 15,000,00 15,000,00 35,000,00 6,000,00 6,000,00 6,000,00 6,000,00 6,000,00 6,000,00	27,000,00 4,000,00 4,000,00 1,200,00 1,200,00 1,000,00 1,	25 000,000 25 000,000 26 000,000 25 000,000
. iği	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		48888 444444	6 th th	0 0 0 0 0 0 0 0 0 0	8 8 8 8 8 8 8 8 8 8	a 8 8 8 8 8 8 8 8 8	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	88686 <del>6</del> 88 10000000000
. NAME	152 Vennez Produtts 202 Bugatti design ing 291 Dine Art Ing. 308 Positech Innovation ing.	316 LIND FURNITURE (CANADA)LT 10020 LBC CANADA ING 10020 ARWAY SURGICAL APPLICAN	10072 KRÁUS CÁNAĎA INC.(HO) 10080 CÚVÁLITY RUGS, CRÁNÁDA LI 10120. BEAVERT CARPET LTEE. 10136. BOBLEN INC.	10222 MATELAS GONDOR ENRG. 10244 MATELAS DAUPHIN.INC. 10273. DOUBLETS INC. 16090 BENNETT FLEET (OUERC) IN		10985 MATERAS LION DOR INC. 10455 COMPONS TO 10465 COUPONS TAHENT INC. 10464 REMBOURRAGE A.N.P. 10502 LITERIE PROVINCIALE L'FEE	10909 AMBUBLAMBUT, FENAISSANG 10910 MATELAS HENE INC. 10621 REMBLOUIS ROUSSEAUNC. 10545 SERWAY PLASTICS LITE 10547 SEBY, CIOHB. DU CANADA 10570 SPORT MASKA INC. 10505 SPORT MASKA INDUSTRIES.	10745 GROUPE GIBOJINO. 10846 PRINCESSE 10846 LITERIE SERUCTION INC. 11046 LITERIE SERUCTION INC. 11150 EROFOW INC. 11157 ANSON INC. 11208: REGMICARIRIER/INC.	11272 INTERACIONE INT. 11272 INTERACIONE INT. 11293 MATELAS SIESTA. 11495 FIGGKLAND TEXTILES ING: 11450 ANTELAS PRESTIGE 11550 QUATRO TECHNOLOSIES INC. 11577 ARTOPEX PLUS.INC. 11577 ARTOPEX PLUS.INC. 11580 FUTQN DESIGN.

(U)

Ľ	വ്
널	Ζ.
₹	궺.
11.	Ż.
	용.
Ħ	3
Ž.	품.
Z,	
	Z.
Ž.	-
ď.	ই
ς.	Ĕ.
3	Ö.
Κ.	ă

Process p	n n in	68.883 (8.888)	கு கு து வஞ்ஞன்ற இது சு ஒது ஒது சு சு சு ச ச ச ச ச ச ச ச ச ச ச ச ச ச
			(\$6°)
· · · · · · · · · · · · · · · · · · ·	(11 128,04)	(740,99)	,
· .	(\$ (603))3).	\$ (58,878,8)	(4.652,89)
3 000,00 30 000,00 10 000,00 75 000,00 2 000,00 12 000,00 12 000,00	44 6 000 000 000 000 000 000 000 000 000	45,000,000 46,000,000 46,000,000 20,000,000 20,000,000 46,000,000	20 000,000 000 000 000 000 000 000 000 0
· 罗鲁杰曼曼坚整 · · · · · · · · · · · · · · · · · · ·	<u>ૣૢૢૢૢૢૢૢૢૢૢઌ૽ૢૢૢૢૢૢૢૢૢૢૢ</u> ઌઌઌઌઌઌઌઌઌઌઌઌઌઌઌ	Post taged 8 5 6 8 6 8 4 7 7 8 8 8 8 9 9 9 8 8 9 9 9 9 9 9 9 9 9	0 0 0 0 6 8 8 8 4 8 8 8 8 4 8 8 8 8 8 8 8 8 8 8 8
1162'F PROTACHOUSTRIES'NC. 1166'FUISTAR, CANADA NC. 1176'FUISTAR, CANADA NC. 1176'YERANDA: ANC. 1176'YERANDA: ANG. 1176'YERANDA: ANG. 1176'YERANDA: ANG. 1181'L BHOLAIN DISTRIBUTORS: UED.	1994 PEL INTERINATIONAL LITEE 1994 PEL INTERINATIONAL LITEE 1994 PEL INTERINATIONAL LITEE 1995 WEST BORD FLOORING INC. 1993 MILLENAIRE WILLIAM INC. 1994 MILLES PELSE MOUSE FOAM (1994 MILLES TOWAL) INC. 1994 MILLES TOWAL) INC. 1994 SITUTE INC. 1997 SITUTE INC. 1999 RELIPRO MEDIOAL INC. 1999 PELIPRO MEDIOAL INC. 19	12021. SEARS DANADA INC. 12021. GRE O PACK INTERACTIONAL 12104. AD WATS CORPORATION 12104. ATLANTIC FABRICS LIMITED 12104. DOWNEAST. PLASTICS. LTD. 12106. MATELAS BEAU REVEINO. 12106. REMBOURFAGE JA WARTINE 12106. REMBOURFAGE JA WARTINE 12106. REMBOURFAGE JA WARTINE 12106. REMBOURFAGE JA WARTINE 12206. WESCO, NA INC. 12202. SPORT SYSTEMS CANADA. IN 12203. BEAULEU JANADA. IN 12203. BEAULEU JANADA. 12203. GEOUTURE CANADA. 12204. COULTURE CANADA. 12204. COULTURE CANADA. 12207. DOMINION CANAGE STATIONAL SERVICE.	12282-VICLINE 2005 INC. 12281-PAITEKING: 12285-PAITEKING: 12285-PAITEKING: 12286-PAITEKING: 12286-PAITEKING: 12308-PAITEKING: 12308-PAITEKING: 12308-PAITEKING: 12308-PAITEKING: 12309-PAITEKING:

SUB TOTAL CON

(

ebate

350 000,00 146 000,00 146 000,00 146 000,00 146 000,00 146 000,00 146 000,00 147 000,00 148 000,00 148 000,00 148 000,00 148 000,00 148 000,00		. 280 000,000		,	
(171, 143,58) (18,633,98) (48,632,79) (17,760,39)	(256.570,577 (10.577,75)	(10,677,75)	(153.942.34) (153.942.34)	30%	ì í
14 17 60 00 00 00 00 00 00 00 00 00 00 00 00	1 1	(10 577.75) 5:	(388,39) S. (388,39) S. (10,656,18) \$ 60%	\$. (17 691,22). \$	ig ig.
(143.702,06); \$. (27.441,52);	(27 441,52)	S. (87,829 Qt.1) 2. (80,828 £81)	60% (50 189.58) \$ (73 414.98) \$ \$ 60% (96.221.24) \$ (16.464.91) \$ 60% (66.221.24) \$ (16.464.91) \$ \$ 5	(116.359,83) \$ (89.879.87) \$	
(30'558 81)	\$ (69'998-79).	\$ (192,956,21)	60% (69 173/71) \$- 60% (40.600.01) \$	3 (62,877,801)	
15. \$\tilde{x}\$ \tilde{x}\$ \tilde	1773 000,00 375,000,00 325,000,00 150,000,00	\$ 1220,000,000 &			45 .\$. 1.759.000,00 80. \$. 125 000,00
10006 ELRAN 10397 SYLCOING. 10397 LIFERIE DENUMMOND L'TEE 10551 GESCO L'TD'PARTNERSHIR 11364 DAR R FOAM 11364 SIEGESIDUCHARME INTERNA 11421 LESMEUBIES VANTAGE 1000 11924 MATELAŞ D'OR 9053-9894 QO 112301 L. LOAVISTEŞTILES (1991) INC	SUB TOTAL CON Insured 10413 MATELAS MARTIN 10585-SUBAR 10127-TISSUS MOTION INC. 11665 BOZANIO INC.	SURTOTAL CON INSURED NO SO .  SUBTOTAL GANNA COUNT  TOTAL GROSS.	SUB TOTAL CONJUSTED SUB TOTAL CONJUSTED SUB TOTAL CONJUSTED SUB TOTAL CONJUSTED SUB TOTAL CON Insured Met 90.	SUBTOTAL GANNA GOUNT.	11682 Value Poam Industries Inc 12102 a z sponge a foam

(t)

(448 435,39) (156.952,39)

. (50°6) \$ . (16¢4) 32).

\$ (20)929 8)

iuso converted to con Exchange ratejan 31, 2012 Total con Ghoss. Total con offer

SUBTIDIAL GANNA COUNT

TOTAL OFFER

SUB TOTAL USD Insured

SUB TOTAL-USD

TOTAL OFFER.

TOTAL, GROSS

SUBTOTAL USP Insured

Rebates

NAME ,	NET	CREDIT LIMIT	CURRENT	31-60 DAYS 61-90 DAYS:	OXERIGO, DAYS	TOTAL	
USD U11771 JEFFGO FIGHES'INC U1273 FRANKUN ANGUST TRADING U1273 FRANKUN ANGUST TRADING U12134 GORM FACTORY INC U12134 GREAT NORTHERN ASSOCIAT U12137 QUINN DISTRIBUTORS U121385 KRAUS USA INC	4.88888848 34.4888848	2 030 325.00 406 055.00 7 089.00 9 862.00 59.172.00 48'310,00 73 864,00 10'008,00.	US .	(260,018,00), \$ (1647/91,35).	44	(444,808,35).	
aus Torra uso	[vi	S. 2.645.787,00 S	S .	(280-018,06) \$ (164.791,35)	5	(444,809,35)	
U11428, NRF DISTRIBUTORS INC.	Ser.	256 470,00 \$	(8:626.04)	•	čģ.	(3 926,04), \$	40.000,00
SÚB TOTAL USD Insured	(i)	.256.410,00 \$	(3.626,04) \$	S	S	(3.626,04) · \$	.40 000 00
METER WOW INC.	<b>3</b> 000	1:300 67.1,00			ų,	4.	5 150,000,00,
sup Total, usp insured Net 30°	w	1.560.673,00 \$	ys.	<i>t</i> 9, 4,	in	, (S)	\$ 150 000,00
SUB TOTAL GANNA COUNT							

35% 35% 35% 35% 35% 35% 35% 35% 35% 35%	•	38%	%50	35%	10%	•
35% 35% 35% 35% 35% (1726 95 (186 95	in	4	\$ (06,300.88)	(57.678,97) \$	Ş, -	*(185 663,27)
35% 35% 35% 35% 35% 35% 35% 35% 35% 35%		.35%	35%	35%	.35%	
35% 35% 35% 5. 5	ş	(1 269,11) · S.			il	(1.269,11)
\$ \$ (2.5,675) \$ (08,800,89)		35%	%98.	.35%	35%	·
\$ (98 006,30) \$ (57 676,97) \$ \$ (156 95	ış		П	4	i)	
\$ (57 678,977) \$ (58 606,30) \$ (57 678,977) \$	Ì	\$**				%0.
\$ (57 676,977) \$ (88 006,30) \$ (57 676,977) \$	,	4.		,	Ŋ	` \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \
\$ (25,676,577) \$ (58,006,30)	_				,	
	43	(1 269,11) \$	\$ (06,300.86)	(57 676,97) \$	· \$ ·-	. (156 952,39)
			·			

(448 435,39)

.(3 626,04) \$

(b)

Consolidated

·lusitrance:	
TOTAL	10.204, 18 10.204, 18 10.204
	я, се
OVER 30 DAYS	
AYS	மாகு முற்ற நடிக்கு நட
G1.90 D.	rri vari vari vari vari vari vari vari v
31-60 DAYS 61-90 DAYS	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
l	HOLD OF THE WAY AND
CURRENT	202. 10. 10. 10. 10. 10. 10. 10. 10. 10. 10
٠٤.	n oo
. СЯБОІТ СІМІГ	25-900.00 15:900.00 7 1000.00 850.000.00 175.000.0
MET	෧ඁ෪෧෪෧෫෧෫෧෫෧෪෪෧෦෪෪෪෧෦෪෪෪෫෧෦෪෪෫෧෦෪෪෫෦෪෧෦෪෪෫෦෪෧෧෪෪෧෦෪෪෧෦
NAME	\$45. YER REZ PRODUJTS \$46. YER REZ PRODUJTS \$26. BUJGATT DESIGN INC \$38. POSTFECH (NNOVATICINIUG) \$40. AIRWAY SURGICAL APPLIÇAN \$40. TOOZA ALPHAYICINIC, \$40. TOOZA SOLFHAYICINIC, \$40. TOOZA SOLFHAYICINIC, \$40. TOOZA SOLFHAYICINIC, \$40. TOOZA MATELAS DAUBHIN INC, \$40. TOOZA MATELAS THERAPEDIO \$40. TOOZA TOOZA TOOZA THE FILED \$40. TOOZA TOOZA TOOZA TOOZA \$40. TOOZA TOOZA TOOZA \$40. TOOZA TOOZA \$40. TOOZA TOOZA \$40. TOOZA TOOZA \$40. TOOZA \$

į-

Consolidated

1,80. 42,927,94. 5,358,46.	8 939,08	10.301.45	13 446,71	4 191,55	1	93 242,34	0.694.80	177,50	31,405,26	3, 133,52	(14,50)	23.288.27 29.284.44	42,203,54	1 1 1	300,000 40,670 B1	(32,80)	17,919,60.	20.862,95	8 00 t	6 505,76	91,376	17.887.75	4 619,79	2346,85	2,002,76	596,03	29 292,85 60 708 77.	1186.25	12/344,90	7,330,86 045.30	4548,77	23,556,06, \$	3.620,86	1 175,80	1.37.12	4 881 DE	11.881,35	9.722,17	5,0,5	45.52	75.305 B
in in in		9. <b>6</b> 5	in.	vs: 41	129	wî (	9	ن د	9 60)	cr.	43·5	A LA	29.	en k	js. 01	FIQ	碤	tr i	) UI	į,	ir.	g- 43	, in	y) (	n (3				69.	iş iç	ž iņ	Şə	eş .	a i	, S	دې		÷ uş	ń.	<b>up</b> 4	io.
f -k t-	(81;TF)		i		•	5 924,94	•	įξ (		1.	4 - 1	* †	ţ	41,0	i . , i	1.	i	4 382,52	F - \$-	· ė·	,	1. 1	- 41	1	ř. j.	. }	2 425,54	1	b	ię g	5,536,65	1 152,75	t.	1,0	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ኘ.	'R'291,06	t. r	į	Þ.	å
w w w	v) a	, vo	1295.4	n 'n	- 6/3	to (	ń f	ne ju	o vo	in.	ń (	n in		or q	_		<b>6</b> 9.	<b>44</b> .9	7 US	'n.	ios e	h 4	. 69	in i	g- 43	.15	(A.D	. 65	vi.	Dr i		W	iņ (	<i>.</i> ).,v	3- <u>(A</u>	£4	u j	s (3	· •	Ġs-t	Ú1.
6 ZO	1 1		,a	583.78		22 (89,94	٠.	e.	3		,		8	: : 1. :	8			5 89	-	. i				Š	) · j.	, 1	17,458.22	2	•	: '	3-4	B 365 47		4 pag.			20	٠.		. r.	•
404,97 \$ (400,30) \$					, (1)	30.248,93 \$	,	420.5	308,67		, i	855,60			181.29		r	781,24	6 1	362,15	699,61		8 62'61B		i i		328,32 119,28					507,75		hzo ne	in the second	- 1	<b>43</b> U			ب الان	
<b>1</b> 4						8.		-CX	147			<u> </u>						រេ -	•	N			4			:	A. 4	•	. 4	,		<b>ω</b> :		g	>						
08-1 08-1 08-1 08-1 08-1 08-1 08-1 08-1												`	۔۔	<7 · 6	, i															9- V 2- E										次· 这[	
1,80 19:314.91 5.758.75	8.716,63	10:301,45	18 946,	1-098,94		24 928,53	7 7 7 7 7 7	23,965	25,496,59	9-183	E	27 397,51	42 500,4	1.	20 573.6	(32)	17.813,60	4.908	.9 dD1,13	4 144.81	8.420,15	17.887.78		1:225,63	2,002,76	596,03	334,08	1.184,25	12.344,90	1.275,	al francis	7.529,0	3.620,89	1 7(9,80	131.15	4.681,05	1.573,48	9 722.17		1.459.0	4.400.5
13 W 14	us v						<b>?</b> ₹	g- i.																									·νī.·	n A	s es	in-	47 4	3, 49	:43	<i>i</i> ) (	ýð;
3 000,00 50 000,00 10 000,00	78 000,00	30,000,00	125 000,00	25.000,00	45 000,00	(ap obj. oa	30,000,00	60,000,00	95 000,00	10.000,00	00000	35.000,00	75 000,00	00'000'34.	.65.000.00	40 000 00	20,000,00	20 000 02	175-000:00	40 000,00	0000	00000	7.800,00	15 000,00	20000	2,000,00	25 05 05 05 05 05 05 05 05 05 05 05 05 05 05 05 05 0	5,000,00	9 000:00	500000	7.500,00	25,000,00	16,000,00	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	000000	30 000,00	20,000,90	10 000,000	10,000,00	15 000,00	
to to in	ده ده ما	, 69	69 6	<b>65</b> 6	· 67	<b>69.</b> (	ą y	٠ ٠	-'41	Ğ.	ı, c	a in	· vy	es e	j ij	S	UŠ	ios d	o va	ÇÝ:	() (	0	<b>164</b>	<b>30</b> 3	3. 63	نوب			•					•		45	in.		į,		
886	<u> </u>	्लुः	₩. <sub>2</sub>	8	ģ.	K,	ने ले	₹,@	ġ	8	F) : 6	3 명	8	B (4	3 12	2	e स्	8.5	1.12	8	ğ.,	₹ %	HO'Dalk	15.	. E	몫	<b>6</b> 5	R	ş	ė, u	2 25	60	Ģ	Y	8	.69	<u>6</u>	ě	S.	Ġ.	
								•															Post d	:	-				ОО		-		i i	3							
j1827. PBO TAC INDUSTRIES INC. 11896. ETHERIE DORMATEX INC. 11894. FUJISTAR: CANADA INC.	TIZ14-TRICA- 11760 VEBANDA JABDIN INC	11779 MANUFACTURE DE MATELAS	11811 BHOLAIN DISTRIBUTORS LTD.	11828 COMPONI FORM & DIBRE 11833 MATELAS SELECT	11844 PEL INTERNATIONAL LIEE	11846 INDUSPACING.	TRADE, VERS GOOD OF CONTINUES IN CONTINUES I	11914 EXCEL BSVC MOUSSE FOAM	11927 MEUBLES BELISLE INC.	11848 MEUBLES TOWALING	11974 STYLTEO ING.	11878 WATELSO CHITCHEUXUE 11979 BELPROMEDICALINO	11982 DIAMOND DIVERSIFIED IND. L.	12021 SEARS CANADA INC.	12104 AD MATS CORPORATION	12122 ATLANTIC FABRICS LIMITED.	12140. DOWINEAST PLASTICS LTD.	12146 MATELAS BEAU REVEINO	12158 BARRY SUPPLIES & PARRIC!	TZZBO WESCO NA.INC.	12242 SPORT SYSTEMS DANADA IN	12255 BEAULIAU KAINAUN 12256 HECHNOVATIONS	12270 COUTURE CARLYLE INC. OW	12277 DOMINION CULTING SERVICE	12290 LOGIFLEX MOB DE BUREAU	12291 PAKTEKING	12296 PAVAR FURNITURE INC. 12298 G BOMANO INC	12303 WES INDUSTRIES INC.	12308 PYRA MEDIC INC. (W)	1.32314 GROUPE SM PELLETIER 1930s Ukelven Leveku kerken	12330 CREATECH INC SOBSO744 QUE	12391, ORTHEX CANADA ENR W	rizask SHOLA INC.	12341 REVOLUTION:SANTE BISEASS	12353 EQUIPMENT SANITABE CHER	12355 ADJ ART DESIGN INTERNATIO	12357 VALENTINO ING W	12357 CONFORT EXCELFLEXING	12373 PPD FOAM SOLLTION INC.	12880 DISTRIBUTION SECURI- SPOR	123gg, condutest carret

SUB TOTAL CON

Committee Control Committee Committe

## AGED ACCOUNTS HECEIVABLE. DOMPOAM INTERNATIONAL INC:

Consolidated

The second secon

340,427,79 -\$ 350,000,00 32,24,78 -\$ 350,000,00 7 615,54 -\$ 14,000,00 55 738,78 -\$ 75,000,00 615,25 -\$ 25,000,00 215,25 -\$ 25,000,00 215,25 -\$ 25,000,00 15,666,26 -\$ 35,000,00 15,666,	758 862.09 'S' 844.000.00 463 655,36 S. 90.00,00 -221 166,00 S 75.000,00 -89 457,391 S 100,00,00 -377 (85,82) S 25,000,00	841, 634, 56 . S . '280, a00, au.	3.344:670,83 3.028.242,23	455 329,25 604,980,74	.30%  .1.988.552,222	254.229.37 71.175.61
683,58 603,58 603,189 64,97 64,97 77,842,53 64,245	36 161,22 5. 67225,28 5. 8 426,10 5 4 297,35	121'109,96 -\$	207.788.37 \$. 20% 8.125.14 \$	84 631,63 S: 84 631,63 S: 77 655,98 S	. 166.422,75 S	in g
\$6 647,92 \$ 1242,99 \$ 1703,56 \$ 410,29 \$ (2769,26) \$ 7755,01 \$ 936,14 \$ 7755,01 \$ 8702,70 \$ 5 695,18 \$ 3178,08 \$ 5 7339,94 \$ 5 3744,45 \$ 5 7339,94 \$ 5 3744,45	19 152.73 \$ 98 939.41 \$ 19 32 32 88 939.5 \$ 5.818.94.5 \$ 25 409.83 \$ 24 157.018.61 \$ 112,323.23 \$ 117,018.61 \$	189 982,90 \$ 282 E64,81 \$	649.552,115 .\$.454.083,62 .\$. 60%: 60%: 55.968.24 .\$	53.491,64: \$ 59,350,05: \$ 50,491,64: \$ 59,350,05: \$ 60%.	389 731.31 - \$ 272 450,17 - \$	\$ 934,27 \$ \$ 92,075.05 \$ 52,059 £1
303777818 303777818 3037818 30578 30578 3057	42 935,08 \$ 55 239,57 \$ 59 746,68 \$ 51 38 756,68 \$	267 671,89 1\$.	741 498,56 S	257 845,94 \$ 257 845,94 \$ 360% 360 803.19 \$	. 1159.948.00 .S	263,896,10 - \$ 27,026,48: \$
15 A 250 000 00 A 350 000 00 A	30.35 375-000.00 \$ 30.35 30.00 \$ 30.35 30.00 \$	\$ 1,550,000,00	<i>ω</i>		a	45 \$ 1750,030,00 \$ 80 \$ 125,030,00 \$
JDDOG ELFAN. TIDST SYLCO INO. TIDST SYLCO INO. TIGGE DE PRIMINGND LITEE TOSSI GESCO LTO PARTNERSHIR TITER DE AF FOAM TITER DE METENA TITER MATELAS D'OR BOSG-9834-QC TITER MATELAS D'OR BOSG-9834-QC	SUBTOTAL CDN Insured 10413 MATELAS MARTIN. 10515 SUBAR 114777 1155US MQRIQQ ING. 11955 BQZANTQ,ING.	SUB TOTAL GON Insured Net 30.	TOTAL GROSS. TOTAL OFFER SUB TOTAL ODN	ŞUBTOTAL CDM İnşured SUBTOTAL CDM,İnşured Net Ap	Sup Total Ganna.count Total Offeb	11682 VALLE FORM INDUSTRIES ING 12102,A 2 SPONGE & FOAM

 $\stackrel{(=)}{=}$ 

L.	Z.
긎	Z
Ē	<u>.</u>
	્ર≰
ń	ð.
H H H	Ξ
	`≾
ž	ã.
z	W
	Έ.
ج	1
ζ.	ÇÎ.
1	*
AGED ACC	õ
٠	$\Box$

Consolidated

, in a con-	TOTAL			12,578,70 12,428,10		765 638,11	11:41£,86 \$- AD'000,00.	: : :11.415,86 · \$ · 40-000:00	00,000,031 'S 04,831 'FA	544.154.40 \$ 150 000,00	The state of the s	1,321,211,37	267.57.234		130.454.04	0%	462 423,98	1. 321.211,37 462.423,98.
974 C 50 OHAY	UVER SULDAYS	<i>1</i> , 1	6 J			49	₩.	S	36.247,43 .15	36.247.43 \$		36.247,43 \$	710%	8	35% 12 €86.60 - 5	5	12 686,60 \$	\$ . 6\$ 28\$ 96.
91-80 DAVE: KL.DO DAVE		78 213,67 .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$ .\$	235,00	43	7 (	136 590,56 \$ 1110,95 \$:	log (à: r	9	\$ 15,18981, \$ 178981,57	160 191,76 \$.178 981,57 \$		296 882,32 \$ 180 092,62 \$	35% 35% 47 841,70 S. 388.83 \$	35% 35%	.36%. 36 057,12 '\$ 62 643,55 \$		103 908;81 \$. 63 032,38 · S	\$ 35'260 081.\$' 28'88'88Z
72000	יאשטעייט	556 732,35 20979,01			. 17.757,90 <b>s</b> 190 <sub>1</sub> 10 <b>s</b>	627.835.60 \$	.11.418.86 S	11.416.86 &	168 733(64. \$	.168 733,64 ·\$		\$. 01,889,10	35%	1 10 .	35% -59 056.77 -8*		282.795,19 - \$	\$. pr'888 109
		45.8. 2 000 325,00. \$ 60.\$. 40,605,00. \$	9 862,00	0:5 i59:172,00 0 \$ 49.810,00	73.984,00	\$ 2 645 787,00 \$	300,000,000,000	\$ 256 410,00 \$	\$ .001.1380.671.00.	\$ 1380-571,00 \$		, s	[ ]				<b> </b>	r -
HWG Z	Triange.	UST TATAL JEFEGO FIBRESINO USTRAL JEFEGO FIBRESINO USTRA GOSTA SERE MANUFACTURING C		UZIB GREAT NORTHERN ASSOCIAT UTZIBY QUINN DISTRIBUTORS	U1218B: KRAUS USA INC. U121803 LATEXINTERNATIONAL	SUB TOTALUSD	U11428, NAF DISTRIBUTORS INC.	SURTOTAL USP Insured	Uziza Wawing	SUB TOTAL USD Insured Net 90	SUBTOTAL GANNA GOUNT	TOTAL GROSS	HÔTAL OFFER SUB TOTAL USD	Sup Tot AL GSD Insured	SUB TOTAL USD losured	SUB TOTAL SANNA COUNT	TOTALOFFER	USD CONVERTED. TO CDN. EXCHANGE RATE JAn 31, 2012 TOTAL, CDN GROSS TOTAL, CDN GROSS

# AGED ACCOUNTS RECEIVABLE DOMFOAM INTERNATIONAL INC.

Dated Jan 31, 2012

		TO NE COME	DOMFOAM IN EKNATIONAL INC.		•
GROSS	-				
Amount Before Rebates (AR)	Ì	Q Q			4
SUBTOTAL	95		-		Totology A A October
SUB TOTAL Insured.	<del>)</del> . 69	1.015.452.66 S.	15.044.90	<b>≯</b> 4€	1.030.407.56
SUB TOTAL Insured Net 90	· <del>LOS</del>				1 000 000 C
SUB TOTAL GANNA COUNT	. <del>ið</del> .	<del>69</del> .		<del>!</del>	7 7 7 7 7
TOTAL GROSS	<del>os</del>	3 901 639,88 \$	1 769 646,76	<del>100</del> /	5 671 286,64
VALLE FOAM INDUSTRIES INC. A Z SPONGE & FOAM			١	<i>€9- €9</i>	254 229,37
MISC. EXCHANGE RATE				<del>⊦                                    </del>	i i
TOTAL GROSS				, <u>ę</u>	5 996 697,62
(any of the second seco		-		,	1.
Rebate Amount (Repates)	***************************************	NEC	LISD of CISH CISH	CDN	Total
SUBTOTAL	<del>93</del>	(289 820,73) \$	€ <del>Ż</del>	€ <del>9</del>	(734-530,08)
SUB TOTAL Insured	<del>rà</del>			i,	(260 186,61)
SUB TOTAL Insured Net 90	₩	(10.577,75)		66 r	(10.577,75)
SUB TOTAL GANNA COUNT	<del>69.</del>		.₩.	: <del>69</del> :	
TOTAL GROSS	LF.	(556.969,05)	(448 435,39) \$	£9;	(1,005,404,44)

Combined Amount (Combined)	1	CDN	dsn	usp to	to CDN	Total
SUB TOTAL	. <del>69</del>	1744 154,18 \$	765.638,11	€9.	<i>ĕ</i> ∌ ÷	2,509,792,29
SUB TOTAL Insured	<del>.éa</del>	758.882,09	11 418,86	64	<del>(19)</del>	770 300,95
SUB TOTAL Insured Net 90	<del>(9</del>	841.634,56 \$	544 154,40	· <del>6</del>	<del>(4)</del>	1.385 788.96
SUB-TOTAL GANNA COUNT	<del>éó</del>	<del>tý</del> ľ	, i.	<del>čo</del> .	<del>())</del>	1
JOTAL GROSS	ca .	3 344 670,83 \$	1321211,37	69-	(4)	4 665 882,20

## AGED ACCOUNTS RECEIVABLE DOMEGAM INTERNATIONAL INC.

- Land

-

-

T.	•
ш	
II.	
بنايا	٠
·	٠

Lancard Vision of the Control of the

Amount Before Rebates (AR)		Z O	dsn	USD to CBN	Total
SUB TOTAL	6 <del>9</del> .		-		1,625,014,50
SUB TOTAL Insured	ĊĄ:		5 265,72 \$	<del>(/3</del>	614 537,31
SUB TOTAL Insured Net 90	<del>ė</del> .	511 327,39	190 454,04 \$	: <b>€∕9</b> :	701 781,43
SUB TOTAL GANNA COUNT	€9-		<del>Ω</del> .	г <b>еэ</b> г	
TOTAL DEFER	69	2,321,956,87	619.376,37 \$	1	2.941 333,24

		÷.			-
Repate Amount (Rebates)				Ţ	•
		ZOO	dsn	USD to CDN	Total
SUB.TOTAL	<del>(</del> Ģ		·(155.683,27) \$		(328 798,93),
SUB TOTAL Insured	ėė.	(153'942,34)	(1.269,11)	- <del>69</del>	(155 211, 46)
SUB TOTAL Insured Net 90	Ġ	(6.346,65), \$		1	(6.346,65)
SUB TOTAL GANNA COUNT	₩.		<del>ta</del>	i i	* * # * * # * * * * * * * * * * * * * *
TÔTAL OFFER	69	(333.404,65) \$	(155,952,39) \$		(490 357,04)
	•				
Combined Amount (Combined)				<del>; -</del>	
	1	CON	äsn	USD to CDN	Total
SUBITOTAL	<b>'69</b>	1:028:242;23 \$	267.973,34 \$		1296 215,57
'SUB'TOTAL Insured	. <del>69</del> .	455 329,25. \$	\$ 09.986 E	<b>9</b>	.459.325,86
SUB TOTAL Insured Net 90	. <del>i.g</del>	504.980,74	190.454,04	1	695 434,78
SUBTOTAL GANNA COUNT	€Э	16 <del>9</del> ;	÷	<del>'⊊⊋</del> i	j.
TOTAL OFFER	c/s	1 988 552,22 . \$	462.423,98 \$	i	2.450.976,20

#### Schedule 2.9(B)

#### Purchased Inventories:

#### Section 1

#### List of Purchased Inventories as of January 11, 2012

(and for some items as of January 31, 2012)

Sèe document attached Total of one (1) page

#### Section 2

Calculation of the Purchase Price attributed to the Purchased Inventories

See document attached Total of two (2) page

454

	Schedul	e 2.9(B) - Sec	tion-1	T
	;		<del>,</del>	
		· · · · · · · · · · · · · · · · · · ·		
<b></b>				
	la seltogities of			
<u> </u>				
<u>`                                    </u>		SEC.	January 11, 2012	
RAW MATERI	IAL	, i		1
SCRAP		1	155 026	1
FILM		2	52 547	+
UNDERLAY TUBI	ING.	3	36 551	+
NON CHEMICALS		.5	26 656	┪
BONDED CHEMI	<del></del>	6,	7354	+
CHEMICALS (MA		7		100
-,	JUNO)	<u>-</u>	810,266	{1
OTHERS		<u>. 8</u>	86756	1
SMALL, CHEMICA		<del>8</del> .	505 644	(1
POLYESTER FIB	KE .	1 1 1 1 1 1 1 1 1	0	
	<del></del>	sub total:	1 680 800	_
			1 680 800.	<u> </u>
WORK IN PRO			- Andread and a second	
BONDED BLOCK		11	10 372	
ĄÖNDED BŪNS I		12	Q	
	/ ETHER BLOCKS:	13	1 010 428	
ROUND ESTER E	ยบทรั	14	25 374	
ROUND, ETHER (	INDERLAY BUNS	15	0	
POLY ETHER BU	NS:	16	143 382	
		sub total:	1 189 556	
			1 189 558	
TOTAL RIM &	W. IN P.	total:	2 870 356	
			2 870 356	1
FINISHED GO	ODS:			1
BONDED UNDER	LAY	17	0	_
SHIPPING INVEN		18,23	113 574	(2)
I	1	sub total.	113 574	+-
		2,32 (3,12.1)	113'574	+
TOTAL INVEN	ITORY	TOTAL:	2 983 930	-
(A) WE WAE!	1 Oica	(O-17AL)	2 983 930	+
	<del> </del>		7 300.370	- <del> -</del> -
1	· ·	·	- 16-	+
SECTION 1	SUPPLIES	1 1	33.407	-
SECTION 2	FIBER	2	3 763	
SECTION:3:	BLOCS:	3.	13:286	<del> </del>
SECTION 4	BUN ENDS	4	1 827	_
SECTION 5	COMMANDES/stoc	5	19798	_ _
<u>, l,</u>				1.
TOTAL INVEN	ITORY	TOTAL:	72 081	
	<u> </u>			_ _
OVERALL I	/ALUE		3 056 011	
				T
(1) As per the	document added	in the Data R	oom on Feb. 14, 2012,	1
			on page 107 of the	
			the Data Room.	1

RAW MATERIAL SGRAP .1. FILM .2. UNDERLAYTUBING .3. NON CHEMICALS .5. GONDED CHEMICAL CHEMICALS (MAJORS) .7 OTHERS .8 SMALL CHEMICALS .9 POLYESTER PIBRÉ SUB TOTAL  WORK'IN PROCESS.	155 026 .52 547 36 551 26 656 7 354 810 266 36 756 07 505 604 0 1 660 800 1 680 800	204168   Şee gotes 1 & 2 belov
RAW MATERIAL  SCRAP  FILM  2  UNDERLAYTUBING  NON CHEMICALS  GONDED CHEMICAL  CHEMICALS (MAJORS)  OTHERS  SMALL CHEMICALS  POLYESTER FIERE  SUB-lotal;	155 026 .52 547 36 551 26 656 10% 7 354 810 266 30% 86 756 0% 505 644 50%	109 190 727;930 lbs @: .15/lb. 2 666 See note 2 below 201 168 See notes 1 & 2 below
SCRAP .1: FILM .2: UNDERLAYTUBING .3: NON CHEMICALS .5: GONDED CHEMICAL6: CHEMICALS (INAIORS) .7 OTHERS .8: SMALL CHEMICALS .9: POLYESTER FIBRE SUB-lotal:	.52 \$47 36 551 26 656 10% 7 3\$4 810 266 30% 86 756 0% 505 644 50%	2666 See note 2 below 204168 See notes 1 & 2 belov
FILM 2 UNDERLAYTUBING 3 NON CHEMICALS 5 RONDED CHEMICAL 5: CHEMICALS (MAJORS) 7 OTHERS 8 SMALL CHEMICALS 9: POLYESTER FIBRE SUB-lotal:	.52 \$47 36 551 26 656 10% 7 3\$4 810 266 30% 86 756 0% 505 644 50%	2666 See note 2 below 204168 See notes 1 & 2 belov
UNDERLAY TUBING 3 NON CHEMICÁLS 5 RONDED CHEMICAL 5: CHEMICALS (MAJORS) 7 OTHERS 8 SMALL CHEMICÁLS 9 POLYESTER PIBRÉ SUD. lotal:	36 551 26 656 10% 7 354 810 266 30% 86 756 0% 505 644 50%	204168   See notes 1 & 2 belov
NON CHEMICAL 5 SOUDED CHEMICAL 15: CHEMICAL 17 OTHERS 8 SMALL CHEMICALS 9: POLYESTER FIBRÉ SUD. lotal:	26 656 10% 7 354 810 266 36% 86 756 0% 505 644 50%	204168   See notes 1 & 2 belov
CHEMICAUS (MAJORS) .7 OTHERS 8 SMALL CHEMICAUS 9 POLYESTER FIBRÉ SUB-lotal;	7 354 810 266 367 86 756 67 505 644 567 0 1 680 800	204168   See notes 1 & 2 belov
CHEMICAUS (MAJORS) .7 OTHERS 8 SMALL CHEMICAUS 9 POLYESTER PIBRÉ SUB-lotal;	810 266 30% 86 756 0% 505 644 50% 0 1 680 800	
OTHERS 8 SMALL CHEMICALS 9 POLYESTER FIBRÉ SUD. total;	86 756 0% 505.644 50% 0 1 680 800	
SMALL CHEMICÁLS 9: POLYESTER FIERÉ SUD. total;	505.644 509 0 1.680.800	11-22-20-20-5-19-20-1
POLYESTER FIERÉ Sub lotal;	0 1 680 800	252 822 See note 2 below
sub total:	1 680 800	(CEC138) 223-337
•		TOTAL SECTION
WORK IN PROCESS	1 680 800	
WORK IN PROCESS		
BONDED BLOCKS square: 11	10,372 <sup>-</sup>	
SONDED BUNS round 12	0.	
SQUAREBLOCK/ETHER BLOCKS. 13	1 010 428 309	303128
AOUND ESTER BUNS 14	2\$974 309	
สอบหม่ ฮาหะล ปะเมื่อสันญ์ อน่หรั 15	0.	
POLY ETHER BUNS 16	143 382 30%	7243015
sub total:	1.189 556	
· · · · · · · · · · · · · · · · · · ·	1 189 556	
TOTAL R.M &.W. IN P. lotal:	Z 870 356	
	2 870 356	
FINISHED GOODS	. "	
BONDEO UNDERLAY 17	0 30%	
SHIPPING INVENTORY (GOODS) 18,23	113:574	Sée note 3 below
sub total:	113 574	
- April 10 million	113 574	
TOTAL INVENTORY TOTALS	2 983 930	
1017611311311	2 983 930	
•		
SECTION 1 SUPPLIES 1	33 107	
	3 763 13 286 309	23277000
	1 827	
SECTION 4 BUN ENDS 4 SECTION 5 COMMANDES/stool 5	19798	
genindus chiwwanneassoci p "	18 790	
TOTAL INVENTORY TOTAL:	70.004	
TOTAL INVENTORY TOTAL:	72 081	
4 1/4		
Add purchase of uncounted Items with no specific	vice anoké	20 000
OVERALL VALUE		1.数据处别的图
OVERALL VALUE	3 056 011	946 586

#### \*\* Désignates counted inventory

	of the latex based on the cost presented on entory <sup>a</sup> posted on February 14, 2012 in the	page 107 of the
	temshave been evaluated as of January 3	ı, 2012. ————————
Note 2		
	Value at 30%	204 168
Remaining inventory		680 561
less TOI in inv (value \$0.00)	93313 lbs @ 1.39/lb	129705
Total Chemical in inventory-		810 266



Saut sur indication contraire, tout le matériel compté est bon et disponible immédiatement.:

21 (LAST SHEET)

#### LATEX INVENTORY

TYPE	QUANTITY	COST	TOTAL
Single Soft (TXL 65)		\$ 115.30	\$ -
Single Medium (TXL 75)		\$: 135.30	\$
Single Firm (TXL 85)		\$ 204,95	\$* -
Single XTRA FIRM (TXL 95	)	\$: 185:00	\$ -
Twin Soft (T 65)		\$ 115,30	\$
Twin Medium (T 75)		\$ 135,30	`\$ -
Twin Firm (T 85)		\$ 204:95	\$ -
Twin XTRA FIRM (T 95)		\$ 185,00	\$ -
Double Soft (F 65)		\$ 216,21	\$ -
Double Medium (F 75)		\$ 230.40	\$ .
Double Firm (F85)		\$ 236.40	\$ -
Double XTRA FIRM (F 95)		\$ 256.00	\$ -
Queen Soft (Q 65)		\$ 185.85	\$ -
Queen Medium (Q-75)		\$ 198.45	\$ -
Queen Firm (Q 85)	, .	\$ 315.31	\$ -
Queen XTRA FIRM (Q95)		\$ 352,40	\$ -
King Soft (K 65)		\$ 235.20	°\$`
King Medium (K75)		\$ 250.95	\$ 7
King Firm (K 85)		\$ 284.55	\$
King XTRA Firm (K95)		\$ 330.75	\$ -
CK SOFT (CK:65)		\$ 325,00	\$ -
CK MED (CK 75)		\$ 325.00	·\$
CK FIRM (CK 85)		\$ 360.00	\$
CK XTRA FIRM (CK 95)		\$ 390,00	.\$ _ ÷
5 ZONE		\$ 39Q.00	\$ -
7 ZONE -		\$ 400.00°	\$
PILLOW-		\$ 20.00	\$

#### Schedule 2.9(E)

#### Excess rebates to enstomers

See document attached Total of two (2) pages 

GROSS REBATES AVAILABLE CDN Rebates Available in (CDN) US. Rebates Available in (CDN)	11. ut		448435,39	. α. μ.	591.556,45 -448.435,39 1 039.991,84
GROSS REBATES APPLIED SUB TOTAL SUB TOTAL Insured SUB TOTAL Insured Net 90 SUB TOTAL GANNA COUNT	w the to	(289 820,73) \$ (256 570,73) \$ (10 577,75)	USD (444 809,35) \$ (3 626,04) \$	USD to GDN.	Total (734.630,08) (260.136,61) (10.577,75)
TOTAL GROSS	Z)	\$ (50'696 959)	. (448 435,39) \$	\$	(1.005 404,44)

les	
S	
•	

EXCESS UNAPPLIED REBATES

Domfoam International Rebates As of Jan 27 2012.

	REBATE AMOUNT
}	OUTSTANDING OFFSET
	AGAINST ACCOUNTS
CUST#	RECEIVABLE
10072	\$127,739.99
10120	\$1,769.76
10192	. \$591.27
10485	\$22,390,75
10551	\$18,833.96
11207	\$2,249.62
U11429	\$3,626.04
12000	\$7,267.85
1,2253	\$3,606.38
00202	\$14,808,00
10005	\$171,143.58
1020	\$24,868.08
10305	-\$1,62,69
10358	\$1,554.44
10379	\$4,939,56
10384	\$7,139.28
10510	\$4,601.00
10545	\$6;193.47
10848	\$514,01
11155	\$4,186.45
11164	\$66,036.80
11209	\$16,269,76
11268	\$5,184.22
11289	\$6,523.36
11457	\$13,380.06
11666	\$10,577.75
11771	\$444,809.35
11844	\$5,922,65
11924	\$17,760.30
11927	\$11,128.04
12158	\$8,822.25
12282	\$5,716.50
· · · · · · · · · · · · · · · · · · ·	
	\$1,039,991.84

#### Schedule 4.1(c)

#### Confidentiality Agreement

See document attached nine (9) pages

#### CONFIDENTIAL

Domfoam International Inc. 8785 Langelier Blvd St Leonard; QC H1P 2C9

#### Altention John Howard

Res

The second second of the second secon

Domfoam International Inc., Valle Foam Industries (1995) Inc. and A-Z Sponge & Foam Products Ltd. (collectively, the "Corporation") — Proposed Sale of Assets ("Proposed Transactions")

We have requested that the Corporation provide us (the "Interested Party") with the Confidential Information (defined below) to assist in determining for ourselves the desirability and feasibility of proceeding with the Proposed Transactions pursuant and subject to the terms and conditions set by the Corporation (the "Permitted Purpose") and contained hearin.

It is acknowledged that but for delivery of this letter the Corporation would not provide the interested Party such Confidential Information.

Upon acceptance of this letter by the Corporation and subject to the provisions thereof, the Corporation shall provide or cause to be provided to the Interested Party such Confidential Information as the Corporation in the exercise of its unfertiered discretion may determine.

"Confidential Information" means all financial statements and information and all data, compilations, analysis, documentation, and other information and every part thereof, whether in electronic form or otherwise, which now or hereafter comes into the possession or under the control of the Interested Party or its Representatives (defined below) including, without limitation, that relating to the Corporation's business organization, business operations and systems, Know-how, business plans, Research Information, Customer Information, suppliers, financial structure, Financial Information, Supplied Information, Marking Information, assets, employees, employees relationships, Software and other information of any kind relating to any of the foregoing and includes all data; compilations, analysis and other documents and information prepared by or on behalf of life Inforested Party or by any of the Representatives based, in whole of in part, on any of the foregoing, but expressly excluding any reports or analysis of the Corporation conducted by any professionals relatined by or with respect to the Corporation, including, without limitation, Deloitte LLP, in connection with the current financial state of the Corporation, any restructuring scenarios available to the Corporation, and any analysis of the value of the Corporation or its assets.

For the purposes of this letter:

(a) "Corporation" shall include and extend to each subsidiary of the Corporation, each partnership controlled by the Corporation or their

respective subsidiaries, each of whom shall be entitled to directly enforce the provisions of this letter against the interested Party not with standing that this letter is not directly addressed to or required to be accepted by any one or more of such subsidiaries and partnerships:

- (b) "Customer Information" includes any information parining to the Corporation's customers, prospective customers, customer base and markets, prospective customer are prospective customer are prospective customer and addresses and the names of customer or prospective customers with whom the Corporation is in contact in its business, or proposes to contact customer or prospective customer requirements and the Corporation's contracts with its customers or prospective customers, including details as to priving, supply of services, and royalty and/or commission rates;
- (c) "Pinancial Information" includes any and all information pertaining to the Corporation's actual or anticipated sales, income, projections, profit, profitability, pricing, salaries and wages;
- (d) "Know-how" includes the accumulated skills, experience, knowledge, patents, intellectual property, information, data, patterns, designs, engineering specifications, schematics and other information with respect to the design, production and provision of the products and/or services offered by the Corporation's business or proposed business and of the Software related thereto;
- (é) "Marketing Information" includes information including but not limited to the Corporation's marketing programs, plans, stratégies and proposed future: products, services, adventising and promotions.
- (f) "person" shall be broadly interpreted to include, without limitation, any individual, corporation, limited liability corporation, company, group, partnership, limited liability partnership or other entity.
- (g) "Representatives" means, collectively, each of the directors, officers, employees, agents and other representatives of the Interested Party of the Corporation (as the case may be) including, without limitation, their respective lawyers, accountants, consultants and financial advisors;

- Suppilmet uid

- (h) "Research Information" includes information pertaining to any research, development, investigation, study analysis, experiment or lest carried on or proposed to be carried on by the Corporation;
- (i) "Software" includes all methods, programs; techniques and other information and materials relating to the handling or treatment of data by computers and all other necessary information utilized in order to provide the services offered by the Corporation's business of proposed business and all documentation thereto, and includes all computer programs, including

ITIME, TAVA, UNIX, LINUX, GNU, C++; and other internet oriented computer programs, systems software and application programs, and all related manuals, documentation and materials relating to the systems software and application programs;

(j) "Supplied Information" includes all business information, computer software and technology which is proprietary to any other person doing business with the Corporation and which is made available to the Corporation under conditions of confidentiality.

At the request of the Corporation and for other good and valuable consideration (the receipt and sufficiency of which is hereby acknowledged) the Interested Party covenants and agrees with the Corporation as follows:

- It Deemed Confidential Property: As between the Corporation and the Interested Party, the Confidential Information, and every part thereof, is and shall at all times be deemed to be the confidential property and constitute valuable trade secrets of the Corporation, whether or not, but for this paragraph, the same would otherwise be considered confidential and/or trade secrets and whether or not it is marked or otherwise expressed on its face to be confidential.
- 2. Ownershipt. The Confidential Information is and shall be deemed to be owned solely by the Corporation and the right to maintain confidential the Confidential Information constitutes an exclusive proprietary right of the Corporation which it is entitled to protect. The Interested Party does not and shall be deemed not to have any right to or proprietary interest in the Confidential Information. All of the Confidential Information which is disclosed to or otherwise comes into the possession or under the control of the Interested Party shall, subject to paragraph 5 below, he received and held by the Interested Party in frust solely for the Corporation notwithstanding the Interested Party's right to use it for the Permitted Purpose.
- 3). Permitted Use: The Confidential Information shall only be used directly by the Interested Party, and its Representatives for the Permitted Purpose. Except as otherwise provided for herein, the Confidential Information will be kept strictly confidential by the Interested Party, whether or not such Confidential Information is marked or otherwise expressed on its face to be confidential. Other than as permitted herein, the Interested Party will not, directly or indirectly, without the prior written consent of the Corporation (which consent may be arbitrarily withheld) disclose, publish, reproduce or otherwise disseminate or communicate the Confidential Information, nor will it permit, cause or acquiesce in such disclosure or other dissemination or communication of the Confidential Information, by the Representatives of the Inforested Party or by any other person, all in any manner whatsoever, in whole or in parts.

The Interested Party agrees to restrict the disclosure, dissemination or other communication of the Confidential Information solely to such of the Interested Party's Representatives who need to know the Confidential Information for the Permitted Purpose and the Interested Party shall inform its Representatives of the ownership and strict, confidential nature of the Confidential Information.

20 pprimer of

Supprime: who sie

Supprime: ed

Supprime: in writing by the interested Party

Supprimé; and who agree in willing will the imerested Party to be bound by the terms of this letter

- Disclosure List & Responsibility, for Representatives: The Interested Party shall maintain a current listing of each of its Representatives to whom any of the Confidential Information has been disclosed or otherwise disseminated or communicated and on each and every request of the Corporation; the Interested Party shall without delay deliver to the Corporation a complete and accurate copy of such list. The Interested Party shall similarly deliver on request, the following information:
  - (a) name of the any law limits, accounting firms and consulting firms (the "Firms") involved with the Interested Party as its Representatives:
  - (b) pame of the person in change for each Pinns
  - (c) business address and telephone number to reach the person in charge for anch

Supprimerofady or all persons set out on the said list.

The Interested Party shall be responsible and liable to the Corporation for any and all acts or omissions of each of its Representatives which if done or omitted directly by the Interested Party would be a breach of the provisions of this letter.

The Interested Party shall safeguard the Confidential Information from disclosing or other dissemination or communication to any person other than as expressly permitted herein.

- Exceptions to Prohibited User. The obligations and limitations imposed on the Interested Party and its Representatives hereunder shall not apply to any Confidential Information:
  - (a) which is or becomes generally available to the public other than as a result of its disclosure or other dissemination or communication by the interested Party or its Representatives; or
  - (b) which becomes available to the Interested Party or its Representatives on a non-confidential basis from a source other than the Corporation or its Representatives, provided that such source is not then bound by a confidentiality agreement with the Corporation or otherwise prohibited from transmitting the Confidential Information to the Interested Party or its Representatives by contractual, legal or fiduciary obligation or
  - (c) which is already in possession of the interested Party: provided that such Confidential Information is not subject to another confidentiality agreement of non-disalesure agreement with the Corporation.

In addition, the obligations and restrictions imposed on the interested Party and its Representatives become shall not extend to prohibit the interested Party from discussing the Proposed Transactions with and disclosing any of the Confidential Information to:

> (i) any other person who has signed a confidentiality agreement with the Corporation in substantially life form of the agreement constituted by this letter regarding any of the Proposed Transactions; and

Supprimë: r

(ii) any governmental or regulatory body in Canada having jurisdiction over the operations of the Corporation or any part thereof or over the Interested Party. in connection with its participation in any of the Proposed Transactions, provided that the disclosure, dissemination or other communication of any of the Confidential Information to such governmental or regulatory body is made on a confidential basis and such confidential basis is either provided for by applicable law or recognized in writing by the relevant body:

Notwillistanding the foregoing provisions of this paragraph 5, in the event that the loterested Party or its Representatives are required by law or governmental action to disclose any of the Confidential Information, the Interested Party and/or its Representatives will on a best effort lawis, deliver to the Corporation prompt prior notification of such requirement(s) so that the Corporation may seek an appropriate protective order and/or waive compliance by the Interested Party or its Representatives with the provisions of this letter. If, failing the obtaining of a protective order or the delivery of such a waiver, the interested Party is, in the opinion of its legal counsel, compelled to disclose, disseminate, or otherwise communicate the Confidential Information, the Interested Party may disclose, disseminate, or otherwise communicate the Confidential Information, but only to the extent so compelled, without liability hereunder.

- 6. Return of Informations. The interested Party shall, on receipt of the written request of the Corporation, promptly deliver to the Corporation all of the Confidential Information which is in recorded form (which such form he in writing or in electropically retrievable form or electropically stored data of otherwise) and each of the Interested Party and its Representatives shall not retain any copies thereof.
- 7. No Liability for Information: At the time(s) any of the Confidential Information is delivered to the Interested Party, the Corporation will attempt to include in such Confidential Information such materials which the Corporation considers to be reliable and relevant for the Permitted Purpose; however, the Corporation shall not have or incur any, and the Interested Party hereby waives any right to claim any, liability to the Interested Party or its Representatives in connection with or arising from the Confidential Information and the use thereof by the Interested Party or its Representatives.

The Interested Party acknowledges and agrees for itself and its Representatives that neither the Corporation nor any of its Representatives in providing the Confidential Information has or will be taken to have made either expressly or impliedly any representations or warranties as to the accuracy or completeness of the Confidential Information and the Interested Party confirms that it is not relying on any representations of warranties of the Corporation or its Representatives in making use of the Confidential Information.

8. Indemnity: The Interested Party hereby unconditionally indemnifies and shall forever save framess the Corporation from and against any and all losses, damages, liabilities (whether actual, contingent of otherwise), proceedings, costs and expenses including, Without limitation, legal fees (on a scale as between a solicitor and his own client), which the Corporation or its Representatives may suffer, incur or sustain its

Supprimé: written-

connection with or as a result of a breach, by the Interested Party or any of its Representatives of any of the provisions contained in this letter.

All of the provisions contained herein are reasonable in the circumstances and valid and the interested Party hereby waives all defences, equities and rights to and against the strict enforcement thereof by the Corporation and its Representatives.

9. Continuity of Agreement: The prohibitions, limitations and obligations of the Interested Party herein contained shall continue in full force and effect notwithstanding the Interested Party censing to proceed with any of the Proposed Transactions and will survive any termination or cancellation of its involvement, if any, in the Proposed Transactions.

10. Use of Information Post Involvement: The Interested Party shall not, at any time after it ceases to proceed with or be involved in the Proposed Transactions and except as otherwise expressly provided for herein; (i) use any of the Confidential Information in furtherance of the business of the Interested Party or the business of any other parson; (ii) interfere in any way with any confractual or other business relationship of the Corporation; (iii) disclose; disseminate or otherwise communicate the Confidential information or any part thereof to any person or utilize the Confidential Information in any way which is adverse to the Corporation; or (iv) directly or indirectly solicit for employment any person who is now employed by the Corporation, unless the Corporation otherwise agrees in writing prior to such solicitation for employment.

11. Right to Enjoin: It is acknowledged that the provisions of this letter are essential for the protection of the Corporation and, in the event of the non-performance of any provisions hereof in strict accordance with their specific terms or any breach of any provisions hereof in strict accordance with their specific terms or any breach of any provisions of this letter, the same would cause immediate and irreparable harm and damage to the Corporation for which monetary relief would be inadequate or impossible to ascertain. The Interested Party agrees that upon any breach or threatened breach of the provisions hereof, the Corporation shall be entitled to obtain from any court of competent jurisdiction, interim and permanent injunctive relief or other appropriate form of equitable relief for effectively enforce the provisions hereof, and an accounting of all profits and benefits which may have been or which may be derived both directly or indirectly, as a direct or indirect result of such breach by the Interested Party or any of its Representatives who have committed or who have threatened to commit such breach, which rights and remedies shall be cumulative and in addition to any other rights or remedies to which the Corporation may be entitled.

No failure or delay by the Corporation in exercising or taking steps to enforce any of its rights or entitlements under the provisions of this letter shall operate as a waiver thereof, unless such waiver is in writing and signed by the Corporation, nor shall any single or partial exercise thereof proclude any other or future exercise of any other right or entitlement of the Corporation hereunder.

12. Notices: Any notice desired, contemplated or required by this letter to be given shall be delivered by hand or facsimile to:

Supprimé: rights of sel off,

(u).	the Corporation of:	•	•
[	,8785 Langelier Blyd.		(Suppriméi.
	St Leonard; QC_HIP	2C9	- Supprimé: .
	Attentions	The second secon	Supprimé: Y
	Altention:	John Howard	à Cara
:	Telephone No. Facsimile No.	514-325-8120	Supprimé:
ŀ	racsuniie No.	\$14-325-6477	Supprimén
	·· :		Supprime:
· (Þ)	the Interested Party at	<b>!</b>	Supprint:
• !	0600 DI- D11		
	8600 Decaric Blyd Suite 200	عام معادر مسام المركز الأمام المحام الم	Supprime:
	Town of Mount Royal	LYNO BLID SNO.	•
	A nim of infiduit Hole	12 Oct 1341, 5142	Superimé:
_	Attention:	Frank I. Gattinger	Supprimé: 9
·	Telephone No.	\$14-341-8600	Supprimé: Questes.
•	Facsimilé No.	Sin-340-1389	Supprimet
,		The state of the s	Supprimés
	With copy to:	:	[Supprimét
	Lamarie Perro	u Lambert Vincent	
	480 Saint-Laur	ent Blvd.	:
	Suite 200		
•	Montreal, QC	112Y 3Y7	
	Attention:	Incures Vincent.	
	Facsimile No:		
			••
13. Applicabl	e Law: The provision	s of this letter shall be governed by and construed and	•
ënfore Canad	ed exclusively in accor a applicable therein.	dance with the laws of Quebec and the federal laws of	Supprimé: Ontario
	•	•	
the letter by s	igning and returning the	ence its agreement and acceptance of the provisions of se enclosed copy thereof which has been signed by the ter will constitute the binding agreement between the with respect to the subject matter thereof.	· · · · · · · · · · · · · · · · · · ·
trireferied t at	is and the corbotation a	and teablest to me applicer things. moreout	
•		Yours very truly,	
• •		S.P. Holdings Canada Inc. or Nominee	
1	ī	Peri	
		TI POMERNATE	
* =		July R. C. V. L. L. L. L. Com. Com. Com.	~ (Company)
			Supprimé: {
			>

TO:

S.P. Holdings Canada Inc. or Nomince

Accepted and agreed to with effect as of the date set out above.

DOMFOAM INTERNATIONAL INC.

Per

Per; JOHN & HOWARD

[ MDW-738

Page 7 : [1] Supprimé		* \$ 2014	-11-25 08:42:00
	Pēr:.		
,	Lei''		*

#### Schedule 5.1(0)

List of assets to be acquired by the Vendor

From Valle Foam Industries (1995) Inc.

Prior to the Closing Date

See document attached one (I) page

#### Equipment to be Acquired From Valle Foam

Bonded foam peeler and boring machine:
Foam shredding line c/w blower
2 shredded foam storage bins c/w fluffer arms, gearboxes and drives (2 units @ \$5,000 each).
Baumer contour saw c/w infeed and outleed conveyors.
Wynter contour saw c/w infeed and outleed conveyors
Convoluter - Edge Saw 88" capacity
Carrousel (of Chinese manufacture)
Automatic wrapping machine (built at Domfoam in 1996).
Automatic verfical saw
Zwick-foam tester

Dismanile, transport and reassemble at Domfoam.

Additional Items.
Foam shredding line dw blower floor scale
Laminator unwind set-up
Laminator heating system.
Underlay sealer
Web allgr unit
Air compressor
H.P. boiler 50 HP.

Dismanile, transport and reassemble at Domfoam

•			

### ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE MR.	) FRIDAY, THE 16 <sup>th</sup> DAY
	)
JUSTICE BROWN	) OF MARCH, 2012

ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

ARRANGEMENT OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES (1995) INC.,

DOMFOLM INTERNATIONAL INC., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### SALE APPROVAL AND VESTING ORDER (re Domfoam International Inc.)

THIS MOTION, made by the Applicants for an Order approving the sale transaction (the "Transaction") contemplated by an asset purchase agreement (the "Sale Agreement") between Domfoam International Inc. (the "Debtor") and 4037057 Canada Inc. (the "Purchaser") dated March 8, 2012 and Confidential Supplement to the Third Report of Deloitte & Touche Inc., in its capacity as Courtappointed monitor (the "Monitor"), dated March 13, 2012 (the "Report"), and vesting in the Purchaser the Debtor's right, title and interest in and to the assets described in the Sale Agreement (the "Purchased Assets"), was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the affidavit of Tony Vallecoccia sworn March 13, 2012, and the exhibits thereto, the Report dated March 13, 2012, and the Confidential Supplement attached thereto, and on hearing the submissions of counsel for the Applicants, counsel for the Monitor, and no one appearing for any other person on the service list, although properly served as appears from the affidavit of Victoria Stewart sworn March 13, 2012, filed,

- 1. THIS COURT ORDERS AND DECLARES that the Transaction is hereby approved, and the execution of the Sale Agreement by the Debtor is hereby authorized and approved, with such minor amendments as the Debtor may deem necessary. The Debtor and the Monitor are hereby authorized and directed to take such additional steps and execute such additional documents as may be necessary or desirable for the completion of the Transaction and for the conveyance of the Purchased Assets to the Purchaser.
- 2. THIS COURT ORDERS AND DECLARES that upon the delivery of a Monitor's certificate to the Purchaser substantially in the form attached as Schedule A hereto (the "Monitor's Certificate"), all of the Debtor's right, title and interest in and to the Purchased Assets described in the Sale Agreement shall vest absolutely in the Purchaser, or such related entity as it may direct, free and clear of and from any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, trusts or deemed trusts (whether contractual, statutory, or otherwise), liens, executions, levies, charges, or other financial or monetary claims (direct, indirect, absolute or contingent), assignments, judgments, executions, writs of seizure and sale, options, adverse claims, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured or otherwise (collectively, the "Claims") including, without limiting the generality of the foregoing: (i) any encumbrances or charges created by the Order of the

Honourable Justice Newbould dated January 12, 2012; (ii) all charges, security interests or claims evidenced by registrations pursuant to the *Personal Property Security Act* (Ontario) or any other personal property registry system; and (iii) the Claims (all of which are collectively referred to as the "Encumbrances", which term shall not include the permitted encumbrances, easements and restrictive covenants listed in the Sale Agreement) and, for greater certainty, this Court orders that all of the Encumbrances affecting or relating to the Purchased Assets are hereby expunged and discharged as against the Purchased Assets.

- 3. THIS COURT ORDERS that for the purposes of determining the nature and priority of Claims, the net proceeds from the sale of the Purchased Assets shall stand in the place and stead of the Purchased Assets, and that from and after the delivery of the Monitor's Certificate all Encumbrances shall attach to the net proceeds from the sale of the Purchased Assets with the same priority as they had with respect to the Purchased Assets immediately prior to the sale, as if the Purchased Assets had not been sold and remained in the possession or control of the person having that possession or control immediately prior to the sale.
- 4. **THIS COURT ORDERS AND DIRECTS** the Monitor to file with the Court a copy of the Monitor's Certificate, forthwith after delivery thereof.
- 5. THIS COURT ORDERS that, pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act, the Debtor and the Monitor are authorized and permitted to disclose and transfer to the Purchaser all human resources and payroll information in the Company's records pertaining to the Debtor's past and current employees, including personal information of those employees listed in the Sale Agreement. The Purchaser shall maintain and protect the privacy of such information and shall be entitled to use the personal

information provided to it in a manner which is in all material respects identical to the prior use of such information by the Debtor.

#### 6. THIS COURT ORDERS that, notwithstanding:

- (a) the pendency of these proceedings;
- (b) any applications for a bankruptcy order now or hereafter issued pursuant to the *Bankruptcy and Insolvency Act* (Canada) in respect of the Debtor and any bankruptcy order issued pursuant to any such applications; and
- (c) any assignment in bankruptcy made in respect of the Debtor;

the vesting of the Purchased Assets in the Purchaser, as well as the execution of all agreements pursuant to this Order shall be binding on any trustee in bankruptcy that may be appointed in respect of the Debtor and shall not be void or voidable by creditors of the Debtor, nor shall it constitute nor be deemed to be a settlement, fraudulent preference, assignment, fraudulent conveyance, transfer at undervalue, or other reviewable transaction under the *Bankruptcy and Insolvency Act* (Canada) or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

7. **THIS COURT ORDERS AND DECLARES** that the Transaction is exempt from the application of the *Bulk Sales Act* (Ontario).

- 8. THIS COURT ORDERS that the Debtor is authorized and directed to change its corporation name from Domfoam International Inc. in order to complete the Transaction, if necessary.
- 9. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States to give effect to this Order and to assist the Debtor and the Monitor and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Debtor and the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order or to assist the Debtor and the Monitor and its agents in carrying out the terms of this Order.

- Mr. - 7.

#1863901 | 4079509

ENTERED AT / INSCRIT À TORONTO ON / BOOK NO: LE / DANS LE REGISTRE NO.:

MAR 1 6 2012

#### Schedule A - Form of Monitor's Certificate

Court File No. CV-12-9545-00CL

### ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES (1995) INC., DOMFOAM INTERNATIONAL INC., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### MONITOR'S CERTIFICATE

#### RECITALS

- A. Pursuant to an Order of the Honourable Justice Newbould of the Ontario Superior Court of Justice (the "Court") dated January 12, 2012, Deloitte & Touche Inc. (the "Monitor") was appointed to monitor the business and financial affairs of Domfoam International Inc. (the "Debtor").
- B. Pursuant to an Order of the Court dated March 16, 2012, the Court approved the asset purchase agreement made as of March 8, 2012 (the "Sale Agreement") between the Debtor and 4037057 Canada Inc. (the "Purchaser") and provided for the vesting in the Purchaser, or such related entity as it may direct, of the Debtor's right, title and interest in and to the Purchased Assets, which vesting is to be effective with respect to the Purchased Assets upon the delivery by the Debtor or Monitor to the Purchaser of a certificate confirming (i) the payment by the

Purchaser of the Purchase Price for the Purchased Assets, save and except for the Holdback; (ii) that the conditions to Closing as set out in the Sale Agreement have been satisfied or waived by the Debtor and the Purchaser; and (iii) save and except for the Holdback and the settlement of the Purchase Price as per Section 2.10 of the Sale Agreement, the Transaction has been completed to the satisfaction of the Debtor and the Monitor.

C. Unless otherwise indicated herein, terms with initial capitals have the meanings set out in the Sale Agreement.

#### THE MONITOR CERTIFIES the following:

- 1. The Purchaser has paid and the Monitor has received the Purchase Price for the Purchased Assets payable on the Closing Date pursuant to the Sale Agreement, save and except for the Holdback;
- 2. The conditions to Closing of the Sale Agreement have been satisfied or waived by the Debtor and the Purchaser; and
- 3. Save and except for the Holdback and the settlement of the Purchase Price as per Section 2.10 of the Sale Agreement, the Transaction has been completed to the satisfaction of the Monitor.
- 4. This Certificate was delivered by the Monitor at [TIME] on \_\_\_\_\_ 2012.

Deloitte & Touche Inc., in its capacity as Court-appointed monitor of the Applicants, and not in its personal capacity

Per:			
	Name:		 
	Title:	4	

#1863901 | 4079509

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF VALLE FOAM INDUSTRIES IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED (1995) INC., DOMFOAM INTERNATIONAL INC., and A-Z SPONGE & FOAM PRODUCTS LTD

Court File No. CV-12-9545-00CL

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

Proceeding commenced at TORONTO

# APPROVAL & VESTING ORDER (re Domfoam International Inc.)

MINDEN GROSS LLP 145 King Street West, Suite 2200 Toronto ON M5H 4G2 Raymond M. Slattery (LSUC #20479L) 416-369-4149
rslattery@mindengross.com

David T. Ullmann (LSUC#423571) 416-369-4148 dullmann@mindengross.com

Sepideh Nassabi (LSUC#60139B) 416-369-4323 snassabi@mindengross.com 416-864-9223 fax

Lawyers for the Applicants

#### ASSIGNMENT AGREEMENT effective as of the Closing Date.

AMONG:

4037057 CANADA INC., corporation duly incorporated, having its head office at 8600 Decarie Blvd., Suite 200, Town of Mont-Royal, in the Province of Québec, H4P 2N2, herein represented by Terry Pomerantz, its President, duly authorized has he so declares;

Hereinafter called (the "Assignor")

AND:

8032858 CANADA INC., corporation duly incorporated, having its head office at 8600 Decarie Blvd., Suite 200, Town of Mont-Royal, in the Province of Québec, H4P 2N2, herein represented by Terry Pomerantz, its President, duly authorized has he so declares;

Hereinafter called (the "Assignee")

AND:

DOMFOAM INTERNATIONAL INC., a corporation duly incorporated, having its head office at 8785 Langelier Blvd. (Saint-Leonard), City of Montreal, herein represented by Tony Vallecoccia, its duly authorised representative;

Hereinaster called (the "Vendor")

#### PREAMBLE:

#### WHEREAS:

- the Assignor entered into an asset purchase agreement as of March 8, 2012 (the "APA") with the Vendor by which the Assignor purchased from the Vendor the Purchased Assets (as described in the APA);
- by Order of Justice Brown dated March 16, 2012 (the "Approval and Vesting Order"), the Ontario Superior
  Court of Justice (the "Court") approved the terms and provisions of the APA and the purchase and sale
  transaction provided for therein and all of the Vendor's right, title and interest, if any, in and to the Purchased
  Assets were vested in the Purchaser, conditional upon the filing by the Monitor of a certificate with the Court;
- the Assignor has the ability to assign the APA (as per Section 7.14 of the APA) if the assignee is a "related entity" to the Assignor within the meaning of that expression under the *Income Tax Act* (Canada). In the event of such assignment, the Assignor shall remain liable to the Vendor in respect of its obligations hereunder;
- the Assignee has agreed to take over all the obligations of the Assignor pursuant to the APA and the Assignee
  declares that it is a "related entity" to the Assignor within the meaning of that expression under the *Income Tax Act* (Canada);
- it is the intention of the parties that all the rights and liabilities as per the APA be assigned from the Assignor to the Assignee;

the Vendor covenants and agrees to this Assignment Agreement;

NOW THEREFORE in consideration of the preamble and the mutual agreements and covenants herein contained, the parties hereto hereby covenant and agree as follows:

#### 1 INTERPRETATION – DEFINED TERMS

- 1.1 Preamble The preamble forms an integral part of the present Agreement.
- 1.2 Defined terms Other than as defined herein, all capitalized terms shall have the meanings ascribed to them in the APA.

#### 2 ASSIGNMENT AND DIRECTION

In consideration of the sum of TWO DOLLARS (\$2.00) and other valuable consideration, the receipt of which is acknowledged, the Assignor assigns and directs to the Assignee all its interest in and to the APA, including all rights of action or other rights accruing to the Assignor, or which might after this assignment takes effect or accrue to the Assignor under the APA. Notwithstanding the assignment of the APA to the Assignee, the Assignor shall remain liable to the Vendor in respect of its obligations as per the APA.

#### 3 INTERVENTION

The Vendor intervenes to this Assignment Agreement to acknowledge and consent to the assignment.

#### 4 GENERAL PROVISIONS

- 4.1 Further Assurances Each of the parties hereby covenants and agrees that at any time and from time to time after the Closing Date it will, upon the request of the others, do, execute, acknowledge and deliver or cause to be done, executed, acknowledged and delivered all such further acts, deeds, assignments, transfers, conveyances and assurances as may be required for the better carrying out and performance of all the terms of this Assignment Agreement.
- 4.2 Counterparts This Assignment Agreement may be executed in several counterparts, each of which so executed shall be deemed to be an original, and such counterparts together shall constitute but one and the same instrument.
- 4.3 Successors and Assigns This Assignment Agreement shall be binding upon and enure to the benefit of the parties hereto and their respective successors and permitted assigns. Nothing herein, express or implied, is intended to confer upon any Person, other than the parties hereto and their respective successors and assigns, any rights, remedies, obligations or liabilities under or by reason of this Assignment Agreement.
- **4.4** Amendments No modification or amendment to this Assignment Agreement may be made unless agreed to by the parties hereto in writing.

4.5 Governing Law - This Assignment Agreement shall be governed by and construed in accordance with the laws of the Province of Quebec and the laws of Canada applicable therein and each of the parties irrevocably attorns to the non-exclusive jurisdiction of the courts of the Province of Quebec. The parties acknowledge that any disputes related to the interpretation of this Assignment Agreement (including but not limited to its interpretation or its execution) shall be sought before the Quebec Superior Court in Montreal.

#### 5 CONFIRMATION OF THE APA

Except as otherwise provided in this Assignment Agreement, the parties confirm and reiterate all the terms and conditions contained in the APA and the parties acknowledge being bound by said terms and conditions.

#### 6 LANGUAGE

The parties acknowledge that they have required that this Assignment Agreement and all related documents be prepared in English. / Les parties reconnaissent avoir exigé que la présente convention et tous les documents connexes soient rédigés en anglais.

(Signature page follows.)

#### IN WITNESS WHEREOF, the parties hereto have executed the present Assignment Agreement:

SIGNED BY THE ASSIGNOR, in Town of Mont-Royal, this 26th day of March 2012.

4037057 Canada Inc.

er:

Terry Pomerantz, President I have authority to bind the corporation

SIGNED BY THE ASSIGNEE, in Town of Mont-Royal, this 26th day of March 2012.

8032858 Canada Inc.

Per:

Terry Pomerantz, President
I have authority to bind the corporation

SIGNED BY THE VENDOR, in Toronto, this 26th day of March 2012.

Domfoam International Inc.

Per:

Tony Vallecoccia, President
I have authority to bind the corporation

K\_Assignment\_of\_APA\_vExec

IN WITNESS WHEREOF, the parties hereto have executed the present Assignment Agreement
SIGNED BY THE ASSIGNOR, in Town of Mont-Royal, this 26th day of March 2012.
Per: Terry Pomerantz, President I have authority to bind the corporation
SIGNED BY THE ASSIGNEE, in Town of Mont-Royal, this 26 <sup>th</sup> day of March 2012.  8032858 Canada Inc.
Per: Terry Pomerantz, President 1 have authority to bind the corporation

SIGNED BY THE VENDOR, in Toronto, this 26th day of March 2012.

Domfoam International Inc.

Per:

Tony Vallecoccia, President I have authority to bind the corporation

K\_Assignment\_of\_APA\_vExec

## Tab G

THIS IS EXHIBIT "G" TO THE AFFIDAVIT OF JACQUES VINCENT SWORN THIS 13<sup>th</sup> DAY OF

SEPTEMBER, 2018

Commissioner, etc.

Anny Groulx

Urethane Litigation Fund

c/o GGG PO Box 10223

Dublin, OH 43017-5723

Citibank, N.A.

1-8/210

CHECK NUMBER: CHECK DATE:

00001733 4 8 7

Three million seven hundred forty one thousand six hundred thirty nine and 62/100 Dollars

\*\*\*\$3,741,639.62

PAY TO THE ORDER OF

DOMFOAM INTERNATIONAL INC

ATTM: DAVID ULLMAN

BLANEY-MCMURTY, LLP

2 QUEEN STREET EAST, SUITE 1500

TORONTO, ONTARIO MSC 3G5

CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

#OD1733# #O21000089# 6783779643#

## Tab H

THIS IS EXHIBIT "H" TO THE AFFIDAVIT OF JACQUES VINCENT

SWORN THIS 13<sup>th</sup> DAY OF SEPTEMBER, 2018

Commissioner, etc.

Anny Groulx

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

### MOTION RECORD OF THE APPLICANTS (Re: Stay Extension, Returnable May 29, 2018)

May 22, 2018

#### **BLANEY McMURTRY LLP**

Barristers and Solicitors Suite 1500 - 2 Queen Street East Toronto, ON M5C 3G5

David T. Ullmann LSUC #42357I

Tel: (416) 596-4289 Fax: (416) 594-2437

Alexandra Teodorescu LSUC #63899D

Tel: (416) 596-4279 Fax: (416) 593-5437

Lawyers for the Applicants

TO: SERVICE LIST

## INDEX

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

### MOTION RECORD OF THE APPLICANTS (Re: Stay Extension, Returnable May 29, 2018)

#### INDEX

Tab		Document	Page Nos.
1		Notice of Motion, dated May 22, 2018	1-5
2		Affidavit of Tony Vallecoccia, sworn May 22, 2018	6-13
	A	Plan Implementation Certificate, dated June 23, 2017	14-16
	В	Letter from US class action counsel in the US Urethane Proceeding to the Applicants enclosing the respective cheques, dated March 21, 2018	17-23
	С	Letter from Stephen Gaudreau to the Monitor, dated May 2, 2018	24-31
	D	Distribution Protocol	32-55
3		Draft Order	56-58

## TAB 1

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

### NOTICE OF MOTION (Re: Stay Extension, Returnable May 29, 2018)

THE MOVING PARTIES, 3113736 Canada Ltd. (formerly known as Valle Foam Industries (1995)) ("Valle Foam"), 4362063 Canada Ltd. (formerly known as Domfoam International Inc.) ("Domfoam"), and A-Z Sponge & Foam Products Ltd. ("A-Z Foam") (collectively, the "Applicants") will make a motion to a judge presiding over the Commercial List at 10:00 a.m. on May 29, 2018, or as soon thereafter as the motion can be heard, at 330 University Avenue, Toronto, Ontario.

#### PROPOSED METHOD OF HEARING:

This motion is to be heard orally.

#### THE MOTION IS FOR:

an Order substantially in the form contained at Tab 3 of the Applicants' Motion Record, extending the Stay Period (as that term is defined in the Initial Order of the Honourable Mr. Justice Newbould dated January 12, 2012) to and including November 30, 2018 and approving the Monitor's report, conduct and fees; and

2. such further and other relief as this Honourable Court may deem just.

#### THE GROUNDS FOR THE MOTION ARE:

- 3. On January 12, 2012, the Applicants sought and were granted protection under the Companies' Creditors Arrangement Act, RSC 1985, c C-36 ("CCAA"), as amended pursuant to the Order of the Honourable Mr. Justice Newbould (the "Initial Order");
- 4. Deloitte & Touche Inc., now known as Deloitte Restructuring Inc., was appointed in the Initial Order to act as monitor in these CCAA proceedings ("Monitor");
- 5. As a result of the sale of assets of the Applicants, Valle Foam changed its name to 3113736 Canada Ltd., and Domfoam changed its name to 4362063 Canada Ltd. The style of cause of these proceedings was changed by the Order of Justice Brown, dated June 15, 2012 to reflect the change of names;
- 6. The Order of the Honourable Mr. Justice Brown, dated June 15, 2012 established a process to identify pre- and post-filing claims against the Applicants and/or their officers and directors ("Claims Process Order")
- 7. The Meeting Order was approved by the Honourable Mr. Justice Penny on September 6, 2016, accepting Domfoam's Plan of Compromise and Arrangement ("Plan") for filing with the Court and authorizing Domfoam to seek approval of the Plan at the meeting of the creditors ("Creditors' Meeting");
- 8. The Creditors Meeting was held on October 19, 2016;

- 9. The Applicants achieved the required statutory "double majority" needed to approve the Plan. Proven Creditors holding 92% in number and 99% in value voted to approve the Resolution in favour of the Plan;
- 10. The Plan was sanctioned by way of Order from the Honourable Mr. Justice Hainey, dated January 24, 2017;
- 11. The conditions precedent to Plan implementation have been satisfied or waived, and the Plan has been implemented;
- 12. Each of the Applicants are claimants in a U.S. class action proceeding relating to price fixing for a product known as "Polyether Polyol" (the "US Urethane Proceeding"). A settlement was entered into with one of the defendants in the US Urethane Proceeding, in which the defendant agreed to pay \$834 million USD for distribution to the class members, including the Applicants ("Polyols Settlement");
- 13. On or about March 21, 2018, an initial distribution representing 85% of the total recovery from the Polyols Settlement was made to the class members, including the Applicants. A second and final tranche of money representing up to 15% is payable to the Applicants from the Polyols Settlement;
- 14. The Applicants may also be class members in a certified class action in Ontario relating to the price fixing of polyether polyols products purchased in Canada ("Canadian Urethane Proceeding"). Settlement funds are being held in trust for the benefit of the class members in the Canadian Urethane Proceeding, and a claims process will be initiated to determine distribution to the class;

15. Valle Foam continues its collection and enforcement efforts to pursue outstanding receivables;

#### **Extension of Stay Period**

- 16. The Initial Order granted a Stay Period until February 10, 2012;
- 17. The Stay Period granted under the Initial Order was subsequently extended for all of the Applicants from time to time by orders of this Honourable Court;
- 18. Most recently, the Stay Period was extended to May 31, 2018, by the Order of the Honourable Mr. Justice Myers, dated November 24, 2017;
- 19. The Applicants have been acting and continue to act in good faith and with due diligence in these CCAA proceedings;
- 20. It is just and convenient and in the interests of the Applicants and their stakeholders that the requested Order be granted and the Stay Period extended;
- 21. Although the Plan has been approved, the continuation of the stay of proceedings in the Domfoam estate is required to ensure the orderly collection and distribution of the remaining assets and settlement funds from the various class actions;
- 22. The proposed extension of the Stay Period is supported by the Monitor and there is no known opposition;

#### Approval of Monitor's fees, conduct and report

23. Following the implementation of the Plan, the Monitor made a distribution of funds on hand to the creditors in accordance with the Plan and the Orders of this Court:

497

24. the provisions of the CCAA and the inherent and equitable jurisdiction of this

Honourable Court;

25. Rule 1.04, 1.05, 2.03, 3.02, 16 and 37 of the Ontario Rules of Civil Procedure, RRO

1990, Reg 194, as amended, and section 106 of the Ontario Courts of Justice Act, RSO

1990, c C 43, as amended; and

26. Such further and other grounds as counsel may advise.

THE FOLLOWING DOCUMENTARY EVIDENCE will be used at the hearing of the

motion:

27. The Affidavit of Tony Vallecoccia, sworn May 22, 2018;

28. The Eighteenth Report of the Monitor, to be filed; and

29. Such further and other material as counsel may advise and this Court may permit.

May 22, 2018

**BLANEY McMURTRY LLP** 

Barristers and Solicitors Suite 1500 - 2 Queen Street East Toronto, ON M5C 3G5

David T. Ullmann LSUC #42357I

Tel: (416) 596-4289

Fax: (416) 594-2437

Alexandra Teodorescu LSUC #63899D

Tel: (416) 596-4279

Fax: (416) 593-5437

Lawyers for the Applicants

TO: SERVICE LIST

## TAB 2

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT,
R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### AFFIDAVIT OF TONY VALLECOCCIA

(Sworn May 22, 2018)

- I, TONY VALLECOCCIA, of the Town of Milton, in the Regional Municipality of Halton, in the Province of Ontario, MAKE OATH AND SAY:
- 1. I am the President and Chief Executive Officer of 3113736 Canada Ltd., formerly known as Valle Foam Industries (1995) Inc. ("Valle Foam"), and of 4362063 Canada Ltd., formerly known as Domfoam International Inc. ("Domfoam"), and a director of Valle Foam, Domfoam and A-Z Sponge & Foam Products Ltd. ("A-Z Foam") (collectively, the "Applicants"), and as such have knowledge of the matters to which I hereinafter depose, except where otherwise stated.
- 2. To the extent that the matters deposed to in this affidavit are based on my review of documents or information and belief, I have stated the source of my information and belief and do verily believe the information to be true.

3. I swear this affidavit in support of the Applicants' motion for an Order, *inter alia*, extending the stay of proceedings for all of the Applicants to and including November 30, 2018, and approving the Eighteenth Report of the Monitor, to be filed separately.

#### Background

- 4. On January 12, 2012, the Applicants sought and were granted protection under the Companies' Creditors Arrangement Act, RSC 1985, c C-36, as amended ("CCAA"), pursuant to the Order of the Honourable Mr. Justice Newbould (the "Initial Order").
- 5. Deloitte & Touche Inc., now known as Deloitte Restructuring Inc., was appointed in the Initial Order to act as monitor in these CCAA proceedings ("Monitor").
- 6. As a result of the sale of assets of the Applicants, Valle Foam changed its name to 3113736 Canada Ltd., and Domfoam changed its name to 4362063 Canada Ltd. The style of cause of these proceedings was changed by the Order of Justice Brown dated June 15, 2012 to reflect the change of names. For the purpose of this affidavit, the said Applicants will still be referred to as Valle Foam, Domfoam and A-Z Foam.
- 7. On September 6, 2016, the Honourable Mr. Justice Penny approved the Applicants' order seeking acceptance of Domfoam's Plan of Compromise and Arrangement, dated August 23, 2016 ("Plan") for filing with the Court and authorizing Domfoam to seek approval of the Plan at a meeting of the creditors ("Meeting Order").
- 8. Pursuant to the Meeting Order, the meeting of the creditors of Domfoam was held on October 19, 2016 ("Creditors' Meeting"). The Plan was approved by an overwhelming majority (92% in number and 99% in value) of creditors at the Creditors' Meeting.

- 9. The Plan was approved and sanctioned by the Honourable Mr. Justice Hainey on January 24, 2017.
- 10. The prerequisites to the implementation of the Plan have all now been satisfied, and, on June 23, 2017, the Monitor filed with the Court its Plan Implementation Certificate, a copy of which is attached hereto and marked as **Exhibit "A"**.
- 11. Following the sanction and implementation of the Plan, the Monitor has the ongoing responsibility to collect funds from the Polyols Settlement (discussed below), and to distribute those funds to creditors with proven claims under the Plan.

#### Extension of the Stay Period

- 12. The Initial Order granted a stay of proceedings ("Stay Period") until February 10, 2012.
- 13. The Stay Period granted under the Initial Order was subsequently extended from time to time by orders of the Court, the most recent being the Order of the Honourable Mr. Justice Myers, dated November 24, 2017, which extended the Stay Period to May 31, 2018.
- 14. The Applicants are seeking to extend the Stay Period up to and including November 30, 2018.
- 15. No cash flow is being provided with this affidavit as the Applicants have limited expenses and no employees. I am confident that the Applicants each have sufficient funds on hand to meet their obligations on a go forward basis for the period of the proposed extension.
- 16. I believe that the Applicants have acted, and continue to act, in good faith and with due diligence in pursuing the orderly wind down of Domfoam and collecting outstanding amounts

owed to Valle Foam (as explained in further detail below). I am informed by the Monitor that it supports the request to extend the Stay Period to November 30, 2018.

17. An extension of the Stay Period is required to allow the Applicants to continue collecting outstanding accounts as well as funds due under the Polyols Settlement (as defined below), and to allow the Monitor to distribute these funds to creditors with proven claims.

#### Collection of the Polyols Settlement

- 18. Each of the Applicants are claimants in a U.S. class action proceeding that relates to price fixing for a product known as "Polyether Polyol" (the "US Urethane Proceeding").
- 19. There was a trial in respect of one of the defendants in the US Urethane Proceeding, the Dow Chemical Company ("Dow"), in which a judgment was rendered against Dow in the amount of \$1.06 billion ("Judgment").
- 20. In March 2016, Dow withdrew its appeal of the Judgment to the United States Supreme Court and accepted a settlement under which it agreed to pay \$834 million USD, for distribution to the class members, including the Applicants (the "Polyols Settlement").
- 21. Refund Recovery Services LLC (now known as Lex Recovery Group) ("Lex Recovery") was retained as the Applicants' exclusive agent to assist in filing the necessary documents to secure their share of the Polyols Settlement funds. Lex Recovery has filed claims with the administrator on behalf of the Applicants in accordance with the deadlines set out in the US Urethane Proceeding.

- 22. A distribution hearing with respect to the Polyols Settlement took place on December 19, 2017 in Kansas City, Kansas, and the Court approved the proposed distribution of the Polyols Settlement funds on that date.
- 23. On or about March 21, 2018, an initial distribution representing 85% of the total recovery from the Polyols Settlement was made to the creditors. The Applicants each received the following amounts from the Polyols Settlement:
  - a) Valle Foam received \$5,542,999.25 USD;
  - b) Domfoam received \$3,741,639.62 USD; and
  - c) A-Z Foam received \$732,651.37 USD.

Attached hereto and marked as **Exhibit "B"** is a copy of the letters from US class action counsel in the US Urethane Proceeding to the Applicants enclosing the respective cheques. I am advised by my counsel, David Ullmann, that these cheques have been sent to the Monitor.

24. The Applicants are required to pay \$2,504,322.56 USD to Lex Recovery from the funds they received from the Polyols Settlement, which represents the 25% fee owing to Lex Recovery based on the retainer with the Applicants to assist and recover their claims in the US Urethane Proceeding. On May 2, 2018, counsel for the Applicants, Blaney McMurtry LLP, wrote to the Monitor enclosing the invoices for Lex Recovery. Attached hereto and marked as **Exhibit "C"** is a copy of a letter from Stephen Gaudreau of Blaney McMurtry LLP to the Monitor, dated May 2, 2018.

- 25. The Applicants are set to receive a second and final tranche of money up to 15% that is payable to them from the Polyols Settlement. As of the swearing of this affidavit, there is no projected final payout statement date.
- 26. An extension of the Stay Period is required to allow for further distributions to be made to the Applicants pursuant to the Polyols Settlement. The funds paid to Domfoam under the Polyols Settlement will be distributed to proven creditors *pro-rata* under the Plan.

#### Canadian Class Action

- 27. A similar class action was initiated and certified against Dow and a number of other defendants in Ontario. The class action was certified on behalf of all persons in Canada who purchased polyether polyol products between January 1, 1999 and December 31, 2004 ("Canadian Urethane Proceeding").
- 28. Settlements have been reached in the Canadian Urethane Proceeding with several defendants wherein the defendants agreed to pay a total of \$13.3 million. Dow agreed to contribute \$5,080,000 CDN into the settlement funds, which are being held in trust for the benefit of the class members.
- 29. Class counsel for the Canadian Urethane Proceeding, Siskinds LLP, intends to implement a claims process in order to determine the class members entitled to a distribution from the Canadian settlement funds. Attached hereto and marked as **Exhibit "D"** is a copy of a summary of the Canadian Urethane Proceeding from the website of class counsel, and the proposed distribution protocol.

30. The Applicants with the assistance of Lex Recovery are currently in the process of determining whether or not they are class members in the Canadian Urethane Proceeding. The Applicants hope to recover additional funds from the Canadian class action for the benefit of the creditors of the respective estates.

#### Valle Foam Collection Efforts

- 31. As set out in my affidavit of November 15, 2017, there were eight actions initiated by Valle Foam to collect various outstanding receivables. Judgment has now been obtained with respect to three of these actions, and Valle Foam has diligently been enforcing these judgments during the stay period. In addition, two of these actions have been settled, and one has been dismissed on consent without costs.
- 32. With respect to the remaining two pieces of litigation, Valle Foam continues to vigorously pursue these actions, including by conducting examinations for discovery and bringing summary judgment motions, where necessary and appropriate. The Monitor has been advised of the status of each of these actions.
- 33. Extending the Stay Period will provide Valle Foam with the breathing room required to continue pursuing its collection and enforcement efforts.

#### A-Z Foam

34. Although the business of A-Z Foam has been ceased for several years at this point in time, it is an affiliated entity of the Applicants, and the continuation of the stay is convenient as there remain amounts to collect from the Polyols Settlement and inter-company accounting to be resolved.

- 35. No one has at any time during the CCAA Proceedings objected to the continuation of the stay with respect to A-Z Foam, and I am not aware of any objections at this time.
- 36. I swear this affidavit in support of the Applicants' motion for an Order, *inter alia*, extending the Stay Period to and including November 30, 2018, and for no improper purpose.

SWORN before me at the Town of Milton in the Province of Ontario, this 22 <sup>nd</sup> day of May, 2018	) ) )	
Mandan	)	a de le
Natasha Rambaran (A commissioner for taking affidavits)	)	TONY VALLECOCCIA

Natasha Devi Ramberan, a Commissioner, etc., Province of Oniario, white a Student-et-Law, Exotres April 17, 2021.

## **EXHIBIT A**

This is Exhibit "A" referred to in the Affidavit of Tony Vallecoccia sworn on the 22<sup>nd</sup> day of May, 2018.

Commissioner for Taking Affidavits (or as may be)

Natasha Rambaran

Netasha Devi Ramberen, a Commissiones, etc., Province of Ontario, while a Student-at-Law, Expires April 17, 2021.

Court File No. CV-12-9545-00CL

#### ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

NOTIFIED THE MATTER OF A PLAN OF COMPROMISE OR NOTIFIED TO A 113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

#### Plan Implementation Certificate

All capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Plan of Compromise and Arrangement concerning, affecting and involving 4362063 Canada Ltd., formerly known as Domfoam International Inc. ("Domfoam") dated August 23, 2016 (as amended, varied or supplemented from time to time in accordance with the terms thereof, and together with all schedules thereto, the "Plan"), which Plan is attached as Schedule "A" to the Order of the Honourable Justice Hainey made in these proceedings on January 24, 2017 (the "Sanction Order").

Pursuant to section 7.1 of the Plan and paragraph 8 of the Sanction Order, Deloitte Restructuring Inc., in its capacity as the Court-appointed monitor of the Applicants (the "Monitor") delivers this certificate and hereby certifies that:

 The conditions precedent set out in section 7.1 of the Plan have been satisfied or waived, as applicable.

- Pursuant to the terms of the Plan, upon the filing of this Plan Implementation Certificate
  with the Court, the Plan Implementation Date shall occur and the Plan shall become
  effective.
- 3. This Plan Implementation Certificate will be filed with the Court.

DATED at the City of Toronto, in the Province of Ontario, this 23rd day of June, 2017.

**DELOITTE RESTRUCTURING INC.**, in its capacity as the Court-appointed Monitor of the Applicants, and not in its personal capacity

Per:

Name: Catherine Hristow

Title:

Vice President - Financial Advisory -

Reorganization Services

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED Z SPONGE & FOAM PRODUCTS LTD.

Court File No.: CV-12-9545-00CL

## SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST) ONTARIO

Proceedings commenced at Toronto

# Plan Implementation Certificate

# Thornton Grout Finnigan LLP

Barristers & Solicitors

Suite 3200, TD West Tower

P.O. Box 329, Toronto-Dominion Centre 100 Wellington Street West

Toronto, ON M5K 1K7 Tel: (416) 304-1616 Fax: (416) 304-1313

# Grant B. Moffat (LSUC# 32380L)

Email: gmoffat@tgf.ca Tel: 416-304-0599

Lawyers for the Monitor, Deloitte Restructuring Inc.

## **EXHIBIT B**

This is Exhibit "B" referred to in the Affidavit of Tony Vallecoccia sworn on the 22<sup>nd</sup> day of May, 2018.

Commissioner for Taking Affidavits (or as may be)

Manlew

Natasha Rambaran

Natasha Devi Ramberan, a Commissioner, etc., Province of Ontario, while a Student-et-Law. Expires April 17, 2021.

#### FINE, KAPLAN AND BLACK, R.P.C.

ATTORNEYS AT LAW

ONE SOUTH BROAD STREET, 23RD FLOOR PHILADELPHIA, PENNSYLVANIA 19107

DONALD L. PERELHAN
ROBERTA D. LIEBENBERG
MICHAEL D. BASCH
JEFFREY S. ISTVAN
GERARD A. DEVER
PAUL COSTA
ADAM J. PESSIN
MARY L. RUSBELL
RIA C. MOMBLANCO
JESSICA D. KHAN

(215) 567-6565 FAX: (215) 568-5872 E-mail:mail@/linekaplan.com www.finekaplan.com ALLEN D. BLACK ARTHUR M. KAPLAN MATTHEW DUNCAN ELIBE E. SINGER ROBERT A. LARSEN JOSEPH J. BORGIA OF COUNSEL

AARON M. FINE 1923-2013

March 21, 2018

To: Members of the Class in the Urethane Antitrust Litigation

From: The Urethane Litigation Team

Enclosed please find your initial distribution from the \$835 million settlement with The Dow Chemical Company. This distribution represents approximately 85% of your total recovery from this settlement.

It has been our pleasure to represent the class over the last 13+ years, and we are gratified that we were able to successfully conclude the litigation on your behalf.

If you have any questions, do not hesitate to contact us.

Donald L. Perelman Roberta D. Liebenberg Gerard A. Dever Paul Costa Matthew Duncan Nancy M. Blakeslee, Paralegal

Allyson L. Katzman, Paralegal Susan J. Hufnagel, Paralegal Fine, Kaplan and Black, R.P.C. One South Broad Street, 23rd Floor

Philadelphia, PA 19107

Co-Lead Counsel

Richard A. Koffman Kit A. Pierson

Christopher J. Cormier Sharon K. Robertson Laura M. Alexander

Cohen Milstein Sellers & Toll, PLLC 1100 New York Ave. NW, Suite 500

Washington, D.C. 20005

Co-Lead Counsel

Joseph Goldberg

Deborah Tope, Paralegal

Freedman Boyd Hollander Goldberg

Urias & Ward, P.A. 20 First Plaza Suite 700 Albuquerque, NM 87102

Lead Trial Counsel



o/o GCG PO Box 10223 Dublin, OH 43017-5723 Claim Number: Check Number: Check Date: Check Amount:

00000621<sup>1</sup> 00001728 03/21/18 \$5,542,999.25

VALLE FOAM ATTN: DAVID ULLMAN BLANEY-MCMURTRY, LLP, 2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5 CANADA

Ref No.: 7364

#### Dear Class Member:

This check represents your initial payment from the **Dow Settlement** fund in the Urethane Antitrust Litigation – Polyether Polyol Cases. This payment is based on the Claim Form(s) you filed in connection with the Dow Settlement and represents 85% of your Recognized Loss as calculated in accordance with the Court-approved Plan of Allocation. This distribution is being made in accordance with the Court's December 19, 2017 Order Approving Distribution from the Dow Settlement Fund.

We cannot provide individual tax advice. The tax treatment of distributions from the Dow Settlement Fund is the responsibility of each recipient. You should consult your tax advisor to determine the tax consequences, if any, of this distribution to you.

Please cash this check promptly as it becomes void and subject to re-distribution 90 days after the date of issue. If you have changed your address from the address on the accompanying check or if you have any questions, you may contact the Settlement Administrator by phone toll-free at 1-877-741-1226, by email at <a href="mailto:questions@polyetherpolyolsettlement.com">questions@polyetherpolyolsettlement.com</a>, or by mail at:

Urethane Antitrust Litigation c/o GCG P.O. Box 10223 Dublin, OH 43017-5723

19

Urethane Litigation Fund c/6 GCG PO Box 10223 Dublin, OH 43017-5723

CRIDERIK, N.A.

516

1-8/210

CHECK NUMBER: CHECK DATE:

00001728 03/21/18

Five million five hundred forty two thousand nine hundred ninety nine and 25/100 Dollars

\*\*\*\$5,542,999.25

PAY TO VALLE FOAM

ATTN: DAVID ULLMAN BLANEY-MCMURTRY, LLP,

2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5 CANADA CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

""OO1728" #021000089# 6783779643#

517



c/o GCG PO Box 10223 Dublin, OH 43017-5723 Claim Number: 01007018
Check Number: 00001733
Check Date: 03/21/18
Check Amount: \$3,741,639.62

DOMFOAM INTERNATIONAL INC ATTM: DAVID ULLMAN BLANEY-MCMURTY, LLP 2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5

Ref No.: 3468

#### Dear Class Member:

This check represents your initial payment from the **Dow Settlement** fund in the Urethane Antitrust Litigation – Polyether Polyol Cases. This payment is based on the Claim Form(s) you filed in connection with the Dow Settlement and represents 85% of your Recognized Loss as calculated in accordance with the Court-approved Plan of Allocation. This distribution is being made in accordance with the Court's December 19, 2017 Order Approving Distribution from the Dow Settlement Fund.

We cannot provide individual tax advice. The tax treatment of distributions from the Dow Settlement Fund is the responsibility of each recipient. You should consult your tax advisor to determine the tax consequences, if any, of this distribution to you.

Please cash this check promptly as it becomes void and subject to re-distribution 90 days after the date of issue. If you have changed your address from the address on the accompanying check or if you have any questions, you may contact the Settlement Administrator by phone toll-free at 1-877-741-1226, by email at questions@polyetherpolyolsettlement.com, or by mail at:

Urethane Antitrust Litigation c/o GCG P.O. Box 10223 Dublin, OH 43017-5723 **Urethane Litigation Fund** c/o GCG PO Box 10223 Dublin, OH 43017-5723

1-6/210

CHECK NUMBER: CHECK DATE:

00001733 03/21/18

Three million seven hundred forty one thousand six hundred thirty nine and 62/100 Dollars

\*\*\*\$3,741,639.62

PAY TO THE ORDER OF

DOMFOAM INTERNATIONAL INC

ATTM: DAVID ULLMAN BLANEY-MCMURTY, LLP

2 QUEEN STREET EAST, SUITE 1500

TORONTO, ONTARIO MSC 3G5

CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

**AUTHORIZED SIGNATURE** 

#OO1733# #O210000B9# 6783779643#

519



c/o GCG PO Box 10223 Dublin, OH 43017-5723 Claim Number: Check Number: Check Date: Check Amount: 01008517 00001736 03/21/18 \$732,651.37

A - Z SPONGE & FOAM PRODUCTS LTD ATTN: DAVID ULLMAN BLANEY-MCMURTY, LLP 2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5 CANADA

Ref No.: 1935

#### Dear Class Member:

This check represents your initial payment from the **Dow Settlement** fund in the Urethane Antitrust Litigation – Polyether Polyol Cases. This payment is based on the Claim Form(s) you filed in connection with the Dow Settlement and represents 85% of your Recognized Loss as calculated in accordance with the Court-approved Plan of Allocation. This distribution is being made in accordance with the Court's December 19, 2017 Order Approving Distribution from the Dow Settlement Fund.

We cannot provide individual tax advice. The tax treatment of distributions from the Dow Settlement Fund is the responsibility of each recipient. You should consult your tax advisor to determine the tax consequences, if any, of this distribution to you.

Please cash this check promptly as it becomes void and subject to re-distribution 90 days after the date of issue. If you have changed your address from the address on the accompanying check or if you have any questions, you may contact the Settlement Administrator by phone toll-free at 1-877-741-1226, by email at questions@polyetherpolyolsettlement.com, or by mail at:

Urethane Antitrust Litigation c/o GCG P.O. Box 10223 Dublin, OH 43017-5723

23

libank, N.A.

1-8/210

Urethane Litigation Fund c/o GCG PO Box 10223 Dublin, OH 43017-5723

CHECK NUMBER: CHECK DATE: 00001736 03/21/18

Seven hundred thirty two thousand six hundred fifty one and 37/100 Dollars

\*\*\*\*\$732,651.37

PAY TO THE ORDER OF

A - Z SPONGE & FOAM PRODUCTS LTD

ATTN: DAVID ULLMAN BLANEY-MCMURTY, LLP

2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5 CANADA CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

#OD1736# #021000089# 6783779643#

# **EXHIBIT C**

This is Exhibit "C" referred to in the Affidavit of Tony Vallecoccia sworn on the 22<sup>nd</sup> day of May, 2018.

Commissioner for Taking Affidavits (or as may be)

Natasha Rambaran

Natasha Devi Rambaran, a Commissioner, etc., Province of Ontario, while a Student-at-Law, Expires April 17, 2021.



Blaney McMurtry LLP | Lawyers 2 Queen Street East | Suite 1500 Toronto, Ontario M5C 3G5 (T) 416-593-1221

(W) Blaney.com

Stephen Gaudreau

D: 416-596-4285 F: 416-594-3594 sgaudreau@blaney.com

May 2, 2018

VIA EMAIL

Catherine Hristow Deloitte 8 Adelaide Street West, Suite 200 Toronto, Ontario, Canada M5H 0A9

Dear Ms. Hristow:

Re: Payments by Dom-Foam International Inc., Valle Foam, and A-Z Sponge & Foam Products Inc. (the "Companies") to Lex Group LLC arising from Urethane Polyether Polyol Antitrust Litigation settlement (the "US Urethane Proceedings")

We are writing with respect to the amounts due and owing to Lex Group LLC ("Lex Group") arising from its retainer with the Companies to assist and recover their claims in the US Urethane Proceedings (the "Retainer"). As a reminder, Lex Group's fee is equal to 25% of all funds paid to the Companies pursuant to the US Urethane Proceedings. The Companies were recently paid a total of \$USD 10,017,290.20 pursuant to settlements reached in the US Urethane Proceedings.

As a result, the Lex Group has delivered invoices to each of the Companies totalling \$USD 2,504,322.56 (being its 25% fee pursuant to the Retainer):

- Invoice #916 to Dom-Foam International Inc. in the amount \$USD 935,409.91;
- Invoice #920 to Valle Form in the amount of \$USD 1,385,749.81; and,
- Invoice #918 to A-Z Sponge & Foam Products Ltd. in the amount of \$USD 183,162.84.

Copies of the invoices are enclosed with this letter.

By way of brief background, the Retainer was originally with Lex Group's wholly owned subsidiary Refund Recovery Service LLC, which was then assumed by the Lex Group. The Valle Foam portion of that retainer was then assigned to the Enterprise Law Group. It is our understanding that by way of agreement between Lex Group and the Enterprise Law Group, all funds payable pursuant to the Retainer are to be paid to the Lex Group. In the past, the Companies have paid Lex Group its 25% fee pursuant to previous settlements in the US Urethane Proceedings. Lex Group's retainer was referenced and approved at paragraphs 38-41 of the Monitor's Seventh Report executed July 12, 2013.

We trust that the information provided is sufficient to provide your authorization to release the Funds to Lex Group.

524

However, should you require further information, please do not hesitate to contact myself or CJ Kishish, President and General Counsel at Lex Group. Mr. Kishish can be reached at 952.224.2152 or <a href="mailto:ckishish@kishishlaw.com">ckishish@kishishlaw.com</a>.

We look forward to your authorization to release the Funds.

Yours very truly,

Blaney McMurtry LLP

Stephen Gaudreau

SG/sg

Enclosures.

cc: David Ullmann, Grant Moffat

Invoice

Lex Acquisition Group, LLC

1555 Southcross Drive West Burnsville, MN 55306

Date	Invoice #		
4/5/2018	916		

Bill To	
Dom-Foam International Inc	T

Terms	Due Date
-	4/5/2018

		<u> </u>
Description	Amou	nt
25% settlement fee - Polyurethane Polyether Polyols ("Dow verdict")		935,409.9
*		
	Total	\$935,409,91
	Payments/Credits	\$0.00
	Balance Due	\$935,409.91

526

Citibank, N.A.

1-8/210

Urelhane Litigation Fund c/o GCG PO Box 10223 Dublin, OH 43017-5723

CHECK NUMBER: CHECK DATE: 00001733

Three million seven hundred forty one thousand six hundred thirty nine and 62/100 Dollars

\*\*\*\$3,741,639.62

PAY TO THE ORDER OF

DOMFOAM INTERNATIONAL INC

ATTM: DAVID ULLMAN

BLANEY-MCMURTY, LLP

2 QUEEN STREET EAST, SUITE 1500

TORONTO, ONTARIO MSC 3G5

CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

M

#001733# #021000089# 6783779643#

### Lex Acquisition Group, LLC

1555 Southcross Drive West Burnsville, MN 55306

527 Invoice	e
-------------	---

Date	Invoice#	
4/5/2018	920	

Bill To		10100	
Valle Foam		 	

Terms	Due Date
	4/5/2018

Description		Amount
% settlement fee - Polyurethane Polyether Polyols ("Dow verdict")		1,385,749.8
	i.	
	Total	\$1,385,749.8
	Payments/0	Credits \$0.0
		Due \$1,385,749.8

Urethane Litigation Fund c/o GCG PO Box 10223 Dublin, OH 43017-5723 Cilibank, N.A.

1-8/210

CHECK NUMBER: CHECK DATE: 00001728 03/21/18

Five million five hundred forty two thousand nine hundred ninety nine and 25/100 Dollars

\*\*\*\$5,542,999.25

PAY TO THE Valle foam

ATTN: DAVID ULLMAN BLANEY-MCMURTRY, LLP,

2 QUEEN STREET EAST, SUITE 1500 TORONTO, ONTARIO MSC 3G5 CANADA CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

AUTHUMIZED SRINATUR

""001728# #021000089# 6783779643#

## Lex Acquisition Group, LLC

1555 Southcross Drive West Burnsville, MN 55306

## <sup>529</sup> Invoice

Date	Invoice #
4/5/2018	918

Bill To	
A-Z Sponge & Foam Products Ltd	
I	

	Terms	Due Date
	,	4/5/2018
Description		Amount
25% settlement fee - Polyurethane Polyether Polyols ("Dow verdict")		183,162.84
		•
l		
	Total	\$183,162.84
	Payments/C	redits \$0.00
	Balance I	Due \$183,162,84

Urethane Litigation Fund c/o GCG PO Box 10223 Dublin, OH 43017-5723 Chibank, N.A.

1-8/210

CHECK NUMBER: CHECK DATE: 00001738

Seven hundred thirty two thousand six hundred fifty one and 37/100 Dollars

\*\*\*\*\$732,651.37

PAY TO THE ORDER A - Z SPONGE & FOAM PRODUCTS LITD

ATTN: DAVID ULLMAN

BLANEY-MCMURTY, LLP

2 QUEEN STREET EAST, SUITE 1500

TORONTO, ONTARIO MSC 3G5 CANADA

CASH PROMPTLY, VOID AND SUBJECT TO RE-DISTRIBUTION IF NOT CASHED WITHIN 90 DAYS AFTER ISSUE DATE.

AUTHORIZED SIGNATURE

- 1

# **EXHIBIT D**

This is Exhibit "D" referred to in the Affidavit of Tony Vallecoccia sworn on the 22<sup>nd</sup> day of May, 2018.

Commissioner for Taking Affidavits (or as may be)

Natasha Rambaran

Natasha Davi Rambaran, a Commissioner, etc., Province of Ontario, while a Student-st-Law, Expires April 17, 2021.

### **Polyether Polyols**

Active Action

The Polyether Polyols class action alleges that the Defendants unlawfully conspired to fix, increase, and/or maintain prices in the market for Polyether Polyols.

Polyether Polyols means polyether polyols, monomeric or polymeric diphenylmethane diisocyanate (MDI), toluene diisocyanate (TDI), and polyether polyol systems. Polyether Polyols are used in a variety of manufacturing applications.

The Plaintiff alleges that from at least January 1999 to December 2004, the Defendants and senior executive of the corporate defendants participated in illegal and secretive meetings and made arrangements relating to price targets, specific price increases, and market share divisions for Polyether Polyols.

Receive Updates on This Case

Veuillez cliquer ici pour accéder la version française de cette page.

#### DevelopmentsFAQs Documents

## **Developments**

#### Settlements

Settlements have been reached with the following defendants, totaling approximately \$13.3 million:

- Bayer Inc, Bayer AG, Bayer MaterialScience LLC (formerly known as Bayer Polymers LLC) and Bayer Corporation (collectively "Bayer") (CDN\$2,500,000)
- Lyondell Chemical Company ("Lyondell") (Up to \$35,000 towards Notice)
- Huntsman International LLC ("Huntsman") (USD\$1,700,000)
- BASF Canada Inc. ("BASF Canada") (CDN\$2,000,000)
- BASF Corporation ("BASF Corp") (CDN\$2,000,000)
- Dow Chemical Company and Dow Chemical Canada Inc. (collectively "Dow") (CDN\$5,080,000)

The Dow settlement approval hearing will be heard before the Ontario court on March 13, 2018. The other settlements have received court approval.

The settlement funds (plus interest, less court-approved fees and expenses) are being held in trust for the benefit of settlement class members. At the March 13, 2018 hearing, the Ontario court will be asked to approve a method of distributing the settlement funds to settlement class members. Please click here to view a copy of the <u>Proposed Distribution Protocol</u>.

#### Contested Litigation

In March 2014, the Ontario action was certified against Dow. The class action was certified on behalf of all persons in Canada (excluding defendants and their respective parents, employees, subsidiaries, affiliates, officers and directors) who purchased Polyether Polyol Products between January 1, 1999 and December 31, 2004. Polyether Polyol Products means polyether polyols, monomeric or polymeric diphenylmethane diisocyanate (MDI), toluene diisocyanate (TDI), and polyether polyol systems.

Dow was denied leave to appeal the certification decision.

If the Dow settlement is approved, it will resolve the Ontario class action in its entirety.

In light of the certification of a national class in Ontario, the Quebec action was discontinued. Please click here to view a copy of the Quebec Discontinue Order and the Quebec Notice of Discontinuance

## Ask a question

First Name	
Last Name	
Email	
Message	
I'm not a robot	
	reCAPTCHA Privacy - Terms
Submit	- Privacy - Jenna

polyetherclassaction@siskinds.com

© Copyright 2018 Siskinds LLP. All rights reserved.

# DISTRIBUTION PROTOCOL IN THE MATTER OF THE POLYETHER POLYOL PRODUCTS CLASS ACTION SETTLEMENTS

### INDEX

GENERAL PRINCIPLES	2
DEFINITIONS	3
DISTRIBUTION OF NET SETTLEMENT FUNDS  Calculation of Payments  Sample Calculations  Distribution	5 8
THE CLAIMS PROCESS	
The ClaimThe Online Claim Portal	9
The Claims Filing Process	12
Assistance in Filing a Claim	
Deficiencies	
Claims Administrator's Decision	15
Appeal of the Claims Administrator's Decision	
THE CLAIMS ADMINISTRATOR'S DUTIES AND RESPONSIBILITIES	
Supervisory Powers of the Ontario Court	
Communication, Languages and Translation	19
Undeliverable Mail Fraudulent Claims	
Taxes20 Reporting	21
Preservation and Disposition of Claim Submissions	21
Assistance to the Claims Administrator	

#### **GENERAL PRINCIPLES**

- 1. The procedures set forth herein are intended to govern the administration of the settlement agreements entered into with the following Defendants:
  - (a) Bayer Inc., Bayer AG, Bayer Material Science LLC (formerly known as Bayer Polymers LLC) and Bayer Corporation, dated June 6, 2007;
  - (b) Lyondell Chemical Company, dated November 22, 2011;
  - (c) Huntsman International LLC, dated April 11, 2012;
  - (d) BASF Corporation, dated April 12, 2012;
  - (e) BASF Canada Inc., dated April 12, 2012; and
  - (f) Dow Chemical Company and Dow Chemical Canada Inc., dated November 17, 2017 (collectively, the "Settlement Agreements").

#### 2. The administration shall:

- (a) implement and conform to the Settlement Agreements, orders of the Ontario Court, and this Distribution Protocol;
- (b) include the establishment and maintenance of the Settlement Website;
- (c) employ secure, paperless, web-based systems with electronic registration and record keeping wherever possible;
- (d) rely on sales information provided by the Defendants wherever possible; and
- (e) be bilingual in all respects.

3. Settlement Class Members seeking compensation must disclose and give credit for any compensation received through other proceedings or private out-of-class settlements in relation to Polyether Polyol Products, unless by such proceedings or private out-of-class settlements the Settlement Class Member's claim was released in its entirety, in which case the Settlement Class Member shall be deemed ineligible for any further compensation.

#### **DEFINITIONS**

- 4. The definitions set out in the Settlement Agreements apply to and are incorporated herein. Where a term is defined in both the Settlement Agreements and in this Distribution Protocol, the definition in this Distribution Protocol shall govern.
- 5. For the purpose of this Distribution Protocol, the following definitions apply:
  - (a) Aggregate Purchases means the aggregate amount paid by a Settlement Class Member for Polyether Polyol Products, less any taxes, delivery or shipping charges, rebates, or other form of discounts.
  - (b) Claim means the paper or electronic form that a Settlement Class Member must complete and submit before the Claims Filing Deadline in order to be considered for settlement benefits under this Distribution Protocol.
  - (c) Claims Administrator means the firm appointed by the Ontario Court to administer the Settlement Amounts in accordance with the provisions of the Settlement Agreements and the Distribution Protocol, and any employees of such firm.

- (d) Claims Filing Deadline means the date by which Claims (and any required supporting documentation) must be postmarked or electronically submitted in order for Settlement Class Members to be considered for settlement benefits under this Distribution Protocol, which date shall be four (4) months after the first publication of the notice advising Settlement Class Members of the claims process.
- (e) Class Period means from January 1, 1999 and December 31, 2004.
- (f) Decision Notice shall have the meaning attributed to it in paragraph 38.
- (g) Net Settlement Amount means the aggregate of the Settlement Amounts recovered pursuant to the Settlement Agreements, plus any costs awards and accrued interest, less:
  - (i) Class Counsel Fees as approved by the Ontario Court or Quebec Court;
  - (ii) Administration Expenses;
  - (iii) taxes accruable with respect to the income earned on the settlement funds prior to distribution (including interest and penalties); and
  - (iv) any other deductions approved by the Ontario Court.
- (h) Notional Entitlement means the total value assigned to Settlement Class Member's purchases of Polyether Polyol Products by the Claims Administrator, in accordance with paragraphs 6 to 13 below.
- (i) Online Claim Portal means a web-based portal created and maintained by the Claims Administrator in accordance with paragraphs 23 to 25 below.

- (j) Polyether Polyol Products means:
  - (i) polyether polyols, monomeric or polymeric diphenylmethane diisocyanate ("MDI") and toluene Diisocyanate ("TDI") purchased in Canada during the Class Period; and

- 5 -

- (ii) polyether polyol systems purchased from the Defendants in Canada during the Class Period.
- (k) Settlement Agreements has the meaning attributed to it in paragraph 1.
- (l) Settlement Class Members means all persons who purchased Polyether Polyol

  Products. The following persons are excluded:
  - (i) each Defendant, Rhodia, Rhodia Inc., and Rhodia Canada Inc., and their respective parents, employees, subsidiaries, affiliates, officers and directors; and
  - (ii) persons who validly opted-out of the Proceedings.
- (m) Settlement Website means the website maintained by or on behalf of the Claims

  Administrator for the purposes of providing Settlement Class Members with

  information on the Settlement Agreements, the Distribution Protocol, and the

  claims process, and access to the Online Claim Portal.

## DISTRIBUTION OF NET SETTLEMENT FUNDS Calculation of Payments

6. The Net Settlement Amount will be distributed to qualifying Settlement Class Members pro rata (proportionally) based on the value of the qualifying Settlement Class Member's Notional Entitlement as against the value of all qualifying Settlement Class Members' Notional Entitlement.

- 7. For the purposes of the *pro rata* distribution, a Settlement Class Member's Notional Entitlement will be calculated based on:
  - (a) the value of the Aggregate Purchases (see paragraph 8);
  - (b) the categorization of the Settlement Class Member (see paragraphs 9 to 11);
  - (c) the type of Polyether Polyol Products purchased (see paragraph 12); and
  - (d) the date of purchase (see paragraph 13).

#### (a) The value of the Aggregate Purchases

8. The Claims Administrator will calculate the value of the Settlement Class Member's Aggregate Purchases based on the information disclosed in the Defendants' sales information and/or information provided by the Settlement Class Member as part of the claims process.

#### (b) The categorization of the Settlement Class Member

- 9. Settlement Class Members will be categorized into the following purchaser groups based on their position in the distribution chain:
  - (a) Direct Purchaser means a Settlement Class Member who purchased Polyether

    Polyol Products from a manufacturer for either (i) its own use and not for

    commercial resale; or (ii) inclusion in a product for commercial resale;

- (b) Distributor means a Settlement Class Member who purchased Polyether Polyol

  Products and resold the Polyether Polyol Products without further processing and
  without including them in any product; and
- (c) Indirect Purchaser means a Settlement Class Member who purchased Polyether

  Polyol Products from a Distributor.
- 10. Settlement Class Members may fall into more than one category.
- 11. For the purposes of calculating a Settlement Class Member's Notional Entitlement, the following values will be applied to account for the capacity in which the Polyether Polyol Products were purchased:
  - (a) Aggregate Purchases made in the capacity of a Direct Purchaser will be valued at 100%;
  - (b) Aggregate Purchases made in the capacity of a Distributor will be valued at 25%;
    and
  - (a) Aggregate Purchases made in the capacity of an Indirect Purchaser will be valued at 75%.

#### (c) The type of Polyether Polyol Products

- 12. For the purposes of calculating a Settlement Class Member's Notional Entitlement, the following values will be applied to account for the type of Polyether Polyol Product:
  - (a) Aggregate Purchases attributable to purchases of MDI, TDI and/or polyether polyols will be valued at 100%; and

(b) Aggregate Purchases attributable to purchases of polyether polyol systems will be valued at 50%.

#### (d) The date of purchase

- 13. For the purposes of calculating a Settlement Class Member's Notional Entitlement, the following values will be applied to account for the date the Polyether Polyol Products were purchased:
  - (a) Aggregate Purchases made between January 1, 2002 and December 31, 2003 will be valued at 100%; and
  - (b) Aggregate Purchases made between January 1, 1999 to December 31, 2001 and January 1, 2004 to December 31, 2004 will be valued at 30%.

#### Sample Calculations

14. If a Distributor purchased \$1,000,000 of MDI and TDI in 2001, its Notional Entitlement would be calculated as follows.

\$1,000,000 (representing the Aggregate Purchases of MDI and TDI) x 0.25 (representing the categorization of the purchaser as a Distributor) x 1.0 (representing the type of Polyether Polyol Product purchased) x 0.3 (representing the timing of the purchase) = \$75,000

15. If an Indirect Purchaser purchased \$2,000,000 worth of polyether polyol systems in 2002 and \$5,000,000 worth of polyether polyols in 2004, its Notional Entitlement would be calculated as follows.

\$2,000,000 (representing the Aggregate Purchases of polyether polyol systems) x 0.75 (representing the categorization of the purchaser as an Indirect Purchaser) x 0.5 (representing the type of

543

Polyether Polyol Product purchased) x 1.0 (representing the timing of the purchase) = \$750,000

plus

\$5,000,000 (representing the Aggregate Purchases of polyether polyols) x 0.75 (representing the categorization of the purchaser as an Indirect Purchaser) x 1.0 (representing the type of Polyether Polyol Product purchased) x 0.3 (representing the timing of the purchase) = \$1,125,000

#### Distribution

- 16. All eligible Settlement Class Members will receive a minimum payment of \$20. The \$20 valuation target is not an estimate of any damages suffered. It is a minimum administrative threshold designed to maintain a feasible economic and administrative platform for the settlement distribution.
- 17. In consultation with Class Counsel, the Claims Administrator can seek directions from the Ontario Court with respect to the distribution of the Net Settlement Amount to ensure a fair and cost effective distribution of the Net Settlement Amount.

#### THE CLAIMS PROCESS

#### The Claim

#### All Settlement Class Members

- 18. All Settlement Class Members will be required to provide the following information:
  - (a) name and contact information;
  - (b) purchase information in accordance with paragraphs 19 to 20 or 21, as applicable;
  - (c) disclosure about whether the Settlement Class Member or any entity related to the Settlement Class Member has received compensation through other proceedings

or private out-of-class settlements and/or provided a release in respect of any of Polyether Polyol Products, and provide details of the compensation received and the claims released;

- (d) authorization to the Claims Administrator to contact the Settlement Class Member or its representative, as the Claims Administrator deems appropriate for more information and/or to audit the Claim;
- (e) a declaration that the information submitted in the Claim is true and correct; and
- (f) if the Claim is submitted by a third-party on behalf of a Settlement Class Member (including a parent company claiming on behalf of a subsidiary or affiliate), the third-party must provide a signed statement from that Settlement Class Member at the time the Claim is filed authorizing the third-party to file the Claim on its behalf.

#### Settlement Class Members relying on Defendant sales data

- 19. Where the Defendants have provided sales information in respect of the Settlement Class Member, the required fields in the Online Claim Portal will be pre-populated with the information provided by the Defendants. The following categories of information shall be provided separately:
  - (a) for the period between January 1, 2002 and December 31, 2003:
    - (i) the value of Aggregate Purchases attributable to polyether polyol systems;
    - (ii) the value of Aggregate Purchases attributable to MDI, TDI or Polyols;
  - (b) for the remainder of the Class Period:

- (i) the value of Aggregate Purchases attributable to polyether polyol systems; and
- (ii) the value of Aggregate Purchases attributable to MDI, TDI or Polyols.
- 20. The Settlement Class Member will be given the opportunity to agree with the purchase information provided by the Defendants. Where the Settlement Class Member agrees with the purchase information provided by the Defendants, no further purchase information is required from the Settlement Class Member. However, the Settlement Class Member must identify which purchases it made in the capacity of a Direct Purchaser or a Distributor.

#### Settlement Class Members claiming for purchases not substantiated by Defendant sales data

- 21. Where the Defendants have not provided sales information in respect of a Settlement Class Member and/or the Settlement Class Member is claiming for purchases of Polyether Polyol Products in addition to those substantiated by the Defendants' data, the Settlement Class Member must provide a declaration attesting to the information described in paragraph 19(a) and (b) above. The Settlement Class Member must also identify which purchases it made in the capacity of a Direct Purchaser, Distributor or Indirect Purchaser.
- Where a Settlement Class Member has purchase records for at least two years during the Class Period, or sales information from the Defendant Bayer, the Settlement Class Member can use such records to extrapolate its purchases of Polyether Polyol Products for the remainder of the Class Period.

#### The Online Claim Portal

- 23. The Claims Administrator shall create an Online Claim Portal that Settlement Class Members can access in order to file a Claim and shall provide the necessary administration support to enable Settlement Class Members to do so.
- 24. The Online Claim Portal shall contain fields that require the Settlement Class Member to provide all applicable information required as part of the Claim, in accordance with paragraphs 18 to 22.
- 25. The Claims Administrator shall develop procedures for tracking and recording in an electronic format the following information, as it is entered into the Online Claim Portal or provided by Settlement Class Members who file hardcopy Claims in accordance with paragraph 28 below:
  - (a) names, addresses, and purchase data of the Settlement Class Members;
  - (b) supporting documents provided by Settlement Class Members as part of the audit process; and
  - (c) any other information that might be useful in the claims administration process.

#### The Claims Filing Process

26. Where a Settlement Class Member has been identified by the Defendants, the Claims

Administrator shall provide to the Settlement Class Member, in writing, by e-mail or regular mail, his, her or its personal user name and password to permit that Settlement Class Member access to the Online Claim Portal.

- 27. Settlement Class Members will be encouraged to complete and submit a Claim electronically using the Online Claim Portal. Subject to paragraphs 28 and 36, or further order of the Ontario Court, Claims must be submitted on the Online Claim Portal on or before the Claim Filing Deadline.
- 28. If a Settlement Class Member does not have internet access or is otherwise unable to submit a Claim using the Online Claim Portal, the Settlement Class Member can register over the telephone with the Claims Administrator and the Claims Administrator shall send the Settlement Class Member a hardcopy claim form by mail. Subject to paragraph 36 or further order of the Ontario Court, the completed and executed hardcopy Claim must be submitted to the Claims Administrator postmarked no later than the Claims Filing Deadline.

#### Assistance in Filing a Claim

- 29. Settlement Class Members can contact the Claims Administrator or Class Counsel, at no charge, with questions about how to complete a Claim.
- 30. Settlement Class Members may utilize third-party claims services, a lawyer of their own choosing, or similar services to file Claims. If a Settlement Class Member chooses to use a third-party claims service, a lawyer of their own choosing, or similar services, the Settlement Class Members will be responsible for any and all expenses incurred in doing so.

#### Audits

31. Where the Settlement Class Member purchased Polyether Polyol Products directly from a Defendant and the Defendant provided sales information in respect of that Settlement

Class Member, the Defendant's sales information shall be *prima facie* proof of the Settlement Class Member's Polyether Polyol Product purchases and those purchases shall not be subject to an audit.

- 32. Where the Settlement Class Member was not identified by Defendants and/or is claiming for additional purchasers, the Claims Administrator shall audit:
  - (i) a random selection of at least 10% of Claims; and
  - (ii) Claims representing the top 15% of Claims (measured by Notional Entitlement).
- 33. The Claims Administrator shall notify the Settlement Class Member, by email or by regular mail, that the Settlement Class Member's Claim is the subject of an audit and the requirement to provide documentary proof:
  - (a) Proof of purchase might include invoices, receipts, delivery or packing slips, purchase records, historical accounting records, credit card statements, bank statements, cancelled cheques, wire transfer confirmations, or comparable verification that is acceptable to the Claims Administrator.
  - (b) where a Settlement Class Member has extrapolated its purchase records in accordance with paragraph 22 above, the Settlement Class Member must provide a declaration explaining the basis for and calculation of the extrapolation of purchases.
- 34. At its sole discretion, the Claims Administrator can elect to audit any Claim and can reject a Claim, in whole or in part, where, in the Claims Administrator's view, the

Settlement Class Member has submitted insufficient or false information or has otherwise engaged in fraudulent conduct.

#### **Deficiencies**

35. If, during claims processing, the Claims Administrator finds that deficiencies exist in a Claim or other required information, the Claims Administrator shall notify the Settlement Class Member, by email or regular mail, of the deficiencies. The Claims Administrator shall allow the Settlement Class Member thirty (30) days from the date of such notice to correct the deficiencies. If the deficiencies are not corrected within the thirty (30) day period, the Claims Administrator may reject the Claim. The Online Claim Portal shall be designed so as to minimize the possibility of deficient claims.

#### Adjustments to Claims Process and Extension of the Claims Filing Deadline

By agreement between the Claims Administrator and Class Counsel, the Claims Filing

Deadline may be extended and the Claims Administrator may adjust the claims process.

Class Counsel and the Claims Administrator shall agree to extend the Claims Filing

Deadline and/or adjust the claims process if, in their opinions, doing so will not adversely

affect the fair and efficient administration of the Net Settlement Amount and it is in the

best interests of the Settlement Class Members to do so.

#### Claims Administrator's Decision

- 37. In respect of each Settlement Class Member who has filed a Claim in accordance with this Distribution Protocol, the Claims Administrator shall:
  - (a) decide whether the Settlement Class Member is eligible to receive settlement benefits payable out of the Net Settlement Amount in accordance with the

- Settlement Agreements, orders of the Ontario Court and this Distribution Protocol; and
- (b) make a determination of the value of the Settlement Class Member's Notional Entitlement in respect of which the Settlement Class Member is entitled to settlement benefits in accordance with the Settlement Agreements, orders of the Ontario Court and this Distribution Protocol.
- 38. The Claims Administrator shall send to the Settlement Class Member, by email or regular mail, a decision as to the approval or rejection of the Claim and the determination of the Notional Entitlement (the "Decision Notice"). Where the Claims Administrator has rejected all or part of the Claim of the Settlement Class Member (or recategorized any purchases), the Claims Administrator shall include in the Decision Notice its grounds for rejecting or recategorizing all or part of the Claim.
- 39. The Claims Administrator's decision will be binding upon the Settlement Class Member, subject to the Settlement Class Member's right to appeal, as outlined in paragraphs 40 to 47 below.

#### Appeal of the Claims Administrator's Decision

- 40. The right to appeal is limited to circumstances where the dispute as to the value of the Notional Entitlement is greater than \$100,000.
- 41. Appeals must be submitted within thirty (30) days from the date of the Decision Notice.
- 42. The following grounds shall not be grounds for appeal:

- (a) the refusal of the Claims Administrator to accept a Claim postmarked or electronically submitted after the Claims Filing Deadline;
- (b) the refusal of the Claims Administrator to accept a Claim where the Settlement Class Member has not cooperated with the Claims Administrator in respect of any audit conducted by the Claims Administrator in respect of that Settlement Class Member's Claim; or
- (c) the refusal of the Claims Administrator to accept a Claim where the Settlement Class Member did not declare that the information submitted in the Claim is true and correct.
- 43. Appeals will be determined by the Ontario Court or a third party designated by the Ontario Court.
- 44. Appeals will be on the basis of written submissions, supported by the documentation provided by the Settlement Class Member as part of the claims process. Settlement Class Members are not permitted to provide any new documentation as part of the appeal. Any new documentation provided as part of the appeal will not be provided to the Ontario Court or its designee for consideration.
- 45. The Claims Administrator must provide to the Ontario Court a copy of the documentation provided by the Settlement Class Member as a part of the claims process, the Decision Notice, and any other information that might be reasonably useful in the determination of the appeal, and make written submissions to the Ontario Court or its designee as is reasonably necessary.

- 46. Notwithstanding the foregoing, the Ontario Court or its designee, acting in its sole discretion, can request oral submissions (to be provided via teleconference or videoconference, as requested by the Ontario Court or its designee) from the Settlement Class Member and/or Claims Administrator.
- 47. The decision on the appeal is final and binding and shall not be subject to any further appeal or review whatsoever.

#### Payment of Claims

- 48. As soon as practicable after the claims evaluations and any appeals are completed, the Claims Administrator shall:
  - (a) report to Class Counsel the particulars of the proposed distribution to each eligible

    Settlement Class Member; and
  - (b) make arrangements to pay approved Claims by cheque.
- 49. To the extent that the full Net Settlement Amount is not paid out due to uncashed cheques, residual interest or otherwise, such monies shall be paid to Pro Bono Canada if the amount is equal or less than \$10,000. For distribution of any amount above \$10,000, further direction of the Ontario Court shall be sought.
- The payment to Pro Bono Canada shall be less any amounts payable to the Fonds d'aide aux actions collectives, pursuant to section 42 of the Act respecting the Fonds d'aide aux actions collectives, CQLR c. F-3.2.0.1.1 and calculated in accordance with Article 1. (1°) of the Regulation respecting the percentage withheld by the Fonds d'aide aux actions collectives, R.S.Q. c. F-3.2.0.1.1, r. 2. For the purposes of calculating the amount

payable to the Fonds d'aide aux actions collectives, 23.6% of the payment to Pro Bono Canada will be notionally allocated to Quebec.

#### THE CLAIMS ADMINISTRATOR'S DUTIES AND RESPONSIBILITIES

#### Supervisory Powers of the Ontario Court

51. The Claims Administrator shall administer the Settlement Agreements and this Distribution Protocol under the ongoing authority and supervision of the Ontario Court.

#### **Investment of Settlement Amounts**

The Settlement Amounts shall be held in guaranteed investment vehicle, liquid money market account or equivalent security with a rating equivalent to or better than that of a Canadian Schedule I bank (a bank listed in Schedule I of the Bank Act, SC 1991, c 46) held at a Canadian financial institution.

#### Communication, Languages and Translation

- Where a Claim is filed by a third-party claims agent or lawyer on behalf of a Settlement Class Member, unless the Settlement Class Member requests otherwise, all communications shall be made to the third-party claims agent or lawyer.
- 54. The Claims Administrator shall establish a toll-free number for calls from Canada.
- 55. The Claims Administrator shall dedicate sufficient personnel to respond to Settlement Class Members' inquiries in English or French, as the Settlement Class Member elects.

<sup>23.6%</sup> represents that portion of the Canadian population that resides in Quebec based on information from Statistics Canada's website.

56. All written communications from the Claims Administrator to a Settlement Class Member shall be transmitted via email if an email address has been provided, or if an email address has not been provided, by regular mail.

## Undeliverable Mail

- 57. The Claims Administrator shall have no responsibility for locating Settlement Class Members for any mailing returned to the Claims Administrator as undeliverable.
- 58. The Claims Administrator shall have the discretion, but is not required, to reissue payments to Settlement Class Member returned as undeliverable under such policies and procedures as the Claims Administrator deems appropriate. Any costs associated with locating current address information for the Settlement Class Member shall be deducted from that Settlement Class Member's settlement benefits.
- 59. Where a Settlement Class Member who is entitled to payment of greater than \$20 requests a cheque be reissued, \$15 shall be deducted from that Settlement Class Member's settlement benefits representing the costs of reissuing payment. Subject to the sole discretion of the Claims Administrator, payments for \$20 will not be reissued.

## Fraudulent Claims

60. The Claims Administrator shall develop and implement processes to detect possible fraudulent conduct, including monitoring claims for unusual activity and multiple claims being filed from the same address.

## Taxes

61. The Claims Administrator shall take all reasonable steps to minimize the imposition of taxes upon the Net Settlement Amount and shall pay any taxes imposed on such monies out of the Net Settlement Amount.

## Reporting

- 62. The Claims Administrator shall provide regular reports to Class Counsel regarding the administration.
- 63. The Claims Administrator shall provide any reports requested by the Ontario Court.

## Preservation and Disposition of Claim Submissions

64. The Claims Administrator shall preserve, in hard copy or electronic form, as the Claims Administrator deems appropriate, the submissions relating to a Claim, until two (2) years after all settlement monies or court awards have been paid out to Settlement Class Members, and at such time shall destroy the submissions by shredding, deleting, or such other means as will render the materials permanently illegible.

## Assistance to the Claims Administrator

65. The Claims Administrator shall have the discretion to enter into such contracts and obtain financial, accounting, and other expert assistance as are reasonably necessary in the implementation of the Settlement Agreements and this Distribution Protocol, provided that related expenses are paid out of the Claims Administrator's fees or approved by the Ontario Court in advance.

## Confidentiality

66. All information received from the Defendants or the Settlement Class Members is collected, used, and retained by the Claims Administrator pursuant to the *Personal Information Protection and Electronic Documents Act*, SC 2000 c 5 for the purposes of administering the Settlement Agreements, including evaluating the Settlement Class Member's eligibility status under the Settlement Agreements. The information provided by the Settlement Class Member is strictly private and confidential and will not be

556

disclosed without the express written consent of the Settlement Class Member, except in accordance with the Settlement Agreements, orders of the Ontario Court and/or this Distribution Protocol.

## **TAB 3**

Court File No. CV-12-9545-00CL

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE	)	TUESDAY, THE 29 <sup>TH</sup> DAY
	)	
	)	OF MAY, 2018

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD. 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

## ORDER

THIS MOTION made by the Applicants pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. c-36, as amended (the "CCAA") for an order, *inter alia*, extending the stay of proceedings in respect of the Applicants to and including November 30, 2018 was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the Affidavit of Tony Vallecoccia sworn May 22, 2018 and the exhibits thereto (the "Vallecoccia Affidavit") and the Eighteenth Report of Deloitte Restructuring Inc. (formerly Deloitte & Touche Inc.) (the "Eighteenth Report") in its capacity as the Courtappointed monitor (the "Monitor") of the Applicants, and on hearing the submissions of counsel for the Applicants, the Monitor and all other counsel listed on the counsel slip, no one appearing for any other person on the service list, although properly served as appears from the Affidavit of Service of • sworn •, filed;

## **SERVICE**

1. **THIS COURT ORDERS** that the time for service of the Notice of Motion, the Motion Record and the Eighteenth Report is hereby abridged and validated and this Motion is properly returnable today without further service or notice thereof.

## **DEFINITIONS**

2. **THIS COURT ORDERS** that capitalized terms not otherwise defined in this Order shall have the meaning set out in the Eighteenth Report.

## **STAY EXTENSION**

3. **THIS COURT ORDERS** that the Stay Period granted under the Initial Order of Justice Newbould dated January 12, 2012 (the "**Initial Order**") and as subsequently extended by, *inter alia*, the Order of the Honourable Mr. Justice Myers, dated November 24, 2017, is hereby extended from May 31, 2018 to and including November 30, 2018.

## MONITOR'S REPORT, ACTIONS AND FEES

- 4. **THIS COURT ORDERS** that the Eighteenth Report and actions, decisions and conduct of the Monitor as set out in the Eighteenth Report are hereby authorized and approved.
- 5. **THIS COURT ORDERS** that the fees and disbursements of the Monitor and its legal counsel as set out in the Eighteenth Report, the Affidavit of •, sworn •, and the exhibits attached thereto, are hereby authorized and approved.

R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z COMPANIES' CREDITORS ARRANGEMENT ACT, SPONGE & FOAM PRODUCTS LTD MATTER OF THE IN THE

Court File No. CV-12-9545-00CL

## SUPERIOR COURT OF JUSTICE ONTARIO

## COMMERCIAL LIST

Proceeding commenced at TORONTO

## ORDER

# BLANEY McMURTRY LLP

Barristers and Solicitors

Suite 1500 - 2 Queen Street East

Toronto, ON M5C 3G5

David T. Ullmann LSUC #423571

(416) 596-4289 (416) 594-2437 Tel: Fax:

Alexandra Teodorescu LSUC #63899D

(416) 596-4279 (416) 593-5437 Tel: Fax:

Lawyers for the Applicants

Court File No. CV-12-9545-00CL

## SUPERIOR COURT OF JUSTICE ONTARIO

## COMMERCIAL LIST

Proceeding commenced at TORONTO

# (Re: Stay Extension, Returnable May 29, 2018) MOTION RECORD OF THE APPLICANTS

# BLANEY McMURTRY LLP

Suite 1500 - 2 Queen Street East Barristers and Solicitors

Toronto, ON M5C 3G5

David T. Ullmann LSUC #423571

(416) 596-4289 (416) 594-2437 Tel: Fax:

Alexandra Teodorescu LSUC #63899D

(416) 596-4279 (416) 593-5437 Tel:

Lawyers for the Applicants

Fax:

561

## Tab 3

Court File No. CV-12-9545-00CL

ŧ

## ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

THE HONOURABLE MR.	)	TUESDAY, THE 29TH DAY
SOUTH A J. WILTON-SOZOT	)	OF MAY, 2018
IN THE MATTER OF THE COM R.S.C. 1985,	<i>(PANIES' (</i> c.C-36, AS	CREDITORS ARRANGEMENT ACT, AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD. 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

(the "Applicants")

## **ORDER**

THIS MOTION made by the Applicants pursuant to the Companies' Creditors Arrangement Act, R.S.C. 1985, c. c-36, as amended (the "CCAA") for an order, inter alia, extending the stay of proceedings in respect of the Applicants to and including November 30, 2018 was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the Affidavit of Tony Vallecoccia sworn May 22, 2018 and the exhibits thereto (the "Vallecoccia Affidavit") and the Eighteenth Report of Deloitte Restructuring Inc. (formerly Deloitte & Touche Inc.) (the "Eighteenth Report") in its capacity as the Court-appointed monitor (the "Monitor") of the Applicants, and on hearing the submissions of counsel for the Applicants, the Monitor and all other counsel listed on the counsel slip, no one appearing for any other person on the service list, although properly served as appears from the Affidavit of Service of Ariyana Botejue sworn May 23, 2018, filed;

## **SERVICE**

 THIS COURT ORDERS that the time for service of the Notice of Motion, the Motion Record and the Eighteenth Report is hereby abridged and validated and this Motion is properly returnable today without further service or notice thereof.

## **DEFINITIONS**

2. THIS COURT ORDERS that capitalized terms not otherwise defined in this Order shall have the meaning set out in the Eighteenth Report or the Order of the Court dated June 15, 2012 (the "Claims Solicitation Procedure Order").

## **STAY EXTENSION**

3. THIS COURT ORDERS that the Stay Period granted under the Initial Order of Justice Newbould dated January 12, 2012 (the "Initial Order") and as subsequently extended by, *inter alia*, the Order of the Honourable Mr. Justice Myers, dated November 24, 2017, is hereby extended from May 31, 2018 to and including November 30, 2018.

## **INTERIM DISTRIBUTIONS**

- 4. THIS COURT ORDERS that the Monitor is hereby authorized to make an interim Distribution of the Valle Foam Proceeds in the amount of \$5,600,000 to the Valle Foam Creditors holding Proven Claims on a pro rata pari parssu basis.
- 5. THIS COURT ORDERS that the Monitor is hereby authorized to make an interim Distribution of the Domfoam Proceeds in the amount of \$3,470,000 to the Domfoam

Creditors holding Proven Claims on a pro rata pari parssu basis.

6. **THIS COURT ORDERS** that the Monitor is hereby authorized to make an interim Distribution of the A-Z Foam Proceeds in the amount of \$708,000 to the A-Z Foam Creditors holding Proven Claims on a pro rata pari parssu basis.

## MONITOR'S REPORT, ACTIONS AND FEES

- 7. **THIS COURT ORDERS** that the Eighteenth Report and the actions, decisions and conduct of the Monitor as set out in the Eighteenth Report are hereby authorized and approved.
- 8. THIS COURT ORDERS that the fees and disbursements of the Monitor and its legal counsel as set out in the Eighteenth Report, the Affidavit of Paul Casey sworn on May 24, 2018 and the Affidavit of Grant B. Moffat sworn on May 23, 2018, and the exhibits attached thereto, are hereby authorized and approved.
- 9. THIS COURT HEREBY requests the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States, to give effect to this Order and to assist the Applicants, the Monitor and their respective agents in carrying out the terms of this order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such Orders and to provide such assistance to the Applicants and to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order, or to assist the Applicants and the Monitor and their respective agents in carrying out the terms of this Order.
- 10. THIS COURT ORDERS that each of the Applicants and the Monitor be at liberty

and are hereby authorized and empowered to apply to any Court, tribunal, regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order.

Wilden-KM J

ENTERED AT / INSCRIT A TORONTO ON / BOOK NO: LE / DANS LE REGISTRE NO:

MAY 2 9 2018

DER / PAR:

AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z c. C-36, THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, SPONGE & FOAM PRODUCTS LTD.

Court File No. CV-12-9545-00CL

## SUPERIOR COURT OF JUSTICE ONTARIO

## COMMERCIAL LIST

Proceeding commenced at TORONTO

## ORDER

# BLANEY McMURTRY LLP

Barristers and Solicitors Suite 1500 - 2 Queen Street East

Toronto, ON M5C 3G5

David T. Ullmann LSUC #42357I

(416) 596-4289 (416) 594-2437 Tel: Fax:

Alexandra Teodorescu LSUC #63899D

(416) 596-4279 (416) 593-5437 Lawyers for the Applicants

## Tab 4

Court File No. CV-12-9545-00CL

## ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

## IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD., 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

**APPLICANTS** 

SEVENTH REPORT OF THE MONITOR DATED JULY 12, 2013

## **EXHIBITS**

EXHIBIT A: Initial Order dated January 12, 2012

EXHIBIT B: U.S. Recognition Order dated February 24, 2012

EXHIBIT C: Claims Solicitation Procedure Order dated June 15, 2012

EXHIBIT D: Canadian Class Action Settlement Agreement

EXHIBIT E: US Direct Purchaser Settlement Approval Order

EXHIBIT F: Trust Agreement

EXHIBIT G: Statement of Receipts and Disbursements for Valle Foam for the period March 29,

2012 to July 11, 2013

EXHIBIT H: Statement of Receipts and Disbursements for Domfoam for the period March 29,

2012 to July 11, 2013

EXHIBIT I: Statement of Receipts and Disbursements for A-Z Foam for the period March 29,

2012 to July 11, 2013

EXHIBIT J: Affidavit of Catherine Hristow of Deloitte & Touche Inc., sworn July 12, 2013

EXHIBIT K: Affidavit of Grant Moffat of Thornton Grout Finnigan LLP, sworn July11, 2013

## TABLE OF CONTENTS

INTRODUCTION	
TERMS OF REFERENCE	3
BACKGROUND	3
CLASS ACTIONS	4
URETHANE ANTITRUST LITIGATION	6
ACTIVITIES OF THE MONITOR	10
SALE OF COMPANIES' ASSETS	10
CLAIMS SOLICITATION PROCEDURE	11
CLAIMS DISALLOWED BY THE MONITOR	12
STATEMENTS OF CASH RECEIPTS AND DISBURSEMENTS	14
PROFESSIONAL FEES	15
ALLOCATION OF PROFESSIONAL FEES	16
EXTENSION OF THE STAY PERIOD	16
MONITOR'S RECOMMENDATIONS	16

## INTRODUCTION

- 1. By Order of the Court dated January 12, 2012 (the "Initial Order"), Valle Foam Industries (1995) Inc. ("Valle Foam"), Domfoam International Inc. ("Domfoam") and A-Z Sponge & Foam Products Ltd. ("A-Z Foam") (collectively, the "Applicants" or the "Companies"), obtained protection from their creditors pursuant to the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"). The CCAA proceeding with respect to the Applicants is referred to herein as the "CCAA Proceeding".
- Pursuant to the Initial Order, Deloitte & Touche Inc. was appointed monitor of the Applicants as part of the CCAA Proceeding (the "Monitor"). Pursuant to the Initial Order, all proceedings against the Applicants were stayed until February 10, 2012, or until such later date as this Court would order (the "Stay Period"). A copy of the Initial Order is attached hereto as Exhibit "A".
- 3. On July 1, 2013, Deloitte & Touche Inc. changed its name to Deloitte Restructuring Inc. (hereafter, "Deloitte").
- 4. As noted in the Monitor's Fourth Report to the Court dated June 12, 2012 (the "Fourth Report"), Valle Foam changed its name to 3113736 Canada Ltd. and Domfoam changed its name to 4362063 Canada Ltd. Throughout this Report, references to Valle Foam mean 3113736 Canada Ltd. and references to Domfoam mean 4362063 Canada Ltd.
- On January 23, 2012, the Monitor in its capacity as foreign representative of the Companies in the CCAA Proceeding filed with the United States Bankruptcy Court, Northern District of Ohio (Western Division) (the "U.S. Bankruptcy Court") a petition for recognition of the CCAA Proceeding as a foreign main proceeding pursuant to Chapter 15 of the U.S. Bankruptcy Code.
- 6. By Order of the U.S. Bankruptcy Court dated January 27, 2012 (the "Provisional U.S. Recognition Order"), all litigation in the United States against the Companies was stayed on a provisional basis (the "U.S. Litigation Stay") until February 10, 2012, and provided that the U.S. Litigation Stay would be automatically extended to correspond to any extension of the Stay Period in the CCAA Proceeding. By Order of the U.S. Bankruptcy Court dated February 24, 2012 (the "U.S. Recognition Order"), the CCAA Proceeding was recognized as a foreign main proceeding. A copy of the U.S. Recognition Order is attached hereto as Exhibit "B".

- 7. By Orders of the Court dated February 8, March 16, June 15, October 25, 2012 and February 28, 2013 (collectively, the "Extension Orders"), the Court has periodically extended the Stay Period, with the most recent extension expiring on July 31, 2013.
- 8. By Order of the Court dated June 15, 2012 (the "Claims Solicitation Procedure Order"), the Court authorized and approved the procedure (the "Claims Solicitation Procedure") to:
  - (i) identify and determine the validity of creditor claims against the Companies as at the date of the Initial Order, as well as any claims which arose subsequent to the date of the Initial Order (collectively, the "Claims"); and
  - (ii) identify claims against any of the current or former directors or officers (collectively, the "Directors and Officers") of the Companies (the "D&O Claims").

The deadline for submission for proofs of claim under the Claims Solicitation Procedure was August 31, 2012 (the "Claims Bar Date"). A copy of the Claims Solicitation Procedure Order is attached hereto as Exhibit "C".

- 9. The Initial Order together with related Court documents, the Notice to Creditors dated January 19, 2012, the Monitor's First Report to the Court dated January 25, 2012 ("First Report"), the Monitor's Second Report to the Court dated February 7, 2012 ("Second Report"), the Monitor's Third Report to the Court dated March 13, 2012 (the "Third Report"), the Monitor's Fourth Report to the Court dated June 12, 2012 ("Fourth Report"), the Monitor's Fifth Report to the Court ("Fifth Report") dated October 22, 2012 and the Monitor's Sixth Report to the Court ("Sixth Report") dated February 25, 2013 (collectively, the "Prior Reports"), have been posted on the Monitor's website at <a href="www.deloitte.com/ca/vallefoam">www.deloitte.com/ca/vallefoam</a> (the "Monitor's Website"). The Monitor has also established a toll free number at 1-855-601-6415 and a dedicated e-mail address at <a href="www.deloitte.ca">vallefoam@deloitte.ca</a> for creditors and other interested parties to contact the Monitor with questions or concerns regarding the CCAA Proceeding.
- 10. The purpose of this report (the "Seventh Report") is to update the Court with respect to the status of the Claims Solicitation Procedure and to provide the Court with the Monitor's recommendation with respect to the Companies' motion for an extension of the Stay Period to December 31, 2013.

## TERMS OF REFERENCE

- 11. In preparing the Seventh Report, the Monitor has relied upon unaudited financial information, the Companies' books and records, the financial information prepared by the Companies, and discussions with management ("Management") and legal counsel for the Companies.
- 12. Unless otherwise stated, all dollar amounts contained in this Seventh Report are expressed in Canadian dollars.
- Capitalized terms not otherwise defined in this Seventh Report are as defined in the Initial Order or the Claims Solicitation Procedure Order.

## BACKGROUND

- 14. The Companies operated together as one of Canada's leading and largest manufacturers and distributors of flexible polyurethane foam products from facilities located in Ontario, Quebec and British Columbia. The operations of Valle Foam and Domfoam historically comprised substantially all of the Companies' operations. A-Z Foam and Valle Foam are wholly owned subsidiaries of Domfoam.
- 15. Mr. Anthony Vallecoccia is the President and Chief Executive Officer of Domfoam, President of Valle Foam, and the sole officer and director of A-Z Foam.
- Other than security interests which may be claimed by certain equipment lessors, the Monitor is not aware of any secured creditors of the Companies. As at January 11, 2012, the total liabilities of Valle Foam, Domfoam and A-Z Foam, not including any claims pursuant to the Class Actions (as defined below), amounted to approximately \$11,218,000, \$11,339,000 and \$368,000 respectively.

## The Competition Bureau (Canada) Fines and Related Litigation

As set out in the First Report, both Domfoam and Valle Foam were charged with, and on January
5, 2012, pled guilty to, certain offences under the Competition Act, R.S.C. 1985, c C¬34 (the "Competition Act") arising from collusion with other manufacturers of slab foam and carpet

- underlay within Canada to lessen competition in the sale or supply of these products and by conspiring with other manufacturers to fix or control the price for these products.
- 18. Domfoam was fined a total of \$6.0 million and Valle Foam was fined a total of \$6.5 million. No fine was assessed against A-Z Foam as no charges were laid against A-Z Foam. In accordance with the terms of the sentence imposed, Valle Foam paid \$500,000 in partial payment of the fines imposed against it on the same day the guilty pleas were entered.
- 19. As a result of the foregoing, each of Valle Foam and Domfoam has an outstanding liability of \$6.0 million in fines payable to the Crown.
- 20. In accordance with the terms of the sentences imposed, Domfoam and Valle Foam are to each pay \$1.0 million on the 1<sup>st</sup> of January of each year, commencing in 2013 and ending in 2018.
- 21. As set out in the Affidavit of Tony Vallecoccia sworn January 11, 2012, (the "Vallecoccia Affidavit"), the Applicants disclosed their financial difficulties to the Crown prior to the entry of their guilty pleas and advised of the Applicants' intention to file for protection under the provisions of a Canadian insolvency regime.
- 22. The Monitor has been advised by the Companies that, as part of the plea arrangement with the Crown, certain officers and directors of the Companies are required to provide sworn testimony to the Competition Bureau.

## **CLASS ACTIONS**

- 23. The Monitor has been advised by the Applicants that some or all of the Applicants have been named as defendants in four class action lawsuits in Canada, and over two dozen class action lawsuits in the United States (together, the "Class Actions"), based upon allegations of price fixing by certain of the Applicants and other manufacturers in the slab foam industry.
- 24. The Canadian Class Actions consist of separate proceedings commenced in each of British Columbia (the "BC Proceeding"), Ontario (the "Ontario Proceeding") and Quebec (the "Quebec Proceeding"). The Canadian Class Actions advance joint and several claims against the Companies and certain other defendants or respondents on behalf of proposed classes

- comprised of all persons or entities who purchased polyurethane foam and polyurethane foam products in Canada from and after January 1, 1999.
- 25. Settlements have been reached with virtually all of the Plaintiffs in both the Canadian and U.S. Class Actions. The terms of the settlement in the Canadian Class Actions are set out in the Canadian Polyurethane Foam Class Actions National Settlement Agreement dated as of January 10, 2012 (the "Canadian Class Action Settlement Agreement"), a copy of which is attached as Exhibit "D". Under both the Canadian Class Action Settlement Agreement and the settlements in the US Class Actions, the Class Actions have been discontinued as against the Companies, provided that the Plaintiffs in the Class Actions may still assert their claims as creditors within the CCAA Proceeding in amounts to be determined. The Canadian Class Action Settlement Agreement is still subject to separate Court approvals in Ontario, British Columbia and Quebec.
- A joint hearing of the Ontario Superior Court of Justice ("Ontario Court") and the Supreme Court of British Columbia ("BC Court") was held on June 28, 2013 to partially certify the Ontario Proceeding and the BC Proceeding, solely for the purpose of approving the Canadian Class Action Settlement Agreement as well as the manner in which all members comprising the class under the Canadian Class Actions will be notified of the foregoing settlement. The Ontario and BC Courts requested that certain modifications be made to the proposed class notification procedure. Although the Plaintiffs filed amendments to the foregoing procedure with the Ontario and BC Courts on July 8, 2013, no Order has yet been issued in connection with the June 28, 2013 motion. The Monitor understands that the Ontario and BC Courts will likely schedule their settlement approval hearings in September or early October 2013.
- 27. The Monitor also understands that a similar partial certification motion took place on July 8, 2013 within the Quebec Proceeding, also for the purpose of approving the manner in which members comprising the class under the Canadian Class Actions will be notified of the foregoing settlement. The Monitor understands that a final order approving the class notification procedure has not yet been issued by the Quebec Superior Court. The Monitor also understands that a final settlement approval hearing in the Quebec Proceeding has been scheduled for September 24, 2013.
- 28. By Order of the United States District Court for the Northern District of Ohio Western Division dated June 21, 2013, the settlement reached between the Companies and the direct purchaser class in the US Class Actions was approved (the "US Direct Purchaser Settlement Approval

- Order"). A copy of the US Direct Purchaser Settlement Approval Order is attached as Exhibit "E". The Monitor has been advised by US counsel to the Companies that the settlement between the Companies and the indirect purchaser class in the United States has not yet been submitted for Court approval.
- 29. Given the issuance of the US Direct Purchaser Settlement Approval Order, the Monitor does not anticipate that the Companies will incur significant additional legal fees in connection with the settlement reached between the Companies and the direct purchaser class in the US Class Actions. The Monitor also does not anticipate that the Companies will incur significant additional legal fees in connection with the settlement with the US indirect purchaser class.
- 30. The Monitor understands that the Companies will continue to incur legal fees in connection with the Court approval process with respect to the Canadian Class Action Settlement as well as the Companies' various continuing obligations under that agreement. In addition, additional legal fees will be incurred by the Companies in connection with the proper valuation to be placed upon the proofs of claim filed by the Plaintiffs in the Canadian Class Actions and the proof of claim field by Revenu Quebec, each as described in more detail below.
- 31. The Canadian Class Action Settlement Agreement and the settlements reached in the US Class Actions require the Companies to make available for examination certain current and former officers and directors of the Companies. Certain officers and directors of the Companies were examined in the United States in January 2013. The Monitor has permitted the Companies to fund the legal fees and expenses of the Companies' former officers and directors in connection with such examinations to ensure that the settlements in the Canadian and US Class Actions are not jeopardized. The Monitor understands that the Companies' former directors and officers will incur additional legal fees and expenses in connection with certain pending examinations of such officers and directors in connection with both the Canadian and US Class Action Settlements, although the Monitor understands that such fees and expenses will not be significant.

### URETHANE ANTITRUST LITIGATION

32. The Monitor has been advised by the Companies that they are each claimants in a class action proceeding pending before the United States District Court for the District of Kansas under the caption In Re Urethane AntiTrust Litigation (the "US Urethane Proceedings"). As described in

more detail in Mr. Vallecoccia's affidavit sworn July 11, 2013, two separate settlements with BASF Corporation and Huntsman International LLC were approved by order of the United States District Court for the District of Kansas dated December 12, 2012. These settlements provide for payment to be made to the claimants in the US Urethane Proceedings. Certain of those funds have been received by the Companies as described in more detail below. The Monitor understands that judgment has been rendered against The Dow Chemical Company in the US Urethane Proceedings and that further payments may be received by the Companies pursuant to this judgment or a settlement thereof, as well as in relation to any additional judgments obtained or settlements reached in the US Urethane Proceedings.

- 33. The affidavit of Mr. Vallecoccia sworn July 11, 2013, provides that each of Domfoam, Valle Foam and A-Z Foam did not intend to sell to the purchaser of its assets its claim in the US Urethane Proceedings (the "Domfoam US Urethane Claim", the "Valle Foam US Urethane Claim", the "A-Z Foam US Urethane Claim" respectively and, collectively, the "US Urethane Claims"), and that the Urethane Claims remain assets of the Companies' estates. The Monitor was not involved in any of the negotiations between the Companies and the purchasers of their assets.
- Mr. Vallecoccia's affidavit sworn July 11, 2013 provides that the Domfoam US Urethane Claim was specifically excluded from the Domfoam assets purchased by 4037057 Canada Inc. Similarly, it appears that the the Valle Foam assets purchased by Fybon Industries Limited pursuant to the Asset Purchase Agreement dated February 22, 2012 would not include the Valle Foam US Urethane Claim since Valle Foam's accounts receivable were excluded from that transaction. As far as the Monitor is aware, neither 4037057 Canada Inc. or Fybon Industries Limited has asserted any claim to the Domfoam US Urethane Claim or the Valle Foam US Urethane Claim respectively. Accordingly, the net proceeds of the Domfoam US Urethane Claim and the Valle Foam US Urethane Claim should be available for distribution to the creditors of Domfoam and Valle Foam respectively.
- 35. Although Mr. Vallecoccia's affidavit sworn July 11, 2013 does indicate that A-Z Foam did not intend to sell the A-Z Foam US Urethane Claim to 0932916 BC Ltd. (the "A-Z Purchaser"), the Asset Purchase Agreement dated February 21, 2012 between A-Z Foam and the A-Z Purchaser does not specifically exclude the A-Z Foam US Urethane Claim from the assets purchased thereunder. As far as the Monitor is aware, the A-Z Purchaser has not asserted a claim

- to the A-Z Foam US Urethane Claim or any of the funds received by A-Z Foam to date pursuant to that claim.
- 36. Barring any claim to the A-Z Foam US Urethane Claim by the A-Z Purchaser, it appears that the net proceeds thereof should be available for distribution to the creditors of A-Z Foam.
- 37. Pursuant to section 4.2 of the Canadian Class Action Settlement Agreement, the Companies agreed to assign to the Canadian Class Action Plaintiffs the Companies' right to receive any proceeds from the US Urethane Proceedings, provided that such assignment is limited to the maximum amount of the first \$200,000 the Companies may receive thereunder.
- 38. Pursuant to a 2008 services agreement between the Companies and Refund Recovery Services, LLC (the "Services Agreement"), the Companies retained Refund Recovery Services, LLC to assist in asserting and recovering their claim in the US Urethane Proceedings in consideration of a fee equal to 25% of all funds paid to the Companies.
- 39. Thereafter, Enterprise Law Group was retained by Refund Recovery Services, LLC to assist in recovering the Valle Foam claim only in the US Urethane Proceedings. Subsequently, Lex Group, LLC, the successor to Refund Recovery Services, LLC assigned to Enterprise Law Group its rights under the Services Agreement to receive the 25% commission in respect of any funds paid to Valle Foam only pursuant to the US Urethane Proceedings.
- 40. In January 2013, the Companies' legal counsel received correspondence from Enterprise Law Group including a cheque in the amount of US\$331,928.29 for Valle Foam in respect of the U.S. Urethane Proceedings. The Monitor understands that no deduction was made from these funds in respect of the 25% commission payable pursuant to the Services Agreement. These funds were delivered to the Monitor. Based on the documentation made available to the Monitor, the Monitor determined that the foregoing payment is subject to a collection fee by Enterprise Law Group pursuant to the Services Agreement equal to 25% of the foregoing amount. That fee has now been paid to Enterprise Law Group.
- 41. Also in January 2013, the Companies' legal counsel received correspondence from Lex Group, LLC, enclosing cheques in the amount of US\$196,802.78 and US\$28,325.87 for Domfoam and A-Z Foam respectively, net of the 25% collection fees payable to Refund Recovery Services, LLC. The Applicants' legal counsel sent the cheques to the Monitor for Domfoam and A-Z Foam which were converted to CAD\$195,248.04 and \$27,821.67 respectively.

- 42. Section 4.2 (2) of the Canadian Class Action Settlement Agreement provides that any distribution of funds to the Companies from the US Urethane Proceedings up to the \$200,000 cap as described above (the "Assigned Urethane Funds"), shall be paid to the escrow agent appointed by the parties (the "Escrow Agent"). Robert Tanner, an Ontario solicitor, has been appointed as Escrow Agent pursuant to the trust agreement dated February 15, 2013 (the "Trust Agreement"), a copy of which is attached as Exhibit "F". The Canadian Class Action Settlement Agreement specifically provides that payment of the Assigned Urethane Funds is subject to any order of the Court in the CCAA Proceeding. The Companies have not made any representation that the foregoing assignment is valid or enforceable.
- 43. In the Sixth Report, the Monitor indicated that it did not object to the Assigned Urethane Funds being delivered to the Escrow Agent in accordance with the terms of the Canadian Class Action Settlement Agreement, provided such funds remain in the Escrow Agent's possession pending resolution of the validity and enforceability of the assignment of such funds to the Canadian Class Action Plaintiffs. The Monitor also indicated its concern that payment of such funds to the Canadian Class Action Plaintiffs would be preferential.
- 44. The Monitor has now had an opportunity to review the Trust Agreement. Under the terms of the Trust Agreement, the Escrow Agent appears to be entitled to charge against the Assigned Urethane Funds the fees and disbursements of the Escrow Agent. The Trust Agreement also appears to provide that the Escrow Agent may apply to Court for directions and/or for an interpleader order in the event there is a dispute with respect to entitlement to those funds.
- 45. Given that the entitlement to the Assigned Urethane Funds by the Canadian Class Action Plaintiffs has not yet been determined, it does not appear appropriate at this stage that such funds be paid to the Escrow Agent, thus becoming subject to various charges, including the costs of any Court attendances, that may arise in connection with any dispute as to entitlement to the Assigned Urethane Funds.
- 46. In addition, since it appears that it will be necessary to obtain an Order of this Court determining entitlement to the Assigned Urethane Funds, it appears to the Monitor to be unnecessarily duplicative to deliver such funds to the Escrow Agent, who will incur costs in connection with steps which will likely be taken by the Companies and the Monitor in any event.

47. Accordingly, it is the Monitor's view that it would be in the best interests of the estate that the Assigned Urethane Funds be retained by the Monitor pending the resolution of the Canadian Class Action Plaintiffs' claims against the Companies. The Canadian Class Action Settlement Agreement provides, and the Monitor has confirmed with the Companies' counsel in the Canadian Class Actions, that the retainer of the Assigned Urethane Funds by the Monitor will not invalidate the Canadian Class Action Settlement Agreement.

## ACTIVITIES OF THE MONITOR

- 48. The Monitor has undertaken the following activities since the date of the Monitor's Sixth Report:
  - (a) met with the Companies' Class Action legal counsel with a view to resolving the value of proofs of claim filed by the Canadian Class Action claimants pursuant to the Claims Solicitation Procedure;
  - (b) engaged in discussions with Revenu Quebec regarding the value of its proof of claim, described in more detail below; and
  - (c) monitored the financial position of the Applicants, and prepared this Seventh Report.

## SALE OF COMPANIES' ASSETS

49. As described in more detail in the Fourth Report, all of the Companies' assets have been sold pursuant to separate transactions. The Monitor is in receipt of the proceeds of sale of these transactions. The Monitor is not aware of any additional assets of the Companies which may be realized upon for the benefit of the Companies' creditors, other than certain accounts receivable of Valle Foam (book value approximately \$2.0 million) of which the Applicants and the Applicants' legal counsel are pursuing collection, and any future payments which may be received in the US Urethane Proceedings. The Monitor has not reviewed the individual accounts receivable invoices and has not made a determination of the collectability of the remaining accounts receivable.

## CLAIMS SOLICITATION PROCEDURE

- 50. The Monitor, with the assistance of the Companies, reviewed all Proofs of Claim delivered to the Monitor by the Claims Bar Date.
- 51. Listed below is a summary of the pre-filing and post-filing Proofs of Claim which have been admitted by the Monitor and those Proofs of Claim that are pending resolution, which are discussed later in this report.

Company	Pre-Filing (Admitted)	Post-Filing (Admitted)	Pending Resolution	Total
Valle				
Foam	\$ 8,525,114.13 (1)	\$ 758,821.94	\$ 292,500,000.00	\$ 301,783,936.07
Domfoam	\$ 8,045,747.98	\$ -	\$ 295,412,919.10	\$ 303,458,667.08
A-Z Foam	\$ 82,398.29	\$ 20,610.40	\$ 292,500,000,00	\$ 292,603,008,69

Note (1): The amount has been reduced by \$4,451.28 as a result of an amended claim being filed reducing the original amount of the claim.

- 52. Of the approximately \$759,000 in Claims filed against Valle Foam for post-filing obligations, approximately \$757,000 is in respect of severance and termination claims by employees who were terminated post January 12, 2012.
- 53. No Proof of Claim forms were filed by the U.S. Class Action claimants.
- 54. Included in the amount of admitted Claims against both Valle Foam and Domfoam are the claims submitted by the Competition Bureau in the amount of \$6.0 million respectively.
- 55. In accordance with the Claims Solicitation Procedure Order, the Monitor sent by September 21, 2012 a Notice of Revision or Disallowance in respect of any Proof of Claim disputed by the Monitor in whole or in part indicating the reasons for the revision or disallowance. Those Proofs of Claim received by the Claims Bar Date in respect of which the Monitor did not send a Notice of Revision or Disallowance by September 21, 2012 are deemed to be Proven Claims.
- Pursuant to the Claims Solicitation Procedure Order, any Creditor who receives a Notice of Revision or Disallowance and who objects to the amount of the Claim set out therein or any other provisions of such notice was required to deliver to the Monitor on or before 5:00 p.m. on October 5, 2012 a Notice of Dispute. If a Creditor failed to deliver a Notice of Dispute to the

Monitor by the foregoing deadline, then the value of such Creditor's claim is deemed to be as set out in the Monitor's Notice of Revision or Disallowance.

—ა.

57. Any creditor who has delivered a Notice of Dispute to the Monitor by the foregoing deadline is required, unless otherwise agreed by the Monitor in writing, to serve on the Monitor and the Applicants a notice of motion in the Court, returnable not less than 30 days after the service of the Notice of Dispute, for determination of the claim in dispute, failing which the value of such Creditor's claim shall be deemed to be as set out in the applicable Notice of Revision or Disallowance.

### CLAIMS DISALLOWED BY THE MONITOR

- 58. In connection with the Notices of Revision or Disallowance issued by the Monitor, only four Notices of Dispute were received as described below.
- 59. Revenu Quebec filed a Proof of Claim against Domfoam in the amount of \$2,912,679.00. The Monitor, after consultation with the Applicants, issued a Notice of Revision or Disallowance disallowing the claim of Revenu Quebec in full on September 21, 2012. On October 5, 2012, Revenu Quebec issued a Notice of Dispute in the full amount of its original claim which has yet to be resolved. The Monitor has agreed to extend the time for Revenu Quebec to bring its motion before the Court to determine its claim to provide the parties an opportunity to resolve same.
- 60. Revenu Quebec has taken the position that, notwithstanding the provisions of the Claims Solicitation Procedure Order, Domfoam was required to file a notice of objection under the Quebec Taxation Act and the Excise Tax Act to the notices of assessment previously issued by Revenu Quebec. The Monitor has advised Revenu Quebec that it does not agree with this position and that Revenu Quebec is bound by the terms of the Claims Solicitation Procedure Order.
- As set out in Mr. Vallecoccia's affidavit dated July 11, 2013, Domfoam has filed a notice of objection to the Revenu Quebec notices of assessment, all without prejudice to Domfoam's position (and that of the Monitor) that Revenu Quebec is nevertheless bound by the terms of the Claims Solicitation Procedure Order. Domfoam was initially advised by Revenu Quebec that its notice of objection was out of time, but Domfoam subsequently filed an appeal to extend the time for filing its notice of objection. On May 10, 2013, Revenu Quebec agreed to extend the time

within which Domfoam may appeal the Revenu Quebec notices of assessment to August 30, 2013. The Monitor understands that the Companies' counsel will be filing materials with Revenu Quebec prior to that date with a view to eliminating the Revenu Quebec claim.

- 62. As noted in the Fifth Report, the most significant Proofs of Claim submitted to the Monitor were filed in respect of the Canadian Class Actions. The Monitor received three separate Proofs of Claim as follows:
  - (a) Proof of Claim from one of the Plaintiffs in the BC Proceedings on behalf of a proposed class of all persons or entities in British Columbia who purchased polyurethane foam or polyurethane foam products, including carpet underlay, directly from the Companies from January 1, 1999 to the present;
  - (b) Proof of Claim by the Plaintiff in the Ontario Proceeding on behalf of a proposed class of all persons or entities in Canada who purchased foam products or carpet underlay directly and/or indirectly from the Companies from January 1, 1999 to the present, except for purchasers in the BC and Quebec Proceedings; and
  - (c) Proof of Claim from one of the Plaintiffs in the Quebec Proceeding, together with a proposed class of all persons and entities in Quebec who purchased polyurethane foam products,

(collectively, the "Canadian Class Action Proofs of Claim").

- 63. The Canadian Class Action Proofs of Claim did not specify the amount claimed but instead indicate such amount is "to be ascertained".
- 64. The Monitor issued a Notice of Revision or Disallowance in respect of each of the Canadian Class Action Proofs of Claim on the following basis:
  - (a) no agreement between the party filing the Proof of Claim and the proposed class or an Order of the Court or any other authority pursuant to which the party filing the Proof of Claim is authorized to file the Proof of Claim on behalf of the subject class was provided to the Monitor; and
  - (b) the Monitor required that the amount claimed under each Canadian Class Action Proof of Claim be quantified.

-14- 583

- 65. In accordance with the Claims Solicitation Procedure Order, the Monitor received Notices of Dispute in respect of each of the foregoing Notices of Revision or Disallowance. As noted above, the Plaintiffs in the Canadian Class Actions have now taken steps to have the class certified for settlement purposes.
- The Notices of Dispute also address quantification of the Claims in the Canadian Class Actions. The Notices of Dispute provide that the amount claimed under each of the Canadian Class Action Proofs of Claim is \$97,500,000 (\$292,500,000 in total) on behalf of the class of plaintiffs referenced in the subject Proof of Claim. The amount claimed is calculated as 10% of the total sales to customers in Canada by Domfoam and Valle Foam during the relevant period. The Notices of Dispute provide that it is the practice of the Competition Bureau and of the U.S. Department of Justice to establish fines using 20% of the volume of commerce to arrive at the appropriate fine. Of that amount, 10% is considered to be the estimate of the damages flowing from the overcharges, whereas the additional 10% is assessed for deterrence purposes.
- In December 2012, counsel to the Canadian Class Action Plaintiffs delivered to the Monitor a report providing an explanation of the amount claimed under the Canadian Class Action Proofs of Claim. The Monitor, together with its counsel and the Companies' counsel, have reviewed the merits of the methodology relied upon by the Canadian Class Action Plaintiffs to justify the amount claimed under the Canadian Class Action Proofs of Claim. The Monitor, the Companies' Class Action counsel and counsel to the Canadian Class Action Plaintiffs intend to meet as soon as possible with a view to resolving the value of the Canadian Class Actions Proofs of Claim, failing which, the Monitor will seek advice and directions from the Court with respect to quantification of the Canadian Class Action Proofs of Claim.

## STATEMENTS OF CASH RECEIPTS AND DISBURSEMENTS

68. Attached as Exhibit "G" is the Statement of Receipts and Disbursements for Valle Foam for the period March 29, 2012 to July 11, 2013. Total cash receipts from the sale of assets, the collection of accounts receivable, reimbursement of legal fees and other receipts are \$6,310,597.18. Total disbursements are \$649,374.26. Net cash on hand as at July 11, 2013 is \$5,661,222.92.

- 69. Attached as Exhibit "H" is the Statement of Receipts and Disbursements for Domfoam for the period March 29, 2012 to July 11, 2013. Total cash receipts from the sale of assets, the collection of accounts receivable, US Urethane settlement funds and other receipts are \$4,568,778.36. Total disbursements are \$1,259,367.27. Net cash on hand as at July 11, 2013 is \$3,309,411.09.
- 70. Attached as Exhibit "T" is the Statement of Receipts and Disbursements for A-Z Foam for the period March 29, 2012 to July 11, 2013. Total cash receipts from the sale of assets, the collection of accounts receivable, US Urethane settlement funds and other receipts are \$1,189,757.03. Total disbursements are \$307,670.37. Net cash on hand as at July 11, 2013 is \$882,086.66.

### PROFESSIONAL FEES

- 71. The Monitor and its independent legal counsel, Thornton Grout Finnigan LLP ("TGF") have maintained detailed records of their professional time and costs since the issuance of the Initial Order. Pursuant to paragraph 29 of the Initial Order, the Monitor and TGF were directed to pass their accounts from time to time before this Court.
- 72. The total fees of the Monitor during the period from October 1, 2012 to July 5, 2013 amount to \$86,231.00, together with expenses and disbursements in the amount of \$128.71 and harmonized sales tax ("HST") in the amount of \$11,226.76, totalling \$97,586.47 (the "Monitor Fees"). The time spent by the Monitor is more particularly described in the Affidavit of Catherine Hristow of Deloitte & Touche Inc., sworn July 12, 2013, sworn in support hereof and attached hereto as Exhibit "J".
- 73. The total legal fees incurred by the Monitor during the period October 1, 2012 to June 30, 2013 for services provided by TGF as the Monitor's independent legal counsel amount to \$68,040.00, together with disbursements in the amount of \$381.50 and HST in the amount of \$8,894.79, totalling \$77,316.29. The time spent by TGF personnel is more particularly described in the Affidavit of Grant Moffat, a partner of TGF, sworn July 11, 2013 in support hereof and attached hereto as Exhibit "K".

## ALLOCATION OF PROFESSIONAL FEES

- As noted in the Fifth Report, Valle Foam paid all of the professional fees incurred by the Companies in connection with the Class Actions and the Competition Bureau proceeding. The Applicants, with the concurrence of the Monitor, determined that the appropriate pro rata allocation of the foregoing professional fees to Valle Foam, Domfoam, and A-Z Foam should be 45%, 45% and 10% respectively. Domfoam and A-Z Foam repaid Valle Foam \$440,750.31 and \$97,944.51 plus HST of \$46,664.25 and \$10,369.84 respectively on account of such professional fees on June 15, 2012. Ongoing payment of the Monitor's fees, and legal fees incurred by the Monitor and the Applicants are paid on the above noted prorated basis.
- 75. Valle Foam also gave retainers to a number of law firms prior to January 12, 2012. As the law firms draw down on their retainers, Domfoam and A-Z Foam repay Valle Foam their respective proportionate share.

## EXTENSION OF THE STAY PERIOD

- 76. The Companies have asked the Court to approve an extension of the Stay Period from July 31, 2013 to December 31, 2013. The basis for this request is to allow time for resolution of the various Notices of Dispute referred to in the Seventh Report and to collect the remaining Valle Foam accounts receivable.
- 77. The Monitor believes that the Companies are acting in good faith and with due diligence and the Monitor therefore supports the stay extension to December 31, 2013.

## MONITOR'S RECOMMENDATIONS

- 78. For the reasons set out above, the Monitor recommends that:
  - (a) the Stay Period be extended until December 31, 2013;
  - (b) the Seventh Report and the activities of the Monitor as described in the Seventh Report be approved; and

the professional fees and disbursements of the Monitor, and TGF be approved and the (c) Companies be authorized to pay all such fees and disbursements.

All of which is respectfully submitted at Toronto, Ontario this 12 day of July, 2013.

**DELOITTE & TOUCHE INC.,** solely in its capacity as the Monitor of the Companies (as defined herein), and without personal or corporate liability

Per:

Senior Vice-President

# IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF 3113736 CANADA LTD. 4362063 CANADA LTD., and A-Z SPONGE & FOAM PRODUCTS LTD.

Court File No.: CV-12-9545-00CL

## ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

PROCEEDING COMMENCED AT TORONTO

# MOTION RECORD VOLUME II

# FRED TAYAR & ASSOCIATES Professional Corporation Barristers and Solicitors 65 Queen Street W, Suite 1200 Toronto, ON M5H 2M5

Fred Tayar (23909N) Tel.: (416) 363-1800 Fax: (416) 363-3356 Lawyers for Domfoam Inc.